40248	
Netional Libra of Canada	ary Bibliothique nationale CARABOAN VARMES 7NÉMES CARABANNES de Canade ON EXENDENCE SUR ANGLEFICARE
. 4	
•	
NAME OF AUTHOR/MON D	DELAUTEUR RICHARD DOUGLAS MILROY
· · · ·	
TITLE OF THESIS/TITRE DE	ELA THÈSE NUMERICAL AND ANALYTIC STUDERS OF KASER - PAASHA
	LATERALTIONS IN A SOLENOIDAL MAGNETICE FEELD
NIVERSITY / UNI VERSITÉ	University of ALBERTA
EGREE FOR WHICH THES	is was presented/ Tette these put presente. Ph. D
	ENNED/ANNIÈE D'OUTENTION DE CE GRADE 1978
iame of supervisor/wo	DA DU DIRECTEUR DE THÈSE DR. C.E. CAPRAGE & Dr. C.R. James
	created to the NATIONAL LIPSARY of the second
armission is heraby g	granted to the NATIONAL LIBRARY OF L'autorisation est, par la présente, éccordée à la BIBLIOTHÈ-
ANADA to microfilm t	this thesis and to lend or sell copies QUE NATIONALE DU CANADA de microfilmer cette thèse et
ANADA to microfilm t f the film.	this thesis and to lend or sell copies QUE NATIONALE DU CANADA de microfilmer cette thèse et de prêter ou de vendre des exemplaires du film.
ANADA to microfilm t f the film. author reserves oth	this thesis and to lend or sell copies QUE NATIONALE DU CANADA de microfilmer cette thèse et de prêter ou de vendre des exemplaires du film. Ner publication rights, and neither the 'L'auteur se réserve les autres droits de publication; ni la
CANADA to microfilm t f the film. author reserves oth mesis nor extensive ext	this thesis and to lend or sell copies QUE NATIONALE DU CANADA de microfilmer cette thèse et de prêter ou de vendre des exemplaires du film. Ner publication rights, and neither the / L'auteur se réserve les autres droits de publication; ni la tracts from it may be printed or other- thèse ni de longs extraits de celle-ci ne doivent être imprimés
CANADA to microfilm t f the film. author reserves oth hesis nor extensive ext	this thesis and to lend or sell copies QUE NATIONALE DU CANADA de microfilmer cette thèse et de prêter ou de vendre des exemplaires du film. Ner publication rights, and neither the / L'auteur se réserve les autres droits de publication; ni la
CANADA to microfilm t f the film. author reserves oth hesis nor extensive ext rise reproduced without	this thesis and to lend or sell copies QUE NATIONALE DU CANADA de microfilmer cette thèse et de prêter ou de vendre des exemplaires du film. Ner publication rights, and neither the / L'auteur se réserve les autres droits de publication; ni la stracts from it may be printed or other- thèse ni de longs extraits de celle-ci ne doivent être imprimés, the author's written permission. ou autrement reproduits sans l'autorisation écrite de l'auteur.
CANADA to microfilm t f the film. author reserves oth hesis nor extensive ext rise reproduced without	this thesis and to lend or sell copies QUE NATIONALE DU CANADA de microfilmer cette thèse et de prêter ou de vendre des exemplaires du film. Ner publication rights, and neither the / L'auteur se réserve les autres droits de publication; ni la stracts from it may be printed or other- thèse ni de longs extraits de celle-ci ne doivent être imprimés, the author's written permission. ou autrement reproduits sans l'autorisation écrite de l'auteur.
CANADA to microfilm t f the film. author reserves oth hesis nor extensive ext rise reproduced without	this thesis and to lend or sell copies QUE NATIONALE DU CANADA de microfilmer cette thèse et de prêter ou de vendre des exemplaires du film. Ner publication rights, and neither the / L'auteur se réserve les autres droits de publication; ni la stracts from it may be printed or other- thèse ni de longs extraits de celle-ci ne doivent être imprimés
CANADA to microfilm t if the film. author reserves oth hesis nor extensive ext rise reproduced without ATED/DATE SEPTE	this thee is and to lend or sell copies QUE NATIONALE DU CANADA de micrafilmer cette thèse et de prêter ou de vendre des exemplaires du film. her publication rights, and neither the / L'auteur se réserve les autres droits de publication; ni la stracts from it may be printed or other- thèse ni de longs extraits de celle-ci ne doivent être imprimés ou autrement reproduits sans l'autorisation écrite de l'auteur. meser les autres droits de l'auteur.
CANADA to microfilm t i the film. I author reserves oth mesis nor extensive ext rise reproduced without ATED/DATE SEPTE	this thesis and to lend or sell copies QUE NATIONALE DU CANADA de micrafilmer cette thèse et de prêter ou de vendre des exemplaires du film. her publication rights, and neither the / L'auteur se réserve les autres droits de publication; ni la stracts from it may be printed or other- thèse ni de longs extraits de celle-ci ne doivent être imprimés au autrement reproduits sans l'autorisation écrite de l'auteur. The suthor's written permission. au autrement reproduits sans l'autorisation écrite de l'auteur. The set of the suthor's straits de celle-ci ne doivent être imprimés au autrement reproduits sans l'autorisation écrite de l'auteur. The set of the suthor's straits de celle-ci ne doivent être imprimés the author's written permission. The set of the suthor's straits de celle-ci ne doivent être imprimés the suthor's written permission. The set of the set
f the film. author reserves oth hesis nor extensive ext rise reproduced without ATED/DATE SEPTE	this thee is and to lend or sell copies QUE NATIONALE DU CANADA de microfilmer cette thèse et de prêter ou de vendre des exemplaires du film. her publication rights, and neither the / L'auteur se réserve les autres droits de publication; ni la stracts from it may be printed or other- thèse ni de longs extraits de celle-ci ne doivent être imprimés ou autrement reproduits sans l'autorisation écrite de l'auteur. meser les autres droits de l'auteur.

NL-91 (8+74



- Cataloguing Branch Canadian Theses Division
- Ottawa, Canada K1A 6N4

# NOTICE

The quality of this microfiche is heavily dependent upon the quality of the original thesis submitted for microfilming. Every effort has been made to ensure the highest quality of reproduction possible.

• \

If pages are missing, contact the university which granted the degree.

Some pages may have indistinct print depectally if the original pages were typed with a poor typewriter ribbon or if the university sent us a poor photocopy.

Previously copyrighted materials (journal articles, published tests, etc.) are not filmed:

Reproduction in full or in part of this film is governed by the Canadian Copyright Act, R.S.C. 1970, c. C-30. Please read the authorization forms which accompany this thesis.

# THIS DISSERTATION HAS BEEN MICROFILMED EXACTLY AS RECEIVED

Dibliothèque nationale du Canada

Direction du catalogage Division des thèses canadiennes

## AVIS

La qualité de cette microfiche dépend grandement de la qualité de la thèse soumise au microfilmage. Nous avons tout tait pour assurer une qualité supérieure de reproduction.

S'il manque des pages, veuillez communiquer avec l'université qui a conféré le grade.

La qualité d'impression de certaines pages peut laisser à désirer, surtout si les pages originales ont été dactylographiées à l'aide d'un ruban usé ou si l'université nous a fait parvenir une photocopie de mauvaise qualité.

Les documents qui font déjà l'objet d'un droit d'auteur (articles de revue, examens publiés, etc.) ne sont pas microfilmés.

La reproduction, même partielle, de ce microfilm est soumise à la Loi canadienne sur le droit d'auteur, SRC 1970, c. C-30. Veuillez prendre connaissance des formules d'autorisation qui accompagnent cette thèse.

# LA THÈSE A ÉTÉ MICROFILMÉE TELLE QUE NOUS L'AVONS REÇUE

## THE UNIVERSITY OF ALBERTA

NUMERICAL AND ANALYTIC STUDIES OF LASER-PLASMA INTERACTIONS IN A SOLENOIDAL MAGNETIC FIELD

BY

4

ŧ.

2 -



## A THESIS

SUBMITTED TO THE FACULTY OF GRADUATE STUDIES AND RESEARCH IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF

DOCTOR OF PHILOSOPHY

DEPARTMENT OF ELECTRICAL ENGINEERING

EDMONTON, ALBERTA

FALL, 1978

## THE UNIVERSITY OF ALBERTA

## FACULTY OF CRADUATE STUDIES AND RESEARCH

The undersigned certify that they have read, and recommend to the Faculty of Graduate Studies and Research, for acceptance, a thesis entitled, "Mumerical and Analytic Studies of Laser-Plasma Interactions in a Solenoidal Magnetic Field", submitted by Richard Douglas Milroy in partial fulfilment of the requirements for the degree of Doctor of Philosophy in Electrical Engineering.

Supervisor

Supe

External Examiner

Date: August 4 J... 1978

#### ABSTRACT

A two-dimensional computergrisulation of a plasme in a solenoidal magnetic field is developed. This simulation is based on a magnetic flux shell model of the plasma. Effects of finite ejectrical conductivity and anisotropic heat conductivity are included. The computer code can optionally calculate the full radial dynamics of the problem or make the approximation of radial pressure balance.

The computer program has been used to do sample calculations on laser heated plasmas confined in a solenoid and on a  $\theta$ -pinch reactor. The program has also been used for a numerical study of the hydrodynamic behaviour of laser heated gas target plasmas.

A theory for the beat frequency mixing of antiparallel laser beams in a homogeneous plasma is developed. This theory includes effects of ion mobility and electron-ion collisions. Based on this theory and MHD simulations of laser heated plasmas in solenoidal magnetic fields, a plasma parametric amplifier of infrared laser radiation is proposed. Analysis of this device indicates that it could be used to efficiently amplify infrared radiation and may also be capable of creating and amplifying extremely short pulses of infrared radiation.

iv

#### ACKNOWLEDGENENTS

The author wishes to express his thanks to Dr. C.E. Capjack, and Dr. C.R. James who supervised this project. A special-thanks is also extended to Dr. J.N. McMullin for the many helpful discussions and the substantial contributions he has made to this work.

A special thanks is also extended to Dr. A.A. Offenberger, Hr. L. Pitt, and Dr. A. Ng for their magy stimulating discussions; and to Ms. Barbara J. Gallaiford for the Speedy and accurate typing of this thesis.

The author is also thankful for the financial support provided by the National Research Council 85 Canada and the University of Alberta.

Most of all the author would like to thank his wife, Kay,

•

÷γ

at .	1	•	1	
•	TAJ	LE OF CONTENTS		
CHAPTER			PAGE	
,1 <sup>a</sup> 1177	OBUCTION		1	
2 A TW	0-D <b>D(E)(</b> 5	IONAL MAGNETIC FLUX SHELL		
A STATE	l for MH	D SINULATIONS	7	•
	t. The Hy	drodynamic Model	9	
	2.1.1	Continuity Equation	10	
¥	2.1.2	Momentum Equation	10	
	2.1.3	Temperature Equation	11	
	2.1.4	Magnetic Field	12	٩
	2.1.5	Laser Absorption and		
		<b>Propagation</b> o	12	
•	2.1.6	Transport Coefficients	_ 13	
2.2	The Ma	agnetic Flux Coordinate System	16	
	2.2.1	Definition of the Coordinate		
		System	16	`
	2.2.2	Tranformation of Derivatives	17	
۰.	2.2.3	Dimensionless Variables	2.0	
	2.2.4	Expansion Parameter 1	21	
	2.2.5	Transformation of Derivatives		
•		and Associated Differential		
		Operators to Dimensionless		
,		Variables	22	

• • •

3

	•	
		<b>K</b> ACZ
2.3	Transformation of NED Equations to the	`
	) Dimensionless Magnetic Flux Coordinate	
	System	25
	2.3.1 Continuity Equation	25
	2.3.2 Momentum Equation	25
,	2.3.3 Temperature Equation	28
	293.4 Magnetic Field	33
	2 Absorption and Propagation	33
2.4	Shell Lquations In the second	35
	2.4.1 Continuity Equation	36
	2.4.2 Momentum Equation	38
	2.4.3 Temperature Equation	39
	2.4.4 Closure of the System of	
	Equations	41
	2.4.4a Radial Pressure	
	Belance	42
	2.4.4b Full Radial Dynamics	45
	2.4.5 Summary of Shell Equations	47
2.5	Numerical Solution	51
COMPI	UTER SIMULATIONS OF PLASMA IN A SOLENOIDAL	
	ETIC FIELD	56
3.1		57
	Production of Long Scale Length	
	Plasma for Beat Frequency Mixing	61
<b>२</b> २	0-Pinch Reactor	64

•

CHAPTER	С	H	P	TI	
---------	---	---	---	----	--

.

٣

CHAPTER		PAGE
4	A COMPUTER SIDERATION OF GAS TARGET	
	EXPERIMENTS	82
	4-1 Computer Model	83
. •	4.2 Results	88
	4.3 Conclusions	93
<b>5</b> '	A THEORY OF BEAT FREQUENCY MIXING	111
	5.1 Beam Coupling in the Presence of	
	Electron Density Fluctuations	114
t	5.2 Density Fluctuations in the	
	Presence of Antiparallel Electro-	
	magnetic Beams	121
	5.3 Case of Slowly Varying Electromagnetic	٠
	Bean Amplitudes	127
	5.4 Transient Analysis	136
	5.5 Results	141
6	SOME APPLICATIONS OF BEAT FREQUENCY MIXING	148
	6.1 A Plasma-Laser Amplifier in the	
	11-14 µm Wavelength Range	149
•	6.2 Short Pulse Amplification	154
·	6.2.1 Quasi-Static Analysis	158
	6.2.2 Transient Analysis	164
		-

•

•

SIONS
ľ

<

. REFERENCES

-

•

- 179

J

177

•

		•	
		•	PAGE
	CHAPTER APPENDIX A.	SCHE PROGRAMMING DETAILS FOR 2-D MED CODE	182
	APPENDIX 3.	LINEAR SUSCEPTATILITIES	213
	APPENDIX C.	PONDERONOTIVE FORCE DUE TO ANTIPARALLEL	
~		BEAMS	216
	APPENDIX D.	LISTING OF COMPUTER ROUTINES	218
	VITAE	•	291

,

.

•

.

.

•

•

,

•

۰.

`

•

## LIST OF THOUSE

٠

739462		MOE
2.1	Orthogonal coordinate system'aligned with	
	magnetic field	54
2.2	Angle of magnetic field lines relative to	
	solenoid axis in magnetic flux coordinate	
	system	54
2.3	Cutaway of Plasma column showing shell	
	structure	55
3.1	Electron temperature in eV at 0.5 µsec;	
	obtained from the first simulation reported	
	in Section 3.1. (Full radial dynamics)	66
3.2	Ion temperature in eV at 0.5 usec; obtained	
	from the first simulation reported in	
	Section 3.1 (Full radial dynamics)	66
3.3	Plasma density in $cm^{-3}$ at 0.5 µsec; obtained	
	from the first simulation reported in	
	Section 3.1. (Full radial dynamics)	67
3.4	Radial plasma velocity in cm/sec at 0.5 usec;	
	obtained from the first simulation reported	
	in Section 3.1. (Full radial dynamics)	67
3.5	Axial plasma velocity in cm/sec at 0.5 usec;	
	obtained from the first simulation reported	•
	in Section 3.1. (Full radial dynamics)	68
3.6	Axial magnetic field strength in kG at 0.5 used	2;
	obtained from the first simulation reported	
	in Section 3.1. (Full radial dynamics)	68

x

-

٩

<b>73608</b>	574. 	THE .	2
3.7	Electron temperature in eV of 0.5 yeac;	•	, <b>H</b>
	distined from the second, simulation reported		
•	in Section 3.1. (Redial propours balance)	69	
3.8	Ion temperature in eV at 0.5 page; obtained		
	from the second simulation reported in		
	Section 3.1. (Redial pressure belance)	69	
3.9	Plasma density in cm <sup>-3</sup> at 0.5 µeec; obtained		
	from the second simulation reported in		
•	Section 3.1. (Redial pressure balance)	70	
3.10	Axial places velocity at 0.5 pase; obtained		
	from the second simulation reported in		
-	Section 3.1. (Radial pressure belance)	70	
3.11	Axial magnetic field strongth at 0.5 usec;		
	obtained from the second simulation		
	reported in Section 3.1. (Radial pressure		
	balance)	71	
3.12	Electron temperature in eV at 0.5 µsec;		
	obtained from the first simulation		
•	reported in Section 3.2	72	
3.13	Ion temperature in eV at 0.5 µsec;		
	obtained from the first simulation		
	reported in Section 3.2	72	
3.14	Plasma density in cm <sup>-3</sup> at 0.5 usec;		
	obtained from the first simulation reported		
	in Section 3.2	73	

x1

.

730R.5		MOR
3.15	Anial values of electrol temperature and	,
-	ion temperature in eV and density in en-3	
	as a function of x; obtained from the	
	first simulation reported in Section 3.2	73
3.16	Electron temperature in eV at 1 usec;	
	obtained from the second simulation reported	
	in Section 3.2	74
3.17	lon temperature in eV at 1, usec; ebtained	
	from the second simulation reported in	
	Section 3.2	74
3.18	Plasma density in cm <sup>-3</sup> at 1 usec; obtained	
	from the second simulation reported in	
	Section 3.2	75
3.19	Axial values of electron temperature and	
	ion temperature in eV and density in cm <sup>-3</sup>	
~	as a function of x; obtained from the	
	second simulation reported in Section 3.2	75
3.20	Radii of magnetic flux surfaces at 0.5 ms;	
	from the simulation of 0-pinch reactor	76
3.21	Electron temperature in eV at 0.5 ms; from	
	the simulation of 8-pinch reactor	77
3.22	Plasma density in $cm^{-3}$ at 0.5 ms; from the	
	simulation of 0-pinch reactor	77
· 3.23	Axial plasma velocity in cm/sec at 0.5 ms;	
	from the simulation of 8-pinch reactor	78
		-

•

•

. \*

• • •

.

.

٠

•

.

,

٠

**xii** 

•

<b>F</b> IGURE	•	PAGE
3.24	A Axial magnetic field strength in kG at 0.5 ms;	
	from the simulation of 0-pinch reactor	78
3.25	Radii of magnetic flux surfaces at 2.0 ms; from	
	the simulation of 0-pinch reactor	79 -
3.26	Electron temperature in eV at 2.0 ms; from the	
	simulation of $\theta$ -pinch reactor	80
3.27	Plasma density in $cm^{-3}$ at 2.0 ms; from the	•
	simulation of $\theta$ -pinch reactor	80
3.28	Axial plasma velocity in cm/sec at 2.0 ms; from	<b>n</b>
	the simulation of $\theta$ -pinch reactor	81
3.29	Axial magnetic field strength in kG at 2.0 ms;	
	from the simulation of $\theta$ -pinch reactor	81
4.1	Initial axial density profile used in gas	
	target simulations #1, #2, and #3. The left	
	* denotes the position of laser focus for	
	simulations #1 and #2. The right * denotes	
	the position of laser focus for simulation	
	#3	94
4.2	Initial axial density profilé used in gas	•
	target simulation #4. The * denotes the	
	position of laser focus	94
4.3	Electron temperature in eV at 20 ns; from	
	simulation #1.	95
4.4	Ion temperature in eV at 20 ns; from	0.5
	simulation #1	95

s ,

٠

.

.

-

•

1

xiii

F	ĩ	G	UŤ	R	E
•	-	•	~	-	~

•

•

.

~

¥.

FIGURE		PAGE
4.5	Plasma density in cm <sup>-3</sup> at 20 ns; from	, ,
	simulation #1	96
4.6	Axial plasma velocity in cm/sec at 20 ns;	
	from simulation #1	<b>96</b>
4.7	Radial plasma velocity in cm/sec at 20 ns;	•
	from simulation #1	97
4.8	Laser beam power and axial laser beam	
	intensity at 20 ns; from simulation #1	98
4.9	Electron and ion temperatures in eV and	
	plasma density in cm <sup>-3</sup> at the laser focal	
	spot as a function of time, from	
	simulation #1	98
4.10	Electron temperature in eV at 20 ns;	
	from simulation #2	99
4.11	Ion temperature in eV at 20 ns; from	
:	simulation #2	99
4.12	Plasma density in $cm^{-3}$ at 20 ns; from	
	simulation #2	100
4.13	Axial plasma velocity in cm/sec at 20 ns;	
· .	from simulation #2	100
4.14	Radial plasma velocity in cm/sec at 20 ns;	
	from simulation #2	101
4.15	Laser beam power and axial laser beam	
	intensity at 20 ns; from simulation #2	102

riv .

FIGURE		PAGE
4.16	Electron and ion temperature in eV and	r
	plasma density in $cm^{-3}$ at the laser focal	
	spot as a function of time; from	
	simulation #2	102
4.17	Electron temperature in eV at 20 ns;	
	from simulation #3	103
4.18	Ion temperature in eV at 20 ns; from	
	simulation #3	103
.19	Plasma density in cm <sup>-3</sup> at 20 ns; from	
	simulation #3	104
. 20	Axial plasma velocity in cm/sec at 20 ns;	
	from simulation #3	104
. 21	Radial plasma velocity in cm/sec at 20 ns;	
	from simulation #3	105
. 22	Laser beam power and axial laser beam	
	intensity at 20 ns; from simulation #3	106
.23	Electron and ion temperatures in eV and	
	plasma density in cm <sup>-3</sup> at the laser focal	
	spot as a function of time; from simulation	
	#3	- 106
.24	Electron temperature in eV at 12 ns; from	
	simulation #4	107
.25	Ion temperature in eV at 12 ns; from	
	simulation #4	107
.26	Plasma density in cm <sup>-3</sup> at 12 ns; from	$\sim$

•

.

~

XV

FIGURE		PAGE
4.27	Axial plasma velocity in cm/sec at 12 ns;	
4.27	from simulation #4	108
4.28	Radial plasma velocity in cm/sec at 12 ng	
4.20	from simulation #4	, 109
4.29	Laser beam power and axial laser beam	
	intensity at 10 ns; from simulation #4	110
4.30	Electron and ion temperatures in eV and	
	plasma density in $cm^{-3}$ at the laser	
	focal spot as a function of time; from	
	simulation #4	110
5.1	a (solid line), $\tilde{N}_{e}(0)$ , and $\tilde{N}_{i}(+)$ as a	$\bigwedge$
J•1	function of $w_3$ for $n_0 = 2.5 \times 10^{17} \text{ cm}^{-3}$ ,	]
	$T_e = 100 \text{ eV}, T_i = 1 \text{ eV}, \text{ and}$	
	ε 1 λ <sub>1</sub> = 10.247 μm	143
5.2	$\alpha$ (solid line), $\hat{N}_{e}(0)$ , and $\tilde{N}_{i}(+)$ as a	
	function of $\omega_3$ for $n_0 = 2.5 \times 10^{17} \text{ cm}^{-3}$ ,	
	$T_{1} = 160 \text{ eV}, T_{1} = 120 \text{ eV}, \text{ and}$	
	$\lambda_1 = 10.247 \ \mu m$	143
5.3	$\alpha$ (solid line), $\hat{N}_{e}(0)$ , and $\hat{N}_{i}(+)$ as a	
	function of $\omega_3$ for $n_0 = 7.5 \times 10^{16} \text{ cm}^{-3}$ ,	
	$T_e = 100 \text{ eV}, T_i = 70 \text{ eV} \text{ and}$	
	$\lambda_1 = 9.6 \ \mu \mathbf{m}$	144
5.4	$\alpha$ as a function of $n_0$ for $\lambda_1 = 9.6$ µm and $\lambda_1$	
~ /	$\lambda_2 = 10.6 \ \mu m$ . The plasma temperature in	
	eV is indicated on the curves	145

xvi

• .

•

FIGURE	•	PAGE
5.5	$\alpha$ as a function of $n_0$ for $\lambda_1 = 10.6 \mu m$ and	
	$\lambda_2 = 12.0 \ \mu\text{m}$ . The plasma temperature in eV is	
	indicated on the curves	149
5.6	$\alpha$ as a function of n for $\lambda_1 = 10.6$ µm and	
	$\lambda_2 = 14.0 \ \mu m$ . The plasma temperature in eV	
	is indicated on the curves	146
5.7	$\alpha$ as a function of n for $\lambda_1 = 10.247 \ \mu$ m and	
<i>*</i>	$\lambda_2 = 10.2605 \text{ µm}.$ (1) $T_e = 100 \text{ eV}, T_1 = 50 \text{ eV};$	
•	(2) $T_e = 200 \text{ eV}, T_i = 150 \text{ eV};$ (3) $T_e = 500 \text{ eV},$	
	$T_{1} = 300 \text{ eV}$	146
5.8	$\alpha$ as a function of $T_e$ for $n_o = 2.5 \times 10^{17} \text{ cm}^{-3}$ ,	140
	$\lambda_1 = 10.247 \ \mu\text{m}, \text{ and } \lambda_2 = 10.2605 \ \mu\text{m}.$ The	
	value of T <sub>i</sub> is indicated on the curves	147
5.9	$\alpha$ as a function of T for n = 2.5 x 10 <sup>17</sup> cm <sup>-3</sup> ,	147
	$\lambda_1 = 10.247$ fm, and $\lambda_2 = 10.2605$ µm. $T_1$	
	is given by (1/a) T where values of a are	
	indicated on the curves	147
5.1	Schematic of optical amplifier. Mirrors in	
	(b) and (c) make an optical resonator and	
	increase the intensity of the low freq-	
	uency beam	169
5.2	$I_2$ in W/cm <sup>2</sup> as a function of axial	
	position, x, at three different times.	
	(1) $t = 1.33$ ns, (2) $t = 2.67$ ns, and	
	(3) t = 4 ns. No beat frequency mixing	
	in this case	169

xvii

ı

FIGURE

		•
.3	I <sub>1</sub> (dashed curves) and I <sub>2</sub> (solid curves)	
	in W/cm <sup>2</sup> as a function of axial position,	
	x, at three different times. (1) t = 1.33 ns,	
	(2) t = 2.67 ms; and (3) t = 4 ms. Initial	• • •
	peak pulse intensity is 10 <sup>8</sup> W/cm <sup>2</sup>	170
5.4 -	I1 (dashed curves) and L2 (solid curves) in	
η.	$W/cm^2$ as a function of axial position, x,	
	at three different times. (1) t = 1.33 ns,	
	(2) $t = 2.67$ ns, and (3) $t = 4$ ns. Initial	•
	peak pulse intensity is 10 <sup>9</sup> W/cm <sup>2</sup>	170
6.5	Real and imaginary parts of D(k <sub>3</sub> , w) vs w	
	for $T_e = 160 \text{ eV}$ , $T_i = 120 \text{ eV}$ and	
	$n_0 = 2.5 \times 10^{17} \text{ cm}^{-3}$	171
6.6	$D(k_3,t)$ as a function of t corresponding	
	to D(k <sub>3</sub> , w) in Figure 6.5	171
6.7	Real and imaginary parts of $D(k_3, \omega)$ vs $\omega$	
	for $T_e = 160 \text{ eV}$ , $T_i = 20 \text{ eV}$ and	
	$n_{o} = 2.5 \times 10^{17} \text{ cm}^{-3}$	172
6.8	$D(k_3,t)$ as a function of t corresponding	
	to D(k <sub>3</sub> , w) in Figure 6.7	172
6.9	$I_1$ , $I_2$ and $ n_e $ as a function of distance	
	from pulse front, x, after pulse front has	
	travelled 40 cm. Initially $I_1 = 1.5 \times 10^{10}$	
	$W/cm^2$ and $I_2 = 10^5 W/cm^2$	173

PAGE

5

ţ

xviii

* •		
· FIGURE		PAGE
6.10	Some as figure 6.9 except pulse Front has	
•	now travelled 80 cm	173
6.11	I1, I2 and in cm-3 as a Sunction of	• • • • • • • • • • • • • • • • • • •
<b>`</b>	distance from pulse front, x, after pulse	· ·
	front has travelled 40 cm. Initially	• •
	$I_1 = 1.8 \times 10^{10} \text{ W/cm}^2 \text{ and } I_2 = 10^{10} \text{ W/cm}^2$	174
6.12	Same as figure 6.11 except pulse front	、
	has now travelled 60 cm	174
6.13	Same as figure 6.11 except pulse front	
	has now travelled 70 cm -	175
6.14	Same as figure 6.11 except pulse front	
	has now travelled 80 cm	175
6.15	$I_1, I_2, \text{ and }   \hat{n}_e   \text{ in cm}^{-3} \text{ as a function of }$	
	distance from pulse front, $\tilde{x}$ , after pulse.	· · ·
	has travelled 40 cm. Initially $I_2$ is of	
	triangular shape with peak intensity of	
	$10^{10}$ W/cm <sup>2</sup> at $\tilde{x} = 1.5$ cm, and	
	$I_2 = 1.8 \times 10^{10}$	176
6.16	Same as figure 6.15 except pulse front has	
	now travelled 80 cm	176
A.1	Spatial positions at which variables are	
	calculated. N, $T_e$ , $T_1$ , $v_2$ , and B are	
	calculated at shell centers denoted by o.	
	R, W, and u are calculated on shell	
	boundaries denoted by x. The indexing of	
	variables is indicated on the bottom and	
	right hand side of the figure	210
	ixx	

FIGURE A	•	PAGE
A. 2	Flow chart illustrating two-step scheme	211
A.3	Position and indexing of numerical grid for	
	calculation of laser power P, and absorbed	
•	energy E, relative to positioning and	
	indexing of MHD variables	212

4-**4** 

·

XX

. **•** 

•

,

Ľ.

#### CHAPTER 1

#### Introduction

٩.

The physics of laser-plasma interactions is an important part of the controlled thermonuclear research program. While most of the effort in laser-plasma interaction research is directed towards the development of laser-pellet implosion fusion, a considerable effort is being devoted to the study of plagma dynamics and laser-plasma interactions in solenoids. In such a device a solenoidal magnetic field provides radial confinement of the plasma and inhibits thermal conduction in the radial direction. Dawson et al have suggested that CO, laser heated plasma columns may offer an alternative approach to controlled thermonuclear fusion. In this scheme the plasms is heated through the absorption of a high intensity CO, laser beam that is directed down the solenoid axis. The beam is guided down the solenoid axis by an onaxis density minimum which is created by a high on-axis heating rate. Additional heating may be provided by an adiabatic compression achieved by increasing the strength of the solenoidal magnetic field after laser heating is terminated<sup>2</sup>. In the simplest concept the plasma confinement time is limited to the time required for the plasma to stream freely out of the solenoid ends. Solenoid lengths of about 1 km are required to achieve adequate confinement times<sup>3</sup>. It has been suggested that endloss could be reduced, and hence the device shortened by plugging the ends with solid material or high density gas. Heat loss to the ends through thermal conduction then becomes a serious problem.

Problems and ideas related to this concept of laser fusion are being investigated in several laboratories using short (< 1 m)

solenoids. However, these experiments and reactor concepts are difficult to analyse theoretically. This is because a good theoretical understanding requires the simultaneous solution to the equations that govern many different but inter-related physical phenomena. Because of the complexity of this type of problem, computer simulations have become an essential tool for plasma physics research. Computer codes can incorporate many different but inter-related physical phenomena. This can be useful in testing simplified theories as well as providing a better understanding of experimental observations. In addition to explaining experimental results, computer models can predict unexpected phenomena, and be useful in the design of new experiments and reactor concepts.

The rapid advance of computer technology in the past few years has resulted in larger and faster computers being available to plasma researchers. This allows for the numerical solution to more sophisticated mathematical models leading to more realistic computer simulations.

In Chapter 2 of this thesis a two-dimensional computational model developed by Dr. J.N. McMullin and the author is described. This model has been designed to simulate the magnetohydrodynamic behaviour of a laser-heated plasma in a solenoidal magnetic field. An attempt has been made to make the computer code sufficiently versatile so that the one program can be used to study short experimental devices as well as reactor design concepts.

Electron thermal conductivity and fluid flow in a plasma in a strong magnetic field is markedly anisotropic. The thermal conductivity is much larger along the field lines than across it and plasma flow is

dominantly along the field lines. Because of this, a large amount of numerical diffusion is experienced when normal differencing techniques are employed to doke the ND equations. An extremity fine computational grid, resulting in an excessive amount of computer time needed to do long simulations, is then required in order to attain reasonable accuracy. This problem has been eliminated in the present model by transforming all of the equations into a moving non-orthogonal coordinate system defined by the magnetic field lines.

ŧ

Another feature built into the present numerical model is the option of assuming radial pressure balance or calculating full radial dynamics. In many cases the time scales of interest are much longer than the time required for a magneto-acoustic wave to propagate across the solenoid radius. When this is true any pressure imbalance in the radial direction is quickly smoothed out and the assumption of radial pressure balance in a numerical model is valid. The main advantage of making this assumption is that it eliminates what can be a very restrictive stability condition. The present model treats diffusion terms implicitly and all other terms are treated explicitly. Consequently, the timestep size is limited by the Courant-Friedricks-Lewy condition<sup>4</sup>. If full radial dynamics are calculated the timestep size is limited by the following two conditions:

$$\Delta t \leq Min \left\{ \frac{\Delta R}{C_{max}} \right\}$$
(1.1)  
$$\Delta t \leq Min \left\{ \frac{\Delta x}{C_{g}} \right\}$$
(1.2)

where At is the timestep size,  $C_{me}$  is the magneto-accustic velocity, At is the numerical grid size in the radial direction,  $C_{m}$  is the sound velocity, and Ax is the numerical grid size in the axial direction. For almost all simulations of plasmas in a solemoidal magnetic field condition (1.1) is much more restrictive than condition (1.2). The assumption of radial pressure balance eliminates condition (1.1) and timestep size is limited only by condition (1.2). For the simulations of long solemoids such as reactors this reduces the required number of calculations by several orders of mignitude.

Flux coordinates were used in a numerical model developed by Hertweck and Schneider<sup>5</sup> and used by Schneider<sup>6</sup> and Bodin <u>et</u> <u>al</u>.<sup>7</sup> in the study of end-loss from 0-pinches. However, their model assumed infinitive electrical conductivity so that the effects of magnetic diffusion in the plasms were not included. Although this approximation was adequate for their studies, it breaks down completely at low temperatures during the early stages of heating of the plasms. Magnetic diffusion is also expected to affect the end-loss from long, high  $\beta$  reactors<sup>8</sup> even at very high temperatures so finite conductivity has been included in the present model. Other improvements in this model are (1) the inclusion of full radial dynamics using an implicit scheme for calculating the motion of flux surfaces; (2) the use of a direct (rather than iterative) method for advancing the equations when radial pressure balance is assumed; (3) the simplification of the equations of motion by assuming only minor distortion of the field lines.

Sample calculations, using this computational model, are presented in Chapter 3. The versatility of the code is demonstrated through the presentation of results from solenoids of lengh 5 cm, 1 m,

and 1 hn.

. /

The code has also been used to provide sumerical backup for experiments performed by Dr. A.A. Offenberger<sup>9</sup>. In Chapter 4 some results from a computer study of the hydrodynamics of gas target experiments are presented. In these simulations a CO<sub>2</sub> laser been is assumed to be focussed into a semi-infinite slab of unmagnetized plasms. The laser heating of the plasma and resulting hydrodynamic expansion are modelled. Because the plasma is unmagnetized and the model uses a coordinate system based on the motion of magnetic flux lines some small alterations had to be made to the code for these simulations.

The results of Chapter 3 suggest that a laser heated plasma in a 1 p solenoid can have very long axial scale lengths. Simulations show that a plasma can be created with parameters ideal for the beat frequency mixing of antiparallel laser radiation. It is suggested in this thesis that this process could be used to create a very efficient broadband amplifier of infared radiation<sup>10</sup> as well as a device capable of creating or amplifying extremely short bursts of infrared radiation.

Krupke<sup>11</sup> has demonstrated that a short pulse of laser radiation can be amplified through the absorption of energy from a longer pulse of laser radiation propagating antiparallel to the short pulse. The radiation in the longer, or pump, pulse has a slightly higher frequency than the short pulse. The process works as follows: a high intensity pulse of laser radiation is directed into a mixing medium, in this case CH<sub>4</sub> gas. A short pulse of lower frequency laser radiation is directed antiparallel to the original pulse. Energy is transferred from the high frequency pulse to the low frequency pulse through a Raman process. Thus

the original pulse energy is depleted and the short pulse is amplified. If the presence is fact enough most of the energy of the pump will be transferred to the front of the amplified pulse. In this way energy from a longer laser pulse is used to create an extremely high intensity short pulse. It is feasible that such a process could be used to efficiently create the short pulses of radiation meeded for laser pallet fusion.

In Erupke's experiment neutral CH<sub>A</sub> gas was the mixing medium for 259 mm and 268 mm wavelength pulses of radiation from ErF lasers. In this thesis it is suggested that a plasma could be used as a mixing medium for infrared radiation in the 10.6 µm range. The mixing can take place between waves with a difference frequency close to the electron plasma frequency (a Raman process) or the ion-acoustic frequency Ta medium process).

Chapter 5 contains a review of some of the theory of the beat frequency mixing of antiparallel laser beams and extends this theory to provide a basis for the analysis of a device capable of amplifying ultra-short pulses of laser radiation. Chapter 6 contains a theoretical analysis of two proposed devices which rely on the beat frequency mixing of laser radiation in a plasms.

#### CHAPTER 2

ł

## A Two-Dimensional Magnetic Flux Shell Model for M.H.D. Simulations

This chapter describes a two-dimensional computational model developed by Dr. J:N. McMullin and the author to simulate the magnetohydrodynamics of a cylindrically symmetric plasma with an imbedded solenoidal magnetic field. This type of numerical simulation is a valuable supplement to current laboratory experiments designed to test the feasibility of using a magnetically confined plasma column as a thermonuclear device. Numerical models are also the only way at present for evaluating proposed designs of fusion reactors and neutron sources based on linear plasma columns. Because the geometries of current / experiments and reactor designs are basically the same, both can be simulated with a single code even though the dimensions and physical parameters vary over wide ranges.

A common feature of all experiments and reactor designs is the attainment of high electron temperatures (>40eV). Due to the resulting large electrical conductivity, the motion of plasma particles across magnetic field lines is strongly inhibited. Strong magnetic fields can therefore be used to restrain the plasma against expansion during heating as well as to inhibit logs of heat by conduction to the surrounding walls. The main losses of internal energy of these plasmas are convection and heat conduction along the field lines which join the confined plasma region to the outside world. If the magnetic field lines are used as coordinates, the motion of the plasma is mainly one-dimensional along these coordinates. Furthermore, there are no mixed derivatives in the parallel heat conduction term of the electron temperature equation so that heat conduction along field

lines may be calculated free of errors due to numerical diffusion across coordinates. For these reasons, the numerical model uses magnetic field line, or flux coordinates, (s,x,t), where s(r,x,t) is the magnetic flux through a circle of radius r centered on the x-axis, instead of the regular cylindrical coordinates, (r,x,t).

c

 $\overline{}$ 

8

.

,

## 2.1 The Hydrodynamic Model

The present model is based on the hydrodynamic equations given by Braginskii<sup>12</sup>. The plasma is approximated as a fully ionized two-temperature ideal magnetohydrodynamic fluid. The electrons and ions are assumed to have the same local number density and the same fluid velocity. Cylindrical symmetry is assumed in the derivation of all the equations. A solenoidal magnetic field is assumed to permeate the plasma although it may have a vanishing field strength. The magnetic field is assumed to diffuse due to a classical electrical resistivity with joule heating being absorbed by the electrons only. The strong anisotropy of thermal conduction perpendicular and parallel to the magnetic field lines is accounted for. Electrons and ions are assumed to exchange energy at the classical equipartition rate. An artifical viscosity is introduced to numerically handle shock waves which may be generated by a strong expansion. The artificial viscosity has the effect of spreading a shock front over several mesh points as described by Richtmeyer and Morton<sup>4</sup>. The model allows for the heating of electrons through the inverse, Bremsstrahlung absorption of a laser beam propagating down the solenoid axis.

The equations used are listed below in cylindrical coordinates (r,x,t). In these equations n<sub>o</sub> is the electron number density, m<sub>i</sub> is the ion mass,  $\rho = m_i n_o$  is the mass density, v<sub>x</sub> and v<sub>r</sub> are the axial and radial components of fluid velocity, T<sub>e</sub> and T<sub>i</sub> are the electron and ion temperatures, and P=n<sub>o</sub>k<sub>B</sub>(T<sub>e</sub>+T<sub>i</sub>) is the scalar pressure. B<sub>x</sub> and B<sub>r</sub> are the axial and radial components of the magnetic field. The total derivative,  $\frac{d}{dt}$ , is the convective derivative:

$$\frac{d}{dt} = \frac{\partial}{\partial t} + v_x \frac{\partial}{\partial x} + v_r \frac{\partial}{\partial r}$$

2.1.1 Continuity Equation:

$$\frac{dn_o}{dt} + n_o \vec{\nabla} \cdot \vec{v} = 0$$

or in cylindrical coordinates:

$$\frac{dn_o}{dt} + n_o \left[ \frac{\partial v_x}{\partial x} + \frac{1}{r} \frac{\partial}{\partial r} (r v_r) \right] = 0$$
 (2.1)

2.1.2 Momentum Equation:

$$\rho \frac{d\vec{v}}{dt} = -\nabla P + \frac{1}{c} (\vec{J}x\vec{B}) - \frac{\partial q_r}{\partial r} \hat{r} - \frac{\partial q_x}{\partial x} \hat{x}$$

where  $q_r$  and  $q_r$  represent artificial viscosities in the radial and axial directions respectively.

$$q_{r} = \begin{cases} a_{r}^{2} \rho \left(\frac{\partial v_{r}}{\partial r}\right)^{2} & \frac{\partial v_{r}}{\partial r} < 0\\ & & \\ 0 & & \frac{\partial v_{r}}{\partial r} \geq 0 \end{cases}$$

a is a variable having the dimensions of length and is chosen to spread a shock over several radial mesh points.

$$q_{\mathbf{x}} = \begin{cases} a_{\mathbf{x}}^{2} \left(\frac{\partial v_{\mathbf{x}}}{\partial \mathbf{x}}\right)^{2} & \frac{\partial v_{\mathbf{x}}}{\partial \mathbf{x}} < 0\\ 0 & \frac{\partial v_{\mathbf{x}}}{\partial \mathbf{x}} \geq 0 \end{cases}$$

a<sub>x</sub> is a variable having the dimensions of length and is chosen to spread a shock over several axial mesh points. Splitting the acceleration equation into axial and radial components and eliminating  $\vec{J}$  with  $\vec{\nabla} x \vec{B} = (4\pi/c) \vec{J}$  yields:

$$\rho \frac{dv_{\mathbf{x}}}{dt} = -\frac{\partial P}{\partial \mathbf{x}} - \frac{B_{\mathbf{r}}}{4\pi} \left( \frac{\partial B_{\mathbf{r}}}{\partial \mathbf{x}} - \frac{\partial B_{\mathbf{x}}}{\partial \mathbf{r}} \right) - \frac{\partial q_{\mathbf{x}}}{\partial \mathbf{x}} \qquad (2.2)$$

$$\rho \frac{dv_{\mathbf{r}}}{dt} = -\frac{\partial P}{\partial \mathbf{r}} + \frac{B_{\mathbf{x}}}{4\pi} \left( \frac{\partial B_{\mathbf{r}}}{\partial \mathbf{x}} - \frac{\partial B_{\mathbf{x}}}{\partial \mathbf{r}} \right) - \frac{\partial q_{\mathbf{r}}}{\partial \mathbf{r}} \qquad (2.3)$$

2.1.3 Temperature Equation: ( $\epsilon = e$  or i for electrons or ions

respectively)  

$$n_{o} \frac{dT_{\varepsilon}}{dt} - (\gamma - 1) T_{\varepsilon} \frac{dn_{o}}{dt} = n_{o} \frac{dT_{\varepsilon}}{dt} \Big|_{coll} + \frac{(\gamma - 1)}{K_{B}} [n_{o} \varepsilon_{L} - \nabla \cdot Q_{\varepsilon} + H_{jul}]$$
(2.4)

The first term on the right gives the rate of energy transfer between electrons and ions by collisions. In the electron temperature equation this term may be approximated as:

$$\frac{dT_{e}}{dt} \Big|_{coll} = \frac{1}{\tau_{eq}} (T_{i} - T_{e})$$
(2.5)

where  $\tau_{eq}$  is the appropriate collision time. The same term with opposite sign appears in the ion equation. The second term represents the inverse Bremsstrahlung absorption of laser energy while the third term represents thermal conduction. The last term, which accounts for electron heating due to magnetic field diffusion can be expressed as:

$$H_{ju1} = \frac{J_{\perp}^2}{\sigma_{\perp}}$$

where  $\sigma_{l}$  is the perpendicular component of the electrical conductivity tensor. Only the perpendicular component of the electrical conductivity tensor is needed since the current is purely azimuthal when the magnetic field is solenoidal. The heat flux vector  $\vec{Q}_{r}$  is given by:

$$\dot{\mathbf{d}}_{\mathbf{\varepsilon}} = -\mathbf{K}_{\mathbf{\varepsilon}} \cdot \nabla \mathbf{T}_{\mathbf{\varepsilon}}$$

where  $K_E$  is the heat conductivity tensor. Shock heating, which can be accounted for through the artificial viscosity term, has been neglected in the present treatment.

## 2.1.4 Magnetic Field Equations:

Equations for the magnetic field components can be written as:

$$B_{r} = -\frac{1}{r} \int_{0}^{r} \frac{\partial B_{x}}{\partial x} r' dr' \qquad (2.6)$$

$$\frac{\partial \mathbf{B}_{\mathbf{x}}}{\partial t} = \frac{1}{r} \frac{\partial}{\partial \mathbf{r}} \left[ \mathbf{r} \left( \mathbf{v}_{\mathbf{x}} \mathbf{B}_{\mathbf{r}} - \mathbf{v}_{\mathbf{r}} \mathbf{B}_{\mathbf{x}} \right) \right] + \frac{c^2}{4\pi r} \frac{\partial}{\partial r} \left[ \frac{\mathbf{r}}{\sigma_{\mathbf{j}}} \left( \frac{\partial \mathbf{B}_{\mathbf{x}}}{\partial \mathbf{r}} - \frac{\partial \mathbf{B}_{\mathbf{r}}}{\partial \mathbf{x}} \right) \right] (2.7)$$

Equation (2.6) can be derived from  $\nabla \cdot B = 0$  while equation (2.7) is obtained from Maxwells equations and Ohm's Law,

$$\dot{J} = \sigma(\dot{E} + \frac{1}{c} \dot{v} \times \dot{B})$$

## 2.1.5 Laser Absorption and Propagation:

The electron heating rate due to the inverse Bremsstrahlung absorption of laser energy used in equation (2.4) can be written as:

$$n_{o} t_{L} = K_{a} I_{1}(r, x, t)$$
 (2.8)

where  $I_1$  is the laser beam intensity and  $K_a$  is the absorption coefficient. The laser beam has a given radial profile, Gaussian for most cases, which is assumed to be unaltered as the beam propagates down the solenoid axis. The total beam power  $P_{\gamma}$  where  $\gamma$ 

$$P_{L}(x,t) = 2\pi \int_{0}^{\infty} I_{1}(r,x,t) r dr$$
 (2.9)

obeys the transfer equation:

$$\frac{1}{c}\frac{\partial}{\partial t}P_{L} + \frac{\partial}{\partial x}P_{L} = -\overline{K}_{a}P_{L} \qquad (2.10)$$

c is the speed of light and  $\overline{K}$  is the absorption coefficient averaged over radial profile with the weights at each radial position being proportional to the beam intensity at that point. This scheme reflects the fact that a light ray is refracted through regions of varying absorption as it propagates through the plasma.

2.1.6 Transport Coefficients:

Values for the transport coefficients that have been used in this model have been taken from the Revised NRL Plasma Formulary<sup>13</sup>. The formula connecting the low and high magnetic field limits for perpendicular thermal conductivity comes from Hain  $\frac{dt}{dt} = \frac{1}{2}$ . In the following temperature is measured in eV and  $k_B = 1.6 \times 10^{-12} \text{ ergs/eV}$ .

Thermal Conduction Parallel to  $\frac{1}{8}$ 

**\$**\_\_\_\_

$$k_{11}^{e} = 3.1 \times 10^{9} \frac{T_{e}}{ln\Lambda} \left(\frac{s^{y}}{e^{y}}\right)$$

$$k_{11}^{1} = 1.25 \times 10^{8} \frac{T_{1}^{5/2}}{in \Lambda} \left(\frac{ergs}{eV \sec cm}\right)$$

# Thermal Conduction Perpendicular to B

$$k_{\perp}^{\varepsilon} = \frac{\delta_{1}^{\varepsilon} T_{\varepsilon}^{5/2}}{(1 + \delta_{2}^{\varepsilon} \frac{T_{\varepsilon}^{3}B^{2}}{n_{o}^{2}})} \quad \left(\frac{ergs}{eV sec cm}\right)$$

where

$$\delta_{1}^{e} = \frac{3.1 \times 10^{9}}{\hbar n \Lambda} \qquad \qquad \delta_{1}^{i} = \frac{1.25 \times 10^{8}}{\hbar n \Lambda}$$
$$\delta_{2}^{e} = \frac{2.45 \times 10^{25}}{(\hbar n \Lambda)^{2}} \qquad \qquad \delta_{2}^{i} = \frac{7.8 \times 10^{22}}{(\hbar n \Lambda)^{2}}$$

In the limit of large magnetic fields,  $\omega_{c\epsilon} >> v_{\epsilon}$ , (where  $\omega_{c\epsilon}$  is the cyclotron frequency of species  $\epsilon$ , and  $v_{\epsilon}$  is the collision frequency of species  $\epsilon$ )

$$k_{\perp}^{\varepsilon} \approx \frac{\delta_{1}^{\varepsilon}}{\delta_{2}^{\varepsilon}} \frac{n_{o}^{2}}{B^{2}\sqrt{T_{\varepsilon}}}$$

In the limit of small magnetic fields,  $\omega_{c\epsilon} < v_{\epsilon}$ ,

$$k_{\perp}^{\varepsilon} = \delta_{1}^{\varepsilon} T_{\varepsilon}^{5/2}$$

Electrical Conductivity  $\sigma_{\perp} = 8.7 \times 10^{13} \quad \frac{T_{e}^{3/2}}{in\Lambda} \quad sec$ 

# Electron-Ion Equipartition of Energy

<u>,</u>

$$\tau_{eq} = \frac{1}{\nu_{e1}}$$
  
 $\nu_{e1} = 3.25 \times 10^{-9} = \frac{n_o \ln \Lambda}{T_o^{-3/2}} = \sec^{-1}$ 

•

# Laser Absorption Coefficient

The value given by Johnston and Dawson  $^{15}$  is used and for  $\lambda = 10.6 \mu m$ 

$$K_{a} = \frac{9.74 \times 10^{-36} n_{o}^{2}}{T_{e}^{3/2}} tnA$$

where

$$\Lambda = \min\{2.3 \cdot T_{e}^{3/2}, 12 \cdot T_{e}\}$$

ų

•
### 2.2 The Magnetic Flux Coordinate System

A number of different numerical schemes may be used to selfconsistently solve the partial differential equations in Section 2.1. Since energy transport due to either particle convection or thermal conduction is strongly anisotropic, with the dominant energy transport being along the magnetic field lines, any such scheme will introduce considerable numerical diffusion in cases where the magnetic field lines are not parallel to the x axis. In the present model the differential equations are transformed into a time dependent non-orthogonal coordinate system defined by the magnetic field lines. In this coordinate system energy transport occurs dominantly along one of the coordinates and the problem of large numerical diffusion is eliminated.

#### 8

#### 2.2.1 Definition of the Coordinate System

The transformation of variables from cylindrical coordinates (r,x,t) to flux coordinates (s,x',t') is defined below.



t' = t

The corresponding inverse transform can be written as:

The radius, R(s,x,t), of a flux surface of constant s now becomes a plasma variable from which the magnetic field components can be determined.

\* 150 X \*\* \*\*\*\*\*

## 2.2.2 Transformation of Derivations

A SA MARKANA SA

10 A.L.

Relations governing the transformation of derivatives between the two coordinate systems will now be obtained.

$$\begin{bmatrix} ds \\ dx' \\ dt' \end{bmatrix} = X \quad \begin{bmatrix} dr \\ dx \\ dt \end{bmatrix} \quad and \quad \begin{bmatrix} dr \\ dx \\ dt \end{bmatrix} = Y \quad \begin{bmatrix} ds \\ dx' \\ dt' \end{bmatrix} \quad (2.13)$$

where

.

$$\mathbf{X} = \begin{bmatrix} \mathbf{\hat{\vartheta}}_{\mathbf{X}}^{\mathbf{S}} & \mathbf{\hat{\vartheta}}_{\mathbf{X}}^{\mathbf{S}} & \mathbf{\hat{\vartheta}}_{\mathbf{L}}^{\mathbf{S}} \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{bmatrix} \text{ and } \mathbf{Y} = \begin{bmatrix} \mathbf{\hat{\vartheta}}_{\mathbf{R}}^{\mathbf{R}} & \mathbf{\hat{\vartheta}}_{\mathbf{X}}^{\mathbf{R}} & \mathbf{\hat{\vartheta}}_{\mathbf{L}}^{\mathbf{R}} \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{bmatrix}$$
(2.14)

From equation (2.13)  $X = Y^{-1}$  and

$$\mathbf{Y}^{-1} = \begin{bmatrix} \frac{1}{\partial_{\mathbf{g}} \mathbf{R}} & -\frac{\partial_{\mathbf{x}} \mathbf{R}}{\partial_{\mathbf{g}} \mathbf{R}} & -\frac{\partial_{\mathbf{t}} \mathbf{R}}{\partial_{\mathbf{g}} \mathbf{R}} \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{bmatrix}$$
(2.15)

C

A term wise comparison of X and  $Y^{-1}$  yields

$$\partial_{\mathbf{r}} S = \frac{1}{\partial_{\mathbf{R}}}, \partial_{\mathbf{x}} S = -\frac{\partial_{\mathbf{x}}, R}{\partial_{\mathbf{R}}}, \text{ and } \partial_{\mathbf{t}} S = -\frac{\partial_{\mathbf{t}}, R}{\partial_{\mathbf{s}} R}$$

17

۰.

#### States and the states of the

18

. 24.

## By using these relations, the partial derivations transform as follows:

$$\frac{\partial}{\partial t} + \frac{\partial}{\partial t} + \frac{\partial S}{\partial t} = \frac{\partial}{\partial t} - \frac{\partial}{\partial t} + \frac{\partial}{\partial t} = \frac{\partial}{\partial t} + \frac{\partial}{\partial t} + \frac{\partial}{\partial t} = \frac{\partial}{\partial t} + \frac{\partial}{\partial t} + \frac{\partial}{\partial t} = \frac{\partial}{\partial t} + \frac{\partial}{\partial t} + \frac{\partial}{\partial t} = \frac{\partial}{\partial t} + \frac{\partial}{\partial t} + \frac{\partial}{\partial t} = \frac{\partial}{\partial t} + \frac{\partial}{\partial t} + \frac{\partial}{\partial t} = \frac{\partial}{\partial t} + \frac{\partial}{\partial t} + \frac{\partial}{\partial t} + \frac{\partial}{\partial t} = \frac{\partial}{\partial t} + \frac{\partial}{\partial t} + \frac{\partial}{\partial t} + \frac{\partial}{\partial t} = \frac{\partial}{\partial t} + \frac{$$

r

The right hand side of equations (2.16) is now a function of the new variables. Some other differential operators and quantities that will be required to express the MHD equations in the new coordinate system are derived below.

> $\frac{d}{dt} = \frac{\partial}{\partial t} + \frac{d}{v} + \frac{\partial}{\partial t} + \frac{\partial}{\partial t} + \frac{\partial}{\partial s} \frac{\partial}{\partial s} + \frac{\partial}{v} \frac{\partial}{\partial x} + \frac{\partial}{v} \frac{\partial}{\partial s} \frac{\partial}{\partial s}$  $+ v_r \frac{\partial S}{\partial r} \frac{\partial}{\partial s}$  $=\frac{\partial}{\partial t^{T}} + v_{x} \frac{\partial}{\partial x^{T}} + (\frac{\partial S}{\partial t} + \dot{v} \cdot \dot{\nabla}S) \frac{\partial}{\partial s}$

οτ

$$\frac{d}{dt} + \frac{\partial}{\partial t'} + v_{x} \frac{\partial}{\partial x'} + \frac{dS}{dt} \frac{\partial}{\partial s}$$
(2.17)

For any vector V defined in cylindrical coordinates

$$\stackrel{+}{\nabla} \cdot \stackrel{+}{V} = \frac{1}{r} \frac{\partial}{\partial r} (r V_r) + \frac{\partial V_x}{\partial x}$$
 (2.18)

By using equations (2.16), this transforms to:

$$\dot{\nabla} \cdot \dot{V} + \partial_{\chi} V_{\chi} = \frac{\partial_{\chi} R}{\partial_{\chi} R} \partial_{\chi} V_{\chi} + \frac{1}{R \partial_{\chi} R} \partial_{\chi} (R V_{\chi})$$
(2.19)

.

An expression for  $v_{T}$  can be obtained.

$$v_{\rm r} = \frac{dR}{dt} = \frac{\partial R}{\partial t^{\rm r}} + v_{\rm r} \frac{\partial R}{\partial x^{\rm r}} + \frac{dS}{dt} \frac{\partial R}{\partial s}$$
(2.20)

Equations (2.17) and (2.20) bo contain the factor  $\frac{dS}{dt}$ . This factor can be expressed in terms of the magnetic field components. In cylindrical coordinates:

$$\frac{dS}{dt} = \frac{\partial S}{\partial t} + v_r \frac{\partial S}{\partial r} + v_x \frac{\partial S}{\partial x}$$
(2.21)

From equation (2.11) and  $\nabla \cdot B = 0$ , it can be shown that:

$$\frac{\partial S}{\partial x} = -2\pi r B_r \qquad (2.22)$$

From equation (2.11):

1

•

$$\frac{\partial S}{\partial r} = 2\pi r B_{\chi}$$
(2.23)

From equations (2.11) and (2.7):

$$\frac{\partial S}{\partial t} = 2\pi r (v_{\mathbf{x}} \frac{B}{r} - v_{\mathbf{r}} \frac{B}{x}) + \frac{c^2}{2\sigma_{\perp}} r (\frac{\partial B_{\mathbf{x}}}{\partial r} - \frac{\partial B_{\mathbf{r}}}{\partial x}) \qquad (2.24)$$

i,

By substituting equations (2.21), (2.22), and (2.24) into equation (2.20):

••

$$\frac{dS}{dt} = \frac{c^2}{2q_1} r \left( \frac{\partial B_{\mu}}{\partial r} - \frac{\partial B_{\mu}}{\partial r} \right)$$
(2.25)

### 2.2.3 Dimensionless Variables

Numerical techniques will be employed to find approximate solutions to the magnetohydrodynamic equations in the magnetic flux coordinate system. It has been found that the algebra can be greatly simplified and computer time can be significantly reduced if certain small terms are neglected in the magnetohydrodynamic equations. Such email terms can most easily be identified if the M.H.D. equations are expressed in terms of a set of dimensionless variables. By using dimensionless variables in the computer program the number of constant multipliers in the difference equations is reduced which in turn reduces the computer time required for a solution. In converting equations to dimensionless units, dimensional quantities Q are replaced by  $Q_0 \tilde{Q}$ where  $Q_0$  is a convenient unit and  $\tilde{Q}$  is the dimensionless variable. The four basic quantities used to define the dimensionless system are defined below.

(i)  $L_o = axial length unit <math>x' + L_o \tilde{x}$ (ii)  $R_{o_i} = radial length unit <math>r' + R_o r$ (iii)  $N_o = number density n_o + N_o N$ (iv)  $T_o = temperature T_{e,i} + T_o T_{e,i}$ 

The other basic units that will be used are defined in terms of the above four variables and are listed below.

(1)	<b>P</b> _0	- NKT	- Pressure
(11)	B	$= (8\pi P_{o})^{1/2}$	- Magnetic field
		$- (k_{B}T_{o}/m_{i})^{1/2}$	- Axial vélocity

(iv) t <sub>o</sub>	$= L_o / v_o$	- Time t'-tot
(v) A <sub>0</sub>	<b>π R<sup>2</sup></b>	- Area -
(vi) S <sub>o</sub>	- A <sub>0</sub> B <sub>0</sub>	- Magnetic flux
(vii) E <sub>o</sub>	- NkTLA	- Energy
(viii)P	= NkTAv	- Power

2.2.4 Expansion Parameter  $\varepsilon_1$ 

The length of a typical solenoidal device that we are interested in modelling will invariably be much greater than its radius. As a result L >> R and a small parameter  $\varepsilon_1 \equiv R_0/L_0$  can be defined. When the MHD equations are expressed in terms of dimensionless units in the magnetic flux coordinate system, many terms appear that are of order  $\epsilon_1^2$ . In the present treatment such terms have been neglected. Neglecting terms of order  $\epsilon_1^2$  is equivalent to assuming that the slope of a field line with respect to the axis of a solenoid is a small quantity  $\varepsilon$  and that terms of the order of  $\epsilon_l^2$  can be neglected. For a magnetic field in a typical solenoid,  $\epsilon_1 << 1$  except for small regions near each end where the field lines are diverging. The axial length of these regions is of the order of the solenoid diameter which for most experiments and all reactor designs is a small fraction of the total plasma. In laser heating experiments in which a bleaching wave propagates through the plasma, the approximation should remain valid if any one of the four following conditions holds:

- (1)  $B^2/8\pi >> P$
- (2) Absorption length, L >> R las, Radius of Laser Beam

٠. 21

(3) Bleaching Wave Velocity, V<sub>bw</sub> >> V<sub>ms</sub>, Magneto-Acoustic Velocity
 (4) c<sup>2</sup>/(4π V<sub>m</sub> R<sub>1as</sub>)·>> σ, Electrical Conductivity.

Condition (1) ensures that the magnetic pressure will prevent significant expansion of the plasma and hence distortion of the field lines. If condition (1) does not hold, then it is reasonable to assume that the region being heated and still expanding is on the order of  $L = \min(V_{bw}\Delta t, L_{abs})$  where  $\Delta t = R_{las}/V_{ma}$  is the approximate expansion time of the plasma under uneven radial heating. Since the displacement of a field line is on the order of the radius of the heated region,  $R_{las}$ , the slope of the field lines is approximately  $\epsilon_1 = R_{las}/L$ . Effect of conditions (2) or (3) ensure that  $\epsilon_1 << 1$ . When condition (4) holds, smoothing of the magnetic field by diffusion dominates over distortion of the field by fluid convection.

# 2.2.5 <u>Transformation of Derivatives and Associated Differential</u> Operators to Dimensionless Variables

The transformation of the variable S, as defined in equation (2.11) into dimensionless form yields:

$$\hat{S} = 2 \int \hat{B}_{\gamma} \hat{r}' d\hat{r}'$$
(2.26)

It then follows that:

B

$$\frac{\partial \hat{S}}{\partial r} = 2\hat{B}_{R}\hat{R} = \frac{1}{-\frac{1}{\sqrt{2}}}$$

or

$$\frac{1}{2R\partial_{n}R}$$
(2.27)

The subscript  $\tilde{x}$  will be dropped on  $\tilde{B}_{\tilde{x}}$  in subsequent analysis. The variable D is defined as follows:

$$D \equiv \frac{dS}{dt} \qquad (2.28)$$

The variable D is converted into dimensionless variables by using equation (2.25) and dropping terms of order  $\epsilon_1^2$ .

$$D = \frac{c^2}{2\sigma_{\perp}} B_{\rho} \hat{R} \frac{\partial \hat{B}}{\partial \hat{r}} = D_{\rho} \hat{D}$$
(2.29)

where

$$D_{o} \equiv \frac{c^{2}B_{o}}{2\sigma_{\perp}}$$

Two other quantities that will be frequently used in subsequent

analysis are:

 $D \frac{\partial}{\partial s} = \frac{V_{o}}{L_{o}} \varepsilon_{2} \tilde{D} \frac{\partial}{\partial s}$   $\frac{\partial D}{\partial s} = \frac{V_{o}}{L_{o}} \frac{\partial}{\partial s} [\varepsilon_{2} \tilde{D}]$ (2.30)

where



The magnetohydrodynamic equations defined in Section 2.1 can be converted directly from cylindrical coordinates to magnetic flux coordinates in dimensionless variables by using the differential operators in equations (2.16) to (2.17) expressed in terms of dimensionless variables.

The operators may be shown to take the following form.

$$\frac{\partial}{\partial t} + \frac{\mathbf{v}_{o}}{\mathbf{L}_{o}} \begin{bmatrix} \partial_{\chi} & - \frac{\partial_{\chi} \mathbf{R}}{\partial \chi} & \partial_{\chi} \end{bmatrix}$$
$$\frac{\partial}{\partial \mathbf{x}} + \frac{1}{\mathbf{L}_{o}} \begin{bmatrix} \partial_{\chi} & - \frac{\partial_{\chi} \mathbf{R}}{\partial \chi} & \partial_{\chi} \end{bmatrix}$$
$$\frac{\partial}{\partial \mathbf{r}} + \frac{1}{\mathbf{R}_{o}} \begin{bmatrix} 1}{\partial \chi \mathbf{R}} & \partial_{\chi} \end{bmatrix}$$
$$\frac{\partial}{\partial t} + \frac{1}{\mathbf{R}_{o}} \begin{bmatrix} \frac{1}{\partial \chi \mathbf{R}} & \partial_{\chi} \end{bmatrix}$$
$$\frac{\partial}{\mathbf{R}} + \frac{1}{\mathbf{R}_{o}} \begin{bmatrix} \frac{1}{\partial \chi \mathbf{R}} & \partial_{\chi} \end{bmatrix}$$

U

(2.31)

where here and in all subsequent analysis the  $\tilde{x}$  subscript will be dropped from  $\tilde{v}_{\chi}$ . Equation (2.19) can be expressed in dimensionless variables as:

r

$$\nabla \cdot V = \frac{V}{L} \left[ \partial_{\chi} \tilde{V}_{\chi} - \frac{\partial_{\chi} \tilde{R}}{\partial_{\chi} \tilde{R}} \partial_{\chi} \tilde{V}_{\chi} \right] + \frac{V}{R_{o}} \left[ \frac{1}{\tilde{R} \partial_{\chi} \tilde{R}} \partial_{\chi} (\tilde{R} \tilde{V}_{\chi}) \right]$$
(2.32)

An expression v<sub>r</sub>, defined in equation (2.19), can be written in terms of the dimensionless variables as:

$$\mathbf{v}_{\mathbf{r}} = \mathbf{v}_{\mathbf{o}} \begin{bmatrix} \mathbf{a}_{\mathbf{c}} \mathbf{\hat{R}} + \mathbf{v}_{\mathbf{a}} \mathbf{\hat{r}} + \mathbf{\varepsilon}_{2} \mathbf{\hat{D}}_{\mathbf{a}} \mathbf{\hat{R}} \end{bmatrix}$$
(2.33)

٩,

## 2.3 <u>Transformation of M.H.D. Equations to the Dimensionless Magnetic</u> Flux Coordinate System

The equations which define the magnetohydrodynamic model of Section 2.1 will now be converted to the coordinate system defined in Section 2.2.

2.3.1 Continuity Equation

By using the transformations indicated in equations (2.31) in equation (2.1) the following expression for the continuity equation in dimensionless magnetic flux coordinates is obtained:

 $\partial_{r}(N R \partial_{R} R) + \partial_{x}(v N R \partial_{R} R) + \partial_{z}(\varepsilon_{2} D N R \partial_{z} R) = 0$  (2.34)

All variables in equation (2.34) are dimensionless. For the sake of clarity the  $\sim$  over the variables has been dropped in this equation.

2.3.2 Momentum Equation

In converting equations (2.2) and (2.3) to dimensionless form the variable  $B_{T}$  may be written in terms of  $B_{X}$  and R(s,x,t). If s and t are held constant, lines defined by R(s,x,t) and x define magnetic field lines. That is

$$\frac{\partial R}{\partial x} = \frac{B}{B}_{x}$$

 $B_r = B_x \frac{\partial R}{\partial x}$ 

Thus,

or in terms of dimensionless variables

$$B_{r} = c_{1} B_{0} \tilde{B} \frac{\partial \tilde{R}}{\partial \tilde{x}}$$

Other relationships that will be used are

$$\rho = \mathbf{m}_{1} \mathbf{N}_{0} \mathbf{\tilde{N}} = \mathbf{N}_{0} \left( \frac{\mathbf{k}_{B} \mathbf{T}_{0}}{\mathbf{v}_{2}^{2}} \right) \mathbf{\tilde{N}}$$

$$\mathbf{v}_{r} = \left( \frac{\mathbf{v}_{0} \mathbf{R}_{0}}{\mathbf{L}_{0}} \right) \mathbf{\tilde{v}}_{r}$$

$$\mathbf{\tilde{v}}_{r}^{\lambda} = \partial_{\chi} \mathbf{\tilde{R}} + \mathbf{\tilde{v}} \partial_{\chi} \mathbf{\tilde{R}} + \varepsilon_{2} \mathbf{\tilde{D}} \partial_{\chi} \mathbf{\tilde{R}}$$

$$\mathbf{\tilde{v}}_{r}^{\lambda} = \partial_{\chi} \mathbf{\tilde{R}} + \mathbf{\tilde{v}} \partial_{\chi} \mathbf{\tilde{R}} + \varepsilon_{2} \mathbf{\tilde{D}} \partial_{\chi} \mathbf{\tilde{R}}$$

By using the above relationships plus the transformations given by equations (2.31), equation (2.3) transforms to

$$N_{O}k_{B}T_{O}\frac{R_{O}}{L_{O}^{2}}\tilde{N}(\partial_{\chi} + \tilde{v}\partial_{\chi} + \epsilon_{2}\tilde{D}\partial_{\chi})\tilde{v}_{r} = -\frac{N_{O}k_{B}T_{O}}{R_{O}}\frac{1}{\partial_{\chi}\tilde{R}}\partial_{\chi}(\tilde{P} + \tilde{q}_{r})$$
$$-\frac{B_{O}^{2}\tilde{B}}{4\pi R_{O}}\frac{1}{\partial_{\chi}\tilde{R}}\partial_{\chi}\tilde{R} + \tilde{v}\partial_{\chi}\tilde{R} +$$

where the term  $\frac{\partial B_r}{\partial x}$  has been neglected since it is of order  $\epsilon_1^2$  smaller than  $\frac{\partial B_x}{\partial r}$ , and  $\hat{q}_r$  is the appropriate expression in dimensionless variables for the artificial viscosity.

(2.35)

26

(2.36)



where  $a_r$  is a dimensionless constant between 1.5 and 2.0 and  $\Delta R$  is the radial grid spacing in the numerical calculations. Equation (2.37) can be re-written as

$$(\partial_{\mathbf{r}} + \mathbf{v}\partial_{\mathbf{x}} + \epsilon_{2}\mathbf{D} \partial_{\mathbf{s}}) \mathbf{v}_{\mathbf{r}} = -\frac{1}{N\epsilon_{1}^{2}} \frac{1}{\partial_{\mathbf{s}}\mathbf{R}} \partial_{\mathbf{s}}(\mathbf{P} + \mathbf{B}^{2} + \mathbf{q}_{\mathbf{r}})$$
 (2.38)

where all variables are dimensionless and the  $\sim$ 's have been dropped for clarity in this equation.

The equation for axial momentum, equation (2.2), is converted to dimensionless variables in a similar manner. By using the transformations given by equations (2.31):

7. Iso

$$\frac{\frac{N_{o}k_{B}T_{o}}{L_{o}}\tilde{N}(\partial_{\chi}^{+}\tilde{v}\partial_{\chi}^{+}\epsilon_{2}\tilde{D}\partial_{\chi})\tilde{v} = -\frac{\frac{N_{o}k_{B}T_{o}}{L_{o}}\frac{\partial}{\partial\tilde{x}}(\tilde{P}+\tilde{q}_{\chi})$$

$$+\frac{\frac{N_{o}k_{B}T_{o}}{L_{o}}\frac{\partial}{\chi}\tilde{R}}{\frac{\chi}{2}}\partial_{\chi}\tilde{P} + \frac{R_{o}}{L_{o}}\frac{\partial}{\chi}\tilde{R}\tilde{B}^{2}_{o}\tilde{B}}{\frac{\chi}{4\pi}R_{o}}\frac{1}{\partial_{\chi}\tilde{R}}\partial_{\chi}\tilde{B}$$
(2.39)

where the term  $\frac{\partial B_r}{\partial x}$  has been neglected since it is of order  $\varepsilon_1^2$  smaller than  $\frac{\partial B_x}{\partial r}$ , and  $\hat{q}_x$  is the appropriate expression for the artificial viscosity.

27 .



where  $a_{X}^{\nu}$  is a dimensionless constant between 1.5 and 2.0 and  $\Delta x$  is the axial grid spacing used in the numerical calculations. Equation (2.39) can be simplified to:

$$N(\partial_{t} + v\partial_{x} + \varepsilon_{2}D\partial_{s}) v = -\frac{\partial}{\partial x} (P+q_{x}) + \frac{\partial_{x}R}{\partial_{s}R} \partial_{s} (P+B^{2}) \qquad (2.40)$$

where all variables are dimensionless and the  $\nu$ 's have been dropped for clarity in this equation.

### 2.3.3 Tempefature Equation

The temperature equation for either species ( $\varepsilon = e$  or i for electrons or ions) in cylindrical coordinates is given by equation (2.4):

$$n_{o} \frac{dT}{dt} - (\gamma - 1) T_{\varepsilon} \frac{dn_{o}}{dt} = \frac{(\gamma - 1)}{k_{B}} \sqrt[\gamma + \frac{\eta}{2} + \frac{\eta_{o}(T_{\varepsilon} - T_{\varepsilon})}{\tau_{eq}}$$
$$+ \delta_{e\varepsilon} \frac{(\gamma - 1)}{k_{B}} n_{o} \varepsilon_{L} + \delta_{e\varepsilon} \frac{\gamma - 1}{k_{B}} H_{jul} \qquad (2.4)$$

where  $\varepsilon' = \varepsilon$  if  $\varepsilon = i$ ,  $\varepsilon' = i$  if  $\varepsilon = \varepsilon$ ,  $\delta_{\varepsilon} = 0$  if  $\varepsilon = i$ , and  $\delta_{\varepsilon} = 1$  if  $\varepsilon = \varepsilon$ . The most difficult term to resolve in equation (2.4) is the heat flow term. In the evaluation of an expression for this term  $\varepsilon$  will be dropped in order to make notation less cumbersome. A technique by which the heat flux in the radial and x-directions may be evaluated is to consider an orthogonal coordinate system which is aligned with the magnetic field as depicted in Figure 2.1. In this figure the unit vectors  $\hat{\epsilon}_1$  and  $\hat{\epsilon}_2$  are defined as:

> $\hat{\mathbf{e}}_1 = \hat{\mathbf{x}} \cos\theta + \hat{\mathbf{r}} \sin\theta$  $\hat{\mathbf{e}}_2 = -\hat{\mathbf{x}} \sin\theta + \hat{\mathbf{r}} \cos\theta$

Then energy flow due to thermal conduction can then be written as:

$$\vec{q} = -k_{11}[\vec{\nabla}T \cdot \hat{e}_1] \hat{e}_1 - k_1[\vec{\nabla}T \cdot \hat{e}_2] \hat{e}_2$$

where  $k_{11}$  and  $k_{1}$  are the thermal conductivities parallel and perpendicular to the magnetic field respectively, and are expressed in units of [ergs/(sec eV cm)]. Other terms in the above expression can be written as:

$$\vec{\nabla}T = \frac{\partial T}{\partial r} \hat{r} + \frac{\partial T}{\partial x} \hat{x}$$
$$\vec{\nabla}T \cdot \hat{e}_1 = \frac{\partial T}{\partial r} \sin\theta + \frac{\partial T}{\partial x} \cos\theta$$
$$\vec{\nabla}T \cdot \hat{e}_2 = \frac{\partial T}{\partial r} \cos\theta - \frac{\partial T}{\partial x} \sin\theta$$

The heat flow  $\vec{Q}$  can then be d vided into a component in the radial direction,  $Q_r$ , and a component in the axial direction,  $Q_x$ .

 $\dot{\vec{q}} = \vec{q_r} \cdot \vec{r} + \vec{q_x} \cdot \vec{x}$ (2.41)

where

$$Q_{r} = -k_{11}[\partial_{r} Tsin\theta + \partial_{x} Tcos\theta]sin\theta - k_{1}[\partial_{r} Tcos\theta - \partial_{x} Tsin\theta]cos\theta$$
(2.42)

$$Q_x = -k_{11}[\partial_r Tsin\theta + \partial_x Tcos\theta]cos\theta + k_1[\partial_r Tcos\theta - \partial_x Tsin\theta]sine$$

The sine and cosine terms may be evaluated by considering Figure 2.2. If terms of order  $\varepsilon_1^2$  are neglected,

cose 
$$\frac{1}{2.43}$$
  
sine  $\frac{1}{x} = \frac{1}{x} = \frac{1}{x}$ 

Equations (2.41) to (2.43) can now be inserted into (2.32) to give an expression for  $(\overline{\nabla} \cdot \overline{Q})$  in terms of dimensionless magnetic flux variables. In this step it is convenient to divide  $(\overline{\nabla} \cdot \overline{Q})$  into two parts,  $(\overline{\nabla} \cdot \overline{Q})_{11}$ which is a term accounting for  $k_{11}$ , and  $(\overline{\nabla} \cdot \overline{Q})_{1}$  which is a term involving  $k_{1}$ .

$$\vec{\nabla} \cdot \vec{Q} = (\vec{\nabla} \cdot \vec{Q})_{11} + (\vec{\nabla} \cdot \vec{Q})_{12}$$
(2.44)

After some algebraic manipulation and dropping terms of order  $\varepsilon_1^2$ 

$$(\nabla \cdot Q)_{11} = -\frac{T_o}{L_o^2} \left\{ \frac{1}{\hat{R} \partial_g \hat{R}} \partial_{\chi} \left[ (k_{11} \partial_{\chi} \hat{T}) (\hat{R} \partial_{\chi} \hat{R}) \right] \right\}, \qquad (2.45)$$

$$(\nabla \cdot Q)_{\perp} = -\frac{T_{o}}{R_{o}^{2}} - \frac{1}{\tilde{R}_{o}^{2}} + \frac{1}{\tilde{R}_{o$$

The diffusion of a magnetic field through a plasma will lead to an ohmic heating of the electrons in the plasma. This is accounted for by the term  $H_{jul}$  in equation (2.4). An expression for this term is evaluated below.

$$H_{jul} = \overline{J} \cdot (\overline{\overline{E}} + \frac{\overline{v} \overline{x} \overline{B}}{c})$$

where J is the induced electrical current, E is the electric field, and  $\vec{v}$  is the fluid velocity. From Ohm's Law and Maxwell's equations the above expression can be written as:

$$H_{jul} = \frac{J^2}{\sigma_{\perp}} = \left(\frac{c}{4\pi}\right)^2 \frac{1}{\sigma_{\perp}} \left(\nabla_{\mathbf{x}\mathbf{B}}\right)^2$$

where  $\sigma_{1}$  is the electrical conductivity perpendicular to the magnetic field. For solenoidal fields J has a component in the  $\hat{\bullet}$  direction only. If terms of order  $\epsilon_{1}^{2}$  are neglected

$$\nabla \mathbf{x} \mathbf{B} = -\frac{\mathbf{B}_{o}}{\mathbf{R}_{o}} \frac{1}{\partial \sqrt{\mathbf{R}}} \frac{1}{\partial \sqrt{\mathbf{B}}}$$

Using the above expression for  $\nabla x B$  and the definition of  $\varepsilon_2$  given by equation (2.30):

. .

$$H_{jul} = \frac{\frac{N}{0} \frac{k}{B} \frac{T}{0} \frac{v}{0} \frac{\varepsilon}{2}}{L_{0}} \left[\frac{1}{2} \frac{1}{2} \frac{v}{2} \frac{v}{B}\right]^{2}$$
(2.46)

The various transport coefficients that appear in equation (2.4) are converted to dimensionless values through the following definitions:

$$\hat{\mathbf{k}}_{1}^{\varepsilon} = \mathbf{k}_{1}^{\varepsilon} \left( \frac{L_{o}}{\mathbf{k}_{B}^{N} \mathbf{v}_{o} \mathbf{k}_{o}^{2}} \right)$$

$$\hat{\mathbf{k}}_{11}^{\varepsilon} = \mathbf{k}_{11}^{\varepsilon} \left( \frac{1}{\mathbf{k}_{B}^{N} \mathbf{v}_{o} \mathbf{k}_{o}} \right)$$

$$\hat{\mathbf{k}}_{11}^{\varepsilon} = \mathbf{k}_{11}^{\varepsilon} \left( \frac{1}{\mathbf{k}_{B}^{N} \mathbf{v}_{o} \mathbf{k}_{o}} \right)$$

$$\hat{\mathbf{k}}_{eq}^{\varepsilon} = \tau_{eq} \left( \frac{\mathbf{v}_{o}}{L_{o}} \right)$$

$$(2.47)$$

$$\hat{\boldsymbol{\varepsilon}}_{L} = \boldsymbol{\varepsilon}_{L} \left( \frac{\boldsymbol{\varepsilon}_{L}}{\boldsymbol{k}_{B} \boldsymbol{T}_{O} \boldsymbol{v}_{O}} \right)$$

.

7

If equation (2.4) is now converted to dimensionless magnetic flux coordinates with the aid of equations (2.44) to (2.47) and (2.31) the following expression is obtained.

$$N(\partial_t + v\partial_x + \varepsilon_2 D\partial_s)T_{\varepsilon} - (\gamma - 1)T_{\varepsilon}(\partial_t + v\partial_x + \varepsilon_2 D\partial_s) N =$$

$$\frac{N(T_{\varepsilon}, -T_{\varepsilon})}{\tau_{eq}} + (\gamma - 1) N \varepsilon_{L} + (\gamma - 1) \varepsilon_{2} \left[\frac{1}{\partial_{g} R} \partial_{g} B\right]^{2} \qquad (2.48)$$

(continued on next page)

.

+ 
$$(\gamma-1) \frac{1}{R\partial_{R}} \partial_{x} [R\partial_{x}Rk_{11}^{c}\partial_{x}T_{c}] + (\gamma-1) \frac{1}{R\partial_{R}} \partial_{x} [\frac{R}{\partial_{R}}k_{1}^{c}\partial_{x}T_{c}]$$

where all variables are assumed to be dimensionless and the  $\sim$ 's have been dropped for clarity.

## 2.3.4 Magnetic Field

~

Rather than solve equations for the magnetic field as given in Section 2.1.4, it is more convenient to solve for a function closely related to  $\hat{R}(s,x,t)$ . The derivation of the appropriate equations will be found in Section 2.4.

## 2.3.5 Laser Absorption and Propagation

The dimensionless units that are defined in connection with the description of the propagation and absorption of the laser beam are:

$$\varepsilon_{L} = \left(\frac{\mathbf{k}_{B} \mathbf{T}_{o} \mathbf{v}_{o}}{\mathbf{L}_{o}}\right) \widetilde{\varepsilon}_{L}$$

$$\mathbf{k}_{a} = \left(\frac{1}{\mathbf{L}_{o}}\right) \widetilde{\mathbf{k}}_{a}$$

$$\mathbf{I}_{1} = \left(\mathbf{N}_{o} \mathbf{k}_{B} \mathbf{T}_{o} \mathbf{v}_{o}\right) \widetilde{\mathbf{I}}_{1}$$

$$\mathbf{P}_{L} = \left(\mathbf{N}_{b} \mathbf{k}_{B} \mathbf{T}_{o} \mathbf{v}_{o}\right) \widetilde{\mathbf{P}}_{L}$$

With these definitions, equations (2.8) and (2.10) transform to the following two equations

$$\varepsilon_{L} = \frac{k_{a} I_{1}}{N}$$

$$\frac{1}{c} \frac{\partial}{\partial t} P_{L} + \frac{\partial}{\partial x} P_{L} = -\overline{k} P_{1}$$
(2.49)

ŝ

4

ſ

0

where all variables are assumed to be dimensionless and the "'s have been dropped.

1

.17



#### 2.4 Shell Equations

The ion density, N, axial fluid velocity, v, electron and ion temperatures,  $T_e$  and  $T_i$ , and axial magnetic field,  $B_g$ , are now approximated as constants between adjacent discrete flux coordinates that have been chosen for the radial grid. Thus the plasma is represented by a finite number of uniform co-axial shells, each with constant magnetic flux as depicted schematically in Figure 2.3. The plasma variables are now functions x,t, and the shell number which may be indicated by an integer subscript, for example,  $N_g(x,t)$ ,  $v_g(x,t)$ , etc. Shell subscripts will be omitted for the sake of clarity. In this approximation, the resulting difference equations are accurate to second order in the differences between adjacent flux coordinates when the values of the plasma variables are taken to be the values at the shell centers.

P;

The advantages of this scheme are two-fold. First, when radial pressure balance is assumed, it is a simple matter to decouple the coupled magnetohydrodynamic equations in each shell by using the constancy of the area inside the cylindrical wall which confines the plasma. This will be described later in this section. Secondly, the program may be used for quasi-one-dimensional calculations of high temperature devices by using a few shells when accuracy in the radial direction is not essential butwhere the effects of diffuse profiles and magnetic diffusion need to be considered. It should be noted here that if a coarse numerical grid in the radial direction were used in a model that did not employ a coordinate system in which one of the coordinates was aligned with the magnetic field, numerical diffusion would be a serious problem.

The shell equations or the magnetohydrodynamic equations

for each shell will be derived below. Since dimensionless variables are used exclusively in this subsection the  $\sim$ 's overball variables are dropped. In these equations,  $\nabla Q$  denotes the difference between the values of a quantity Q evaluated at the upper and lower boundaries of a shell. Values on a boundary can be calculated as linear interpolations or averages of adjacent shell values. The symbol  $\langle Q \rangle$  represents the average value of Q inside the shell when Q is not a simple plasma variable.

2.4.1 Continuity Equation

Multiplication of equation (2.34) by two yields:

$$\partial_{\mu} (N 2R\partial_{R}R) + \partial_{\mu} (v N2R\partial_{R}R) + \partial_{\mu} (\epsilon_{2}DN2R\partial_{R}R) = 0$$
 (2.50)

This equation is now integrated over s from  $s_L$  to  $s_u$  where  $s_L$  is defined as the lower boundary of the s'th shell and  $s_u$  is the upper boundary of the s'th shell. Since N and v are assumed constant over the integration the first term becomes:

$$\frac{\partial}{\partial t} \left[ N \int_{s}^{s} 2R \partial_{s} R ds \right] = \frac{\partial}{\partial t} \left[ N \int_{s}^{s} \partial_{s} (R^{2}) ds \right] = \frac{\partial}{\partial t} (NA)$$

where

$$A \equiv \int_{a}^{a} \partial_{a}(R^{2}) ds = R^{2} |_{a}^{a}$$

is the area of the shell being integrated over. Similarly the second term in equation (2.50) becomes:

$$\partial_{\mathbf{x}} \mathbf{N} \mathbf{v} \begin{bmatrix} \mathbf{s} \\ \mathbf{u} \\ \mathbf{s} \end{bmatrix} = \partial_{\mathbf{x}} (\mathbf{N} \mathbf{A} \mathbf{v})$$

A,

With the use of equation (2.27) the last term in equation (2.50) can be written as:

$$\partial_{\mathbf{s}}(\epsilon_2 \mathbf{D} \mathbf{N} 2\mathbf{R} \partial_{\mathbf{s}} \mathbf{R}) = \partial_{\mathbf{s}}(\frac{\epsilon_2 \mathbf{D} \mathbf{N}}{\mathbf{B}})$$

thus,

$$\int_{a}^{a} \partial_{g} (\epsilon_{2} DN2R \partial_{g} R) ds = \left(\frac{\epsilon_{2} DN}{B}\right) |_{u}^{s}$$

The following definitions are made.

$$F_{g} = \left(\frac{\varepsilon_{2}^{DN}}{B}\right)$$
  
s+1/2

and

$$\Delta F = F_s - F_{s-1}$$

where a subscript s denotes the s'th shell and the subscript (s + 1/2) we denotes the boundary between the s'th and the (s+1)'th shell.

The continuity equation can now be expressed as:

$$\frac{\partial}{\partial t}$$
 (NA) +  $\frac{\partial}{\partial x}$  (NAv) +  $\Delta F = 0$  (2.51)

It is noted that since the integration has been performed the continuity equation has become a function of the area A of each shell rather than s.' The third term can be physically interpreted as a particle flux



across the magnetic field lines.

### 2.4.2 Momentum Equation

If equation (2.40) is multiplied by 2R? R and the result added to 2v times equation (2.34), the following equation is obtained:

Ð

 $\cdot 2NR\partial_{\mathbf{R}} [\partial_{\mathbf{t}} + v\partial_{\mathbf{x}} + \varepsilon_{2}D\partial_{\mathbf{s}}]v + 2v[\partial_{\mathbf{t}}(NR\partial_{\mathbf{s}}R) + \partial_{\mathbf{x}}(vNR\partial_{\mathbf{s}}R)$ 

+  $\partial_{\mathbf{s}}(\mathbf{c}_{2} \text{DNR} \partial_{\mathbf{s}} \mathbf{R})$ ] =  $\nabla_{\mathbf{s}} 2\mathbf{R} \partial_{\mathbf{s}} \mathbf{R} \partial_{\mathbf{x}}(\mathbf{P} + \mathbf{q}_{x}) + 2\mathbf{R} \partial_{\mathbf{x}} \mathbf{R} \partial_{\mathbf{s}}(\mathbf{P} + \mathbf{B}^{2})$ 

The above equation can be written as:

$$\frac{\partial}{\partial t} \left[ Nv \frac{\partial R^2}{\partial s} \right] + \frac{\partial}{\partial x} \left[ Nv^2 \frac{\partial R^2}{\partial s} \right] + \frac{\partial}{\partial s} \left[ \epsilon_2 D \frac{NV}{B} \right]$$
$$= \frac{\partial R^2}{\partial s} \frac{\partial}{\partial x} (P+q_x) + 2(R\partial_z P) \frac{\partial}{\partial s} (P+B^2)$$

If the above equation is now integrated over a shell s assuming that N, v, P, and B are constant over the shell it is found that:

$$\frac{\partial}{\partial t} [NAv] + \frac{\partial}{\partial x} [NAv^{2}] + \Delta(\epsilon_{2} \frac{DNv}{B}) = -A \frac{\partial}{\partial x} [P+q_{x}] + \langle \partial_{x} (R^{2}) \rangle \Delta(P+B^{2})$$

If the derivatives on the left hand side are expanded and use is made of equation (2.51) the following equation is obtained. ,

$$\frac{\partial \mathbf{v}}{\partial t} = -\mathbf{v} \frac{\partial}{\partial \mathbf{x}} \mathbf{v} - \frac{\left[\Delta \mathbf{G} - \mathbf{v} \Delta \mathbf{F}\right]}{\mathbf{N} \mathbf{A}} - \frac{1}{\mathbf{N}} \frac{\partial}{\partial \mathbf{x}} (\mathbf{P} + \mathbf{q}_{\mathbf{x}}) + \frac{\langle \partial_{\mathbf{x}} \mathbf{R}^{\mathbf{z}} \rangle}{\mathbf{N} \mathbf{A}} \Delta (\mathbf{P} + \mathbf{B}^{2})$$
(2.52)

where 
$$G_{\rm s} \equiv \left(\frac{\epsilon_2 D N v}{B}\right)_{\rm s+1/2} = v_{\rm s+1/2} F_{\rm s}$$
 (2.53)

### 2.4.3 Temperature Equation

Equation (2.48) can be expressed as:

$$N^{\gamma}\left(\frac{\partial}{\partial t} + v \frac{\partial}{\partial x} + \varepsilon_{2} D \frac{\partial}{\partial s}\right) (TN^{-(\gamma-1)}) = \frac{N(T_{\varepsilon}, -T_{\varepsilon})}{\tau_{eq}} + (\gamma-1)N \varepsilon_{1}$$

+ 
$$(\gamma-1)\varepsilon_{2}\left[\frac{1}{\partial R}\partial_{s}B\right]^{2}$$
 +  $(\gamma-1)\frac{1}{R\partial_{s}R}\partial_{x}\left[R\partial_{s}Rk_{11}^{\varepsilon}\partial_{x}T_{\varepsilon}\right]$ 

$$\stackrel{\sim}{+} (\gamma-1) \frac{1}{R\partial_{\mathbf{R}} R} \partial_{\mathbf{s}} \left[ \frac{R}{\partial_{\mathbf{R}} R} \mathbf{k}_{\perp}^{\mathbf{c}} \partial_{\mathbf{s}} \mathbf{T}_{\mathbf{c}} \right]$$

For the remainder of this subsection the right hand side of the above equation will be referred to as RHS. The above equation is multiplied by  $2N^{(1-\gamma)}R\partial_{s}R$  to obtain the following:

$$2N(R\partial_{\mathbf{s}}R)\left[\frac{\partial}{\partial t} + v \frac{\partial}{\partial x} + \epsilon_2 D \frac{\partial}{\partial s}\right](TN^{1-\gamma}) = 2N^{1-\gamma} R\partial_{\mathbf{s}}R \cdot RHS$$

From the continuity equation:

÷,

$$\Gamma_{\varepsilon} N^{1-\gamma} [\partial_{t} (2NR\partial_{s}R) + \partial_{x} (2vNR\partial_{s}R) + \partial_{s} (2\varepsilon_{2}DNR\partial_{s}R)] = 0$$

By adding the last two equations the following equation can be obtained.

J

$$\frac{\partial}{\partial t} \left[ \partial_{\mathbf{s}} R^{2} T_{\varepsilon} N^{(2-\gamma)} \right] + \frac{\partial}{\partial x} \left[ \partial_{\mathbf{s}} R^{2} N^{(2-\gamma)} v T_{\varepsilon} \right] + \frac{\partial}{\partial s} \left[ \frac{\varepsilon_{2} D N^{(2-\gamma)}}{B} T_{\varepsilon} \right]$$
$$- \partial_{\mathbf{s}} R^{2} N^{(1-\gamma)} \cdot RHS$$

If the above equation is integrated over a shell s, noting that N, v,  $T_{\varepsilon}$ , and B are constants over a shell the following equation is obtained.

$$\frac{\partial}{\partial t} \left[ AN^{(2-\gamma)}T_{\varepsilon} \right] + \frac{\partial}{\partial x} \left[ AN^{(2-\gamma)}vT_{\varepsilon} \right] + \Delta \left[ \frac{\varepsilon_2 DN^{(2-\gamma)}T_{\alpha}}{B} \right]$$
$$= \int_{L}^{s} u \partial_{s} R^2 N^{(1-\gamma)} \cdot (RHS) ds$$

The variable I is defined as:

1.

$$I_{s} = \frac{\epsilon_{2} D N}{B} T_{\epsilon} N^{(1-\gamma)} |_{s+1/2} = (N^{(1-\gamma)} T_{\epsilon}) |_{s+1/2} F_{s} (2.54)$$

Now, by expanding derivatives the above equation can be written as:

$$N^{(1-\gamma)} T_{\varepsilon}[\partial_{t}(NA) + \partial_{x}(NAV)] + NA[\partial_{t} + v\partial_{x}](N^{(1-\gamma)} T_{\varepsilon})$$
$$+ \Delta I = \int_{s_{T}}^{s_{u}} \partial_{s}R^{2} N^{(1-\gamma)} (RHS) ds$$

A value for the first term in brackets may be obtained from equation (2.51) and is equal to  $-\Delta F$ . By using this expression and dividing the above equations by  $AN^{(1-\gamma)}$  and evaluating RHS, the following equation is  $Q_{\lambda}$ obtained:

$$H[\partial_{t} + v\partial_{x}] T_{\epsilon} - (\gamma-1) T_{\epsilon}[\partial_{t} + v\partial_{x}]H + \frac{1}{MA} [H^{\gamma} \Delta I - HT_{\epsilon} \Delta F]$$

$$= \frac{N(T_{\epsilon}, - T_{\epsilon})}{T_{\epsilon q}} + (\gamma-1) N \epsilon_{L} + (\gamma-1) \epsilon_{2} [\frac{1}{\partial_{s} R} - \partial_{s} B]^{2} \qquad (2.55)$$

۰,

$$+ \frac{(\gamma-1)}{A} \partial_{\mathbf{x}} [\mathbf{A} \mathbf{k}_{11}^{\varepsilon} \partial_{\mathbf{x}} \mathbf{T}_{\varepsilon}] + \frac{(\gamma-1)}{A} \Delta [\frac{2R}{\partial_{\mathbf{x}} R} \mathbf{k}_{1}^{\varepsilon} \partial_{\mathbf{x}} \mathbf{T}_{\varepsilon}]$$

It is convenient to rewrite this equation as:

$$\partial_{t} T_{\varepsilon} = \frac{(\gamma-1) T_{\varepsilon} \partial_{t} N}{N} + H_{\varepsilon}^{+} + \frac{(\gamma-1)}{NA} \Delta [2R k^{\varepsilon} \frac{1}{\partial_{s} R} \partial_{s} T_{\varepsilon}] + \frac{(\gamma-1)}{NA} \partial_{x} [A k_{11}^{\varepsilon} \partial_{x} T_{\varepsilon}]$$
(2.56)

where

$$H_{\varepsilon}^{+} = -v \partial_{x} T_{\varepsilon} + (\gamma - 1) T_{\varepsilon} v \frac{\partial_{x} N}{N} - \frac{1}{NA} [N^{(\gamma - 1)} \Delta I - T_{\varepsilon} \Delta F] + (\gamma - 1) \varepsilon_{L}$$

$$+ \frac{(T_{\varepsilon} - T_{\varepsilon})}{\tau_{eq}} + \frac{(\gamma - 1) \varepsilon_2}{N} \left(\frac{1}{\partial_s R} \partial_s B\right)^2 \qquad (2.57)$$

The third term on the right hand side of equation (2.56) represents diffusion of heat across the magnetic field while the fourth term represents heat flow along the field lines.

2.4.4 Closure of the System of Equations

Equations (2.51), (2.52), and (2.56) describe four dynamic

equations for five plasma variables, namely N, v,  $T_e$ ,  $T_i$ , and A. To complete the set of equations, an equation for A must be derived from the equations describing the radial dynamics of the plasma. That is, an equation for  $\partial A/\partial t$  is required. If the time scale over which effects occur is much longer than that required for a magnetosonic wave to travel across the plasma, then the assumption that pressure balance exists is valid. In this case, any pressure imbalance is communicated sufficiently rapidly across the plasma column so as to insure that pressure balance is essentially being maintained. This assumption would apply to physical situations like that of a long reactor plasma. In the case where a plasma is rapidly heated by a laser, causing both radial and axial shocks, full radial dynamics must be included. The computer program has been developed so that two options exist, namely that of using full radial dynamics or that of pressure balance. The assumption of radial pressure balance, where it is valid, leads to considerable savings in CPU time.

## 4 2.4.4a Radial Pressure Balance

'In this subsection radial pressure balance is assumed in the derivation for an expression for  $\partial A/\partial t$ . For this case equation (2.38) for  $v_r$  is not used. The assumption of radial pressure balance is expressed as:

$$\frac{\partial}{\partial a}$$
 (P + B<sup>2</sup>) = 0 or P + B<sup>2</sup> =  $\pi$ (x,t) (2.58)

where  $\pi(\mathbf{x}, \mathbf{t})$  is the total kinetic plus magnetic pressure and is not a function of s. If  $\phi$  is the magnetic flux in a shell, then  $B = \phi/A$ where  $\phi$  is a constant in each shell. By using this expression for B in equation (2.58), and differentiating with respect to time yields:

$$\partial_t P = 2 \frac{\phi^2}{\Lambda^3} \partial_t \Lambda = \partial_t \pi(x,t)$$

.

OT

.

$$A(\partial_t P) = 2 B^2(\partial_t A) + A(\partial_t \pi)$$
(2.59)

~

)

Now the temperature equation can be expressed as:

$$AN(\partial_t T_{\epsilon}) = (\gamma - 1) T_{\epsilon} A(\partial_t N) + H_{\epsilon}$$

where

,

$$H_{\varepsilon} = H_{\varepsilon}^{+} + \frac{(\gamma-1)}{NA} \Delta [2R \ k_{\perp}^{\varepsilon} \ \frac{1}{\partial R} \ \partial_{s} T_{\varepsilon}] + \frac{(\gamma-1)}{NA} \partial_{x} [A \ k_{\perp}^{\varepsilon} \ \partial_{x} T_{\varepsilon}]$$

•

.

.

,

Define,

$$H \equiv H_e + H_i$$
 and  $T = T_e + T_i$ 

and add the two temperature equation to obtain:

$$AN(\partial_t T) = (\gamma - 1) T A(\partial_t N) + H$$

Now,

$$\partial_t(P) = \partial_t(NT) = N \partial_t T + T \partial_t N$$

•

Combining this with the above equation yields:

.

$$A(\partial_{\mu}P) - \gamma TA(\partial_{\mu}N) = H$$
 (2.60)

The continuity equation is used to define C as:

$$C \equiv \partial_{\mu}(NA) = - [\partial_{\mu}(NAv) + \Delta F]$$

or

$$9^N = [C - N 9^A]/A$$

Substituting this into equation (2.60) yields:

$$\mathbf{A}(\partial_{\mathbf{P}}\mathbf{P}) = \mathbf{H} + \gamma \mathbf{T}[\mathbf{C} - \mathbf{N}(\partial_{\mathbf{A}}\mathbf{A})]$$
(2.61)

Comparing equations (2.59) and (2.61) yields an expression for  $\partial_{\mu}A$ .

$$(\partial_{t}A) = \left[\frac{H + \gamma TC}{2B^{2} + \gamma P}\right] - \frac{A \partial_{t}^{\pi}}{2B^{2} + \gamma P} \qquad (2.62)$$

Before equation (2.62) can be used an expression for  $\beta_{t}\pi$  must be derived. This may be done by noting that the total cross sectional area of the solenoid is a constant. If equation (2.62) is summed over all shells.

$$\sum_{All shells}^{c} (\partial_{t}^{A}) = 0$$

Thus,

$$(\partial_t \pi) = \sum_{All Shells} \frac{H + \gamma TC}{2B^2 + \gamma P} / \sum_{All Shells} \frac{A}{2B^2 + \gamma P}$$
 (2.63)

### 2.4.4b Full Radial Dynamics

In this sub-section an expression for  $\frac{\partial A}{\partial t}$  is derived by using the appropriate equations to incorporate full radial dynamics into the model. First an expression for  $\frac{\partial}{\partial t}$  (R<sup>2</sup>) is found. It is then a trivial matter to find an expression for  $\frac{\partial A}{\partial t}$ . In order to obtain an equation for R<sup>2</sup>, equation (2.19) is multiplied by 2R. The resulting 'equation can be written as:

$$\partial_{t}(\mathbf{R}^{2}) + \mathbf{y} \partial_{\mathbf{x}}(\mathbf{R}^{2}) + 2\mathbf{R} \varepsilon_{2} D \partial_{s}\mathbf{R} = 2\mathbf{R}\mathbf{v}_{r}$$

The third term in the above equation can be reduced by expanding D through the use of equations (2.29) and (2.27).

$$D = R \frac{\partial B}{\partial r} = \frac{R}{\partial_{g}R} \partial_{g}B = \frac{R}{\partial_{g}R} \partial_{g} (\frac{1}{2R\partial_{g}R}) = \frac{R}{\partial_{g}R} \partial_{g} (\frac{1}{\partial_{g}R^{2}})$$

The third term therefore becomes:

$$2R^{2} \epsilon_{2} \partial_{\mathfrak{s}} \left(\frac{1}{\partial_{\mathfrak{s}}R^{2}}\right) = \frac{-2R^{2} \epsilon_{2} \partial_{\mathfrak{s}\mathfrak{s}}(R^{2})}{(2R \partial_{\mathfrak{s}}R)^{2}} = -2R^{2}B^{2} \epsilon_{2} \partial_{\mathfrak{s}\mathfrak{s}}(R^{2})$$

The equation for  $R^2$  can now be written as:

$$\frac{\partial}{\partial t} (R^2) = -v \frac{\partial}{\partial x} (R^2) + 2Rv_r + 2 \varepsilon_2 R^2 B^2 \partial_{ss} (R^2) \qquad (2.64)$$

The solution to equation (2.64) requires an expression for  $Rv_r$ . An equation for  $Rv_r$  can be found as follows. Note that

$$\frac{dR}{dt} = v_r$$

Thus,

$$N \frac{d}{dt} (Rv_r) = Nv_r^2 + NR \frac{dv_r}{dt}$$

An expression for  $\frac{dv_r}{dt}$  is given by equation (2.38) so that the above expression can be written as:

$$N[\partial_{t} + v\partial_{x} + \varepsilon_{2}D\partial_{s}] (Rv_{r}) = \frac{N(Rv_{r})^{2}}{R^{2}} - \frac{2B(r)}{\varepsilon_{1}}(P + B^{2} + q_{r})$$
(2.65)

VE I

An expression for D is required for equation (2.65). By using equations (2.29) and (2.27), D can be written as:

$$D = R \frac{\partial B}{\partial r} = \frac{R}{\partial R} \partial_{s} B = \frac{2R^{2}}{2R\partial_{R}R} \partial_{s} B = 2R^{2} B \partial_{s} B. \qquad (2.66)$$

Thus equation (2.65) can be used to find  $\mathbb{R}v_r$ , and equation (2.64) can be used to find  $\mathbb{R}^2(\mathbf{s},\mathbf{x},\mathbf{t})$ . In a shell model  $\frac{\partial}{\partial t} (\mathbb{R}^2)_{\mathbf{s}+1/2}$  is found where  $(\mathbb{R}^2)_{\mathbf{s}+1/2}$  is the outer radius of a shell. Then the desired quantity  $\frac{\partial A}{\partial t}$  can be found.

$$\frac{\partial}{\partial t} A = \frac{\partial}{\partial t} (\mathbf{R}^2)_{s+1/2} - \frac{\partial}{\partial t} (\mathbf{R}^2)_{s-v_{1/2}}$$
(2.67)

5

## 2.4.5 Summary of Shell Equations

In this sub-section the shell equations are summarized and written in a form suggestive of how they might be solved through numerical techniques. In this sub-section the notation  $\hat{Q}$  and  $\hat{Q}'$  will be used for  $\partial_t \hat{Q}$  and  $\partial_x \hat{Q}$  respectively. The first step is a solution for the area,  $A_1$ , of a shell.

If radial pressure balance is assumed equation (2.62) is used.

$$A = \frac{H + \gamma TC - A \pi (x, t)}{2B^2 + \gamma P}$$
(2.68)

where

and C E

$$T = \begin{bmatrix} \frac{H + \gamma TC^{2}}{2B^{2} + \gamma P} & 12B^{2} + \gamma P \\ Shells & Shells & Shells & (2.69) \\ (2.69) & (2.69) & (2.70) \\ \end{array}$$

If full radial diamics are used an equation for A comes from W defined on shell bowndaries where

(2.71)

From equation (2.67)

 $w \equiv R^2$ 

$$A = W_{a+1/2} - W_{a-1/2}$$

and from equation (2.64)

 $U \equiv R v_r$ 

$$W = -v W' + 2U + 2\varepsilon_2 W B^2 \frac{a^2}{a} W$$
 (2.73)

(2.72)

(2.74)

where

expressed as:

$$U = -v U' - \epsilon_2 D \partial_{\mu} U + \frac{U^2}{W} - \frac{2BW}{\epsilon_1^2 N} \partial_{\mu} \left[P + B \right] \left[P + B \right] \left[2.75\right]$$

Once A has be found, W can be found through solution of the following equation, which comes from equations (2.51) and (2.70):

$$\dot{N} = \frac{(C - NA)}{A}$$
(2.76)

Now the temperature equations can be solved. With the new notation equations (2.56) and (2.57) can be written as:

$$\dot{T}_{\varepsilon} = H_{\varepsilon}^{*} + \theta_{11} + (\gamma - 1) T_{\varepsilon} \dot{N}/N \qquad (2.77)_{\varepsilon}$$

where

\$

$$H_{\varepsilon}^{\bullet} \equiv -v T_{\varepsilon}^{\dagger} + (\gamma - 1) T_{\varepsilon} v N^{\varepsilon} / N - \frac{1}{NA} [N^{(\gamma - 1)} \Delta I - T_{\varepsilon} \Delta F]$$

$$+ (\gamma - 1) \varepsilon_{L}^{\dagger} + \frac{(T_{\varepsilon} + T_{\varepsilon})}{\tau_{eq}} + \frac{(\gamma - 1)\varepsilon_{2}}{N} (\frac{1}{\partial_{s}R} \partial_{s}B)^{2}$$

$$+ \frac{(\gamma - 1)}{NA} \Delta [2R k_{L}^{\varepsilon} \frac{1}{\partial_{s}R} \partial_{s} T_{\varepsilon}]$$
(2.78)

and

$$\theta_{11} = \frac{(\gamma - 1)}{NA} \left[ A k_{11}^{\varepsilon} T_{\varepsilon}^{*} \right]'$$
(2.79)

.

Finally, the axial velocity can be found. Equation (2.52) can be expressed as:

•,

$$v = -v v' - \frac{1}{NA} [\Delta G - v \Delta F] - \frac{1}{N} [P + q_x]' + \frac{\langle W' \rangle}{NA} \Delta (P + B^2)$$
  
(2.80)

This set of equations requires the evaluation of  $F_s$ ,  $I_s$ , and  $G_s$  on shell boundaries. From equation (2.66):

 $\frac{D}{B} = 2R^2 \partial_s B$ 

Substituting this into the definition of F:

 $\mathbf{F}_{\mathbf{s}} = 2\mathbf{R}^2 \epsilon_2 \mathbf{N} \partial_{\mathbf{s}} \mathbf{B}$ (2.81)

G and I can be evaluated from:

, ,

v<sub>s+1/2</sub> F<sub>s</sub> (2.82) 0  $\int_{\mathbf{N}} (1-\gamma) \mathbf{T}_{c} = \int_{c} \mathbf{s} + 1/2 \mathbf{F}_{s}$ 

A computer model of a plasma in a solenoidal magnetic

field can be made through the numerical solution of the coupled equations presented in this sub-section.



### 2.5 Numerical Solution

An computer simulation based on the shell equations of the previous sub-section has been developed. The main computer routine of this simulation solves the appropriate difference equations which have been written in dimensionless units. As the solution is advanced in time the internal variables of the routine are output on magnetic tape or disk. (This is essentially the only output of the main routine. Subsequently, post-processor routines read the output of the main routine and calculate values for various plasma parameters in a more familiar unit system. Different post-processor routines have been developed to print out values for plasma parameters and perform checks on the validity of the simulation. Two checks that are done to test the validity of a simulation are conservation of energy and conservation of mass. These will be discussed later in this section. In addition to these post-processor routines two graphics routines have been developed. One of these routines plots various two-dimensional plots of plasma parameters as functions of x, r, or t. The other makes three dimensional plots of various plasma parameters as functions of x and r at given times. Examples of output from these routines will be presented in chapters there and four.

An attempt has been made to make a package of programs that will allow a user with a minimal familiarity with the programming details to solve a variety of problems with little effort. Various options have been built into the routines and desired options are specified in an input file. A more detailed description of the routines and some details of the numerical techniques can be found in Appendix A.
The shell equations have been solved using a two-step second order Euler method. In the first step, the solution at t is used to calculate temporary values of the variables at  $t + \Delta t/2$ . These temporary values are used to calculate the coefficients and spatial derivatives in the equations and the solution is then advanced from t to t +  $\Delta t$  in the second step.

Central differences are used to calculate all spatial derivatives. The continuity and axial acceleration equations are advanced explicitly. When radial pressure balance is assumed, the equation for  $\frac{\partial A}{\partial t}$  is also advanced explicitly. However the decoupling procedure used to find  $\frac{\partial A}{\partial t}$  is similar to an implicit scheme in the sense that changes in the central pressure due to heating are felt instantly in all the shells. When full radial dynamics are taken into account, the radial acceleration equation is solved explicitly but the shell boundary equation for W, equation (2.73), is solved implicitly. This is necessary since large magnetic diffusion in regions of low electron temperature severely limits the timestep of an explicit scheme.

For each timestep, the temperature equations are advanced in two stages. In the first step, paralial hear conduction is turned off and the equations are advanced using implicit differences for the perpendicular conduction terms. The difference between the new and old values of T<sub>c</sub> is taken to be the contribution to  $\frac{\partial T_c}{\partial t}$   $\Delta t$  from perpendicular conduction. This quantity is substituted for the perpendicular terms in the second stage in which the equations are advanced using implicit differences for the parallel conduction terms.

The boundary conditions used at the outer wall are  $V_{T} = 0$ ,  $\frac{\partial T_{E}}{\partial s} = 0$ . At the open ends, the plasma variables are assumed to vary linearly with x. The exception to this rule is that  $\frac{\partial P}{\partial x}$  is set equal to a constant in order to get the flow started at t = 0. The results are insensitive to the particular value of  $\frac{\partial P}{\partial x}$  chosen.

The code was tested in several ways. Axial flow without heat conduction was compared to analytic similarity solutions from a one-shell model<sup>18</sup>.Axial heat conduction was tested by freezing the plasma (v=0, N=N<sub>0</sub>) and assuming a constant conduction coefficent for which an analytic solution is easily obtainable from reasonable initial conditions. In both cases, excellent agreement was obtained. The full radial dynamics under axially uniform laser heating was compared with results from a one-dimensional code described by Burnett and Offenberger<sup>37</sup>. Essentially identical results were obtained.

A check on conservation of energy and mass is made for each simulation. This is done by calculating total internal energy or mass of the plasma at various times and comparing changes in internal energy or mass with the calculated energy or mass flow through the solenoid ends. Typically the simulations conserve energy to better than 5% and mass to better than 1%.

A more detailed description of numerical techniques are presented in Appendix A.



¥

Figure 2.1 Orthogonal coordinate system aligned with magnetic field.



Figure 2.2 Angle of magnetic field lines relative to solenoid axis in magnetic Trux coordinate system



#### CHAPTER 3

#### Computer Simulations of Plasma in a Selencidal Magnetic Field

In this chapter some results from the computer program described in the previous chapter are presented. The versatility of the code is demonstrated through the simulation of short solenoids (5 cm length) in Section 3.1, intermediate length solenoids (1 m length) in Section 3.2, and long solenoids (1 to length) in Section 3.3. A comparison of a simulation in which full radial dynamics have been included to a simulation that assumes radial pressure balance is presented in Section 3.1. Radial pressure balance is assumed for simulations in Section 3.2 and Section 3.3. The parameters used for the simulations in Section 3.2 were chosen to test the feasibility of creating plasms conditions ideally suited for the beat frequency mixing of antiparallel laser beams in a laser heated solenoid. The required plasms conditions will be discussed in Chapter 6 where the results from Section 3.2 will be referred to. Results of Section 3.3 demonstrate the ability of this type of code to simulate the magnetohydrodynamic behaviour of a 1 km solenoid reactor.

### 3.1 Laser Beated Solenoid /

In the first simulation, a preionised hydrogen plasma of radius 1.5 cm is heated by a  $CO_2$  laser with Taussian cross-section and half-power radius 1.77 mm. The laser power rises linearly from 0 at t=0 to 100 MW at t=10 nsec and then remains constant. The initial plasma variables are:

 $n = 2 \times 10^{18} \text{ cm}^{-3}$ 

 $T_e = T_i = 1 eV$ 

B = 100 kgauss

The parameters were chosen to be the same as those used in an experimental and theoretical study by Scudder et  $al^{16}$  with the important difference that their plasma was formed by the breakdown of neutral gas by the laser.

The code was run with 60 axial points and 30 shells until t=0.5 µsec was reached. This run required 20 minutes of CPU time on an Amdahl 470 V/6. The size of each timestep was limited by the hydrodynamic stability condition,

 $\Delta t < \Delta r/V$ 

where  $\Delta r$  is the shell thickness and  $V_{ma}$  is the magneto-acoustic velocity. Three dimensional plots of  $T_e$ ,  $T_i$ ,  $n_o$ ,  $v_r$ ,  $v_x$  and  $B_x$  are presented in Figures 3.1 to 3.6. In each plot, the lines joining the plotted values

- A 😫

57

Ł

of the plasms variables are drawn at a constant mutal or radial position so that linear interpolation between shell values was necessary.

The electron temperature attained, 40-45 eV, is similar to those calculated by Scudder <u>et</u> <u>al</u>.<sup>16</sup> in a two-dimensional Lagrangian simulation. The ion temperature distribution is almost identical to that of the electrons since the collision time is short compared to the heating\_time at these densities. The temperature plateau behind the bleaching front is apparently due to adiabatic cooling which follows expansion of the heated plasma.

The most significant difference between this simulation and that of Scudder et al. is the nature of the plasma dynamics at the bleaching front. They have found that after t=140 nsec the laser driven shock on the axis attenuates and has disappeared at t=280 nsec. Figure 3.3 indicates, however, the presence of a strong shock at the bleaching front at t=500 nsec. (The jagged edge on the curved shock front is a plotting routine effect.) The shock wave calculated by the shell code has been found to propagate more as a travelling wave rather than a damped wave. The discrepancy between the two codes may be due to the different methods of calculation of the laser beam propagation. Scudder et al. assumed that the propagation was parallel to the axis so that the rays in the outer edges are always in a region of higher absorption than those in the center. The effect was that the beam became progressively narrower as it propagated through the plasma. Eventually, the heated region at the bleaching front may have been too slender to drive a forward moving shock wave. On the other hand, since the shell code assumes a constant

beam radius, the heated region always has the same width. To got a more accurate picture of the dynamics at the bleaching fromt, a better laser model will have to be developed.

The radial velocity distribution is plotted in Figure 3.4. The interesting points are that (1) the velocity is small compared to the magneto-acoustic velocity so that not much energy goes into radial motion and (2) the radial velocity quickly becomes negligible after the passage of the region of fast heating indicating the return to radial equilibrium. These observations suggest that similar densities and temperatures will be obtained if the plasma is assumed to be in radial pressure balance at all times.

The computer simulation was repeated with all of the same input parameters as above, except that radial pressure balance was assumed and the number of shells was reduced to 15. Three dimensional photos of  $T_e$ ,  $T_i$ ,  $n_{c'}$  v and  $B_z$  are presented in Figures 3.7 to 3.11. The temperatures attained are very similar in the two runs. The densities are also similar, especially in the quiescent region between x = 0 and x=3 cm. The main differences with pressure balance are that (1) the bleaching wave propagates about 10% farther in 500 nsec and (2) the amplitude of the shock is smaller.

The advantage of assuming pressure balance is the significant reduction of computing costs. The run with pressure balance required about 6 minutes of CPU time compared to 20 minutes for full radial dynamics. This reduction of computing time is possible because the timestep is not restricted by the hydrodynamic stability condition when radial pressure balance is assumed. The criterion used for limiting the timestep in this

run was that the temperature should not rise by more than 30% at any point in a single step. The assumption of radial pressure balance leads to even greater savings in modelling longer devices and when a laser beam bleaching front is not pushing a shock wave. 3.2 Production of Long Scale Longth Places for Jean Tremency Mining

In Chapter 5 a theory for the best-dreamy mixing of antiperallel laser beams in a homogeneous plasms will be presented. In Chapter 6 the possibility of using this process to create a useful amplifier of infrared radiation will be examined. The possibility of making such an amplifier is dependent on the feasibility of creating a hot plasms with a uniform density and temperature in the axial direction. In this section computer simulations are used to examine the feasibility of creating such a plasms through the inverse bremsstrahlung absorption of laser radiation in a plasma radially confined by a solenoidal magnetic field.

Results from two simulations are presented. The input parmeters for both simulations are identical except for initial plasma densities. In both cases a preionized hydrogen plasma is assumed to be contained by a solenoid of radius 1.5 cm, length 1 m, and is heated by a  $CO_2$  laser with Gaussian cross-section and half-power radius of 1.77 m. The laser power rises linearly from 0 at t=0 to 1 GW at t=10 ns and then remains constant. Initially the temperature, density, and magnetic field are uniform. The temperature is assumed to be 1 eV and the magnetic field strength is assumed to be 100 kG. The computational grid consists of 30 axial points and 15 shells. Under these conditions the code requires approximately 1 minute of CPU time on the Amdahl 470 V/6 computer to simulate the plasma for 1 usec.

In the first simulation the initial plasma density was  $7.8 \times 10^{16}$  cm<sup>-3</sup>. Figures 3.12 to 3.14 are three dimensional plots of  $T_e$ ,  $T_i$ , and  $n_o$  at t=0.5 µsec. Figure 3.15 is a plot of the axial values of  $T_e$ ,  $T_i$  and  $n_o$  as a function of x. These figures show that the axial

values of the plasma temperature and density show little variation down most of the column length. For about 80% of the selensit length the plasma density is almost exactly 7.5x10<sup>16</sup> cm.<sup>-3</sup> while electron temperature varies from about 95 eV to 100 eV and the ion temperature varies from about 67 eV to 72 eV. It will be shown in Chapter 6 that such plasma conditions are ideally suited to the beat frequency mixing of antiparallel electromagnetic beams with wavelengths of 9.6 µm and 10.6 µm.

The uniform please density and temperatures in the axial direction can be attributed to two main factors. First, the high laser powers, and low initial plasma densities lead to high laser blenching velocities. In the above simulation the laser has bleached its way through the plasma column in less than 40 ns. This leads to a relatively " uniform laser heating rate down the column axis. The second factor leading to uniform axial profiles is the large electron thermal conductivity in the axial direction which tends to smooth out any nonuniformities.

In the second simulation the initial plasma density was chosen to be  $3 \times 10^{17}$  cm<sup>-3</sup>. This is a much higher initial density than that of the previous run. Consequently a much lower bleaching velocity is expected. It was found that it took about 300 ns for the laser beam to bleach its way through the solenoid. Because of the slower bleaching velocity, the plasma must be heated for a longer period of time for its axial parameters to become uniform. Figures 3.16 to 3.18 are three dimensional plots of T<sub>a</sub>, T<sub>i</sub>, and n<sub>a</sub> after 1 usec of laser heating. Ligure 3.19 is a plot of the axial values of  $T_a$ ,  $T_j$ , and no as a function of x. These figures show that the plasme has a density of between  $2.4 \times 10^{17}$  and  $2.5 \times 10^{17}$  cm<sup>-3</sup>, an electron temperature of between 120 and 180 eV and an ion temperature of between 100 and 140 eV

over an 80 cm length of the 1 meter solenoid. While these parameters are not as uniform as those in the previous simulation it will be shown in Chapter 6 that this plasma would serve very well as the mixing a medium for antiperallel radiations with difference frequencies close to the ion-acoustic frequency.

The results of these simulations suggest that a laser heated plasma, radially confined by a solenoidal magnetic field, will have a uniform axial density and temperature profile at a time t after the laser is turned on if:

$$\frac{L}{v_{b1}} \ll t \ll \frac{L}{v_{ac}}$$

where  $L_0$  is the solenoid length,  $v_{b1}$  is the bleaching wave velocity and  $v_{ac}$  is the acoustic velocity. The relation  $L_0/v_{b1} \ll t$  allows 3 laser energy to be deposited uniformly along the solenoid axis while the relation t  $\ll L_0/v_{ac}$  ensures that the effects of end loss are experienced only near to the solenoid ends. The above relation can obviously be satisfied only if the laser intensity and plasma density are such that

v<sub>b1</sub> >> v<sub>ac</sub>.

#### 3.3 0-Pinch Reactor

The magnetodydrodynamic behaviour of a 1 km 0-pinch reactor in radial equilibrium was simulated with a 10 shell model and 60 axial points. The initial density and temperature profiles were taken to be:

$$n_o(r,x) = \frac{4x10^{17} \text{ cm}^{-3}}{1 + \exp((r-1.5)/0.15)}$$

1

and

$$T_{e,f}(r,x) = \frac{5000 \text{ eV}}{1 + \exp((r^2.07/4).10)}$$

The external magnetic field was set at 405 kguass corresponding to a value of  $\beta = 0.98$  on the axis and the magnetic field in each shell was calculated assuming radial pressure balance. Symmetric boundary conditions around x=500 m were assumed. There was no external heating and a particle effects were not taken into account.

The results for two times (0.5 ms and 2 ms) are presented in Figures 3.20 to 3.29. Because symmetry is assumed around the point x=500 m, MHD quantities are only plotted in the region defined by  $0 \le x \le 500$  m. Figures 3.20 and 3.25 show the radius of the shell boundaries used in the numerical calculations as a function of x. These shell boundaries can also be thought of as magnetic flux lines... In the remaining figures three dimensional plots of  $T_{eno}$ ,  $v_x$ , and  $B_x$  are presented. The ion temperature is almost identical to that of the electrons and has not been plotted. At t=0.5 ms, the effects of self-mirroring are clearly evident in Figure 3.20. Figures 3.22 and 3.23 for the plasma density and velocity show that the greatest loss of shell mass occurs between 1 and 2 cm off-axis. This is in agreement with a theoretical one shell model<sup>17</sup> which predicts that the mass flux at the end is proportional to  $(1-\beta)^{1/2} T_e^{1/2}$ . This function reaches a maximum ' for r > 0 since  $\beta$  and  $T_e$  decrease off-axis. It is also observed from Figure 3.23 that the effects of the open ends propagate faster in the region of lower  $\beta$  as expected from one-dimensional theory.

At t=2 ms, the effects of the open solenoid ends have reached the midplane in all shells and 72% of the mass has flowed from the end of the reactor. Figure 3.25 shows how the shell boundaries, or magnetic flux lines, have moved in towards the solenoid center as the plasma flows out the ends. Figure 3.26 shows that adiabatic expansion has led to a decrease in plasma temperature from 5000 eV to about 3200 eV. Figure 3.27 shows that while 72% of the mass has flowed from the reactor ends, the on axis density has only decreased by about 12%. This is because, as the mass flows from central portions of the reactor, the magnetic field compresses the plasma radially towards the center. This effect could be important since the thermonuclear reaction rate is proportional to  $n_{\perp}^2$ . The velocity distribution function, Figure 3.28, as a function of x is found to be very nearly linear in all shells as suggested in the one shell theory of McMullin and Capjack<sup>18</sup>. Figures 3.24 and 3.29 illustrate how the magnetic field fills the central portion of the solenoid as the plasma flows out.



.

R











pressure balance).









5

in Section 3.2.

ì

?













•••

·.

.

·//

\$

79

.





#### CHAPTER 4

# A Computer Simulation of Gas Target Experiments

•

Laser-induced parametric instabilities in gas target plasmas have recently been studied experimentally by A.A. Offenberger et al.  $^{19-21}$  and J.J. Schuss et al.  $^{22}$ . Three main advantages of using gas targets over solid targets for the study of laser-induced parametric instabilities can be summarized from apaper by A. Ng et al.  $^{23}$  l) The nonlinear interactions can be studied at controllable plasma densities which can range all the way from critical to well subcritical. 2) Nonlinearities in underdense plasmas may not be obscured by a critical layer as in solid target experiments. 3) The relatively long characteristic scale lengths in gas target experiments (as compared to solid target experiments) make spatial and temporally resolved diagnostics more accessible.

In this chapter some results from two dimensional hydrodynamic simulations of  $CO_2$  laser heated gas target plasmas are reported. These models can be a valuable supplement to experiments since they give detailed predictions for the spatial and temporal profiles of the various hydrodynamic variables calculated. Since these quantities are very difficult to measure with good spatial and temporal resolution the simulations can be very valuable in the interpretation of experimental results. An attempt has been made to make the simulations correspond as closely as possible to the experiments of A.A. Offenberger <u>et al.</u><sup>19-21</sup>. The experimental details of this gas target can be found in the paper by A. Ng <u>et al.<sup>22</sup></u>.

82

## 4.1 Computer Model

The computer routine used for these simulations has been excised in Chapter 2. This program was developed for the purpose of solenoidal magnetic field. For reasons described in Chapter 2 the appropriate MHD equations are solved in a moving coordinate system defined by magnetic field lines, consequently the routine will only model a plasma with an imbedded solenoidal magnetic field. However, in the gas target experiments no magnetic field exists. This problem is easily overcome by setting the solenoidal field to a sufficiently small value that it has essentially no effect on the plasma dynamics. Also; since the routine works best for  $(B_r/B_x)$  small, where  $B_r$  is the magnetic field in the radial direction, and  $B_x$  is the magnetic field in the axial direction, the electrical conductivity in the model is artificially set to a small value so that movement of the field lines is small and the factor  $(B_r/B_x)$  remains very small.

In the simulations the laser beam intensity is assumed to be of the form:

$$I(x,r,t) = \frac{P(x,t)}{2\pi\sigma(x)^{2}} \exp \left[-\frac{r^{2}}{2\sigma(x)^{2}}\right]$$
(4.1)

where P(x,t) is the calculated total laser power. P(x,t) is calculated through a numerical solution to equation (2.10). The beam width is defined through the parameter  $\sigma(x)$  which has been defined to reflect the beam focussing in free space. Refraction of the beam by the plasma has been neglected in the present model. The parameter  $\sigma(x)$  has been set to conform to the optics of A.A. Offenberger <u>et</u> <u>al</u>.:

$$\sigma(\mathbf{x}) = \sigma_0 + \frac{1}{2f} |\mathbf{x} - \mathbf{L}_3|$$
 (4.2)

where  $\sigma_0$  gives the beam radius at the focal spot, f is the f number of the optics (which is set to 2 for these simulations), and L<sub>3</sub> specifies the position of the focal spot. The laser power input, as a function of time, is assumed to be  $\tau_L$  sec long and of triangular shape. The pulse 'starts at zero, the constant starts at zero, the constant set power P<sub>0</sub> in 1/2  $\tau_L$  secs and falls off to zero again at starts

In these simulations the appropriate differential equations are solved numerically in the cylindrical region defined by the following three boundaries:  $r = R_0$ ,  $x = L_0$ , and x = 0. The initial density profile is assumed to be defined as:

$$n_{o}(r, x, t=0) = N_{o}[1 + exp(\frac{L_{1}-x}{\lambda})]^{-1}$$

where  $L_1$  defines the position at which the density rises to half of  $N_0$ and  $\lambda$  defines the scale length over which the density rises.

For these simulations an attempt has been made to place the boundaries sufficiently far away from the region of interest that the boundary conditions do not play a significant role in the evolution of the solution. In fact  $R_0$  and  $L_0$  are sufficiently large that no plasma motion reaches these boundaries during the simulations. This is not true however for the x=0 boundary. Here we make the boundary condition that all quantities are linear  $(\partial^2/\partial x^2 = 0)$  with two exceptions. 1) Thermal conductivity is set to zero at x = 0. This is reasonable since there is

nall amount of plasma in the region x < 0 to absorb the heat.

2)  $\partial P/\partial x = C$  where P is the plasma pressure  $(Nk_B(T_e + T_i))$  and C is a pre-set constant. We find the solution is very insensitive to the value of C chosen.

In this chapter results from four different simulations are presented. The input parameters for the computer program are specified below. For all the simulations the plasma is assumed to be pre-ionized with an initial temperature of 1 eV. All of the simulations assume  $N_o = 9 \times 10^{18} \text{ cm}^{-3}$  and  $\lambda = 0.0145 \text{ cm}$ . These two parameters lead to the initial density profile rising from 3% of  $N_o$  at  $(L_1 = 0.05)$  cm to 97% of  $n_o$  at  $(L_1 + 0.05)$  cm as illustrated graphically in Figures 4.1 and 4.2. The initial plasma density of  $9 \times 10^{18} \text{ cm}^{-3}$  is just below the critical density for 10.6 um wavelength radiation from a  $CO_2$  lager. Other input parameters for the four simulations are:

Simulation #1  $P_{o} = 0.1GW$ ,  $L_{3} = 0.2cm$ ,  $\sigma_{o} = 0.0125cm$ ,  $L_{1} = 0.15cm$ ,  $R_{o} = 0.3cm$ ,  $L_{o} = 0.6 cm$ , and  $\tau_{L} = 40 \times 10^{-9} sec$ Simulation #2  $P_{o} = 1.0GW$ ,  $L_{3} = 0.2cm$ ;  $\sigma_{o} = 0.0125 cm$ ,  $L_{1} = 0.15cm$ ,  $R_{o} = 0.5cm$ ,  $L_{o} = 0.7cm$ , and  $\tau_{L} = 40 \times 10^{-9} sec$ Simulation #3  $P_{o} = 1.0GW$ ,  $L_{3} = 0.3cm$ ,  $\sigma_{o} = 0.0125cm$ ,  $L_{1} = 0.15cm$ ,  $R_{o} = 0.5cm$ ,  $L_{o} = 0.7cm$ , and  $\tau_{L} = 40 \times 10^{-9} sec$ Simulation #4  $P_{o} = 0.3GW$ ,  $L_{3} = 0.5cm$ ,  $L_{o} = 0.7cm$ , and  $\tau_{L} = 40 \times 10^{-9} sec$ Simulation #4  $P_{o} = 0.3GW$ ,  $L_{3} = 0.2cm$ ,  $\sigma_{o} = 0.0035cm$ ,  $L_{1} = 0.05cm$ ,  $R_{o} = 0.3cm$ ,  $L_{o} = 0.3cm$ , and  $\tau_{L} = 30 \times 10^{-9} sec$ 

In the first three simulations an attempt has been made to match the experimental conditions of A.A. Offenberger <u>et al</u>. when a stable resonator was used on the  $CO_2$  laser. When an unstable resonator 85

s !

was used the angular divergence of the output beam was somewhat smaller. This leads to a smaller laser beam radius at the focal spot. In simulation #4  $\sigma_0$  has been set to 35 µm to see what effect this may have on the hydrodynamics.

In all of these simulations a two-dimensional computational grid consisting of thirty radial shells and sixty axial points was used. The small value of  $\sigma_0$  in simulation #4 leads to very high laser intensities and therefore very high heating rates in a very small region in the vicinity of the focal spot. Because of this a finer computational mesh is needed for this simulation. This is achieved by making the region of solution smaller (smaller values of  $L_0$  and  $R_0$ ) and concentrating the shell structure near the solenoid axis. That is the shell spacing is smaller for r close to zero than it is for r close to  $R_0$ . The first three simulations ran with a timestep size of close to  $5 \times 10^{-11}$  sec and required approximately 20 minutes CPU time on the Amdahl 470 V/6 computer. Simulation #4 had to be run with a much smaller timestep size (down to  $2 \times 10^{-12}$  sec as the laser beam bleached through the focal spot) and required a total of 70 minutes CPU time.

The peak laser intensity in the 4'th simulation is much higher than in the first three simulations due to the smaller value of  $\sigma_0$ . This can be very desirable for the experimental investigation of laser-induced parametric instabilities, however, it adds complications to the simulation of the hydrodynamic expansion since ponderomotive forces now become significant. Following the derivation of Chen<sup>24</sup> the ponderomotive force can be expressed as:

$$\mathbf{\dot{f}}_{p} = -\frac{1}{2} \frac{\boldsymbol{\omega}_{pe}^{2}}{\boldsymbol{\omega}_{1}} \quad \forall \quad (\frac{\mathbf{I}}{\mathbf{c}_{1}})$$

where  $\omega_1$  is the frequency of the laser beam, I is its intensity, and  $c_1$  its group velocity. The ponderomotive force is added to the hydrodynamic model by adding the term:

$$f_{r} = -\frac{1}{2} \frac{\frac{\omega_{pe}^{2}}{pe}}{\frac{2}{2} \frac{3}{3r}} \left(\frac{I}{c_{1}}\right)$$

٠

to the right hand side of equation (2.2) and adding the term:

$$f_{\mathbf{x}} = -\frac{1}{2} \frac{\omega_{\mathbf{pe}}^2}{\omega_1^2} \frac{\partial}{\partial \mathbf{x}} \left(\frac{\mathbf{I}}{c_1}\right)$$

to the right hand side of equation (2.3). The appropriate modifications have been made to the code for simulation #4.
## 4.2 Results

The results of these simulations are summarized graphically in Figures 4.3 to 4.30. The plots are of three types. (1) At a specified time three dimensional plots of the hydrodynamic quantities  $(T_e, T_i, n_o, v_x \text{ and } v_x)$  are plotted as functions of r and x. (2) At the same time the laser power P(x) and the axial laser intensity I(o,x)are plotted as functions of x. (3) The values of  $T_e, T_i$ , and  $n_o$  at the laser focal spot are plotted as functions of time.

In the three dimensional plots the density peak in the shock fronts has been limited to values of less than  $2 \times 10^{19} \text{ cm}^{-3}$  in order to make scaling convenient. This is achieved by setting n to  $2 \times 10^{19} \text{ cm}^{-3}$  if it exceeds  $2 \times 10^{19} \text{ cm}^{-3}$  in the plot routine.

The three dimensional plots of plasma density all show a "hole" in the density profile around the region of laser heating. A strong axial shock is seen to propagate just in front of the laser beam, and strong shocks are also seen to be propagating in the radial direction. The radial shocks propagate outwards with a velocity of about  $4 \times 10^6$  cm/sec when the peak laser power is 0.1 GW and at about  $6.5 \times 10^6$  cm/sec when peak laser power is 0.1 GW and at about  $6.5 \times 10^6$  cm/sec when peak laser power is 1.0 GW. The velocity of the axial shock is seen to be dependent on the laser beam width but a typical shock velocity of about  $7.5 \times 10^6$  cm/sec is found for the case of the 0.1 GW laser. The higher powered laser beam is seen to push the axial shock at a higher velocity of between  $10^7$  and  $2 \times 10^7$  cm/sec.

Plots of the density profiles show a hump in the region just behind the shock wave. The formation seems to be dependent on the beam

width as indefined by the fact that it is found to develop later in the third eimulation, where the laser is focussed further into the plasma, than it does in the second simulation. The fact that the hump is spread over about seven axial points is an indication that this feature is real and not due to the discrete numerical model.

An important feature of gas target experiments is the long characteristic scale lengths over which hydrodynamic quantities vary significantly. Density plots show that at 20 ns the density changes by 50% over axial distances of about 0.05 cm or radial distances of about 0.1 cm for the 0.1 GW laser. For the simulations with the higher powered lasers both the radial and axial scale lengths are doubled. Characteristic scale lengths for both T and T are considerably longer than for density.

The temperature plots show that electron temperature rises very rapidly to its maximum value in the region just behind the axial shock. This can be explained by the high laser absorption rate in the high density region at the back end of the shock. Further back the laser heating rate is reduced by the lower densities and higher electron temperatures, and must compete with the cooling effect of adiabatic expansion. The ion temperature profile is similar, although the ions remain considerably colder than the electrons. The ions are heated rapidly in the high density region just behind the axial shock due to electron-ion collisions. The subsequent adiabatic expansion causes the ions to cool somewhat in the region behind the shock. Typical electron temperatures of 40 to 50 eV and ion temperatures of about 30 eV are whieved in the first simulation with the 0.1 GW laser. The second simulation shows that increasing the laser power by a factor of 10 leads to an increase in typical electron and ion temperatures by a factor of about two. Increasing the laser power by a factor of ten leads to a large change in the volume of the plasma heated rather than making a large change in the plasma temperature. The third simulation, where the laser is focussed further into the plasma, shows significantly higher electron temperatures can be achieved in the region of the focal volume if the laser beam is focussed sufficiently far into the plasma to cause an axial shock to be driven through the focal volume.

The plots of laser power versus x show that the beam power is fairly constant up to the shock wave. This shows that almost all of the laser energy is being absorbed in the high density region at the back of the axial shock.

A knowledge of plasma fluid velocities can be important in the interpretation of experiments since the fluid velocity can lead to a doppler shift in the frequency of plasma light emission. The axial velocity plots show typical axial velocities in the focal volume of about 10<sup>7</sup> cm/sec.

The plots of  $T_e$ ,  $T_1$ , and  $n_o$  as a function of time provide a summary of how the values of these variables in the focal volume change over the duration of the experiment. It is seen that for the simulation with the 0.1 GW laser the electron temperature in the focal volume rapidly rises to 55 eV, and then slowly cools to 40 eV at 30 ns. The ion temperature, however, rises rapidly to 30 eV and maintains this value throughout the time of the simulation. The ion heating, due to electron-

ion collisions, is presumably balanced by the ion cooling, due to adiabatic expansion. Increasing the laser power by a factor of ten, as in the second simulation, leads to electrons in the focal volume heating rapidly to 95 eV and then cooling to 75 eV at 30 ns. The ions rapidly heat to about 35 eV, and then continue to rise in temperature until they reach 50 eV at 30 ns. It is interesting to note that the fluid densities at the focal volume for these two simulations are almost identical over the period of 12 ns to 30 ns. These plots also show the importance of the laser focus position. In the third simulation, where the beam was focussed further into the plasma, the electron temperature rapidly rose to 155 eV as the shock wave propagated through the plasma. The plots indicate that at about 7 ns a hot plasma of critical density existed in the focal volume. Since this critical layer is part of the shock wave, characteristic scale lengths are very short at this time. After the shock passes through the focal volume, the electron temperature cooled much more rapidly than was found in the other simulations. By 30 ns the electron temperature is only 75 eV which is the same as the electron temperature of the second simulation at 30 ns. The fluid density also fell rapidly and became essentially the same as that of the second simulation by 22 ns. The ion temperature in the focal volume of the third simulation rose rapidly to about 50 eV and remained constant for the duration of the simulation. Another important feature seen in these plots is that the density curve goes through the quarter critical value fairly quickly before becoming more level at a density of about  $10^{17}$  cm<sup>-3</sup>. The time at which the density curve goes through the quarter critical value depends on both the laser power and the laser focus position.

91

ŕ

The three dimensional plots for the fourth simulation are shown at a time of 12 ns. This is close to the time of peak laser power and is just after the laser bgam has bleached its way through the focal volume. A peak electron temperature in excess of 200 eV is seen at this time while the peak ion temperature is close to 60 eV. The three dimensional plot of plasma density shows no axial shock, however other plots not included in this thesis show that an axial shock exists at both earlier and later times. The reason for this effect is that as the laser beam bleaching wave propagates through the focal volume, the bleaching velocity exceeds the acoustic velocity in the hot plasma behind the bleaching front. This high velocity bleaching wave temporarily detroys the shock wave. The plots of  $v_r$  and  $v_z$  show much larger fluid velocities in the region of the bleaching front than that of previous simulations. This is a reflection of the large pressure gradients that have been set up.

.1

In Figure 4.30,  $n_0$ ,  $T_e$ , and  $T_i$  at the position of the laser focal spot are plotted as a function of time. This plot shows that the axial shock reaches the focal spot before it is destroyed as indicated by Figure 4.27. As the back end of the shock wave passes the focal point of the laser beam, the electron temperature rises very rapidly to reach a peak of 265 eV. Just after the shock wave passes through the focal spot the plasma expands rapidly and the electron temperature falls to about 100 eV in a period of about 1 ns, and then continues to drop but at a much slower rate. This rapid cooling can be attributed to the rapid adiabatic expansion as well as thermal conduction due to large temperature gradients.

### 4.3 Conclusions

of 1

The computer simulations of laser heated gas target plasmas predict that the hydrodynamic quantities vary over relatively long characteristic scale lengths. Both axial and radial shocks propagate away from the region of laser heating. Most of the laser energy is found to be deposited in a small volume of plasma just behind the axial shock, leading to both electrons and ions reaching their peak temperature at a short distance behind the shock. The plasma temperature and density in the laser focal volume is found to be a relatively weak function of input laser power.

It is found that the position of the laser focal spot relative to the initial density profile can be important to the resulting hydrodynamics. If the laser is focussed sufficiently far into the initial density profile that an axial shock wave is driven through the focal volume, the early time electron temperatures and densities are found to be much higher than temperatures and densities at corresponding time for the case where the laser is focussed at the front edge of the density profile. These differences could lead to significantly different observations of laser induced parametric instabilities. The hydrodynamic behaviour of the plasma at later times (> 25 ns), however, is found to be relatively insensitive to the position of the laser focal spot.

The reduction of the beam radius at the focal spot can have dramatic effects on the resulting hydrodynamics. While the electron temperature at the focal spot rose to a much higher value for the case of  $\sigma_0$  = 35 µm as compared to when  $\sigma_0$  = 125 µm, it also decreased much more rapidly. The plasma density also drops more quickly for smaller



position of Iaser focus.

94 `



95

t





•

.

1

•

Figure 4.7 Radial plasma velocity in cm/sec at 20 ns; from simulation #1.

And a support of the second second













ررس

¥0.0

Te.

2

8

8

•

ł





Figure 4.13 Axial plasma velocity in cm/sec at 20 ns; from simulation 02.

100



ţ

Figure 4.14 Radial plasma velocity in cm/sec at 20 ns; from simulation #2.





•







x + 0 m



103







:

105

r





÷ . •





. 2









110

Ċ

#### CHAPTER 5

## A Theory of Beat Frequency Mixing

In this chapter a theory for the beat frequency mixing of antiparallel electromagnetic beams in a plasma is presented. This theory assumes that the plasma is uniform and has an isotropic Maxwellian distribution function. The magnetic field, if present, is assumed to be parallel to the directions of propagation of the interacting beams.

Theoretical studies on the beat frequency mixing of antiparallel electromagnetic beams, where the difference frequency of the interacting beams is close to the electron plasma frequency, have been made by Schmidt<sup>25</sup>, Fuchs <u>et al</u><sup>26</sup>, Kaufman and Cohen<sup>27</sup>, Cohen<sup>28</sup>, Cohen <u>et al</u><sup>29</sup>, and Capjack and James<sup>30</sup>. Much of the work in references 25-29 is devoted to the theory of beat frequency mixing in an inhomogeneous plasma. Capjack and James present a theory for beat frequency mixing in a homogeneous plasma. It was found, however, that the results of Capjack and James differed by a factor of about  $(\omega_{pe}/\omega_3)^4$  from the calculation made by Cohen <u>et al</u><sup>29</sup>. The source of this discrepancy is explained in this chapter.

Results from Chapter 3 of this thesis indicate that it may be possible to create a plasma which is radially confined by a solenoidal magnetic field with very long density and temperature scale lengths in the axial direction. For these constructions the theory for beat frequency mixing in a homogeneous plasma is val suggested that the beat frequency solenois to part parallel laser beams in a plasma radially confined by a solenois magnetic field could be used to create a useful broad-band infrared amplifier. A good theory for the **111** (

beat-frequency mixing process in homogeneous plasmas is a valuable tool for the theoretical study of such a device.

In this chapter beat frequency mixing rates for a homogeneous plasma are derived. The effects of ion mobility and electron-ion collisions have been included in this derivation. Inclusion of ion mobility leads to the calculation of high mixing rates for the case when the beat frequency is close to the ion-acoustic frequency. This approach leads to an expression that is valid for difference frequencies ranging from zero to much greater than the electron-plasma frequency. Inclusion of the effects of electron-ion collisions makes the results equally valid for cases where the dominant mechanism for the damping of the driven electrostatic mode is collisional, collisionless, or the intermediate case where both effects are important.

In Section 5.1 differential equations which describe the coupling of antiparallel electromagnetic beams in the presence of electron density fluctuations are derived. This derivation follows Cohen<sup>28</sup> and has been included here for completeness. In Section 5.2 an integral equation which defines the electron density fluctuation caused by the beating of antiparallel electromagnetic beams is derived. The two differential equations from Section 5.1 and the integral equation from Section 5.2 form a set of three coupled equations which describe the beat frequency mixing process. In Section 5.3 these three equations are simplified through an approximation based on the assumption that the amplitudes of the fields in the electromagnetic beams vary slowly on the time and length scales defined by the beat frequency period and wavelength. Beat frequency heating rates are derived for this particular case. In Section 5.4 the equations of Section 5.1 and Section 5.2 are cast in an approximate form

which allows for easy numerical solution. These new equations are useful since they can be used to describe the beat frequency mixing of a beam with a short pulse while properly accounting for transient effects in the electrostatic mode. In Section 5.5 the dependence of the beat frequency mixing rates on plasma parameters are displayed graphically and discussed.

In the derivation that follows it is assumed that the interacting beams are plane polarized, however it can be shown that all the results are equally valid for the case of circular polarization.

# 5.1 Beam Coupling in the Presence of Electron Density Fluctuations

In this section an expression for the coupling of two.antiparallel, linearly polarized, electromagnetic beams in the presence of electron density fluctuations is derived. Effects of collisional damping (inverse Bremsstrahlung absorption) are included. The frequency of the electromagnetic beams,  $\omega_1$  (i=1,2) is assumed to be greater than the electron plasma frequency,  $\omega_{pe}$ .

The vector potential in the transverse gauge due to the two interacting beams can be written as:

$$\partial_{\mathbf{x}}^2 \vec{\mathbf{A}} - \frac{1}{c^2} \partial_{\mathbf{t}}^2 \vec{\mathbf{A}} = -\frac{4\pi}{c} J_{\mathrm{T}}$$
(5.1)

where  $\vec{J}_T$  is the transverse current. From here on it is assumed that  $\vec{A}$  is in the  $\hat{y}$  direction and that the wave vectors  $\vec{k}_1$  (i=1,2) are in the  $\hat{x}$  direction so that vector notation can be dropped.

Neglecting higher order terms, A and  $J_T$  can be expanded as:

$$A = A_{1}(\mathbf{x}, t) e^{i\xi_{1}} + A_{2}(\mathbf{x}, t) e^{i\xi_{2}} + c.c,$$

$$J_{T} = J_{1}(\mathbf{x}, t) e^{i\xi_{1}} + J_{2}(\mathbf{x}, t) e^{i\xi_{2}} + c.c.$$
(5.2)

where n is the equilibrium density and  $\xi_1 = (k_1 x - \omega_1 t)$  and  $\xi_2 = (k_2 x - \omega_2 t)$ . It is also assumed that electron density can be expanded as:

$$n_{e} = n_{o} + (\hat{n}_{e}(\mathbf{x}, t) e^{i\xi_{3}} + c.c.)$$
 (5.3)

where  $\xi_3 = \xi_1 - \xi_2$ . Terms of the form  $\exp[i(\xi_1 + \xi_2)]$  have been neglected since they are off resonant.

The transverse current can be found from the transverse electron fluid velocity u.

$$J_{T} = -n_{e} e u \qquad (5.4)$$

The ion contribution to the transverse current has been neglected since it is smaller than the electron contribution by a factor of  $(\frac{m}{e}/\frac{m}{i})$ . The transverse electron fluid velocity u is expanded as:

$$u = u_1(\mathbf{x},t) e^{i\xi_1} + u_2(\mathbf{x},t) e^{i\xi_2} + c.c.$$
 (5.5)

The electron fluid velocity can be derived from:

$$\partial_t \mathbf{u} = -v_{e1} \mathbf{u} - \frac{\mathbf{e}}{\mathbf{n}_e} \mathbf{E}$$
 (5.6)

where  $v_{ei}$  is the electron-ion collision frequency and  $E = -\frac{1}{c} \partial_t A$  is the electric field due to the electromagnetic beams. Magnetic terms are of order (u/c) and have been neglected. Equation (5.6) can be split into two parts and written as:

$$\partial_t (u_i e^{i\xi_i}) = - v_{ei} u_i e^{i\xi_i} + \frac{e}{m_e c} \partial_t (A_i e^{i\xi_i})$$
 (5.7)

or

$$(\partial_t - i\omega_i) u_i = -v_{ei} u_i + \frac{e}{m_e c} (\partial_t - i\omega_i) A_i$$

for i = 1, 2.

 $\partial_t u_1 \ll \omega_1 u_1$  the above equation can be approximated as:

or

If

$$u_{i} = n_{i} \frac{\mathbf{a} \mathbf{A}_{i}}{\mathbf{m}_{e}^{c}}$$
(5.8)

where

$$n_{i} = \frac{\omega_{i}^{2} - i v_{ei} \omega_{i}}{\omega_{i}^{2} + v_{ei}^{2}} = 1 - \frac{i v_{ei}}{\omega_{i}}$$
(5.9)

where it is assumed that  $(v_{ei}/\omega_i)^2 \ll 1$ .

 $-i\omega_{i}u_{i} = -v_{ei}u_{i} - i\omega_{i}\frac{e}{m_{e}c}A_{i}$ 

From (5.3) and (5.4) the transverse component of the

current may be shown to be:

.

$$J_{T} = - eu[n_{o} + (\hat{n}_{e} e^{i\xi_{3}} + c.c.)]$$
 (5.10).

Combining (5.8) and (5.10) the resonant current terms are found to be:

$$J_{1}(\mathbf{x},t) = -\frac{\eta_{1}}{4\pi c} \omega_{pe}^{2} \mathbf{A}_{1} - \frac{\eta_{2}}{4\pi c} \omega_{pe}^{2} \mathbf{A}_{2} \left(\frac{n}{n_{o}}\right)$$
(5.11)  
$$J_{2}(\mathbf{x},t) = -\frac{\eta_{2}}{4\pi c} \omega_{pe}^{2} \mathbf{A}_{2} - \frac{\eta_{1}}{4\pi c} \omega_{pe}^{2} \mathbf{A}_{1} \left(\frac{n}{n_{o}}\right)$$

Equation (5.1) is split into two parts and its harmonic dependence is inserted. Then:

$$\partial_t^2 + (\partial_t - i\omega_i)^2 = \partial_t^2 - \omega_i^2 - 2i\omega_i \partial_t$$
$$\partial_x^2 + (\partial_x - ik_i)^2 = \partial_x^2 - k_i^2 - 2ik_i \partial_x$$

•

It is assumed that  $\vartheta_t^2 A_1 \ll u_1^2 A_1$  and  $\vartheta_x^2 A_1 \ll k_1^2 A_1$  and that small terms can be neglected. Then:

$$(\partial_t^2 - c^2 \partial_x^2) + (-\omega_1^2 + k_1^2 c^2 - 2i\omega_1 \partial_t - 2ic^2 k_1 \partial_x)$$
 (5.12)

By using this relation the component of equation (5.1) with an  $e_{1}^{i\xi}$  dependence can be written  $e^{i\xi}$ :

$$[(-\omega_{1}^{2} + k_{1}^{2} c^{2} + \omega_{pe}^{2}) - 2i\omega_{1}^{3} t - 2ic^{2}k_{1}^{3} t - \frac{i\nu_{ei}\omega_{pe}^{2}}{\omega_{1}}] A_{1}$$

$$= -\eta_{2} \omega_{pe}^{2} A_{2} (\frac{\tilde{n}_{e}}{\eta_{o}})$$
(5.13)

It is assumed that the waves in the electromagnetic beams obey the dispersion relation:

$$- \omega_{1}^{2} + k_{1}^{2} c^{2} + \omega_{pe}^{2} = 0$$

so the first term in (5.13) can be set to zero. Any error in this assumption due to nonlinear frequency shifts does not lead to an error in the solution. Instead it shows up in the complex solution of  $A_1(x,t)$ -rotating in the complex plane.

Equation (5.13) and the corresponding equation that can be derived from the component of (5.1) with a  $e^{i\xi_2}$  dependence can be expressed as:

$$\begin{bmatrix} \partial_{t} + c_{1} & \partial_{x} + \Gamma_{1} \end{bmatrix} A_{1}(x,t) = - (\frac{1}{2}) \frac{\omega_{pe}^{2}}{\omega_{1}} \frac{\hat{n}}{e} A_{2}(x,t)$$
(5.14)

$$\begin{bmatrix} \partial_t + c_2 & \partial_x + \Gamma_2 \end{bmatrix} \mathbf{A}_2(\mathbf{x}, t) = - \begin{pmatrix} \frac{1}{2} \end{pmatrix} \frac{\omega^2}{\omega_2} - \frac{\mathbf{n}}{\mathbf{n}} \mathbf{A}_1(\mathbf{x}, t)$$

where  $c_1^{\circ} = \frac{c_1^2 k_1}{\omega_1}$  is the group velocity of the electromagnetic beams in the plasmas,  $\Gamma_1 = \frac{v_{e1}^2 \omega_{pe}^2}{2\omega_1^2}$  are the collisional absorption rates for the

beams, and \* denotes the complex conjugate.

It should be noted that (5.14) makes no assumption on the magnitude of, or rates of change of the density fluctuation  $\tilde{n}_{e}^{\nu}(\mathbf{x},t)$ . In fact the only restriction on the validity of equations (5.14) is that the magnitude of  $A_{i}(\mathbf{x},t)$  does not change significantly in one period or wavelength of the electromagnetic beams.

The Manley-Rowe relations (photon conservation) can be easily derived from equations (5.14). Neglect collisional absorption and multiply the first equation by  $\omega_1 A_1^{\dagger}$  and the second equation by  $\omega_2 A_2^{\dagger}$ . Add each of these equations to its complex conjugate and then add the two resulting equations. The result is:

$$\omega_{1}[\partial_{t} + c_{1}\partial_{x}] |A_{1}|^{2} + \omega_{2}[\partial_{t} + c_{2}\partial_{x}] |A_{2}|^{2} = 0 \qquad (5.15)$$

The intensity in each beam can be expressed as:

$$I_{i} = \frac{\omega_{i} |k_{i}|}{2\pi} |A_{i}|^{2} \qquad (\frac{\text{ergs}}{\text{sec cm}^{2}})$$

Substituting this into the above equation yields:

$$\begin{bmatrix}\mathbf{a}_{\mathbf{b}} + \mathbf{c}_{1} \ \mathbf{a}_{\mathbf{x}}\end{bmatrix} \begin{pmatrix} \mathbf{I}_{1} \\ |\mathbf{k}_{1}| \end{pmatrix} + \begin{bmatrix}\mathbf{a}_{t} + \mathbf{c}_{2} \ \mathbf{a}_{\mathbf{x}}\end{bmatrix} \begin{pmatrix} \mathbf{I}_{2} \\ |\mathbf{k}_{2}| \end{pmatrix} = 0$$
(5.16)

Since the photon density in each beam is proportional to  $(I_i/k_i)$  this equation implies photon conservation. An equivalent form for (5.16) is (use  $k_i = (\omega_i c_i)/c^2$ )

•)

$$[\partial_{t} + c_{1} \partial_{x}] (I_{1} / |c_{1}|\omega_{1}) + [\partial_{t} + c_{2}\partial_{x}] (I_{2} / |c_{2}|\omega_{2}) = 0$$

The Manley-Rowe relations can be cast in a simpler form. The energy density of a beam in the plasma is  $E_1 = I_1 / |c_1|$ . Equation (5.16) can be rewritten as:

$$\frac{c_1}{k_1} (\partial_t + c_1 \partial_x) E_1 + \frac{c_2}{k_2} (\partial_t + c_2 \partial_x) E_2 = 0$$

If  $W_1$  is defined as the rate at which energy is being transferred out of the high frequency beam and  $W_2$  is the rate at which energy is being transferred into the low frequency beam,

$$W_1 = - (\partial_t + c_1 \partial_x) E_1$$
  

$$W_2 = + (\partial_t + c_2 \partial_x) E_2$$

Using the above definitions and the fact that  $\frac{c_1}{k_1} = \frac{c^2}{\omega_1}$ , the above equation can be expressed as,

$$\frac{\frac{w_1}{\omega_1}}{\frac{w_2}{\omega_1}} = \frac{\frac{w_2}{\omega_2}}{\frac{w_2}{\omega_2}}$$

If the rate at which energy is transferred to the beat frequency electro-

static wave is defined as  $W_3$ , the conservation of energy principle implies that:

$$w_3 = w_1 - w_2 = w_1(1 - \omega_2/\omega_1) = \frac{\omega_3}{\omega_1} w_1$$

٢

Combining the above two equations yields:

•

$$\frac{w_1}{w_1} = \frac{w_2}{w_2} = \frac{w_3}{w_3}$$
(5.17)

.

.

.

i

# 5.2 Density Fluctuations in the Presence of Antiparallel Electro-

### magnetic Beams

Equations (5.14) can be used to describe the coupling of two antiparallel electromagnetic beams in the presence of an electron density fluctuation. The ponderomotive force from these two beams, however drives a density fluctuation of frequency  $\omega_3$  and wave-vector  $\vec{k}_3$  where  $\omega_3 = \omega_1 - \omega_2$  and  $\vec{k}_3 = \vec{k}_1 - \vec{k}_2$ . In this section a general relation for the electron density fluctuation resulting from the mixing of two antiparallel electromagnetic beams is derived. The effects of ion mobility and electron-ion collisions are included in this analysis.

In this treatment the following four forces acting on electrons and ions are accounted for.

- (1)  $F_{c\epsilon}$  Force experienced by specieve  $\epsilon$  ( $\epsilon$  = e or i for electrons or ions) due to the coulomb electric field resulting from electron and ion density fluctuations.
- (2) Ponderomotive force on electrons.
- (3) F<sub>1</sub> Ponderomotive force on ions.
   (4) F<sub>col</sub> Phenomenological force on electrons to account for electron-ion collisions. An equal and opposite force is applied to ions.

In addition to these four forces, we also define:

- (1)  $F_{Te} = F_{ce} + F_{e} + F_{col}$  Total force acting on electrons.
- (2)  $F_{Ti} = F_{ci} + F_{i} F_{col}$  Total force acting on ions.

The double Fourier transform of all quantities Q(x,t)

that vary as  $\exp[i(k_3x - \omega_3t)]$  is taken. The transform has been defined symmetrically as:

$$Q(\mathbf{k},\omega) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} Q(\mathbf{x},t) e^{-i(\mathbf{k}\mathbf{x}-\omega t)} dt dx$$
$$Q(\mathbf{x},t) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} Q(\mathbf{k},\omega) e^{i(\mathbf{k}\mathbf{x}-\omega t)} dk d\omega$$

With this definition taking the transform of a differential equation in x and t means that:

where  $q_{\varepsilon}$  is the charge of species  $\varepsilon$ , and  $E_{c\varepsilon}$  is the component of the coulomb field due to the density fluctuations of species  $\varepsilon$ .

The ponderomotive force on species  $\epsilon$  has been derived in Appendix C as:

$$\mathbf{F}_{\varepsilon}(\mathbf{x},\mathbf{t}) = -\frac{\mathbf{i}q_{\varepsilon}^{2}}{\mathbf{m}_{\varepsilon}c^{2}} \mathbf{k}_{3}\mathbf{A}_{1}(\mathbf{x},\mathbf{t})\mathbf{A}_{2}^{*}(\mathbf{x},\mathbf{t})\mathbf{exp}[\mathbf{j}(\mathbf{k}_{3}\mathbf{x} - \boldsymbol{\omega}_{3}\mathbf{t})] + c.c.$$
(5.19)

where  $m_{\epsilon}$  is the particle mass of species  $\epsilon$  and c = speed of light.

 $F_{col}(k,\omega)$  can be expressed as a function of  $E_{ce}(k,\omega)$  and  $E_{ci}(k,\omega)$ . The average force on electrons due to electron-ion collisions can be expressed as:

$$\mathbf{F}_{col} = \mathbf{m}_{e} (\mathbf{v}_{i} - \mathbf{v}_{e}) \cdot \mathbf{v}_{ei}$$
(5.20)

where  $v_{\varepsilon}$  is the average velocity of particles of species  $\varepsilon$ , and  $v_{\varepsilon}$ is the electron-ion collision frequency. However from the relations: •

$$v_{\varepsilon} = \frac{J_{\varepsilon}}{n_{o}q_{\varepsilon}}$$
;  $\nabla \cdot E_{c\varepsilon}(\mathbf{x},t) = 4\pi \rho_{\varepsilon}(\mathbf{x},t)$ ; and  
 $\frac{\partial \rho_{\varepsilon}}{\partial t} + \nabla \cdot J_{\varepsilon} = 0$  where  $J_{\varepsilon}$  is the current due to species  
is the charge density due to perturbations in the number

density of species  $\varepsilon$  it is found that

 $\varepsilon$ , and  $\rho_{\varepsilon}$ 

or

$$\mathbf{x}_{\varepsilon}(\mathbf{k},\omega) = \frac{\mathbf{i}\omega\mathbf{E}_{\varepsilon\varepsilon}(\mathbf{k},\omega)}{4\pi\mathbf{n}_{o}\mathbf{q}_{\varepsilon}}$$
(5.21)

Combining (5.20) and (5.21) yields:

$$F_{col}(k,\omega) = \frac{im_{e}\omega v_{el}}{4\pi n_{o}e} \{ E_{ci}(k,\omega) + E_{ce}(k,\omega) \}$$

$$F_{col}(k,\omega) = ie \frac{\omega v_{el}}{\omega_{pe}} \{ E_{ci}(k,\omega) + E_{ce}(k,\omega) \}$$
(5.22)

where  $\omega_{pe} = \left(\frac{4\pi n_{o}e^{2}}{m_{e}}\right)^{1/2}$  is the electron plasma frequency. Combining

equations (5.18) and (5.22):

$$F_{col}(k,\omega) = -\frac{i\omega v_{el}}{2} F_{ce}(k,\omega)$$
(5.23)  
<sup>w</sup>pe

To simplify algebra the following constants are defined:

$$\mathbf{a} \equiv \frac{1\omega \mathbf{v} \mathbf{e} \mathbf{i}}{2} \qquad \mathbf{b} \equiv 1 - \mathbf{a}$$
Then from the definitions of  $\mathbf{F}_{Te}$  and  $\mathbf{F}_{Ti}$ :

× 575.

$$F_{Te} = [F_{e} + b - F_{ce}]$$
  
(5.24)
  
 $F_{Ti} = [F_{i} - b F_{ce}]$ 

124

۱

Using a kinetic approach, and solving the linearized Vlasov equation (see Appendix B) it is found that the coulomb field resulting from fluctuations of species  $\alpha$  can be expressed as:

$$\mathbf{E}_{c\varepsilon}(\mathbf{w},\mathbf{k}) = -\frac{1}{q_{\varepsilon}} \chi_{\varepsilon}(\omega,\mathbf{k}) \mathbf{F}_{T\varepsilon}(\omega,\mathbf{k})$$
(5.25)

where  $\chi_\epsilon$  is the linear susceptability of species  $\epsilon.$  From Appendix B:

$$\chi_{\varepsilon}(\omega,k) = 2 \frac{\omega_{p\varepsilon}^{2}}{k^{2}} \frac{1}{v_{\theta\varepsilon}^{2}} \{1 + i \alpha_{o\varepsilon} F_{o}^{\varepsilon}\}$$
(5.26)

where

.

$$\omega_{p\epsilon}^{2} = \frac{4\pi n_{o}e^{2}}{m_{\epsilon}}, \quad v_{\theta\epsilon} = \left(\frac{2k_{B}T_{\epsilon}}{m_{\epsilon}}\right)^{1/2},$$
$$\alpha_{o\epsilon} = \frac{\omega}{kv_{\theta\epsilon}}, \quad F_{o}^{\epsilon} = \frac{1k}{\sqrt{\pi}} \int_{-\infty}^{\infty} \frac{e^{-v^{2}/v_{\theta\epsilon}}}{(\omega_{3} - k_{3}v)} dv,$$

and  $T_{\epsilon}$  is the temperature of species  $\epsilon$ . Techniques of evaluating  $F_{o}^{\epsilon}$  are discussed in Stix<sup>31</sup> (Chapter 8).From equations (5.18), (5.24), and (5.25):

$$E_{ce}(k,\omega) = -\chi_{e}\left\{-\frac{F_{e}}{e} + b(E_{ce} + E_{ci})\right\}$$

$$E_{ci}(k,\omega) = -\chi_{i}\left\{\frac{F_{i}}{e} + b(E_{ce} + E_{ci})\right\}$$
(5.27)

Solving equations (5.27) for E and E yields:

$$E_{ce} = + \frac{\chi_{e} \{F_{e}(1 + b \chi_{i}) + b \chi_{i} F_{i}\}}{e(1 + b \chi_{e} + b \chi_{i})} \qquad (5.28)$$

$$E_{ci} = - \frac{\chi_{i} \{F_{i}(1 + b \chi_{e}) + b \chi_{e} F_{e}\}}{e(1 + b \chi_{e} + b \chi_{i})}$$

Writing the total species density  $n_{TE}(x,t)$  as:

$$n_{Te}(x,t) = n_0 + n_e(x,t)$$

and using,

43

$$\nabla \cdot \mathbf{E}_{c\varepsilon}(\mathbf{x},t) = 4\pi q n_{\varepsilon}(\mathbf{x},t)$$

yields,

•

$$n_{\varepsilon}(k,\omega) = \frac{ik E_{c\varepsilon}(k,\omega)}{4\pi q_{\varepsilon}}$$
(5.29)

Combining (5.28) and (5.29) and ignoring  $F_i$  since  $F_i \leq F_e$  yields:

$$n_{e}(k,\omega) = -\frac{1k}{4\pi e^{2}} \frac{\chi_{e}^{F} F_{e}(1 + b \chi_{1})}{(1 + b \chi_{e} + b \chi_{1})}$$
(5.30)

The Fourier transform of  $F_e(x,t)$  defined in equation (5.19) may be shown to be:

$$F_{e}(k,\omega) = -\frac{1e^{2}k_{3}}{m_{e}c^{2}}f(k,\omega) + \frac{1e^{2}k_{3}}{m_{e}c^{2}}g(k,\omega)$$

where

$$f(k,\omega) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} A_{1}(x,t) A_{2}^{*}(x,t) e^{i(k_{3}x-\omega_{3}t)} e^{-i(kx-\omega t)} dx dt$$

and

1

$$g(-k,-\omega) = f^{(k,\omega)}$$

If  $D(k, \omega)$  is defined by:

$$D(k,\omega) = -\frac{kk_3}{4\pi m_e c^2} \frac{\chi_e(1+b\chi_1)}{(1+b\chi_1+b\chi_2)}$$
(5.32)

it may be shown that  $D(-k,-\omega) = -D^{*}(k,\omega)$ . If the inverse Fourier transform of equation (5.30) is now taken and use made of equations (5.31) and (5.32), the electron density perturbation may be written as:

$$n_{e}(\mathbf{x},\mathbf{t}) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} D(\mathbf{k},\omega) f(\mathbf{k},\omega) e^{\mathbf{i}(\mathbf{k}\mathbf{x}-\omega\mathbf{t})} d\mathbf{k} d\omega + c.c.$$
(5.33)

The expression for  $n_e^{\nu}(\mathbf{x},t)$  as required for equation (5.14) may be written as:

$$\hat{n}_{e}^{-i(k_{3}x-\omega_{3}t)} = e^{\int_{-\infty}^{\infty} \int_{-\infty}^{\infty} D(k,\omega) f(k,\omega)e^{i(kx-\omega t)} dk d\omega}$$
(5.34)

## 5.3 Case of Slowly Varying Electromagnetic Beam Amplitudes

If both  $A_1(x,t)$  and  $A_2(x,t)$  vary slowly on a time scale defined by the damping rate of the beat wave and on a length scale defined by the wavelength of the beat wave, equation (5.34) can be approximated. Under these conditions  $f(k,\omega)$  is sharply resonant and has significant values only for  $k = +k_3$  and  $\omega = +\omega_3$ . Equation (5.34) can be approximated as:

$$n_{e}^{-1(k_{3}x-\omega_{3}t)} D(k_{3},\omega_{3}) f(x,t)$$

where f(x,t) is the double Fourier transform of  $f(k,\omega)$  defined in equation (5.31). Using this definition:

$$\hat{n}_{e}^{\nu}(\mathbf{x},t) = D(k_{3},\omega_{3}) A_{1}(\mathbf{x},t) A_{2}^{*}(\mathbf{x},t)$$
(5.35)

Equations (5.14) and (5.35) form a set of three coupled equations that describe the beat-frequency mixing process for slowly varying  $A_1$  and  $A_2$ .

By inserting (5.35) into (5.14) it is found that:

$$\begin{bmatrix} \partial_{t} + c_{1} \partial_{x} + \Gamma_{1} \end{bmatrix} \mathbf{A}_{1}(\mathbf{x}, t) = - (\frac{1}{2}) \frac{\omega_{pe}}{\omega_{1}}^{2} \cdot \frac{1}{n_{o}} D(k_{3}, \omega_{3}) \mathbf{A}_{1} \mathbf{A}_{2} \mathbf{A}_{2}^{*}$$

$$\begin{bmatrix} \partial_{t} + c_{2} \partial_{x} + \Gamma_{2} \end{bmatrix} \mathbf{A}_{2}(\mathbf{x}, t) = - (\frac{1}{2}) \frac{\omega_{pe}}{\omega_{2}} \cdot \frac{1}{n_{o}} D^{*}(k_{3}, \omega_{3}) \mathbf{A}_{1} \mathbf{A}_{1}^{*} \mathbf{A}_{2}$$
(5.36)

Multiplying the top equation by  $A_1^*$  and the bottom equation by  $A_2^*$ , and adding each to its complex conjugate yields:

$$\begin{bmatrix} \partial_{t} + c_{1} \partial_{x} + 2\Gamma_{1} \end{bmatrix} |A_{1}(x,t)|^{2} = \frac{1}{\omega_{1}} (B - B^{*}) |A_{1}|^{2} |A_{2}|^{2}$$

$$\begin{bmatrix} \partial_{t} + c_{2} \partial_{x} + 2\Gamma_{2} \end{bmatrix} |A_{2}(x,t)|^{2} = -\frac{1}{\omega_{2}} (B - B^{*}) |A_{1}|^{2} |A_{2}|^{2}$$
(5.37)

where,

$$\beta = -\frac{1}{2} \omega_{pe}^{2} \cdot \frac{1}{n_{o}} D(k_{3}, \omega_{3})$$

Now using:

$$|A_{i}|^{2} = \frac{2\pi}{\omega_{i}|k_{i}|} I_{i}$$
 (5.38)

where  $I_i$  is the beam intensity;

$$\begin{bmatrix} \partial_{t} + c_{1} \partial_{x} + 2\Gamma_{1} \end{bmatrix} \mathbf{I}_{1} = \frac{2\pi \mathbf{i}}{\omega_{1} \omega_{2} |\mathbf{k}_{2}|} \quad (\beta - \beta^{\star}) \mathbf{I}_{1} \mathbf{I}_{2}$$

$$[\partial_{t} + c_{2}\partial_{x} + 2\Gamma_{2}] I_{2} = - \frac{2\pi i}{\omega_{1}\omega_{2}|k_{1}|} (\beta - \beta^{*}) I_{1}I_{2}$$

or if,

.

$$\alpha = \frac{2\pi 1}{\omega_{1}\omega_{2}|k_{2}|c_{1}} \qquad (\beta - \beta^{\star}) \qquad (5.39)$$

$$[\partial_{t} + c_{1}\partial_{x} + 2\Gamma_{1}] I_{1} = \alpha c_{1} I_{1}I_{2} \qquad (5.40)$$

$$[\partial_{t} + c_{2}\partial_{x} + 2\Gamma_{2}] I_{2} = - |\frac{k_{2}}{k_{1}}| \alpha c_{1} I_{1}I_{2}$$

The coupling coefficient  $\alpha$  can be rewritten as:

1 m

$$\alpha = \frac{2\pi\omega_{pe}^{2}}{n_{o}\omega_{1}\omega_{2}k_{2}c_{1}} \operatorname{Im} \{D(k_{3},\omega_{3})\}$$
(5.41)

The parameter a must be multiplied by  $10^{-7}$  if I is expressed in  $W/cm^2$ .

The rate that energy is deposited in the plasma due to beat frequency mixing,  $W_3$ , can be calculated from the conservation of energy principle through the use of equations (5.40) and (5.41). The deposited energy  $W_3$  is set equal to the difference in the transfer rate of energy from the first beam and the transfer rate of energy into the second beam. The energy density of a beam in the plasma is  $E_1 = I_1/|c_1|$ ergs/cm<sup>3</sup>. Neglecting collisional absorption of the beams, equations (5.40) can be rewritten as:

$$(\partial_{t} + c_{1}\partial_{x}) E_{1} = \alpha I_{1}I_{2}$$
  
 $(\partial_{t} + c_{2}\partial_{x}) E_{2} = - |\frac{k_{2}}{k_{1}}| |\frac{c_{1}}{c_{2}}| \alpha I_{1}I_{2}$ 

The left hand side of these equations is clearly the rate of energy transfer into the beams. Thus, from the definition of  $W_3$ :

$$W_3 = - [(\partial_t + c_1 \partial_x) E_1 + (\partial_t + c_2 \partial_x) E_2]$$

$$W_3 = -\alpha I_1 I_2 (1 - |\frac{k_2}{k_1}| |\frac{c_1}{c_2}|)$$

Using  $c_1 = \frac{c^2 k_1}{\omega_1}$ ,  $|\frac{k_2}{k_1}| |\frac{c_1}{c_2}|$  can be rewritten as  $\frac{\omega_2}{\omega_1}$ . Then:

٥r

$$W_3 = -\frac{w_3}{w_1} = I_1 I_2 = -\frac{w_3 c_1}{c^2 k_1} = I_1 I_2$$

Using the definition of a from equation (5.41) this expression becomes:

$$W_{3} = \frac{-2 \pi \omega_{pe}^{2} \omega_{3}}{n_{o}c^{2} \omega_{1} \omega_{2} k_{1} k_{2}} Im\{D(k_{3}, \omega_{3})\} I_{1}(x, t) I_{2}(x, t)$$
(5.42)

It is interesting to note that equation (5.42) can also be derived by the more direct method of setting the plasma heating rate to the ponderomotive force on the electrons times the electron particle flux. The electron particle flux I can be found from the continuity equation:

$$\frac{\partial}{\partial t} n_{e}(x,t) = -\frac{\partial}{\partial x} I_{e}(x,t)$$
 (5.43)

and equation (5.30):

$$I_{e}(k,\omega) = -\frac{i\omega}{4\pi e^{2}} - \frac{\chi_{e}(1 + b \chi_{i})}{1 + b \chi_{e} + b \chi_{i}} F_{e}(k,\omega)$$

or, using (5.32):

$$I_{e}(k,\omega) = \frac{i\omega m_{e}c^{2}}{k k_{3}e^{2}} D(k,\omega) F_{e}(k,\omega) \qquad (5.44)$$

For slowly varying  $F_e(x,t)$ ,  $F_e(k,\omega)$  is sharply resonant in k and  $\omega$ , and the inverse transform of (5.44) can be approximated so:

$$I_{e}(x,t) = I_{e}(x,t) \exp\{i(k_{3}x - \omega_{3}t)\} + c.c.$$
 (5.45)

where,

The main the barres

$$\tilde{I}_{e}(x,t) = \frac{1 \omega_{3} \omega_{e}^{2}}{k_{3}^{2} e^{2}} D(k_{3}, \omega_{3}) \tilde{F}_{e}(x,t)$$
(5.46)

and,

$$\hat{\mathbf{F}}_{e}(\mathbf{x},t) = -\frac{ie^{2}}{n_{e}c^{2}} \mathbf{k}_{3} \mathbf{A}_{1}(\mathbf{x},t) \mathbf{A}_{2}^{e}(\mathbf{x},t)$$
(5.47)

The rate that energy is deposited in the plasma,  $W_3$ , is set to the ponderomotive force on the electrons times the electron particle flux.

.

. .

$$W_3 = 2 \operatorname{Re}\{I_e F_e\}$$
 (5.48)  
ing equations (5.46) and (5.47) in equation (5.48) yields:

$$W_{3} = 2 \operatorname{Re} \left\{ \frac{i \omega_{3} e^{2}}{m_{e} c^{2}} D(k_{3}, \omega_{3}) A_{1} A_{1}^{*} A_{2} A_{2}^{*} \right\}$$

Using equation (5.38)

$$W_3 = 2 \operatorname{Re} \left\{ \frac{4\pi^2 i\omega_3 e^2}{m_e c^2 \omega_1 \omega_2 k_1 k_2} D(k_3, \omega_3) \right\} I_1(x, t) I_2(x, t)$$

٥r

$$W_{3} = -\frac{\frac{2\pi\omega_{pe}^{2}\omega_{3}}{n_{o}c^{2}\omega_{1}\omega_{2}k_{1}k_{2}}}{I_{a}(x,t)} I_{a}(x,t) I_{a}(x,t) I_{a}(x,t)$$

which is identical to (5.42).

It is interesting to compare the results of this derivation with the results of Capjack and James 30 where a different type of derivation was used. For this comparison collisional effects are neglected since they are accounted for in a different manner in the two

treatments. The present work accounts for electron-ion collisions in a more consistent manner. First, the beat frequency electric field  $E_3 \equiv (E_{ce} + E_{ci})$  is found. From equations (5.28):

$$E_{3}(\omega,k) = \frac{\chi_{e}}{1 + \chi_{i} + \chi_{e}} - \frac{F_{e}}{e}$$
(5.49)

where b has been set to 1 and  $F_i$  has been set to zero. Again, if  $F_e(\omega, k)$  is sharply resonant around  $\omega = \pm \omega_3$  and  $k = \pm k_3$  the inverse Fourier transform of (5.49) can be approximated.

$$E_3(x,t) = \tilde{E}_3(x,t) \exp[i(k_3x - w_3t)] + c.c.$$
 (5.50)

where

8

$$\tilde{E}_{3}(x,t) = \frac{x'_{e}}{1 + x'_{2} + x'_{e}} - \frac{\tilde{F}_{e}(x,t)}{e}$$
 (5.51)

N

$$\chi'_{\alpha} = \chi_{\alpha}(\omega_3, k_3)$$

and  $\tilde{F}_{e}(\mathbf{x},t)$  is defined in (5.47).  $\tilde{F}_{e}(\mathbf{x},t)$  must be cast in a different form to compare with the results of Capjack and James. Assuming  $\omega_{i} \gg \omega_{pe}$  or  $\omega_{i} = ck_{i}$ , equation (5.38) can be approximated as:

$$|A_{i}| = (\frac{2\pi c}{\omega_{i}^{2}})^{1/2} \sqrt{I_{i}}$$

Then:

$$A_1 A_2 = \frac{2\pi c}{\omega_1 \omega_2} \sqrt{I_1 I_2} e^{i\phi}$$

N.

where  $\phi$  is a phase factor which can be neglected in this treatment.

Substitution of the above equation into equation (5.47) yields:

$$\hat{F}_{e}(x,t) = -i \frac{2\pi c e^{2}}{m_{e} c^{2}} \qquad \frac{k_{3}}{\omega_{1} \omega_{2}} \left[I_{1}(x,t) I_{2}(x,t)\right]^{1/2}$$

Again the assumption that  $\omega_i \gg \omega_{pe}$  or  $\omega_i = c k_i$  is made and  $(\omega_1 \omega_2)^{-1}$  is approximated as:

$$(\omega_1 \omega_2)^{-1} = \frac{\lambda_1 \lambda_2}{(2\pi c)^2}$$

Then:

$$\hat{F}_{e}^{(x,t)} = -\frac{ie^{2}}{2\pi c^{3}m_{e}} k_{3} \lambda_{1} \lambda_{2} \sqrt{I_{1}I_{2}}$$

or:

ພ່

$$\hat{F}_{e}(x,t)/e = -i(3.11 \times 10^{-16}) k_{3} \lambda_{1} \lambda_{2} \sqrt{I_{1}I_{2}}$$
(5.52)

where  $\lambda_{i}$  is expressed in  $\mu m$  and  $I_{i}$  is expressed in  $W/cm^{2}$ . From (5.26):

$$x'_{\mathbf{s}} = 2 \frac{\frac{\omega_{\mathbf{p}} \mathbf{\epsilon}}{2}}{k_{3}^{2}} \frac{1}{v_{\theta \epsilon}^{2}} \{1 + \mathbf{i} \alpha_{0}^{\mathbf{\epsilon}} \mathbf{F}_{0}^{\mathbf{\epsilon}}\}$$

After some algebra:

$$\frac{\chi'_{e}}{1 + \chi'_{e} + \chi'_{1}} = \frac{1 + i\alpha_{o}^{e} F_{o}^{e}}{[K^{2} + \{1 + i\alpha_{o}^{e} F_{o}^{e}\} + \frac{T_{e}}{T_{1}} \{1 + i\alpha_{o}^{1} F_{o}^{1}\}]}$$
(5.53)

where  $K \equiv k_3^{\lambda} \lambda_D$  and  $\lambda_D = v_{\theta_e}^{\lambda} / \sqrt{2\omega_{pe}}$  is the Debye wavelength. From equations (5.51), (5.52), and (5.53):

$$\widetilde{E}_{3}(\mathbf{x},t) = \frac{\mathbf{i}(3.11 \times 10^{-16})\mathbf{k}_{3} \lambda_{1}\lambda_{2}(1 + \mathbf{i}\alpha_{o}\mathbf{F}_{o}^{e})\sqrt{P_{1}P_{2}}}{[\mathbf{K}^{2} + \{1 + \mathbf{i}\alpha_{o}^{e}\mathbf{F}_{o}^{e}\} + \frac{\mathbf{e}}{\mathbf{T}_{i}}\{1 + \mathbf{i}\alpha_{o}^{i}\mathbf{F}_{p}^{i}\}\}}$$
(5.54)

For  $T_e = T_i$  this is in exact agreement with the value for the coulomb field found by Capjack and James<sup>30</sup>.

The plasma heating rate  $W_3$  can also be cast into a form similar to that of Capjack and James<sup>30</sup>. It is most convenient to express  $\tilde{I}_e(x,t)$  and  $\tilde{F}_e(x,t)$  in terms of  $\tilde{E}_{ce}(x,t)$  before using (5.48) to find  $W_3$ . By using the continuity equation:

$$\partial_n n_n(\mathbf{x}, \mathbf{t}) = -\partial_n I_n(\mathbf{x}, \mathbf{t})$$

and Gauss' Law:

$$\Theta = E_{x,t}(x,t) = -4\pi e n e$$

it is found that:

 $I_{e}(k,\omega) = -\frac{i\omega}{4\pi e} E_{ce}(k,\omega) \qquad (5.55)$ 

Taking the inverse Fourier transform yields:

$$\int_{e}^{v} (\mathbf{x}, t) = - \frac{i\omega_{3}}{4\pi e} \int_{e}^{v} E_{e}(\mathbf{x}, t)$$

where  $I_e^{(x,t)}$  is defined in (5.45).

Letting b=l and  $\chi_1=0$  in equation (5.28) and approximating the inverse Fourier transform yields:

$$\hat{F}_{e}(x,t) = \frac{\chi_{e}^{+} + 1}{\chi_{e}^{+}} e \tilde{E}_{ce}(x,t)$$
(5.56)

where  $\chi'_e \equiv \chi_e(k_3, \omega_3)$ . Substituting (5.55) and (5.56) into (5.48) yields:



$$W_3 = \frac{1}{2\pi} I_m(\omega_3 \chi_e^{+}) \frac{\left| \frac{E_{ce} \right|^2}{|\chi_e^{+}|^2}}{|\chi_e^{+}|^2}$$

Substitution of values for  $\chi_e^1$  yields:

$$W_{3} = \frac{\omega_{pe}^{2} \alpha_{oe}^{2}}{k_{3}^{v} \theta e^{\pi}} \sqrt{\pi} \operatorname{sgn}(k_{3}) e^{-\alpha_{oe}^{2}} \frac{|E_{ce}|^{2}}{|x_{a}'|^{2}}$$
(5.57)

The heating rate given by (5.57) is smaller by a factor of  $|\chi_e^+|^{-2}$  from that derived by Capjack and James<sup>30</sup>. For a cold plasma  $(\alpha_{oe} \ll 1)$  the two results differ by a factor of  $(\omega_3/\omega_{pe})^4$ . The discrepancy can be attributed to an error in the definition of the heating rate used by Capjack and James<sup>30</sup>.

### 5.4 Transient Analysis

In cases where the amplitude of one of the electromagnetic waves changes rapidly in time the quasi-static approximation breaks down. In this section, equations (5.14) and (5.34) are cast in a form such that they can easily be solved numerically.

$$\begin{bmatrix} \partial_{t} + c_{1} \partial_{x} + \Gamma_{1} \end{bmatrix} \mathbf{A}_{1}(\mathbf{x}, t) = - (\frac{1}{2}) \frac{\omega_{pe}^{2}}{\omega_{1}} \frac{\tilde{\mathbf{n}}_{e}(\mathbf{x}, t)}{n_{o}} \mathbf{A}_{2}(\mathbf{x}, t)$$

$$\begin{bmatrix} \partial_{t} + c_{2} \partial_{x} + \Gamma_{2} \end{bmatrix} \mathbf{A}_{2}(\mathbf{x}, t) = - (\frac{1}{2}) \frac{\omega_{pe}^{2}}{\omega_{2}} \frac{\tilde{\mathbf{n}}_{e}^{*}(\mathbf{x}, t)}{n_{o}} \mathbf{A}_{1}(\mathbf{x}, t)$$

$$\tilde{\mathbf{n}}_{0}(\mathbf{x}, t) = e^{-\mathbf{i} \{\mathbf{k}_{3} \mathbf{x} - \omega_{3} t\}} \frac{1}{2} = \int_{0}^{\infty} \int_{0}^{\infty} \mathbf{D}(\mathbf{k}, \omega) f(\mathbf{k}, \omega) e^{\mathbf{i} (\mathbf{k} \mathbf{x} - \omega t)} d\mathbf{k} d\omega$$

$$\tilde{n}_{e}(\mathbf{x},t) = e^{-1(k_{3}\mathbf{x}-\omega_{3}t)} \cdot \frac{1}{2\pi} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} D(k,\omega)f(k,\omega)e^{1(kx-\omega t)}dkd\omega$$
(5.34)

where:

$$f(k,\omega)' = \frac{1}{2\pi} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} A_1(x,t) A_2^{\dagger}(x,t) e^{i(k_3x-\omega t)} e^{-i(kx-\omega t)} dx dt$$

and

$$D(k,\omega) = -\frac{k k_3}{4\pi m_e c^2} \frac{\chi_e (1 + b \chi_1)}{(1 + b\chi_e + b\chi_1)}$$
(5.32)

where  $\chi_e$  and  $\chi_i$  are the usual electron and ion linear susceptabilities as calculated from the linear Vlasov equations. The validity of these equations is dependent on the following inequalities.

$$\begin{vmatrix} \partial_{t} A_{i}(\mathbf{x},t) \end{vmatrix} << |\omega_{i} \cdot A_{i}(\mathbf{x},t)|$$

$$i=1,2$$

$$\begin{vmatrix} \partial_{\mathbf{x}} A_{i}(\mathbf{x},t) \end{vmatrix} << |k_{i} \cdot A_{i}(\mathbf{x},t)|$$

Since for the mixing of antiparallel beams  $|k_3| = |k_1| + |k_2|$ , the

.• J

second of the above inequalities implies that in the region of validity of the above equations  $f(k,\omega)$  will be sharply resonant in k, and has significant values only for k near  $\pm k_3$ . The function  $f(k,\omega)$  is not necessarily sharply resonant in  $\omega$ .

As an example consider the mixing of a beam of wavelength 10.6µm with a pulse with a wavelength of 10.612µm. For this case the frequency and wavevector of the driven electrostatic wave almost satisfies the dispersion relation of an ion-acoustic wave in a 100eV plasma. The frequency  $\omega_3 = 2 \times 10^{11} \text{ sec}^{-1}$  and the wavevector  $k_3 = 1.2 \times 10^4 \text{ cm}^{-1}$  or  $T_3 = 3.1 \times 10^{-11} \text{ sec}$ , and  $\lambda_3 = 5.3 \times 10^{-4} \text{ cm}$  where  $T_3$ and  $\lambda_3$  are the period and wavelength of the driven electro-static wave. If the pulse is  $10^{-11}$  seconds or 0.3 cm long,

$$\Delta \omega \propto \frac{1}{\Delta t} = 10^{11} + \Delta \omega / \omega_3 = 0.5$$

$$\Delta \mathbf{k} \alpha \frac{1}{\Delta \mathbf{x}} = 3.3 + \Delta \mathbf{k}/\mathbf{k}_3 = 2.8 \times 10^{-4}$$

Such a pulse is highly resonant in k but the frequency spread is almost as large as the frequency itself.

When  $f(k,\omega)$  is sharply resonant in k around  $k_3$  so that D(k, $\omega$ ) does not change much over the range of k, the k integration in (5.34) can be approximated:

$$\hat{n}_{e}(\mathbf{x},t) = e^{-\mathbf{i}(\mathbf{k}_{3}\mathbf{x}-\mathbf{\omega}_{3}t)} \frac{1}{\sqrt{2\pi}} \int_{-\infty}^{\infty} f(\mathbf{x},\mathbf{\omega})D(\mathbf{k}_{3},\mathbf{\omega})e^{-\mathbf{i}\omega t} d\omega \quad (5.58)$$

where

$$f(\mathbf{x},\omega) \equiv \frac{1}{\sqrt{2\pi}} \int_{-\infty}^{\infty} f(\mathbf{k},\omega) e^{\mathbf{i}\mathbf{k}\cdot\mathbf{x}} d\mathbf{k}$$
 (5.59)

$$\hat{n}_{e}(\mathbf{x},t) = e^{-\mathbf{i}(\mathbf{k}_{3}\mathbf{x}-\mathbf{u}_{3}t)} \frac{1}{\sqrt{2\pi}} \int_{-\infty}^{t} f(\mathbf{x},\tau) D(\mathbf{k}_{3},t-\tau)d\tau \qquad (5.60)$$

where

$$D(k_{3},t) = \frac{1}{\sqrt{2\pi}} \int D(k_{3},\omega) e^{-i\omega t} d\omega$$
 (5.61)

Now if

 $\hat{n}_{e}(x,t)$  can be expressed as:

$$\hat{n}_{e}(\mathbf{x},t) = e^{i\omega_{3}t} \frac{1}{\sqrt{2\pi}} \int_{-\infty}^{t} \hat{f}(\mathbf{x},\tau) D(k_{3},t-\tau) d\tau \qquad (5.63)$$

In principle equations (5.14) and (5.63) can be solved self-consistently to yield an approximate solution to the evolution of two interacting beams.

In order to numerically calculate the solution to these equations for a short pulse, it is most convenient to transform the equations into a coordinate system where the pulse is stationary. The appropriate variables are t' and x' where:

$$t' = t \qquad \partial_t \neq \partial_t - c_2 \partial_x'$$

$$x' = x - c_2 t \qquad \partial_x \neq \partial_x'$$
(5.64)

Equations (5.14) transform to:

3

$$[\partial_{t'} + (c_1 - c_2)\partial_{x'} + \Gamma_1]A_1'(x', t') = R_1 \tilde{n}_e''(x', t')A_2'(x', t') \quad (5.65)$$
$$[\partial_{t'} + \Gamma_2]A_2'(x', t') = R_2 \tilde{n}_e^{*'}(x', t')A_1'(x', t') \quad (5.66)$$

where

$$R_{1} = -(\frac{1}{2}) \frac{\frac{\omega^{2}}{pe}}{\omega_{1}^{n}o}$$

$$R_{2} = -(\frac{1}{2}) \frac{\frac{\omega^{2}}{pe}}{\omega_{2}^{n}o}$$
(5.67)

The variables  $A_1, A_2$ , and ne have been primed to emphasize the fact that they have a new functional form in the new coordinate system.

In terms of the new variables, equation (5.63) can be written as:

$$\sum_{n_{e}}^{\nu} (\mathbf{x}' + c_{2}t', t') = e^{i\omega_{3}t'} \frac{1}{\sqrt{2\pi}} \int_{-\infty}^{t'} f(\mathbf{x}' + c_{2}t', \tau) D(k_{3}, t' - \tau) d\tau$$

The functional dependence of,  $n_{\mathbf{n}}$  and f is changed in the primed coordinate system so that:

$$\hat{n}_{e}^{\nu}(\mathbf{x}^{\prime}, t^{\prime}) = \hat{n}_{e}^{\nu}(\mathbf{x}^{\prime} + c_{2}t^{\prime}, t^{\prime})$$

$$\hat{f}^{\prime}(\mathbf{x}^{\prime}, t^{\prime}) = \hat{f}(\mathbf{x}^{\prime} + c_{2}t^{\prime}, t^{\prime}) \text{ or } \hat{f}(\mathbf{x}^{\prime}, t^{\prime}) = \hat{f}^{\prime}(\mathbf{x}^{\prime} - c_{2}t^{\prime}, t^{\prime})$$

$$\hat{f}(\mathbf{x}^{\prime} + c_{2}t^{\prime}, \tau) = \hat{f}^{\prime}(\xi, \tau) = \mathbf{A}_{1}^{\prime}(\xi, \tau) \mathbf{A}_{2}^{\nu}(\xi, \tau) \mathbf{e}^{-i\omega_{3}t^{\prime}}$$
(5.68)

Thus,

where  

$$t = x' + c_2(t' - \tau)$$
The integral equation for  $\hat{n}_e$  can now be expressed as:  
 $\hat{n}_e'(x', t') = e^{i\omega_3 t'} \frac{1}{\sqrt{2\pi}} \int_{t'+x'/c_2}^{t'} \hat{f}'(\xi, \tau) D(k_3, t' - \tau) d\tau$ 
(5.69)

The lower limit on the integral comes from assuming  $A_2^{\prime}(x^{\prime},t^{\prime}) = 0$  for  $x^{\prime} < 0$  and also assuming that  $c_2 < 0$  and  $c_1 > 0$ . Substitution of (5.68) into (5.69) yields:

$$\hat{n}_{e}^{\prime}(\mathbf{x}',t') = \frac{1}{\sqrt{2\pi}} \int_{t'+\mathbf{x}'/c_{2}}^{t'} A_{1}^{f}(\xi,\tau) A_{2}^{*'}(\xi,\tau) e^{i\omega_{3}(t'-\tau)} D(k_{3},t'-\tau) d\tau$$

Changing the variable of integration from  $\tau$  to  $\tau' = t' - \tau$  yields:

$$\sum_{n}^{n} (\mathbf{x}', t') = \frac{1}{\sqrt{2\pi}} \int_{0}^{-\mathbf{x}'/c_{2}} \mathbf{A}_{1}'(\mathbf{x}' + c_{2}\tau', t' - \tau') \mathbf{A}_{2}'^{*}(\mathbf{x}' + c_{2}\tau', t' - \tau') \cdot \frac{i\omega_{3}\tau'}{e} D(k_{3}, \tau') d\tau'$$
(5.70)

Equations (5.65), (5.66) and (5.70) form a set of coupled

equations from which an approximate solution can be obtained numerically. A numerical solution to these equations will be presented in Chapter 6.

### 5.5 Results

In this section the dependence of the beat frequency mixing rate (defined through the parameter  $\alpha$ ) on plasma parameters and the laser difference frequency is displayed graphically. The figures in this chapter were chosen to show the dependence of  $\alpha$  on plasma parameters, and to provide numerical data that will be needed in Chapter 6.

Figures 5.1, 5.2, and 5.3 show the beat frequency mixing rate a, the electron density fluctuation level  $\tilde{N}_{e}$ , and the ion density fluctuation level  $\tilde{N}_{i}$ , as a function of the laser difference frequency  $\omega_{3}$ . Here the density fluctuation levels are normalized to be independent of laser beam intensities. That is:

$$\hat{\mathbf{N}}_{\varepsilon} \equiv \frac{2\left|\hat{\mathbf{n}}_{\varepsilon}\right|}{\sqrt{\mathbf{I}_{1}\mathbf{I}_{2}}\mathbf{n}_{o}}$$

where  $I_1$  and  $I_2$  are expressed in  $W \cdot cm^{-2}$ . All of these plots show a sharp peak in a, and  $\tilde{N}_e$  for  $\omega_3$  close to the Langmuir frequency. In Figure 5.1,  $T_e \gg T_1$  and there is a sharp peak in  $\tilde{N}_e$ ,  $\tilde{N}_1$  and a for  $\omega_3$ close to the ion-acoustic frequency. It is interesting to note that in Figures 5.2 and 5.3, the plasma density fluctuations are almost independent of  $\omega_3$  over a wide range while the parameter a increases monotonically with increasing  $\omega_3$ . Another interesting feature of these plots is the sharp minimum in  $\tilde{N}_e$  and a for  $\omega_3$  close to  $\omega_{pi}$  where  $\omega_{pi} = (4\pi n_0 e^2/m_1)^{1/2}$ is the ion plasma frequency. The minimum results from the fact that near the ion plasma frequency, the ions move in such a way that their coulomb field cancels the driving field on the electrons.

Figures 5.4, 5.5, and 5.6 show a versus n for various

difference frequencies, and plasma temperatures. These plots show that at higher temperatures, where Landau damping is strong, a is not nearly as strongly dependent on density as it is at lower temperatures where damping is weaker. Also, for larger difference frequencies, higher temperatures are needed to broaden the resonance.

Figures 5.7, 5.8, and 5.9 show the dependence of a on density and temperature for beat frequency mixing near the ion-acoustic frequency. In these plots it was assumed that  $\lambda_1 = 10.247$  µm and  $\lambda_2 = 10.2605$ µm. These wavelengths correspond to the R(20) and R(18) transitions in a CO<sub>2</sub> laser respectively. Figure 5.7 shows that a increases monotonically with increasing density over a wide range of densities. Figures 5.8 and 5.9 show a as a function of temperatures. For cases of  $T_i \ll T_e$  sharp peaks are seen in a for  $\omega_3$  close to the ion-acoustic frequency.





Ķ.









eV is indicated











#### CHAPTER 6

# Some Applications of Beat Frequency Mixing

The beat frequency mixing of antiparallel laser beams leads to energy being transferred from the high frequency electromagnetic wave to the lower frequency electromagnetic wave and to the beat frequency electrostatic wave. The electrostatic wave is damped by collisional, and collisionless mechanisms and is eventually converted to plasma thermal energy. The ratio of energy transferred to the electromagnetic wave to energy transferred to the electrostatic wave is given by the Manley-Rowe relations (5.17) as  $(w_2/w_3)$ . This process has often been considered as a possible mechanism for heating underdense plasmas. From the Manley-Rowe relations beat frequency heating can be much more efficient for  $w_3$  close to the Langmuir frequency than it is for  $w_3$  close to the ion-acoustic frequency.

In this chapter, some applications of the amplification of the lower frequency electromagnetic wave will be considered. A device is proposed which could be capable of efficiently amplifying coherent radiation in the ll-14µm range. This same device could also be capable of amplify very short pulses of infrared radiation.

## 6.1 A Plasme-Laser Amplifier in the 11-14µm Wavelength Range

A laser amplifier in the infrared is of interest for applications in laser heating of plasmas confined magnetically in solenoids and recently for nuclear isotope separation. In this study a laser amplifier is proposed which relies on a 10.6 $\mu$ m laser to serve as a pump for amplifying laser beams in the range 11-14  $\mu$ m with a high, conversion efficiency. The proposed scheme is shown to be ideally suited for amplification of radiation in the range 12-13 $\mu$ m which includes the 12.1 $\mu$ m line which is of interest in nuclear isotope separation (see Jensen et al.<sup>32</sup>).

The basis of operation for the proposed amplifier is the nonlinear interaction in a plasma between a strong 10.6µm laser beam which serves as a pump and a lower frequency laser beam which is to be amplified. The two beams force an electron plasma wave at their beat frequency.

The analysis of Chapter 5 shows that for a plasma with a temperature greater than  $\sim 100$  eV, the mechanism for the transfer of energy to the lower frequency laser beam is very insensitive to variations in the plasma density and temperature. That is, this technique for transferring energy to a lower frequency laser beam does not need to be restricted only to cases where the difference frequency of the laser beams is very close to the Langmuir frequency. Based on this observation, a laser amplifier in the far infrared with plasma serving as a nonlinear mixing medium appears experimentally realizable. The proposed scheme is essentially a parametric amplifier with the crystal replaced by a plasma.

.

than perametric amplifiers which use a crystal as their been mixing medium. The amount of energy transferred from the pump to the lower frequency waves is found to obey a Manley-Rowe type of relationship, equation (5.17).

$$\frac{W_1}{\omega_1} = \frac{W_2}{\omega_2} = \frac{W_3}{\omega_3}$$

where  $W_1$  is the energy transferred from the high frequency beam,  $W_2$  and  $W_3$  are the energies transferred into the low frequency beam and the plasma wave respectively.  $\omega_1$  and  $\omega_2$  are the frequencies of the high and low frequency electromagnetic beams respectively and  $\omega_3^{-\omega_1^{-\omega_2^{-1}}}$  is the frequency of the driven plasma wave.

A schematic of a simple optical amplifier is shown in Figure 6.1(a). In this device plasma is radially confined by a solenoidal magnetic field. Even though the 10.6µm laser beam enters the solenoid at a slight angle it will be guided down the axis of the solenoid due to a density minimum on the axis. Such a density minimum could be created during the plasma formation or by the CO<sub>2</sub> laser itself (see Steinhauer<sup>33</sup>, 1976). The required physical length of such a device is dependent on the plasma density and temperature profiles as well as the wavelength and initial intensity of the beam to be amplified. These factors are discussed in detail later.

Certain advantages can be gained by placing the amplifier in an optical resonator. Two possible configurations are shown in Figures 6.1(b) and 6.1(c). The mirror configuration in Figure 6.1(b) is the

same as that of a laser. Part of the low frequency been is reflected back into the plasma by partially reflecting mirror M2. The low frequency beam propagating antiparallel to the 10.6um beam is amplified at the expense of the 10.6µm beam. The portion of the low frequency beam propagating parallel to the 10.6um beam does not intriject with the 10.6µm beam (see Capjack and James<sup>30</sup>). An optical resonator enables the low frequency beam (frequency  $u_2$ ) to be of relatively high intensity and thus enhances the energy transfer process. A second major advantage of placing the amplifier in an optical resonator is that the low frequency beam can continue to be emitted long after the initial low frequency beam has shut off. Thus a short Q-switched pulse signal can be used to start the process which will then continue long after the initial low frequency signal has died away. There is also a possibility that the low frequency signal could be generated spontaneously, thus eliminating the need for the injected signal. In this case however precise frequency control could not be achieved. The principle of operation of the device with the mirror configuration shown in Figure 6.1(c) is much the same as that of Figure 6.1(b). The main difference is that the low frequency beam now only traverses the plasma in one direction. This has the advantage of reducing the absorption of the beam by the plasma through inverse Bremsstrahlung absorption.

For either of the mirror configurations the ratio of the intensity of the low frequency beam as it enters the plasma to that of the high frequency beam as it enters the plasma is given by:

**\*** 

 $\frac{I_{2}^{(L)}}{I_{1}(0)} = \frac{u_{2}}{u_{1}} \frac{R}{(1-R)}$ 

where R is the fraction of energy reflected by the partially reflecting mirror,  $I_1(x)$  and  $I_2(x)$  are the intensities of the pump beam and the beam to be amplified respectively in  $W/cm^2$ , and L is the length of the solenoid. In deriving this relation it has been assumed that inverse Bremsstrahlung absorption is negligible and that all of the energy in the high frequency wave has been transferred to the two lower frequency waves.

li

Optimal efficiency in energy conversion from the high frequency beam can be achieved by making the physical length of the solenoid longer than the absorption length of the first beam. By neglecting absorption through inverse Bremsstrahlung (valid for high plasma temperatures since absorption length as a  $T_e^{3/2}$  dependence) a theoretical estimate of the spatial dependence of the intensities of the two beams in the plasma can be written as:

$$\frac{\partial I_1(\mathbf{x})}{\partial \mathbf{x}} = -\alpha I_1(\mathbf{x}) I_2(\mathbf{x}), \qquad (6.1)$$

$$\frac{\partial I_2(\mathbf{x})}{\partial \mathbf{x}} = -\frac{\omega_2}{\omega_1} \alpha I_1(\mathbf{x}) I_2(\mathbf{x})$$

These equations are obtained from (5.40) by assuming  $\Gamma_1 = \Gamma_2 = 0$ ,  $c_1 = -c_2 = c_2$ and  $|k_2/k_1| = (\omega_2/\omega_1)$ . The last two relations are valid if  $\omega_{1,2}^{>>\omega}$  pe

Note that the high frequency beam propagates in the +x direction and is attenuated while the low frequency beam propagates in the -x direction and grows. Solving (6.1) for  $I_1(x)$ 



$$I_{1}(x) = \frac{abI_{10}(0)exp(-abx)}{ab+a(w_{2}/w_{1})I_{1}(0)[1-exp(-abx)]}$$
(6.2)

where  $b = [I_2(L) - (\omega_2/\omega_1)I_1(L)].$ 

Theoretical estimates of the parameter a have been made in Chapter 5. Figure 5.5 indicates that for a 12 um beam mixing with a 10.6 um beam, a could be expected to be greater than  $10^{-12}$ for a density near  $10^{17}$  cm<sup>-3</sup> and a temperature near 200 eV. Figure 5.6 shows that much higher temperatures and densities are needed to achieve a broad resonance and high value of a for the amplification of a 14 µm line. However high temperatures may be achieved in such a device since the plasma is heated by both inverse Bremsstrahlung absorption of the electromagnetic waves and by the beat frequency mixing process.

If a 14 µm beam is being amplified and  $\alpha = 10^{-12}$ , equation (6.2) predicts that the first beam will be 90% absorbed in 20 cm if both beams enter the plasma with an intensity of  $10^{11}$  W/cm<sup>2</sup>. If the first beam enters the plasma with an intensity of  $10^{12}$  W/cm<sup>2</sup> and the second beam enters with an intensity of  $10^{9}$  W/cm<sup>2</sup> the first beam will be 90% absorbed in 60 cm.

This work could be extended through the development of a computer subroutine which numerically solves equations (5.40) in a time and spatially varying plasma and which is then interfaced with the two dimensional simulation described in Chapter 2. In this way the plasma parameters could be calculated self-consistently with the beat frequency heating rate.

### 6.2 Short Pulse Amplification

In this section, the possibility of using the best frequency mixing process to amplify and temporally shorten pulses of infrared Imper radiation is considered. Consideration is given to cases where the heat frequency is close to the electron plasma frequency and to cases where the beat frequency is close to the ion-acoustic frequency. The basis of operation for such a device is as follows: A signal oscillator injects a pulse of electromagnetic radiation into the plasma antiparallel to the pump beam. The beat frequency mixing process then causes an energy transfer out of the pump beam, and into the signal to be amplified and the electrostatic mixing mode. If this mixing occurs rapidly the pump beam may be rapidly depleted causing only the leading edge of the signal beam to be amplified. This leads to a comparatively long signal pulse growing into a sharp burst of intense radiation. Also considered is the case where this device is used to amplify an ultrashorf ( $\lesssim$  100ps) pulse of laser radiation. Ways of generating such short pulses have been developed by Yablonovitch and Goldhar<sup>34</sup>, and later by Fisher and Fieldman<sup>35</sup>

In the previous section it was suggested that a plasma could be a useful nonlinear medium for the parametric amplification of infrared radiation. This analysis assumed a steady state condition (no temporal dependence) for the plasma and interacting beams and consequently is not valid for describing the amplification of short pulses of radiation. While the former analysis considered only the case where the beat frequency mode was an electron plasma mode, the present analysis concentrates more on the case where the beat frequency mode is an ionacoustic mode.

Again, Figure 6.1(a) can serve as a schematic diagram for the proposed amplifier. For purposes of this thesis, it is assumed that a pulse of radiation with a wavelength of approximately 10.6µm is to be amplified. If the pump beam has a wavelength of approximately 9.6µm and the pulse to be amplified has a wavelength of 10.6µm, the beat frequency of these two waves is close to the electron plasma frequency of a plasma of density  $10^{17}$  cm<sup>-3</sup>. If the appropriate plasma conditions exist an electron plasma wave can be driven and the 10.6µm pulse is amplified at the expense of the 9.6µm beam. Similarly, if the pump beam has a wavelength of 10.247µm (R(20) transition for a CO<sub>2</sub> laser), and the pulse to be amplified has a wavelength of 10.2605µm (R(18) transition for a CO<sub>2</sub> laser), the beat frequency of these two waves is close to the ion-acoustic frequency of a plasma of 350eV. If the appropriate plasma conditions exist a large ion-acoustic wave can be driven as a beat frequency mixing mode.

The energy that is transferred from the pump beam to the pulse to be amplified is dependent on the device length, the mixing rate, the frequencies of the two beams as well as inverse Bremsstrahlung absorption rates. If inverse Bremsstrahlung absorption is neglected, and it is assumed that the mixing rate is sufficiently rapid so that essentially all of the energy is transferred out of the higher frequency beam, the principle of conservation of energy and the Manley-Rowe relations yield the following expression for the energy gained by the pulse.

$$\Delta E = 2I_1 A \frac{\omega_2}{\omega_1} \qquad \frac{1}{c} L \qquad (6.3)$$

where  $\Delta E$  is the energy gained by the pulse,  $I_1$  is the average crosssectional intensity of the pump beam, A is the cross-sectional area of the pump beam, c is the speed of light and L is the plasma length. In the design of this device, the device length, the pump beam intensity, and the plasma conditions must be chosen in such a way that Brillouin backscatter is not a serious problem. Since this device is intended to amplify back-travelling radiation, threshold conditions for parametric instabilities must, by definition, be exceeded. Since these instabilities must grow from thermal plasma fluctuations (which may be incoherent) no problem will be created if the device is sufficiently short. Exactly what conditions must be met before these instabilities create a problem cannot be determined at the present time. For the purposes of this analysis, it is assumed that a sufficient condition to prevent problems with parametric instabilities is:

$$\exp(\alpha_{I_{1}}L) \lesssim 10^{4}$$
(6.4)

where  $\alpha_s$  is the growth rate for the back-scattered radiation.  $\alpha_s$  can be found as the maximum on a curve of  $\alpha$  versus  $\omega_3$  for given plasma conditions. Since this analysis uses the linearized Vlasov equation, we must be careful to ensure that plasma and beam conditions are always such that:

$$\frac{n}{n} << 1$$
(6.5)

where  $n_e$  is the magnitude of the density fluctuations due to the beat frequency mixing process and  $n_o$  is the equilibrium density.

Again, certain advantages can be gained by placing the amplifier in an optical resonator such as those illustrated schematically in Figures 1(b) and 1(c). This can greatly increase the interaction length for a pulse allowing it to grow to larger amplitudes. Since higher amplitude pulses tend to sharpen quicker due to pump depletion, optical resonators could lead to shorter pulses. After the signal oscillator has injected one pulse into the amplifier a whole series of pulses would be emitted. The temporal separation of these pulses would be  $L_p/c$  where  $L_p$  is the optical pathlength of one round trip in the device. In order to prevent parametric instabilities from becoming a problem in this case, a saturable absorber would have to be placed in the path of the pulse.

# 6.2.1 Quasi-Static Analysis

From Section 5.3, equations (3.40) are valid if the resonance width of  $f(k, \omega)$  is much narrower than the resonance width of  $D(k, \omega)$  so that equation (5.34) can be approximated. The resonance width of  $D(k, \omega)$  is of the order of the damping rate of the electrostatic mode  $\gamma_L$ , and the resonance width of  $f(k, \omega)$  is  $\Delta \omega_3$ . Therefore, equations (5.4) are valid for:

$$\Delta \omega_3 << \gamma_L$$

 $\Delta \omega_{\gamma}$  can be approximated as:

$$\Delta \omega_3 = \Delta \omega_1 + \Delta \omega_2$$

where  $\Delta \omega_1$  is the frequency spread of the pump beam and  $\Delta \omega_2$  is the frequency spread of the pulse being amplified. Considering the pump beam and the pulse to have minimum frequency spreads,  $\Delta \omega_1 << \Delta \omega_2$  so:

Thus the quasi-static analysis is valid if:

 $\Delta \omega_2 << \gamma_L$   $\tau_B >> \gamma_L^{-1}$ 

(6.6)

۰,

or

where  $\tau_p$  is the temporal width of the pulse. When (6.6) is true, the polynomial of the pulse can be found by solving (5.40):

$$\begin{bmatrix} \partial_{t} + c_{1} & \partial_{x} + 2r_{1} \end{bmatrix} I_{1} = \alpha c_{1} I_{1}I_{2}$$

$$\begin{bmatrix} \partial_{t} + c_{2} & \partial_{x} + 2r_{2} \end{bmatrix} I_{2} = - \left| \frac{k_{2}}{k_{1}} \right| \alpha c_{1} I_{1}I_{2}$$
(6.7)

In this analysis it is assumed that the plasma is well underdense so that:

$$c_1 = -c_2 = c = 3 \times 10^{10} \text{ cm/sec}$$
  
 $\left| \frac{k_2}{k_1} \right| = \frac{\omega_2}{\omega_1}$ 

Equations (6.7) can be integrated numerically, however it is useful to notice that the amplification factor of the back travelling pulse in the pulse front (before pump depletion becomes important) is simply given by  $\exp[\alpha I_1 t]$  where t is the distance this portion of the pulse has travelled in the plasma.

Figure 5.4 is a plot of  $\alpha$  as a function of n for various plasma temperatures for the case of  $\lambda_1 = 9.6\mu$ m and  $\lambda_2 = 10.6\mu$ m. A value of  $\alpha > 4x10^{-12}$  is expected for the case of T close to 100eV and plasma density between  $7x10^{16}$  and  $8x10^{16}$  cm<sup>-3</sup>. Chapter 3 discusses a M.H.D. computer simulation of a laser heated plasma in a one meter solenoid with an initial density of  $7.8x10^{16}$  cm<sup>-3</sup>. After  $0.5\mu$ -sec of heating with a 1GW laser the plasma had an on axis density of  $7.5x10^{16}$  cm<sup>-3</sup> and an electron temperature between 95 and 100eV over a distance of more than

(6.8)
80cm. In other words, the plasma conditions are such that almost optimum values of a exist over most of the length of the one mater solenoid. Figure 5.3 shows a as a function of u<sub>3</sub> for these plasma conditions. It is seen that the pask value of a due to the Brillouin process is less than the value of a predicted for the Raman process. Consequently, Brillioun scattering would not be expected to be a problem for this case.

Another M.H.D. simulation reported in Chapter 3 is that of a laser heated plasms in a one meter solenoid and an initial density of  $3.0 \times 10^{17} \text{ cm}^{-3}$ . After one usec of heating with a one GW laser the plasms has a density between  $2.4 \times 10^{17}$  and  $2.5 \times 10^{17} \text{ cm}^{-3}$ , an electron temperature of between 120 and 180eV, and an ion temperature of between 100 and , 140eV over an 80cm length of the one meter solenoid. Figure 5.2 shows a as a function of  $w_3$  for these plasms conditions. It is seen that a large value of a is achieved due to mixing with the ion-acoustic mode if  $w_3$  is close to  $2.4 \times 10^{11} \text{ sec}^{-1}$ . Beat frequency mixing of  $CO_2$  laser beams with wavelengths of  $\lambda_1 = 10.2470$  (R(20)) and  $\lambda_2 = 10.2605$ , R(18)) yields  $w_3 = 2.42 \times 10^{11} \text{ sec}^{-1}$ . From Figures 5.2, 5.7, 5.8 and 5.9 it is seen that the resonance for this mode is very broad and a value of  $a > 4 \times 10^{-12}$  could be expected over at least 80cm of the 1 meter solenoid.

The above discussion indicates that suitable plasma conditions could be attained in a laboratory to yield values of  $\alpha > 4 \times 10^{-12}$ for the mixing of radiation with invelengths of either 9.6 m and 10.6 mm, or 10.247 m and 10.2605 mm. For either case the evolution of the interacting usves should be similar, as predicted by equations (6.7). A

condition for the validity of the quasi-static approximation is given by equation (6.6). The plasma conditions assumed for the beat frequency mixing of 9.6µm and 10.6µm redistion ( $T_e$ =100eV and  $n_e$ =7.75x10<sup>16</sup> cm<sup>-3</sup>) imply a value of  $k_3\lambda_p$ =0.33 and  $\gamma_L$ =3.8x10<sup>11</sup> sec<sup>-1</sup>. For the case of mixing radiation with wavelengths of 10.247µm a quasi mode (damping rate is almost as large as the frequency) is driven and  $\gamma_L$  3.3x10<sup>11</sup> sec<sup>-1</sup>. Thus for either of these cases the quasi-static approximation is valid if the temporal width of the pulse to be amplified,  $\tau_p$ , satisfies

$$t_p >> 3x10^{-12} \sec 5$$
 (6.9)

In order to illustrate the amplification of a pulse using this process, equations (6.7) have been integrated numerically assuming the following conditions:

$$\lambda_{1} = 9.6\mu m$$

$$\lambda_{2} = 10.6\mu m$$

$$n_{0} = 2.5 \times 10^{16} cm^{-y} ,$$
Plasma length = 80cm
Initial pump intensity = 1.6\times 10^{10} W/cm^{2}
Input pulse intensity = I<sub>2</sub>(t) where
$$I_{2}(t) = \begin{cases} 2\times 10^{9} I_{2p} \cdot t & \frac{1}{2} < .5\times 10^{-9} \sec t \\ 2\times 10^{9} I_{2p}(10^{-9} - t) & .5\times 10^{-9} < t < 10^{-9} \\ 0 & t > 10^{-9} \sec t \end{cases}$$

That is the injected pulse is of triangular form rising to a peak power

X

161

;\*

of  $I_{2p}$  W/cm<sup>2</sup> at 0.5 ns and falling off to zero again at 1 ns. The above parameters yield  $\alpha$ =5.7x10<sup>-12</sup> and in the absence of pump depletion the whole pulse would be amplified by a factor of 1500?

£

The results of these simulations are illustrated graphically in Figures 6.2, 6.3, and 6.4. In each of these figures the intensity is plotted on a logarithmic scale as a function of axial distance down the solenoid axis at three different times. Plasma is assumed to exist only in the interval 0-80cm and a is therefore set to zero in the interval of 80-120cm (i.e. no mixing takes place in this region). Figure 6.2 illustrates the pulse propagating down the solenoid for a case where there was no beat frequency mixing. The pump beam intensity was set to zero for this case. Figure 6.3 illustrates the growth of a pulse which initially had a peak intensity of  $10^8$  W/cm<sup>2</sup>. In this case the peak pulse power has been amplified by a factor of about 250. This is considerably less than what would be attained if pump depletion were not important. The pump intensity at the three different times is also plotted in this figure. Figure 6.4 is a similar plot except the peak power in the pulse was 10<sup>9</sup> W/cm<sup>2</sup> before it was injected into the amplifier. The interesting feature of this calculation is that the front end of the pulse has sharpened considerably during amplification. Initially the pulse rose from zero to its peak power in 0.5 ns. After amplification it rises to peak power in 0.13 ns. In fact the rise time of this pulse is becoming sufficiently rapid that the validity of the quasi-static approximation is becoming questionable. The reason the pulse sharpens in this method of amplification is that the pump is depleted near the leading edge of the pulse. As a result only the leading edge

of the pulse is suplified. The higher the intensity of the injected pulse the steeper the leading edge becomes during amplification. For pulses with very sharp leading edges, the quasi-static approximation breaks down and a more sophisticated transient analysis must be done.

163

••

# 6.2.2 Trensient Analysis

In Chapter 5 a set of equations which describe the evolution of a palse of electromagnetic radiation propagating antiparallel to another electromagnetic beam with a different frequency in a plasme were derived. These equations ((5.65), (5.66), and (5.70)) account for transient effects in the electrostatic mode. Before these equations are solved two additional approximations are made.

In the integration of (5.70)
 (x+c<sub>2</sub>τ,t-τ) & A<sub>1</sub>(x+c<sub>2</sub>τ,t), This is valid when A<sub>1</sub>
 and n are slowly varying functions in the moving coordinate system.
 c<sub>2</sub><sup>x</sup> -c<sub>1</sub>. This is valideif w<sub>2</sub> >> w<sub>pe</sub>.

With these approximations, the equations to be solved are summarized below. Primes have been dropped for clarity.

$$[\partial_{t}^{+2c_{1}}\partial_{x}^{+}\Gamma_{1}] \wedge_{1}(x,t) = R_{1}^{n} (x,t) \wedge_{2}(x,t)$$
(6.10)

$$[\partial_t + \Gamma_2] A_2(x,t) = R_2 \hat{n}_e^*(x,t) A_1(x,t)$$
 (6.11)

$$A_{e}(x,t) = \frac{1}{\sqrt{2\pi}} \int_{0}^{x/c_{1}} A_{1}(x-c_{1}x) A_{2}(x-c_{1}\tau,t)e^{i\omega_{3}\tau} D(k_{3},\tau)d\tau$$

(6.12)



and  $\chi_{1}$  and  $\chi_{1}$  are defined in Appendix B.

The above equations are equally valid for beat frequency mixing a difference frequency near the electron plasma frequency or the ion-Ecoustic frequency. However, in the numerical interation a finestep is limited to being much less than the period of the beat wave and consequently a prohibitive amount of computer time would be required to integrate the equations if the beat frequency is close to the electron plasma frequency. In this section solutions for the case of the beat frequency being close to the ion-acoustic frequency are reported.

L

For the purposes of these computations the assumed plasma conditions correspond to those of the MHD simulation reported in Chapter 3 with an initial density of S. Culf. Dince Figures 5.8 and 5.9 Micate

that the ion mode is very broad, it is reasonable to approximate the plasma being uniform. A plasma with  $T_1=1000$ ,  $T_1=1200$ , and  $v_0=2.5\times10^{17}$  cm<sup>-3</sup> is assumed. The versionization of the pump beam is 10.247 µm ( $\lambda_1$ ) and the wavelength of the pulse to be amplified is met to 10.2605 µm ( $\lambda_2$ ). This yields a difference frequency of  $u_3=2.42\times10^{11}$  sec<sup>-1</sup> and closely corresponds to a maximum in the plot of a versus  $u_0$  in Figure 5.2. For these conditions  $a=6.06\times10^{-12}$ . By assuming an interaction length of 80 cm, the input pump intensity must be less than  $1.9\times10^{10}$  W/cm<sup>2</sup> to spatiation equation (6.4).

117

Figure 6.5 shows the real and imaginary parts of  $D(k_3, \theta)$ . as a function of  $\tau$  for  $T_1=160eV$ ,  $T_1=120eV$  and  $n_0=2.5\times10^{17}cm^{-3}$ . Figure 6.6 shows  $D(k_3, \tau)$  as a function of  $\tau$  for these same conditions. It is noted that the real part of  $D(k_3, \omega)$  is even, while the imaginary part is odd as is required to make  $D(k_3, \tau)$  real. The plot of  $D(k_3, \tau)$  versus  $\tau$  shows that the density occutation level at time t is only dependent on the driving forces in the time interval from t-T to t where T is the period of the beat frequency wave. This is characteristic of a heavily damped oscillator. To illustrate the effects of reducing the damping rate similar plots of  $D(k_3, \omega)$  and  $D(k_3, \tau)$  are shown in Figure 6. and 6.8 for the case of  $T_1=160eV$ ,  $T_1=20eV$ , and  $n_0=2.5\times10^{17}$ . Here a much sharper structure is seen in the frequency spectrum and  $D(k_3, \tau)$  is seen to damp out after several periods.

To illustrate the importance of transient effects without the complications of pump depletion and inverse Brensstrahlung absorption, a calculation has been done which assumes an input pulse which rises instantaneously to a constant value of  $10^5$  W/cm<sup>2</sup> was injected into the

solenoid antiparallel to a beam with an intensity of  $1.5 \times 10^{10} \, \text{W/cm}^2$ . For this calculation the effects of inverse Brensstrahlung absorption have been neglected. The initial pulse intensity  $(10^5 \text{ W/cm}^2)$  is sufficiently low that the effects of pump depletion are negligible. Figures 6.9 and 6.10 show I<sub>1</sub>, I<sub>2</sub>, and  $|\tilde{n}_a|$  as a function of  $\tilde{x}$  (where  $\tilde{x}$ is the distance in cm from the pulse front) after the pulse front has travelled 40cm and 80cm in the plasma. For the case of Figure 6.10 a quasi-static analysis predicts that the pulse front should have experienced an amplification of a factor of 1440 and that the portion of the pulse near  $\frac{1}{2}$  cm, which has only travelled 77 cm in the plasma, should be amplified by a factor of 1090. The transient analysis and quasi-static analysis agree at  $\frac{2}{2}$  cm but transient effects lead to a smaller amplification in the pulse front. Comparison of Figures 6.9 and 6.10 shows that the further the pulse has travelled in the plasma the larger the value  $\widetilde{\mathbf{x}}$  must be before the transient analysis agrees with the quasi-static analysis.

{

€

1

A similar calculation has been made to illustrate the importance of pump depletion. Again, it is assumed that the injected pulse rises to a constant value instantaneously, however this time the initial intensity of the injected beam is  $10^{10}$  W/cm<sup>2</sup> and the intensity of the pump is  $1.8 \times 10^{10}$  W/cm<sup>2</sup>. For these conditions a quasi-static analysis of the pulse front predicts an amplification of a factor of 6163 after it has interacted over a region of 80cm. Figures 6.11 to 6.14 show  $I_1$ ,  $I_2$  and  $2 |\tilde{n}_e|$  has a function of  $\tilde{x}$  after the pulse has travelled 40, 60, 70 and 80cm respectively. From Figure 6.14 it is seen that the maximum amplification is a factor of 100 and occurs at  $\tilde{x}=0.6$ cm. It is

interesting to note that up to the time the pulse has travelled 70cm in the plasme  $I_1(x)$  is a monotonically decreasing function, whereas after the pulse has travelled 80cm  $I_1(x)$  experiences some regrowth behind the peak of the pulse. If the pulse is sufficient where the phase relationship between the electrostatic wave and the two electromagnetic waves can be such that energy is transferred from the lower frequency waves to the high frequency wave in this region.

A calculation with any pjected pulse with a more realistic initial profile has also been made. The injected pulse is assumed to be of triangular shape rising from 0 at t=0 to  $10^{10}$  W/G at t=50ps and falling off to zero at 100 ps. The intensity of the pump wave is again assumed to be  $1.8 \times 10^{10}$  W/cm<sup>2</sup>. Figures 6.15 and 6.16 abow I<sub>1</sub>, I<sub>2</sub> and  $|\hat{n}_e|$  as a function of  $\hat{x}$  after the pulse has travelled 40 and 80 cm in the plasma. It can be seen that the maximum value of the signal intensity has moved ahead and has a value of about 80 times the peak value before amplification. It is also seen that the pulse width has decreased from 50 ps to about 30 ps during amplification.

The above analysis indicates that the process of beat frequency mixing in a plasma could be used for the amplification, and in some cases, the shortening of pulses of electromagnetic radiation. This type of analysis could also be used to examine Brillioun backscattering from a laser heated plasma.

١









c







171 |(``

•

١

.....

















triangula ntensity of

ront has trave

pue

**7=1.8x10<sup>10</sup>** 

ape vit nitially

010

#### HAPTER 7

Conclusions of

A two dimensional numerical model of a laser heated plasms in a solenoidal magnetic field has been successfully developed. The model is based on one fluid two temperature HHD equations. By solving the MHD equations in a moving coordinate system defined by the magnetic field lines, numerical diffusion arising from the strong anisotropy of the plasma has been eliminated. This allows for a more coarse computational grid to be used which leads to savings in CPU time.

Care has been taken to make the computer routine as versatile as possible. One example of this is that, to a large extent, initial conditions as well as parameters that control some of the internal processing are specified through an input file. In this way the routine can be a flexible and useful tool for users not familiar with the numerical techniques used to solve the equations.

Results from sample calculations made by this routine indicate that the assumption of radial pressure balance is valid for a wide variety of problems. The routine has been constructed so that full radial dynamics can be calculated or radial pressure balance can be assumed.

The routine has been used to provide numerical backup for gas target experiments performed by A.A. Offenberger <u>et</u>  $al^{19-21}$  Results from this study indicate that laser heated gas targets can provide an attractive plasma source for the study of laser-plasma interactions. Current near future plans for this routine include more gas target studies

and the numerical modelling of the short (1/2 m) solenoid experiments planned by A.A. Offenberger <u>et</u>  $al^9$ .

A plasma parametric amplifier has been proposed for amplifying laser beams in the 11-14 um wavelength range to very high intensities. The plasma section of the amplifying device can be relatively short (less than one meter) and only a very coarse control on plasma density and temperature needs to be maintained. Because the idler wave is an electron plasma wave and is Landau damped the process of energy transfer from the pump beam to the signal beam becomes irreversible. Further analysis suggests that this device could also be used to amplify and temporally shorten ultra-short pulses of infrared laser radiation. Fulse amplification could take place through beat frequency mixing with a beat frequency.

t

178

#### BIBLICCRAPHY

- J.M. Dawson, A. Hertsberg, R.E. Kidder, G.C. Vlases,
   H.G. Ahlstrom, and L.C. Steinhauer; <u>Proceedings of</u> the Fourth Conference on Plasma Physics and Controlled <u>Nuclear Fusion Research</u>, Vol. 1 (IAEA, Vienna, 1971).
- A.L. Hoffman, and G.C. Vlases; Appl. Phys. Lett., <u>12</u>, 690, (1973).
- L.C. Steinhauer and P.H. Rose, in <u>Proceedings of the First</u> <u>Topical Meeting on Technology and Controlled Thermonuclear</u> <u>Reactions</u>, edited by G.R. Hopkins, United States Atomic Energy Commission Report CONF-740402 (1974), Vol. 2, p. 63.
- R.D. Richtmyer and K.W. Morton, <u>Difference Methods for</u> <u>Initial Value Problems</u>, Interscience Publishers (1967).
- 5. F. Hertweck and W. Schneider, Institut fur Plasmaphysik Report IPP6/90 (1970).
- 6. W. Schneider, Institut fur Plasmaphysik Report IPP1/24 (1972).
- H.A.B. Bodin, J. McCartan, (I.K. Pasco, W.H. Schneider, Phys. Fluids <u>15</u>, 1341 (1972).
- 8. L.C. Steinhauer, Phys. Fluids 19, 738 (1976).
- 9. A.A. Offenberger Private Communication.
- R.D. Milroy, C.E. Capjack, and C.R. James, Plasma Physics <u>19</u>, 989 (1977).
- W.F. Krupke, Nineteenth Annual Meeting of the Division of Plasma Physics of the American Physical Society, 6K3.
- S.I. Braginskii, "Transport Processes in a Plasma", in
   M.A. Legntovich, Ed., <u>Reviews of Plasma Physics</u>, Vol. 1, Consultants Bureau Enterprises (1965).

- 13. D.L. Book, "Revised NRL Formulary", Naval Research Laboratories.
- 14, K. Hain, G. Hain, K.V. Roberts, S.J. Roberts, and W. Koppendorfer, Z. Naturforsch. A <u>15</u>, 1039 (1960).

T.W. Johnston and J.M. Dawson, Phys. Fluids, 16, 722 (1973).

- 16. D.W. Scudder, Z.A. Pietrzyk, G.C. Vlases, and A.H. Makomaski, 10'th International Shock Tubes and Waves Symposium, Seattle, 1977.
- J.A. Wesson, in <u>Plasma Physics and Controlled Nuclear Fusion</u> <u>Research</u>, (International Atomic Energy Agency, Vienna, 1966) Vol. 1, p. 233.
- 18. J.N. McMullin and C.E. Capjack, Phys. Fluids, 20, 1566 (1977).
- A.A. Offenberger, A. Ng, and M.R. Cervenan, Can. J. Phys. <u>56</u>, 381 (1978).
- A.A. Offenberger, A. Ng, and L. Pitt, Phys. Rev. Lett. <u>40</u>, 873 (1978).
- 21. A.A. Offenberger, A. Ng, L. Pitt, and M.R. Cervenan, (In Press - Phys. Rev. A (1978)).
- 22. J.J. Schuss, T.K. Chu, and L.C. Johnston, Princeton Report, PPPL-1394.
- A. Ng, M.R. Cervenan, and A.A. Offenberger, Optics Communications, <u>25</u>, 116 (1978).
- F.F. Chen, <u>Introduction to Plasma Physics</u>, Plenum Press, Chapter 8 (1974).
- 25. G. Schmidt, Phys. Fluids, 16, 1676 (1973).
- V. Fuchs, C.R. Neufeld, J. Teichmann, and A.G. Engelhardt, Phys. Rev. Lett., <u>31</u>, 1110 (1973).

B.I. Cohen, Ph.D. Thesis, University of California, Berkeley (1975). 28. B.I. Cohen, M.A. Mostrom, D.R. Nicholson, A.N. Kaufman, C.E. Max, 29. and A.B. Langdon, Phys. Fluids, 18, 470 (1975). C.E. Capjack, and C.R. James, Can. J. Phys. <u>53</u>, 2606 (1975). 30. T.H. Martin Theory of Plasma Waves, McGraw Hill (1962). 31. R.J. Jenser; J.G. Marinuzzi, Schemingon, and S.D. Rockwood, 32. Laser Focus, May 1976. L.C. Steinhauer, Phys. Fluids, 19, 1740 (1976). 33. E. Yablonovitch and J. Goldhar, Appl. Phys. Lett. 25, 580 (1974). 34. R.A. Fisher and B.J. Feldman, Optics Letters, 1, 161 (1977) 35. I. Lindemuth and J. Killeen, J. Comp. Phys. 13, 181 (1973). 36.

A.N. Kaufman and B.I. Cohen, Phys. Rev. Lett. 30, 1306 (1973).

27.

37. N.H. Burnett and A.A. Offenberger, J. Appl. Phys. <u>45</u>, 2155 (1974).

101

 $\left( \right)$ 

Appendix A Some Programming Details for 2-D MHD Code

This Appendix contains some of the programming details of numerical model described in Chapter 2. All equations are written in J dimensionless form as described in Chapter 2. The derived quantities are:

N - Plasma density, n<sub>o</sub>
TE - Electron temperature, T<sub>e</sub>
TI - Ion temperature, T<sub>i</sub>
V - Axial fluid velocity, v<sub>z</sub>
A - Area between shells
R - Shell radii
W - Square of shell radii, R<sup>2</sup>
B - Magnetic field, B<sub>z</sub>
U - Radial fluid velocity times radius, Rv

Any two of A, R, and W can be derived from the third however all three are kept as internal variables in the numerical solution for convenience. When the solution employs full radial dynamics, W is solved for, and A and R are derived from W. When radial pressure balance is assumed A is solved for, and R and W are derived from A. The magnetic field B can also be derived directly from A. Initial conditions specify the magnetic flux in each shell,  $\phi_g$ , and this is a constant. B can then be derived from

$$B = \frac{\Phi_{s}}{A}$$

The difference equations are solved on a computational grid consisting of M shells and NX x-points.

## A-1 Grid Structure and Two-Step Scheme

The position at which the variables are calculated on the x, s grid is illustrated in Figure A.1. Note, x and s have been declared as integers in the routine and are used to specify the location of a quantity on the computational grid. The quantities N, V, TE, and TI are calculated at the cell centers while U, W, and R are calculated on the shell boundaries. The quantity A denotes the cross sectional area of a cell at a particular x-point. Quantities that are calculated at the shell centers and are subscripted with an (s,x) are assumed to be at the x'th x-point and between the (s-1)'th and s'th shell boundaries as illustrated in Figure A.1.

All variables are calculated at the same time t. In order to obtain second order accuracy in the timestep size,  $\Delta t$ , a two-step method was used to solve the equations. First the solution at time t is used to find temporary values at time t +  $\Delta t/2$ . These temporary values are used to calculate transport coefficients and spatial derivatives in the equations. The solution is then advanced a full timestep from t to t +  $\Delta t$ .

At time t the value of all quantities Q are stored in arrays denoted by  $Q_{s,x}$  and  $(Q_2)_{s,x}$ . After the first step in the two step scheme  $Q_{s,x}$  contains the temporary solution at time  $t + \Delta t/2$ . Based on these values of  $Q_{s,x}$ , transport coefficients and spatial derivatives are calculated. Based on these, and the values in  $(Q_2)_{s,x}$ , the solution is advanced from time t to  $t + \Delta t$  and placed in array  $Q_{s,x}$ .  $(Q_2)_{s,x}$  is then set equal to  $Q_{s,x}$  and the process is repeated. A flow-chart illustrating the gross features of the two-step scheme can be found in Figure A.2.

~~

## A.2 Difference Equations

The difference equations used in the routine are based on the partial differential equations in Section 2.4.5. All of the equations are of the form:

$$\hat{Q} = RHS$$
 (A.1)

where RHS has terms involving first and second order spatial derivatives and time derivatives of other variables. In the code the equations are solved in a sequence such that the required time derivatives of other variables are evaluated first. Expressing the time derivative of the above equation:

$$\frac{2'}{2} = \frac{2'}{\delta t} = RHS$$

or

$$Q'_{\mathbf{x},\mathbf{x}} = Q^{\bullet}_{\mathbf{x},\mathbf{x}} + \delta t \cdot RHS^{\bullet}$$

or

$$Q_{\mathbf{s},\mathbf{x}}^{*} = Q_{\mathbf{s},\mathbf{x}}^{*} + \delta t \cdot Q_{\mathbf{s},\mathbf{x}}$$

where  $Q_{\mathbf{s},\mathbf{x}}^{\bullet}$  is the value of  $Q_{\mathbf{s},\mathbf{x}}$  at time t and  $Q_{\mathbf{s},\mathbf{x}}^{\dagger}$  is the value of  $Q_{\mathbf{s},\mathbf{x}}$  at time t +  $\delta$ t and  $\delta$ t is the timestep size. In the two-step scheme  $\delta t = \frac{1}{2} \Delta t$  for a half step and  $\delta t = \Delta t$  for a full step where  $\Delta t$  is the timestep size for a full step.

Contral differences are employed throughout when evaluating spatial derivatives. In all cases the partial derivative of Q with respect to x is written as:

$$\left(\frac{\partial Q}{\partial x}\right)_{s,x} = \frac{Q_{s,x+1} - Q_{s,x-1}}{2\Delta x}$$
 (A.3)

where  $\Delta x$  is the grid spacing in the x-direction. In the present version the computational grid is uniform in the axial direction so  $\Delta x$  is a constant. Differencing in the radial direction is somewhat less straight-forward since shell spacing is not uniform and derivatives on the shell boundaries as well as at the cell centers are required. When the value of a quantity that is known on shell boundaries must be calculated at a shell center, simple averaging is used. When the value of a quantity that is known in shell centers must be calculated on shell boundaries a weighted average is used. Two variables are defined as:

$$R1 = \frac{(\Delta R)_{s+1}}{[(\Delta R)_{s} + (\Delta R)_{s+1}]} \quad R2 = \frac{(\Delta R)_{s}}{[(\Delta R)_{s} + (\Delta R)_{s+1}]} \quad (A.4)$$

Here  $(\Delta R)_{g}$  is the radial grid spacing of the s'th shell. The value of Q on the s'th shell is given by  $Q_{g+1/2,x}$ 

$$Q_{s+1/2,x} = R1 \cdot Q_{s,x} + R2 \cdot Q_{s+1,x}$$
 (A.5)

When Q is known in shell centers and  $\frac{\partial Q}{\partial s}$  is required on a shell boundary, the necessary differencing is straight forward.

$$\frac{12}{2} \Big|_{0+1/2} = \frac{2 \cdot (2_{0+1,0} - 2_{0,0})}{\frac{1}{2} + \frac{1}{2} + \frac{1$$

Here  $\phi_{g}$  is the dimensionless value of magnetic flux in the s'th shell, and is equal to be across the s'th shell.  $\phi_{g}$  is specified by initial conditions and is independent of x and t is the flux coordinate model. For cases where Q is known at shell conterp and  $\frac{\partial Q}{\partial g}$  is required at shell centers, the derivative is first found on the boundary and then averages taken to find values in cell centers. The quantity  $(1/\partial_{g}R)$ appears in several expressions in Section 2.4.5. This is easily evaluated from equation (2.27).

$$\frac{1}{\partial_{\mathbf{R}}\mathbf{R}} = 2 \mathbf{R}\mathbf{B} \tag{A.7}$$

Equations which do not involve diffusion terms (second order spatial derivatives) are evaluated using a straight forward explicit differencing scheme. That is RHS of (A.1) is evaluated using differences of the types described above and then the value of the desired quantity is found at the new timestep through applications of equation (A.2).

• The use of an explicit formation on the equations that have diffusion terms would introduce a very restrictive stability condition on the program. If an equation of the form:

$$\frac{\partial}{\partial t} Q = \sigma \frac{\partial^2}{\partial x^2} Q$$

is solved using explicit differencing, the timestep size in whe difference equations must satisfy (see Richtmyer and Morton<sup>4</sup>)

 $\Delta t \leq \frac{\Delta x^2}{2\sigma}$ 

RHS = a  $Q_{\mathbf{s},\mathbf{x}-1}$  + b  $Q_{\mathbf{s},\mathbf{x}}$  + c  $Q_{\mathbf{s},\mathbf{x}+1}$  + other terms

where the coefficients a, b, c may be spatially dependent and may also depend on other variables. Thus equation (A.2) can be expressed as:

$$Q_{\mathbf{s},\mathbf{x}}^{h} = Q_{\mathbf{s},\mathbf{x}}^{\bullet} + \delta t \{ \mathbf{a} \ Q_{\mathbf{s},\mathbf{x}-1} + b \ Q_{\mathbf{s},\mathbf{x}} + c \ Q_{\mathbf{s},\mathbf{x}+1} + other terms \}$$

¢

In an explicit formalism the terms  $Q_{\mathbf{s},\mathbf{x}}$  in brackets would be the same as  $Q_{\mathbf{s},\mathbf{x}}^{\bullet}$  in the first step of the two-step scheme. In the second step of the two step scheme, these as well as (other terms) would be based on the results of the first step.

In the implicit formalism used in the present routine however the former equation is re-written as:

$$Q_{b,x}^{*} = Q_{b,x}^{*} + \delta t \{ a \ Q_{b,x}^{1/2} + b \ Q_{b,x}^{1/2} + c \ Q_{b,x+1}^{1/2} + other terms \}$$

where

۲

٩.

$$Q_{n,x}^{1/2} = \frac{1}{2} (Q_{n,x}^{*} + Q_{n,x}^{*})$$

Here the value of  $Q_{B,X}$  as found in the first step is not directly used to advance the equation in the second step, however it may be required for transport coefficients and for other equations. The results from the first step are used in the evaluation of other terms in the second step.

The above equation can be re-written in a form:

$$-A_{s,x} Q'_{s,x+1} + B_{s,x} Q'_{s,x} - C_{s,x} Q'_{s,x-1} = D_{s,x}$$

where

$$A_{s,x} = \frac{a}{2} \delta t$$

$$B_{s,x} = 1 - \frac{b}{2} \delta t$$

$$C_{s,x} = \frac{c}{2} \delta t$$

$$D_{s,x} = Q_{s,x}^{\circ} + \frac{1}{2} \delta t \{a \ Q_{s,x-1}^{\circ} + b \ Q_{s,x}^{\circ} + c \ Q_{s,x+1}^{\circ} + other \ terms\}$$

The solution of this system of equations and the application of boundary conditions followed the procedures outlined by Richtmyer and Morton<sup>4</sup>.

A completely analogous technique can be employed to solve equations that contain a diffusion term in the radial direction. However, a somewhat more complex technique must be employed to solve the electron

temperature equation which has diffusion terms in both the radial and axial directions. The ion temperature equation has been simplified by neglecting thermal conduction in the axial direction.

.

One method of numerically solving partial differential equations that have diffusion terms in two directions is through the use of Alternating Direction Implicit (ADI) techniques. An example of the use of such a technique can be found in Lindemuth and Killeen<sup>36</sup>. Basically this technique works by setting up two levels of difference equations. In one level the equation is advanced implicitly impedirection and in the other level it is advanced implicitly in the other direction. Then one timestep uses one set of equations and the next uses the other. It is not clear how such a scheme could be used in the present two step scheme and so another closely related technique was used to advance the electron temperature equation. An equation of this form can be written in the simplified form:

$$\hat{Q} = RD + AD + OT$$

١

where RD represents terms in RHS of (A.1) that involve radial diffusion terms (second order derivatives in the radial direction), AD represents terms that involve axial diffusion terms (second order derivatives in the axial direction), and OT represents other terms. The above equation then split into two parts:

 $\dot{q} = (\dot{q}1) + AD + OT$ 

A solution for (Q1) is found by using implicit techniques and this solution is inserted into the second equation replacing RD. The solution for (Q1) involves advancing Q a mimestep assuming radial diffusion terms are the only terms on the RHS.

### A.3 Leser Heating

The equations that govern the propagation and absorption of the laser beam are given in dimensionless units in Section 2.3.5. The MHD equations are solved on a computational grid with HX axial points. The laser beam is assumed to be input at the end of the solenoid specified by the HX'th axial point. Laser beam power, specified in dimensionless units P(x), is calculated on axial grid boundaries, and the power absorbed per unit length, E(x) is calculated in cell centers. The indexing of  $P_x$  and  $E_x$  relative to the indexing of MHD quantities is illustrated in Figure A.3.

The routine contains the option of two types of axial boundary conditions at the point x=1.

- (1) Boundary conditions are the same as those at the other end of the solenoid. This boundary condition allows for free streaming of plasma out the end.
- (2) Symmetric boundary conditions. This boundary condition assumes that the solution is symmetric about the point x=2. i.e. the point x=2 represents solenoid center and by symmetry arguments, the solution is required only for half of the solenoid.

A more detailed description of these boundary conditions can be found in Section A.4. When symmetric boundary conditions are employed the beam is assumed to fold over on itself around the point x=2. This is because in a symmetric problem a beam is assumed to be injected from both ends of the solenoid.

The total cross-sectional laser power is given on the axial grid points as  $P_x$ . The relative intensity of the beam in each shell is specified by an array  $(PR)_{s,x}$  which is used to define the radial profile of the beam (Gaussian for examples in this thesis) and allows for an axial dependence as was required in Chapter 4.

•

Using the laser absorption coefficient given by Johnston and Dawson<sup>15</sup> and results of Section 2.3.5,  $k_a$  is given in dimensionless units as:

$$k_{a} = K \frac{N^{2}}{T_{a}^{3/2}} \ln N$$

where

$$K = \frac{9.74 \times 10^{-36} \text{ N}^2}{T_0^{3/2}} L_c$$

As described in Section 2.1.5 the absorption coefficient was obtained by averaging over the radial profile. A weighted average is used with the weight at each radial position proportional to the beam intensity at that point. In the finite difference scheme the averaging is over discrete shells and weighting is proportional to total beam power in each shell. The beam power in each shell is proportional  $to[(PR)_{s,x} \cdot A_{s,x}]$ , thus:

$$\overline{k}_{a} = \frac{K}{\xi_{x}} \sum_{s} \frac{y_{s,x}^{2}}{(TE)_{s,x}^{3/2}} \ln \Lambda (PR)_{s,x} \cdot A_{s,x}$$

where

$$\xi_{\mathbf{x}} = \sum_{\mathbf{s}} (PR)_{\mathbf{s},\mathbf{x}} \cdot \mathbf{A}_{\mathbf{s},\mathbf{x}}$$

۲

The propagation of the laser beam is described by:

$$\frac{1}{c}\frac{\partial}{\partial t}P + \frac{\partial}{\partial x}P = -\overline{k}P \qquad (2.49)$$

Normal differencing techniques described earlier in this appendix can be employed to solve this equation, however it is found that it is sharp discontinuities expected in the bleaching front lead to great difficulties for this case. There, is also a problem in the propagation of a sharp wavefront through a semi-transparent plasma. Instead the following approach is employed. The timestep size is restricted to:

$$\Delta \tau = \frac{m\Delta x}{c}$$

where m is an even integer.  $\Delta \tau$  is not the same size as  $\Delta t$ , the timestep size for hydrodynamic quantities, however m is continually adjusted so i the time in laser calculations never differs from the time in MHD calculations by more than  $\frac{1}{2} \frac{\Delta x}{c}$ . With this definition the beam, propagating at a speed c, propagates an integer number of space-step in each timestep  $\Delta \tau$ .

The beam profile  $P_x$  is calculated by advancing it by m small timesteps of size  $\frac{\Delta \tau}{m}$  for each full timestep of the main routine and by  $\frac{m}{2}$  small timestep for each half step of the main routine. The laser power  $P_x$  is advanced in each of the small timesteps through the solution of:

$$\frac{\mathbf{P}_{\mathbf{x}}^{\dagger} - \mathbf{P}_{\mathbf{x}-1}^{\bullet}}{\Delta \mathbf{x}} = -\frac{\overline{\mathbf{k}}}{2} \{\mathbf{P}_{\mathbf{x}}^{\dagger} + \mathbf{P}_{\mathbf{x}-1}^{\bullet}\}$$

or

1

$$P'_{\mathbf{x}} = P^{\bullet}_{\mathbf{x}-1} \frac{\left[1 - \frac{1}{2} \overline{k}_{\mathbf{a}} \Delta \mathbf{x}\right]}{\left[1 + \frac{1}{2} \overline{k}_{\mathbf{a}} \Delta \mathbf{x}\right]}$$

where  $P'_{x}$  represents the new value of  $P_{x}$  and  $P^{\bullet}_{x}$  represents the old value.

The laser power that is absorbed per unit length by the plasma, E, is defined in such a way that energy is conserved in spite of numerical errors which may be made in the first two or three shells of the beam front. E is defined as:

$$E = \overline{k}_{a} P$$

In each of the m small timesteps an  $E_x^i$  is defined as:

$$E_{x}^{i} = (P_{x}^{i-1} - P_{x+1}^{i}) / \Delta x$$

E for a full timestep is then given by:

$$E_{\mathbf{x}} = \frac{1}{m} \sum_{i=1}^{m} E_{\mathbf{x}}^{i}$$

Now the energy absorption rate per particle  $(\epsilon_L)$  must be defined. The total power absorbed in a shell is proportional to both N<sub>s,x</sub> · A<sub>s,x</sub> ·  $(\epsilon_L)_{s,x}$  and  $[N_{s,x}^2/(TE)_{s,x}^{3/2}]$  ·  $(PR)_{s,x}$  · A<sub>s,x</sub> · ln  $\Lambda$ . Also:

•

,

195

$$E_{\mathbf{x}} = \sum_{\mathbf{s}} [(c_{\mathbf{L}})_{\mathbf{s},\mathbf{x}} \cdot \mathbf{N}_{\mathbf{s},\mathbf{x}} \cdot \mathbf{A}_{\mathbf{s},\mathbf{x}}]$$

All of these conditions can be satisfied only if:

e

.

.

$$N_{s,x} \cdot A_{s,x} \cdot (\varepsilon_{L})_{s,x} = \frac{\left(\frac{N_{s,x}^{2}}{TE^{3/2}}\right) PR_{s,x} A_{s,x}}{\sum_{s,x} \left(\frac{s_{s,x}}{TE^{3/2}}\right) PR_{s,x} A_{s,x}} E_{x}$$

The denominator of this equation can be rewritten as,  $\frac{\xi \overline{k}}{K}$  so:

$$(\epsilon_{L})_{s,x} = \frac{\kappa}{\xi_{x}(\overline{k}_{a})_{x}} \frac{\sum_{x} (PR)_{s,x} + E_{x}}{(TE)_{s,x}^{3/2}}$$
# A.4 A Users Manual for MHD-2D Routine

In this section some of the various features and options incorporated into the computer routine are described. An explanation of the parameters that are specified through the main input file are given. A listing of the input file, which contains a brief description of each of the parameters, can be found at the end of this section.

# A.4.1 Start, Stop and Restart

When a simulation is started at a time of t=0 the parameter NTIME must be set to 0. The simulation will then run until either NTOTAL timesteps have been taken, or the simulation time reaches TMAX seconds. If NTIME # 0 the routine attempts to read input data from Fortran unit #2 so that it can start from the NTIME'th timestep of a previous simulation. Fortran unit #2 must be attached to a file or device which contains the unformatted output of a previous simulation.

## A.4.2 Initial Conditions

If the simulation is started at a time of t=0 the initial conditions are calculated from various parameters input through the input file. If the routine is being restarted from a previous simulation these parameters are ignored and the initial conditions are read directly from Fortran unit #2 as explained in Section A.4.1.

The computational grid used for the numerical calculation consists of NX x-points and M shells. The spacing of the x-points is always uniform however the shell spacing can vary. The initial shell spacing is specified through the parameters RO1 and W. The first shell has a radius of RO1 cm and the width of the second shell is W cm. The width of the i'th shell is  $a^{1-1}$  W where a is calculated by the routine We such as to make the outer radius of the last shell equal to the specified radius of the solenoid. The solenoid is LO cm long and has a radius of RO. These values of LO and RO are also used as the basic units of length and radius for the transformation to dimensionless variables. The other two basic units needed for the transformation to dimensionless variables (see Section 2.2.3), are given by NO cm<sup>-3</sup> and TO eV. If the initial density and temperature profile is to be uniform, the plasma will have an initial density of NO cm<sup>-3</sup>, an initial temp-erature of TINT eV and an initial magnetic field of BINT Gauss.

The density and temperature can be given an initial profile. The density is specified through parameters NO, R1N, and R2N as:

$$N(r, t=0) = \frac{NO}{1 + exp[(r - R1N)/R2N]} cm^{-3}$$

RIN, and R2N are input in units of cm. The temperature profile is specified through parameters TINT, RIT, and R2T as:

$$T_{e,i}(r, t=0) = \frac{TINT}{1 + exp[(r - R1T)/R2T]} eV$$

۲

A uniform density and temperature profile in the radial direction is specified by setting RlN >> RO, R2N << RO, RlT >> RO and R2T << RO. The magnetic field is set to BINT Gauss at r = RO and is adjusted for r < RO so that radial pressure balance exists.

An initial density profile can be specified in the axial direction through parameters L1, L2, L4, and L5. The initial axial density profile is given by:

# $H(z, t=0) = \left\{ \frac{.999 \cdot H0}{1 + \exp[(z - L1)/L2]} + .001 \right\} = \left\{ \frac{.999 \cdot H0}{1 + \exp[(L0 - z - L4)/L5]} \right\}$

#### + .001 }

L1, L2, L4 and L5 are input in units of cm. Initial density profiles such as those used in Chapter 4 can be specified with these parameters. A uniform axial profile can be specified by setting L1 = - L0, L4 = - L0,  $L2 \ll L0$ , and L5  $\ll L0$ .

#### A.4.3 Output

The main output from the routine is written in unformatted form on Fortran unit #3. It is intended that this unit be attached to a disk file or a magnetic tape. This output consists of the calculated hydrodynamic and laser quantities as well as various other parameters that may be needed to restart the simulation or by other post-processor routines. This data is output after the initial conditions are calculated and after every subsequent NTAPE'th timestep. The output is suppressed if NTAPE = 999.

Formatted output intended for a printer is written on Fortran unit #6. This output consists of the hydrodynamic and laser variables written in dimensionless form. The output is in matrix form and provision is made for only part of the information to be printed. This is controlled by parameters NXI, NXF, M1, MF and MS. Output is from x-points specified by  $MXI \leq x \leq NXF$  and from every MS'th shell for  $Ml \leq s \leq MF$ . This data is output after initial conditions are calculated, and after every subsequent NPRT'th timestep. Output is suppressed if

#### NPRT - 999.

# A.4.4 Timestep Control

The timestep size is adjusted after every timestep. The timestep size  $\Delta t$  is the maximum allowed by the following four conditions:

- 1.  $\Delta t$  must be less than a fraction Al of the maximum allowed by the stability condition. The stability condition is specified by equations (1.1) and (1.2).
- The electron temperature in any cell cannot change by more than a fraction A2 in a single timestep.
- 3. Timestep size cannot increase to more than A3 times the previous timestep size.
- 4. At must be less than DTMAX seconds.

The first timestep must only satisfy conditions 1 (with Al replaced by All) and 4.

#### A.4.5 Laser Parameters

The laser beam power as it enters the solenoid is specified through parameters PWR1, PWR2, PWR3, T1, T2, T3, and T4. The beam power is 0 at t = 0, PWR1 at t = T1, PWR2 at t = T2, PWR3 at t = T3, and 0 at t = T4. Power levels between the specified times are calculated by linear interpolation. Power is specified in Watts and time is specified in seconds.

The radial intensity profile is Gaussian as specified by equations (4.1) and (4.2). Parameters needed in equation (4.2) are input as  $\sigma_0 = RLO$ ,  $\frac{1}{2f} = BS$ , and  $L_3 = L3$ .

Sometimes it may be desirable to artificially control the bleaching

velocity of the laser front. This is accomplished by setting VLAS to the desired bleaching velocity. If so limit is desired VLAS is set to the speed of light.

In Section A.3, an explanation is given of how the laser beam transport equation is advanced in small timesteps for every timestep of the main routine. Under some conditions H can be a very large number and essentially no changes result from limiting H to a smaller number. This can result in significant agvings in CPU time. H is limited to being less than MONAX. Care must be taken when setting this parameter to a small number since it can lead to errors including loss of energy conservation. It is very unlikely for significant errors to result from this feature if MONAX.  $\geq NX$ .

#### A.4.6 Boundary Conditions

Options exist on some boundary conditions as specified in Section 2.5. If TRBC = . TRUE ., the temperature T remains a constant at r = RO. If TRBC = . FALSE .  $\partial T / \partial s = 0$  at r = RO. This is equivalent to assuming a thermally insulated boundary. The thermal pressure P is assumed to satisfy a boundary condition at the solenoid ends of the form  $\partial P/\partial x$  = constant. This boundary condition is imposed by assuming the pressure at x-point 0 is (1 - TDP1) times the pressure at x-point 2 and the pressure at x-point (NX+1) is (1-TDP2) times the pressure at x-point (NX-1). If NBOUND = 1 the solenoid is assumed to be open ended and similar boundary conditions are imposed on each end. If NBOUND = 2 symmetric axial boundary conditions are imposed around x-point 2.

#### A. 4.7 Miscellaneous

í

Artificial viscosity coefficients  $a_r$  and  $a_x$  defined in Section 2.3, are set to values AVR and AVX. If these quantities are set to values between 1.5 and 2.0 shocks will be spread over several mesh points. Artificial viscosity is turned off by setting these values to zero.

ú

When a laser beam is bleaching its way through a cold plasma there is no MHD motion well ahead of the beam front, and it is just a waste of CPU time to do calculations in this region. Initially, calculations are not done for x-points 1 to NXMIN1. Then as the beam bleaches its way through the plasma more x-points are added to the region in which calculations are made. Calculations are made up to NXMIN x ints ahead of the laser front. This feature is dissabled if NXMIN1  $\leq$  NBOUND.

The ratio of ion mass to proton mass is specified by parameter RM. A deuterium plasma can be simulated by setting RM = 2. Before the program can be used for higher z gasses the transport coefficients must be modified to include the z dependence.

Calculations on laser propagation and heating are made only for simulation time t < T4. Thus, if the simulation is to be run with no laser heating, setting T4 to a value less than 0 will save CPU time.

#### A.4.8 Post Processor Routines

Readable output from the main routine is very limited. Additional information can be obtained from other routines which read the unformatted data from the main routine and do additional processing.

Routine TAPRINT reads unformatted data from disk or tape

and prints results from pre-selected timesteps.

 $\mathbf{a}$ 

Routine ENERGY checkes validity of the results of a simulation by calculating the extent to which energy and mass were conserved during the simulation. A more complete description of this routine can be found in the next section.

Routine MPLT3D reads unformatted data from disk or tape and creates three dimensional plots of  $T_e$ ,  $T_i$ ,  $v_z$ ,  $v_r$ , B, and  $n_o$  as well as a plot of the radius of shell boundaries as a function of x. All variables are converted to natural units and scaling is done automatically. Features such as perspecive plotting and removal of hidden lines are included. The routine works interactively on a Tektronix 4013 or 4015 terminal or will create Calcomp files. The routine automatically determines if output is to a Tektronix terminal or a Calcomp file. If it is writing on a terminal each plot is scaled to fill the screen. By pressing return on the terminal the plot is erased and the next one is drawn. If output is to a Calcomp file, plots are automatically spaced to use a minimum amount of plotter paper.

Routine MPLTID reads unformatted data from disk or tape, and creates the following plots at specified times; (1) laser beam power and axial beam intensity as a function of x; <sup>(2)</sup>/<sub>2</sub>) axial values of  $T_e$ ,  $T_i$ , and  $n_o$  as a function of x; (3) values of  $T_e$ ,  $T_i$ , and  $n_o$  at a given x as a function  $f_i$ ; and (4) the axial values of  $v_z$  are plotted as a function of x. In addition, the axial values of  $T_e$ ,  $T_i$ , and  $n_o$ at a specified x are plotted as a function of simulation time. In the same manner as MPLT3D, the routine determines if the output is to a Tektronix terminal or a Calcomp file and adjusts its format appropriately.

## A.4.8 Listing of Input File

Initial timestop. If WTINE-0 program starts at t=0. If WTINE-s program starts from s'th timestep. Initial conditions BTINE are read off Fortras Unit #2. Harisus assber of timesteps to take. Program will take HTOTAL (WTOTAL-WTINE) timesteps waless toteax first. Sarinus time (in seconds) for calculation to run. TRAX Print output frequency; - on Fortran Unit #6. XP 2T No output if MPRT-999. Unformatted output frequency on Fortran Unit #3. Used for NTAPE plotting, energy conservation check and restarting program. 10 output if STAPE-999. Laser powers in Watts at times Ti. Values are calculated by linear interpolation between values of Ti. Laser power is 0.0 at PWRI time=0, PWR1 at T1, PWR2 at T2, PWR3 at T3, and 0.0 at time=T4. See PHEL. τi Bleaching velocity of laser in cm/mec. Value is artificially lisited to account for energy used in plasma formation. If no limit VLAS is desired set VLAS=3.0D+10. Parameter for averaging on half timesteps. Quantity Q given by P P Q (S, X) = PP+ (Q2 (S, XH) + Q2 (S, XP) ) +QQ+Q2 (S, X) +DT 1+QDOT where QU=1.0-2.0+PP. Dimensionless constant multiple for artificial viscosity AVR in the radial direction. (Good value between 1.5 and 2.0). Dimensionless constant multiple for artificial viscomity XYX is the axial direction. (Good value between 1.5 and 2.0). Fraction of stability condition that first timestep must satisfy. A11 Praction of stability condition that other timesteps must satisfy. 11 Max. fraction electron temperature in central shells can 12 increase by in a single timestep. dar. fraction timestep can increase by in a single timestep. ¥3 ` Distance from input end of solemoid (in cm) that laser is focussed. L3 Hin. beam radius. (at position L3). R L Å Beam slope. i.e. rate of convergence or divergence of beam. 85 Bax. timestep size that is allowed. (is seconds) DTHAI Sumber of x points in front of beam front that calculations NXHIH are to be sade. Harisum number of timesteps to be taken, in LHEAT per sain tisestep. Small values save CPU time but can lead to sumerical MBHAY error including loss of energy conservation. Small values are

good if the laser beam profile does not change much is a single timestep. There is only need for this to be greater than IX is extreme cases such as a pulse travelling through a plasme that is optically this before laser beating. If TRBC-. IEUR., Temperature on Badial Boundary in Constant. 1110 If TEBC-. TALSE., Temperature ous. assumes thermal insulation on radial boundary. Ratio of (ion mass)/(proton mass). (i.e. RH=2 for deuterium) 26 Pressure at x-point 0 is assumed to be (1.0-TDP1) times TDP1 pressure at x-point 2. Pressure at x-poist (HI+1) is assumed to be (1.0-TDP2) 7**3** P 2 times pressure at x-point (#X-1). Initial x-point at which data should be printed. IXI Initial output for X=XHIN if NX I=999. Final x-point at which data should be printed. #XP Final output for X-BX if MXF-999. Initial shell for which output is printed. # 1 Final shell for which output is printed. A P Final shell for S=H if HF-999. Every HS'th shell is output on printer. 83 Type of boundary condition applied to L.H.S. RBGUND Independent if #BOUND=1, Symetric if #BOUND=2. Bumber of x - points. II Number of shells. . Length of solenoid in cm. LQ Badium of solemoid in cm. RO Initial plasma density and value used in normalization; in NO particles/cs++3. Value used in mormalization of plasma temperature in eV. TO. RO1 is the radius of the first sheli. W is the width of R01 & W the second shell in cm. Width of other shells increase linearly in such a way as to make radius of last shell BO. Initial value of B - field in Gauss. BINT Initial values for electron and ion temperatures in eV. TINT Radius at which temperature decreases to 1/2 of its value in the R IT initial conditions. Scale length over which temperature decreases. R2T Radius at which density decreases to 1/2 of its value in the 211

205

initial conditions.

N. K

- E28 Scale length over which density decreases.
- L1 Distance from laser input and of solepoid that initial density increases to 1/2 of its full value.
- L2 Scale length over which density increases.
- L4 Distance from other end of solenoid that initial density increases to 1/2 of its full value.

55 Scale length ever which density increases.

·, '

- MINING Initial value for the sumber of x-points BOT to calculate. If this value is less than or equal to -MSOUND this feature is dissabled.
- Redial DyNamics. If RDN=.TRVE. full radial-dynamics are calculated. If RDN=.PALSE. Radial Pressure Balasce is assumed. This can save CPU time since the stability condition is much less severe so larger timesteps can be taken. If something other than radial hydrodynamic stability is the limiting factor on the timestep aims this will NOT lead to significant savings.

GINPUT NTINE=0 GENN GINPUT NTOTAL=1000, THAN=1.5D-06, NPRT=20, NTAPB=10, PNR1=1.0D+09, PNR2=1.0D+09; PNR3=1.0D+09, T1=10.0D-9, T2=40.0D-09, T3=2.0D-6, T4=2.2D-6, VLAS=3.0D+10, PP=0.0, AVR=0.0, AVR=0.0, A11=0.01, A1=0.4, A2=0.5, A3=1.2, L3=25.0, RL0=0.15, BS=0.0 DTHAX=1.0D-08, NININ=6, REHAX=40, TRDC=.PALSE., Em=1.0, DTHAX=1.0D-08, NININ=6, REHAX=40, TRDC=.PALSE., Em=1.0, TDP 1=0.25, TDP2=0.25, NII=999, NIP=999, M1=1, AF=999, NS=2 GEND GINPUT NBOUND=1, NI=30, H=15, L0=100.0, E0=1.5, ND=3.817, TO=1.0D02, R01=0.045D0, Z=0.045D0, BINT=1.0D+05, TINT=1.0, R 1T=10.0, N2T=0.1, R1N=10.0, R2N=0.15, L1=-100.0, L2=10.0, L4=-100.0, L5=10.0, NININ1=25) RDN=.PALSE.

۰.

÷

## A.5 <u>Conservation of Energy and Mass</u>

The routine EMERGY reads unformatted data from disk or tape, and calculates the total energy and mass of the plasma. After every I'th input record (where I is an input parameter) the following data is printed; (1) total plasma energy; (2) total plasma mass; (3) the amount the plasma energy has changed from previous timestep; (4) the amount of energy that has escaped through solenoid ends; (5) the amount the plasma mass has changed from previous timestep; (6) the amount of mass that has escaped through solenoid ends; (7) the difference between (3) and (4) divided by total plasma energy; and (8) the difference between (5) and (6) divided by total plasma mass.

Energy and mass are calculated and printed out in dimensionless units. A unit of energy is given by  $E_0 = N_k T_L A_0$  and a unit of mass is  $M_0 = A_0 L_0 M_1 N_0$ . Two parameters, N1 and N4, specify the axial region in which mass and energy conservation is checked. The rate at which energy and mass is flowing out of the left end of the solenoid is calculated at the x-point N1 while the rate at which energy and mass are flowing out of the right end of the solenoid is calculated at the x-point N4. The total energy and mass is the sum of that calculated at each x-point in the region specified by  $(NI + 1) \le x \le (N4 - 1)$  plus half of the energy at x-points N1 and N4. If NBOUND = 2 an axis of symmetry exists around the point x = 2 and there is no energy flow past this point. For this case the routine automatically sets N1 = 2 and calculates energy and mass flow past the point x = N4 only.

The total energy (or mass) is found by summing the energy (or mass) calculated in each of the computational cells in the region of interest. In dimensionless units the mass in a cell specified by  $M_{n,x}$  is:

$$M_{s,x} = (NA)_{s,x} \Delta x$$

In dimensionless units the energy in a cell is the sum of the following components:

Thermal Energy = 
$$(\gamma-1)^{-1} N \cdot (T + T_1) \cdot A \cdot \Delta x$$
  
s,x e is,x  $\delta x$ 

Magnetic Energy =  $(B_{s,x})^2 \cdot A_{s,x} \cdot \Delta x$ 

Kinetic Energy =  $\frac{1}{2} N_{s,x} [v_{s,x}^2 + \epsilon_1^2 (v_r)_{s,x}^2] A_{s,x} \cdot \Delta x$ 

Cross-Sectional Laser Energy =  $c^{-1} P_x \Delta x$ ;  $(c = c/v_0)$ 

The total energy U is:

4

$$\int_{1}^{\infty} U = \Delta x \sum_{x} \sum_{x}^{\infty-1} P_{x} \sum_{s}^{\infty} A_{s,x} \{ (\gamma-1)^{-1} N_{s,x} \cdot (T_{e}+T_{1})_{s,x} + B_{s,x}^{2} + \frac{1}{2} N_{s,x} [v_{s,x}^{2} + \varepsilon_{1}^{2} (v_{r})_{s,x}^{2}] \}$$

The mass flow out of a solenoid end in a time  $\Delta t$  is (mass density)  $\cdot$  ( $\Delta V$ ) where  $\Delta V$  is the volume of plasma in a shell that has flowed past the solenoid end.  $\Delta V$  for a computational cell is given by:

$$(\Delta V)_{s,x} = (Av)_{s,x} \cdot \Delta t$$

In dimensionless units the mass flow past a x-point in a time  $\Delta t$  is given by  $(\Delta M)_x$  where:

$$(\Delta M)_{x} = \Delta t \sum_{s} (NAv)_{s,x}$$

The energy flow past an x-point can be found by summing the following terms:

Thermal Energy = 
$$\sum_{s} (\gamma-1)^{-1} [N(T_e + T_i) Av]_{s,x} \Delta t$$

Kinetic Energy =  $\sum_{s} \frac{1}{2} [N(v^2 + \epsilon_1^2 v_r^2) Av]_{s,x} \Delta t$ 

(Pressure)  $\Delta V$  term = work done in pushing gas out the end of the solenoid:

$$= \sum_{s} [N(T_e + T_i) Av]_{s,x} \Delta t$$

Thermal Conduction =  $-\delta \sum_{\mathbf{s}} (\mathbf{A} T_{\mathbf{e}}^{5/2})_{\mathbf{s},\mathbf{x}} [(T_{\mathbf{e}})_{\mathbf{s},\mathbf{x}+1} - (T_{\mathbf{e}})_{\mathbf{s},\mathbf{x}-1}] \frac{\Delta t}{2\Delta x}$ 

where 
$$\delta = (k_{11} T_{e}^{-5/2})_{s,x}$$

Laser Energy Flow =  $\frac{1}{2} [P_{nx+1-x}]$ .

if NBOUND = 1, and  
= 
$$\frac{1}{2} [P_{nx+1-x} + P_{nx+2-x} - P_{nx-2+x} - P_{nx-3+x}]$$

if NBOUND = 2.

(∆U)<sub>x</sub>.

$$(\Delta U)_{\mathbf{x}} = \sum_{\mathbf{a}} \left\{ \left[ \frac{\gamma}{\gamma - 1} N(T_{\mathbf{e}} + T_{\mathbf{i}}) + \frac{1}{2} N(v^2 + \varepsilon_1^2 v_r^2) \right] \mathbf{A} v \right\}_{\mathbf{s}, \mathbf{x}} \Delta t$$

$$-\delta \sum_{\mathbf{s}} (\mathbf{A} \mathbf{T}_{\mathbf{e}}^{5/2})_{\mathbf{s},\mathbf{x}} [(\mathbf{T}_{\mathbf{e}})_{\mathbf{s},\mathbf{x}+1} - (\mathbf{T}_{\mathbf{e}})_{\mathbf{s},\mathbf{x}-1}] \frac{\Delta t}{2\Delta \mathbf{x}}$$

•

+ Laser Energy Flow.



Figure A.1 Spatial positions at which variables are calculated. N, T<sub>e</sub>, T<sub>i</sub>, v<sub>x</sub>, and B are calculated at shell centers denoted by o. R, W, and u are calculated on shell boundaries denoted by x. The indexing of variables is indicated on the bottom and right hand side of the figure.



Figure A.2 Flow chart illustrating two-step scheme.



ı.

Figure A.3 Position and indexing of numerical grid for calculation of laser power P, and absorbed energy E, relative to positioning and indexing of MHD variables.

.

### Appendix B Linear Susceptabilities

In the following analysis,  $F_{TE}(x,t)$  is the total force acting species  $\epsilon$  ( $\epsilon = e$  or 1) and  $E_{ce}(x,t)$  is the component of the Coulomb field resulting from density fluctuations of species  $\epsilon$ . The distribution function  $f_{\epsilon}(x, v, t)$  can be split into two parts; the equilibrium distribution function  $f_{\epsilon}^{0}(v)$ , and the perturbation  $f_{\epsilon}'(x, v, t)$ .

$$f_{\varepsilon}(x,v,t) = f_{\varepsilon}^{*}(v) + f_{\varepsilon}^{\dagger}(x,v,t)$$

The linearized Vlasov equation can be written as: lacksquare

$$\partial_t f' + v \partial_x f' + \frac{F_{Te}}{m_e} \partial_v f^{\bullet} = 0$$

Taking a double Fourier transform in x and t yields:

$$-1\omega f_{\varepsilon} + 1 k v f_{\varepsilon} = -\frac{F_{T\varepsilon}}{m_{\varepsilon}} \partial_{\varepsilon} f_{\varepsilon}^{*}$$

where  $f_{\epsilon}$  and  $F_{T\epsilon}$  are the double Fourier transforms of  $f_{\epsilon}'$  and  $F_{T\epsilon}$  respectively. Thus:

$$\int_{\epsilon}^{\infty} = -\frac{iF_{T\epsilon}}{m_{\epsilon}(\omega - kv)} \quad \partial_{v} f_{\epsilon}^{\bullet}$$
(B.1)

From Gauss' Law:

$$\partial E_{ce}(x,t) = 4\pi B_{e}q_{e}$$

where  $q_E$  is the species charge and  $n_E$  is the density perturbation defined by:

$$\mathbf{n}_{\varepsilon}(\mathbf{x},t) = \mathbf{n}_{0} \int f'_{\varepsilon}(\mathbf{x},\mathbf{v},t) d\mathbf{v}$$

which implies:

$$\sum_{c \in a}^{b} -\frac{-1}{k} \frac{4\pi n_{o} q_{c}}{k} \int_{c}^{a} f_{c} dv \qquad (B.2)$$

where  $E_{CE}^{(k,\omega)}$  is the double Fourier transform of  $E_{CE}^{(x,t)}$ . Substitution of (B.1) into (B.2) yields:

$$\sum_{c \in (\omega, k)}^{n} = -\frac{1}{q_{\varepsilon}} \chi_{\varepsilon}(\omega, k) F_{T\varepsilon}(\omega, k)$$
(B.3)

where

$$\chi_{\varepsilon}(\omega, \mathbf{k}) = \frac{\sum_{k=1}^{2} \sum_{k=1}^{\infty} \frac{\mathbf{k} \partial \mathbf{f}}{(\omega - \mathbf{k}\mathbf{v})} d\mathbf{v} \qquad (\mathbf{B}.4)$$

and  $\omega_{pe} = (4\pi n_{p} q_{e}/m_{e})^{1/2}$ . Assuming  $f_{e}^{*}(v^{b})$  is Maxwellian:

$$f_{\varepsilon}^{\bullet}(v) = \frac{1}{\sqrt{\pi} v_{\theta \varepsilon}} exp[-v^2/v_{\theta \varepsilon}^2]$$

where  $v_{\theta \varepsilon} = (2k_B \tau/m_{\varepsilon})^{1/2}$  is the thermal velocity of species  $\varepsilon$ . With this value of  $f_{\varepsilon}^*$ ,  $\chi_{\varepsilon}$  can be expressed as:

$$x_{\varepsilon} = -\frac{2\omega_{p\varepsilon}^{2}}{k\sqrt{\pi}} \int_{\theta\varepsilon}^{\infty} \frac{v \exp[-v^{2}/v_{\theta\varepsilon}^{2}]}{\omega - kv} dv$$

Using techniques discussed in Stix,  $\chi_{\epsilon}$  can be re-written as:

$$\chi_{\varepsilon}(\omega, \mathbf{k}) = 2 \frac{\frac{\omega^2}{p\varepsilon}}{\mathbf{k}^2} \cdot \frac{1}{\mathbf{v}_{\theta\varepsilon}^2} \{1 + i \alpha_{o\varepsilon} F_o^{\varepsilon}\}$$
(B.5)

where  $\alpha_{0\varepsilon} = \omega/(kv_{\theta\varepsilon})$  and

L.

$$\mathbf{F}_{0}^{\varepsilon} = \frac{\mathbf{i}\mathbf{k}}{\sqrt{\pi}} \int_{-\infty}^{\infty} \frac{\exp[-\mathbf{v}^{2}/\mathbf{v}_{\theta\varepsilon}^{2}]}{(\omega - \mathbf{k}\mathbf{v})} d\mathbf{v}$$
(B.6)

Techniques for evaluating  $F_0^{\varepsilon}$  are discussed in Stix <sup>31</sup> (Chapter 8).

Appendix C Ponderomotive Force Due to Antiparallel Beams

In this Appendix the ponderomotive force resulting from the mixing of two antiparallel electromagnetic beams is found. Only terms at the beat frequency of the beams are retained.

The force on a particle (electron or ion) is given by

$$\vec{F} = q_{\varepsilon} (\vec{E} + \frac{1}{c} \vec{v} \times \vec{B}) \qquad \varepsilon = e \text{ or } i \qquad (C.1)$$

where  $q_E$  is the species charge,  $\vec{E}$  and  $\vec{B}$  are the total electric and magnetic fields due to the interacting beams and  $\vec{v}$  is the particle velocity. The total vector potential due to the interacting beams can be written as

$$\mathbf{\dot{x}} = \mathbf{A}_{1} \ \mathbf{e}^{\mathbf{1}\xi_{1}} \ \hat{\mathbf{y}} + \mathbf{A}_{2} \ \mathbf{e}^{\mathbf{1}\xi_{2}} \ \hat{\mathbf{y}} + \mathbf{c.c.}$$

where  $\xi_i = (k_i x - \omega_i t)$ , i = 1, 2. Then to first order the particle velocity is given by

$$\mathbf{m}_{\varepsilon} \frac{\partial \mathbf{v}}{\partial t} = -\frac{\mathbf{q}_{\varepsilon}}{c} \frac{\partial \mathbf{A}}{\partial t}$$

or

$$\dot{\mathbf{v}} = -\frac{\mathbf{q}_{\varepsilon}^{\mathbf{A}}}{\mathbf{m}_{\varepsilon}^{\mathbf{C}}}$$
(C.2)

.

The ponderomotive force now comes from the second term in (C.1) where v



is approximated as in (C.2). This force is given by

r

$$\mathbf{F}_{\varepsilon} = \frac{\mathbf{q}_{\varepsilon}}{c} \stackrel{+}{\mathbf{v}} \stackrel{+}{\mathbf{x}} \stackrel{B}{\mathbf{B}} = \frac{\mathbf{q}_{\varepsilon}}{c} \stackrel{+}{\mathbf{v}} \stackrel{+}{\mathbf{x}} (\stackrel{+}{\mathbf{v}} \stackrel{+}{\mathbf{x}} \stackrel{+}{\mathbf{A}})$$

or

$$\dot{F}_{\varepsilon} = -\frac{q_{\varepsilon}^{2}}{mc^{2}} (A_{1}e^{i\xi_{1}} + A_{2}e^{i\xi_{2}} + c.c.) (ik_{1}A_{1}e^{i\xi_{1}} + ik_{2}A_{2}e^{i\xi_{2}} + c.c.) \hat{x}$$

Retaining only terms at the beat frequency in this expression leads to

$$\mathbf{F}_{\varepsilon} = -\frac{\mathbf{i}q_{\varepsilon}^{2}}{\mathbf{w}_{\varepsilon}c^{2}} \mathbf{A}_{1}\mathbf{A}_{2}^{*}\mathbf{k}_{3}\mathbf{exp}[\mathbf{i}(\mathbf{k}_{3}\mathbf{x} - \omega_{3}t)] \hat{\mathbf{x}} + c.c.$$
where  $\mathbf{k}_{3} = \mathbf{k}_{1} - \mathbf{k}_{2}$  and  $\omega_{3} = \omega_{1} - \omega_{2}$ .

۰.

```
С
     .
            A TWO DIRENSIONAL MAGNETORYD RODYNAMIC SIMULATION OF A
С
     .
             PLASMA IN A SOLEWOIDAL MAGNETIC FILLD.
             RICHARD D. HILROY & J. H. HCHULLIN 1977.
С
     ٠
INPLICIT REAL (A-H, 0-Z)
      BEAL SS, NEND, N732, NDOT, SQRT, LO, NO
      BILL # (30,60) , A (30,60) , Y (30,60) , TE (30,60) , TI (30,60) ,
         #2(30,60), A2(30,60), Y2(30,60), TE2(30,60), TI2(30,60),
     ٠
         AA(30,60),BB(30,60),CC(30,60),DD(30,60),
         PHI (31), P (31), G (31), IE (31), II (31), C (31), TDOT (31),
HI (31), HE (31), GH (31), GP (31), HISTAR (31), (31),
     ٠
     .
         ATER (31) , HTER (31) , TTER (31) , TITER (31) , EP (30, 60) , KE, KI,
         P (120) , P2 (120) , TE 15 (30, 60)
         , TE3 (31) , TI3 (31) , DDE (31) , DDI (31) , AE (31) , AI (31) , BE (31) , BI (31) , CE (31) , CI (31) , DE (31) , DI (31) , ZE (31) , ZI (31) , PE (31) , /I (31)
          , PE (30, 60), U (30, 60), U2 (30, 60), H (30, 60), H2 (30, 60), UDOT (31)
     .
          , UTER (31) , R (30, 60) , ATER 1 (31) , EP 52 (31) , EH (31) , PH (31) , N50, N5H
      INTEGER S, SP, SH, X, XH, XP, ST 2P, XMIN, XMP1, XMM1
      LOGICAL TROC, RDN, RPD
      DATA P/120+0.0D0/, P2/120+0.0D0/, ST EP/1/, PE/1800+0.0D0/
     1, EP/1800+0.0D0/, EPB/. TEUE./
      COM
          ON LO,RO,NO,TO,PWR1,PWR2,PWR3,T1,T2,T3,T4,THAX,VLAS,TDP1,TDP2
          XV, XVX, PP1, HX, H, HTOTAL, HBOUHD, HTIME, HPRT, HTAPE, HIMIH, RDH, TEBC
               C+4
                                                                          ....
С
      STINE - SUNDER OF TIMESTEPS TAKES SO FAR.
С
      MPRT = PRINT OUTPUT PREQUENCY.
                                           NO OUTPUT IP SPRT-999.
С
      NTAPE - TAPE OUTPUT PREQUENCY.
                                           NO OUTPUT IF HIAPE-999.
                                                                            .
С
      STOTAL = TOTAL TIMESTEPS TO TAKE.
     C***
      CALL INIT (DT, DX, DEL, PCOLL, EPS20, PHI, V, H, A, TE, TI
     1, DELE1, DELE2, DELI1, DELI2, ABO, CS, TIME, THO, XLAS, B, U, W, XHIH)
      IT (MTIRE. ME. 0)
     +CALL REST (DT, DX, DEL, PCOLL, EPS20, PHI, V, N, A, TE, TI
      1, DELE1, DELE2, DELI1, DELI2, ABO, CS, P, P2, TIME, THO, XLAS, R, U, W, XHIW)
      IF(2DS) RP3=.FALSE.
      GAN=5.0D0/3.0D0
      EPS1 = (10/L0)
      EPS12= (20/L0) ++2
      AVE2 - (AT+EPS 1) ++2
      ##1=#-1
      BIS1=SI-1
      #IP1=#I+1
      #IP2=#I+2
      #DT=0.5+DT
      TDI=2.0+DI
      #12=2+#X
```

#12#1=#12-1 #12#2=#12#1-1 #X#1=#X-1 GH 1=GAH-1 PPP=1.0 QQQ=2.0-PPP P(1)=0.0 P (8+1)=0.0 6 (1) =0.0 G (H+1)=0.0 IE(1)=0.0 IE (A+1) =0.0 II(1)=0.0 II (#+1)=0.0 0.0 = (B) TOOU C-----INITIALIZE TEMPORARY ARRAYS DO 50 S=1,8 ATER (5) = 0. NTER (S) = 0. VTER (S) = 0. 50 TITEN(S) = 0. GOTO425 C-75 CONTINUE TIME - TIME+ DT HTINE = HTINE+1 NDT = 0.5+DT DT3=1.0D0/(3.0D0+DT) C----- CALCULATE XHIN. IF (IMIN. LT. BBOUND) STOP IP (IMIN. GT. MBOUND) CALL ND PTS (XRIN, P, NX, NBOUND, KYRIN) 1 XEP1 = XHIN+1 XHH1 = XHIN-1 BPHY = BX+1-IHIB .85 CONTINUE DT 1= HDT+STEP GDT 1=GE 1+DT 1 PP= (2-STEP) + PP 1 QQ=1.0D0-2.0D0\*PP C----- DEFINE TE\*\*1,5 AS TE15. DO 14 X=1, MX 14 TE15(S,X) = TE(S,X)+SQRT(TE(S,X)) DO 14 5=1,8 ICALL LHEAT (H, NX, STEP, TIBE, T1, T2, T3, T4, NBOUND, TB15, XHIN 2, PURI, PERZ, PURS, XLAS, VLAS, RO, ABO, P, P2, PR, M, H, A, TE, EP, DX, MIP1, TO) C-1-----AXIAL LOOP DO 300 X=XBIN, NX IBR = 1• IF (X.EQ. 1) IBR = 2 IP (X.BQ. #X) IBR = 3 X8 = X-1  $\mathbf{XP} = \mathbf{X+1}$ SUN1 = 0. SU#2 = 0. ABHD = 0.

#J1=0.0 #J2=0.0

> , . 1

ESP=0. 5+R (1, X)

4

219

```
V_{PS} = U(1, I) / R(1, I)
Q2 = 0.0
         IF (VPS. LT. 0. 0) Q2 = # (1,2) *ATE2*VPS**2
         PTL = = (1, Z) + (TE (1, Z) +TI (1, Z)) + (MI (9 /A (1, Z)) ++2
         DEDIL = 0.0
          ---- FIRST SIELL LOOP
  C-2--
        DO 150 S=1, H
         SP 4 5+1
        SH = 5-1
         ¥S = ¥(S,X)
        AS = A (S, X)
        TS = Y(S,X)
        TES = TE(S, I)
        TIS - TI (S, I)
        IS = TES + TIS
        PS = #5+TS
        PHS = PHI(S)
        35 - PES/AS
                                                                        •
        ¥5 = ¥(5,1)
 US = D(S, X)
C-2-1----CALCULATION OF PLUXES THEU UPPER BOUNDART
        IF (S. EQ. 8) GO TO 101
     ٦.
        BSP = PHI(SP)/\lambda(SP,I)
        PH2 = PHI(SP)
        PH12 = PHS+PH2
        ABND = ABND + AS
        15=152
       RSP=0.5* (R (S, I) +R (S+1, I))
        DE 1=AS/ (2.0+ ES)
       DR2=A (S+1, I) / (2.0+RSP)
       DE12=DE1+DE2
       # 1=D#2/D#12
       #2=D#1/D#12
       BBHD=#1+#$+#2+P#2/A(SP,X)
       DBTB = 4. +AB HD+ (PH 2/A (SP, X) - BS) / PH 12
      YBED = #1+VS + #2+V(SP,I)
       TIBBD = RITTIS + R2+TI (SP, I)
       TEBND = RI+TES + R2+TE (SP,X)
       SQTEBD = SQRT (TEBH D)
       EPS2 (S) = EPS20/ (TEBHD+SQTEBD)
       F(SP) = EPS2 (S) *#BHD*DBIB
       G (SP) = VBID+F (SP)
       PBYH = F (SP) /HBHD++GH1
       II(SP) = TIBND+PBYN
       IE (SP) = TEBND+PBTN
      HJ2=16.0+EPS2(S) +ABHD+(((PH2/A(S+1,X))-(PHS/AS))/(AS+A(S+1,X)))++2
  101 HJ2=0.0
  100 #T32 = #S/TE 15 (5, I)
      HJUL=0.5+ (HJ 1+HJ 2)
       NJ1=8J2
      XHU=PCOLL+HT32
      IF(XNU.GT.DT3) XNU=DT3
      TEQ=INU+ (TES-TIS)
C-2-2----CALCULATE SPACE DERIV'S AND HET SHELL FLUXES
      DELP = P(SP) - P(S)
      DELG = G(SP)-G(S)
      PEGRI = ES++GRI
      DELIE - PHGE 1+ (IE(SP)-IE(S))
```

•

.

```
DELII = PHGR1+ (II (SP)-II (S))
   AU - AS+US
   GO TO (30,40,20) , IBR
30 TDH = #(S,XP)-#(S,XR)
   TDV = V (S, XP)-V (S, XH)
   TDTE = TE(S, IP)-TE(S, IH)
   TDTI = TI(S, XP) - TI(S, XR) 
TDP = H(S, XP) + (TE(S, XP) + TI(S, XP)) - H(S, XR) + (TE(S, XR) + TI(S, XR)) 
   IDNAY = + (NS+AS+ (V (S, XP) - V (S, XH)) + NS+VS+ (A (S, XP) - A (S, XH))
           + AS+YS+ (# (S,XP) -# (S,XH)))
  1
   \Delta T = A (S, XH) + TE (S, IH) + TE (S, XH)
\Delta T = A S + TE S + TE (S, I)
    ATP = A(S, XP) + TE(S, XP) + TE15(S, XP)
   TDU = U(S, XP) - U(S, XH)
    TDN = W(S,XP) - W(S,XH)
    VIN=V(S,XA)
    YXP=Y (S, IP)
    Q==0.0
    Q#=0.0
    IF (VIP.LT.VS) QP= (AVI+ (VIP-VS)) ++2 + 0.5+(US+U(S,IP))
    IP (VS. LT. VXH) QH= (AVX+ (VS-VXH)) ++2 + 0.5+(H(S, XH)+HS)
    GO TO 21
20 TDH = 2.0+ (H (S,I)-H(S,IH))
    TDV = 2.0+ (V(S,I)-V(S,IH))
    TDTE = 2.0+ (TE (S,X)-TE (S,XH))
    IDTI = 2.0+ (II (S, I)-II (S, IN))
    TDP = -# ($, IH) + (TR ($, IH) +TI ($, IH)) +TDP2
    TDHAY=2.000+ (#S+AS+ (YS-T (S,XH)) + #S+YS+ (AS-A (S,XH))
   1+15+YS+ (#3-# (5,XA) ))
    ATE = A (S, XH) +TE (S, XH) +TE15 (S, XH)
    ATO = AS+TES+TE15(S,X)
    ATP = ATO
    \begin{array}{l} TDU = 2.0 + (U(S,X) - U(S,XH)) \\ TDu = 2.0 + (U(S,X) - V(S,XH)) \end{array}
    QP = 0.0
    QH = 0.0
    GO TO 21
40 TDH = 2.0+ (H (S, IP) -H (S, I))
TDV = 2.0+ (V (S, IP) -V (S, I))
    TDTE = 2.0+ (TE (S, IP) -TE (S, I) )
    TDTI = 2.0* (TI (S, XP) -TI (S, X))
    TDP = #($, 1P) + (TE($, 1P) +TI($, 1P) ) +TDP1
    TDHAY=2.0D0+ (HS+AS+ (V (S, IP) - VS) + HS+VS+ (A (S, IP) - AS)
   1+AS+VS+ (N (S, XP)-NS))
    ATH = AS*TES*TE15(5,X)
    ATO = ATH
    ATP = A (S, XP) +TE (S, XP) +TE15 (S, XP)
    TDU = 2.0+ (U (S, IP) -U (S, I))
    TOW = 2.0+ (# (S,XP) -# (S,X))
    QP = 0.0
     QH = 0.0
 21 CONTINUE
     PTU = NBND+ (TEBND+TIBND) + BBND++2
     DUDIU = TDW/TDI
     DPT = PTU - PTL
     DWDIN = 0.5+ (DWDIL+DWDIU)
     PTL = PTU
     DUDIL = DUDIU
     ST = DYDIN+D PT/AN
     DQDI = (QP-QH) /DI
```

```
C(S) = -( TDHAY/TDI + DELF )
VDOT(S) = -( VS*IDV+TDP/HS )/TDI - (DELG-VS*DHLF)/AH - DQDI/HS
     1 + BT
      HE(S) = -VS+ (TDTE-OR 1+TES+TDE/HS)/TDI - (DELIE-TES+DELP)/AB
        -TEQ + GH1+EP(S, I) + GH1+HJUL/HS
     1
      HI(S) = -VS*(IDTI-GR1*TIS*TDE/BS)/IDI - (DELII-TIS*DELF)/AB
        +TBQ
     1
      GF = 0.5+GH1+DEL+DE/(HS+AS+DX++2)
      GH (5) = GF+ (ATH+ATO)
      GP (S) = GP+ (ATP+ATO)
C----- CALCULATE COBPICIENTS FOR QUASI-INPLICIT PERP. MEAT FLOW.
      IF(S.EQ.H) GO TO 5
      KE = DELE1+TEBHD+TEBHD+SQTEBD
             / (1.0 + DELE2+TEBHD++3+BBHD++2/(HBHD++2))
      1
      KI = DELI1+TIBHD+TIBHD+SQRT (TIBHD)
            / (1.0 + DELI2+TI BHD++3+BBHD++2/(HBHD++2))
      1
      1=4.000+ABHD/(AS+A (SP, I))
       DDE(S)=Z+KE
       DD1(5)=2*KI
       IF (S. 2011) 60 TO 5
       Z = GDT1 / (#S+LS)
       AE (S) = 2+DDE (S) +PPP
       A1 (5)=2+DDI (5) +PPP
       BE (S) = 1. 0+Z* (DDE (S) +DDE (S-1) ) *PPP
      BI (S) = 1.0+2* (DDI (S) + DDI (S-1) ) * PPP
    .
       CE(S) = Z+DDE(S-1) +PPP
       CI (S) = Z + DDI (S-1) + P PP
       DE (S) =TE2 (S, I) + I+ (DDE (S) +TE2 (S+1, I) - (DDE (S) + DDE (S-1)) +TE2 (S, I)
           +DDE (S-1) +TE2 (S-1, X) ) +QQQ
     1
       DI (S) = TI2 (S, X) + Z* (DDI (S) +TI2 (S+1, X) - (DDI (S) + DDI (S-1)) +TI2 (S, X)
  1
          +DDI(S-1) +TI2(S-1,I)) +QQQ + HI(S) +DT1
      1
     5 CONTINUE
       IF (RPB) GO TO 150
C----- SOLVE ZONS. ASSOCIATED WITH BADIAL DYNABICS.
C----- FIRST FIND ARTIFICIAL VISCOSITY.
       IF (S. BQ. H) GO TO 150
       VPSP = U(SP, I)/R(SP, I)
       01 = 0.0
       IF (VPSP.LT.VPS) Q1 = NS*AVE2* (VPSP-VPS) **2
       DODS = 2.0*(Q1-Q2)/PH12
       Q2 = Q1
        YPS = YPSP
       IF (S. ME. 1) GO TO 161
        DUDS = U(2, X) / PH 12
                                                      .
        GO TO 162
   161 DUDS = (U(SP,X)-U(SH,X))/PH12
162 DUDX = TDU/TDX
        D = 4.0+#S+BBND+ (BSP-BS) /PH12
        Z = 2.0+BBND+WS/(EPS12+HBND)
        DPDS = ((#(SP,X)*(TE(SP,X)+TI(SP,X))) - (#S*TS))+2.0/PH12
        DB2DS = ((BSP++2) - (BS++2))+2.0/PH12
UDOT(S) = -VBHD+DUDX - EPS2(S)+D+DUDS + (US++2)/WS
             -1+ (DPDS + DB2DS + DQDS)
       1
 C----- SOLVE FOR W SENI-IMPLICITLY.
        DUDI = TDU/TDI
        I = 2.0+7.252 (S) + (BB# D++2)+¥5
        IF (S. ME. 1) GO TO 163
        AH = DT1+1/(PH2+PH12)
        BE = 1.0 + DI1+I+ (1.0/(PH2+PH12) + 1.0/(PHS+PH12))
        DE = W2(3,X) + DT1+(2.0+US - YBHD+DHDX)
```

....

```
+ A #+#2 (SP, X) - (BH-1) +#2 (S,X)
     1
      28 (S) = A8/B8
      P#(S) = D#/88
      GO TO 150
  163 AB = DT1+Z/(PH2+PH12)
      CH = DT1+Z/(FHS+PH12)
       BH = 1.0 + AH + CH
      DH = W2(S,X) + DT1+(2.0+US - TBHD+DWDX)
           + AH+W2 (SP, X) - (BH-1) +W2 (S,X) + CH+W2 (SH,X)
      1
      EH(S) = AH/(BH - CH+EH(SH))
      FE(5) = (DE + CH+FE(SH))/(BH -CH+EH(SH))
  150 CONTINUE
      IF(RPB) GO TO 102
C----- SOLVE FOR W(S,X).
       ¥(8,X) = ¥2(8,X)
       DO 164 J=1,881
       S = H-J
       SP = S+1
       # (S, X) = EH (S) +# (SP, X) +FE (S)
  164 CONTINUE
      --- USE NEW VALUE OF W TO FIND ATEN 1(5).
c--
      ATER1(1) = W(1,X)
       DO 165 5=2,8
  165 \text{ ATEN1}(S) = (P(S, X) - P(S-1, X))
  ----- FIND HORE COEFICIENTS FOR IMPLICIT EQUATIONS.
С
  102 Z=GDT1/(H(1,X) +A(1,X))
       EEE=2+DDE(1) +PPP
       FFF=1.0+EEE
       GGG=TE2(1,1)+EEE+(TE2(2,1)-TE2(1,1))+QQQ/PPP
       EE(1) = EEE/FFP
       PE(1)=GGG/FFF
       EEE=2+DDI(1)+PPP
       FFF=1.0+BEE
       GGG=TI2(1,X) +EEE+(TI2(2,X)-TI2(1,X))+QQQ/PPP + HI(1)+DT1
       EI(1)=BEE/PPP
       FI(1)=GGG/FFF
       DO 6 5=2,881
       EE(S) = AE(S) / (BE(S) - CE(S) + EE(S-1))
        \begin{array}{l} \textbf{EI(S) = AI(S) / (BI(S) - CI(S) * EI(S - 1)) } \\ \textbf{FE(S) = (DE(S) + CE(S) * FE(S - 1)) / (BE(S) - CE(S) * EE(S - 1)) } \\ \end{array} 
       PI(S) = (DI(S) + CI(S) + PI(S-1)) / (BI(S) - CI(S) + EI(S-1))
     6 CONTINUE
C----- FIND TZ3(H) AND TI3(H).
       IF (TRBC) GO TO 15
       Z = GDT1/(B(H, X) + A(H, X))
       FFF=2+DDE (N-1) +PPP
       EEE=1.0+FF7
       GGG=TE2(B, X) - FFF=(TE2(H, X) - TE2(B-1, X)) = QQQ/PPP
       TE3 (8) = (GGG+FFF+FE (881))/(EEE-FFF+EE (881))
       FFF=2+DDI (681) +PPP
       EBE=1.0+FFF
       GGG=TI2(8,1)-FFP+(TI2(8,1)-TI2(881,1))+QQQ/PPP + HI(8)+DT1
       TI3(8) = (GGG+FFF+FI(881))/(EEE-FFF+EI(881))
       GO TO 16
    15 TE3(H) = TE2(H,I)
       TI3(B) = TI2(B,X)
    16 CONTINUE
C----- FIND TE3(5) AND TI3(S) - S=HE1 TO 1.
       DO 7 J=1,881
       S=#-J
```

```
TE3(S) = EE (S) +TE3 (S+1) + FE (S)
      TI3(S) = BI (S) +TI3 (S+1) + FI (S)
    7 CONTINUE
C----- ADD RATE OF CHANGE OF TE AND TI TO HE(S) AND HI(S).
      DO 8 5=1,8
      IBDI= (IS3(S)-IE2(S,I)) /DI1
      TIDT= (TI3 (S) -TI2 (S, I) ) /DT1
      HE(S) = HE(S) + TEDT
      HI(S)=TIDT
    8 CONTINUE
      IF (RDN) GO TO 103
C-2-4----CALCULATE TERMS FOR ADOT SUMS
      DO 9 5=1,#
      GO TO (34,44,24) , IBR
   34 THETA = (GP(S) *TE(S, IP) - (GH(S) +GP(S)) *TE(S, X) +GH(S) *TE(S, IH)) /DT
      GO TO 25
   24 IRETA = (GP(S) *TE(S, X) - (GR(S) +GP(S)) *TE(S, X) +GE(S) *TE(S, X8) ) /DT
      GO TO 25
   44 THETA = (GP(S) *TE(S, IP) - (GH(S) + GP(S)) *TE(S, I) + GH(S) *TE(S, I) ) / DT
   25 CONTINUE
       BSTAR(S) = H(S, X) + A(S, X) + (BE(S) + HI(S) + THETA)
       PISTAR = 2.* (PHI (S) / A (S, X) ) **2 + GAR*N (S, X) * (TE (S, X) +TI (S, X))
       SUE1 = SUE1 + (HSTAR (S) + GAH+ (TE (S, X) +TI (S, X) ) +C (S) ) /PISTAR
       SUR2 = SUB2 + A(S,X) /PISTAR
    9 CONTINUE
       PIDOT = SUH1/SUH2
C-3----SECOND SHELL LOOP
  103 CONTINUE
       ATOT=0.0
       DO 275 S=1,8
       HS = H(S, I)
       AS = A(S, X)
       YS = Y(S, I)
       TES = TE (S,I)
       TIS = TI(S, I)
       TS = TES + TIS
       PS = HS*TS
       PHS = PHI(S)
       BS = PHS/AS
C-3-1----CALCULATE TIME DEBIVATIVES
      IF(RPB)
      1ADOT = (HSTAR(S)+GAM*TS*C(S)-AS*PIDOT)/(2.*BS*BS+GAM*PS)
       IP(RDN) ADOT = (ATEN1(S)-A2(S,X))/DT1
       HDOT = (C(S) -HS+ADOT) /AS
       DINDOT = GHI+NDOT/HS
       TIDOT = HI(S) + TIS*DLNDOT
C-3-2-----Q IS NON-CONDUCTIVE PART OF TEDOT
      Q(S) = HE(S) + TES + DLHDOT
C----- TRANSPER TEMP. TO PERS.
       IF(I.EQ.ININ) GO TO 60
       A(S, IH) = ATEH(S)
       ATOT = ATOT+A(S,XH)
       IP(BPB) W(S,IN) = ATOT
       # (S, IH) = #TEH (S)
       ¥ (S, XH) = ¥TEH (S)
       TI (S, IN) = TITEN (S)
       IF (BDN) U (S, IH) = UTEH (S)
R (S, IH) = SQRT (W (S, IH))
C----- ADVANCE TEMP. QUANTITIES TO T+(HDT+STEP).
    60 CONTINUE
```

```
GU TO (37,47,47) ,IBR
   37 CONTINUE
      IF (RDH) ATER (S) = ATER1 (S)
      IF (BDH) UTEN (S) = PP+ (U2 (S, XH) +U2 (S, XP) ) + QQ+U2 (S, X) +DT 1+UDOT (S)
      IF (MPB) ATER (S) = PP+ (A2 (S, IH) + A2 (S, IP) ) +QQ+ A2 (S, I) + DT 1+ ADOT
      HIER (S) = PP+ (H2 (S,XH)+H2 (S,XP))+QC+H2 (S,X) + UT 1+HDOT
      YTER (S) = PP+ (V2 (S, XR) + V2 (S, XP) ) +QQ+V2 (S, X) + DT1+VDUT (S)
      TITER(5) = PP+ (T12 (S,IH) +T12 (S,XP) ) + QQ+T12 (S,X) + DT1+T1DOT
      GO TO 27
   47 CONTINUE
      IF(RDH) ATEN (S) = ATEN1 (S)
      IF (RDH) UTEN (S) = U2 (S, X) + DT1+UDOT (S)
      IF (RPB) ATER (S) = A2 (S, X) + DT 1+ ADOT
      \texttt{MTER}(S) = \texttt{M2}(S, \texttt{X}) + \texttt{DT1+MDOT}
      YTER (S) = Y2 (S, I) + DT 1+ YDOT (S)
      TITZA(S) = TI2(S,I) + DT1+TIDOT
   27 CONTINUE
C-3-5-----COEFF'S FOR INPLICIT SCHENE
      GO TO (38,48,28) , IBR
   38 CC(S,X) =+0.25+5TEP+GH(S)
      AA(S,X) =+0.25+5TEP+GP(S)
      BB(S,I) = 1. +AA(S,I) +CC(S,I)
      DD(S,X) = 0.5*ST2P*DT*Q(S) + (PP+CC(S,X)) *TE2(S,XE) + (PP+AA(S,X))*
     C TE2(S, IP) + (QQ-CC(S, I) -AA (S, I) ) + TE2(S, I)
      GO TO 29
   28 AA (S,X) =0.0
      BB(S,I)=1.0+0.25+STEP+GH(S)
      CC(S,I) =0.25*STEP+GN(S)
      DD(S,I) = 0.5*STEP*DT*Q(S)+(CC(S,I))*TE2(S,IH)+(AA(S,I))*
     C TE2(S,X) + (1.0-CC(S,X)-AA(S,X))+TE2(S,X)
      GO TO 29
   48 AA(S,I) = 0.25 + STEP + GP(S)
      BB(S,X) = 1.0+AA(S,X)
      CC(S, X) = 0.0
      DD(S, I) = 0.5*STEP*DT*Q(S) + (CC(S, I)) *TE2(S, I) + (AA(S, I)) *
     C TE2(S, IP) + (1.0-CC (S, I) - AA (S, X)) +TE2 (S, X)
   29 CONTINUE
  275 CONTINUE
  300 CONTINUE
C-3-6----TRANSPER TEMP TO PERM AT MX
      ATOT = 0.0
      DO 310 S=1,M
      N(S, NX) = NTEM(S)
      A(S, MI) = ATEN(S)
      ATOT = ATOT+A(S, MX)
      IF (RPB) W (S, NX) = ATOT
      V(S,HI) = VIEM(S)
      IF(RDN) U(S,NX) = UTER(S)
      R(S, HX) = SQRT(W(S, HX))
  310 TI(S.HI) = TITEN(S)
¢*
      *******
С
                        IMPLICIT EQUATIONS
                                                                          .
С
      * AS SOLVED BY RICHTHYER AND BORGAN, - PAGE 198.
                                                                          .
        T (S,X) = E (S,X) +T (S,X+1)+P (S,X)
С
      ۰
      * SET E(S,X) TO AA(S,X) AND P(S,X) TO BB(S,X)
С
                                                                          .
81 = 8
      IF (TEBC) H1 = H-1
      GO TO (52,51) , MBOUND
   51 CONTINUE
```

,

.

```
----- CASE WITH STRETEIC BOUNDARY CONDITIONS.
C---
       IF (ININ. EQ. 2) GO TO 2
DO 1 $=1.8
                                                                             •
       AA(S,XRIH) = AA(S,XHIH)/DB(S,XHIH)
     1 BB(S, XHIN) = (DD(S, XHIN) +CC(S, XHIN) +TE2(S, XHIN)) /BB(S, XHIN)
       GO TO J
     2 DU 340 S=1,8
                                                                      4
       AA(S,2) = \{AA(S,2) + CC(S,2)\} / BB(S,2)
  340 BB (5,2) =DD (5,2)/BB (5,2)
     3 DO 350 X+X3P1, XX#1
       DO 350 S=1,8
       AA(S,X) = AA(S,X) / (BB(S,X) - CC(S,X) + AA(S,X-1))
  350 BB (S, X) = (DD (S, X) +CC (S, X) +BB (S, X-1) )/(BB (S, X) -CC (S, X) +AA (S, X-1))
C-2----APPLY RIGHT B.C.
       DO 370 S=1,#1
   370 TE(S,HX) = (DĎ(S,HX)+CC(S,HX)+BB(S,HX-1))/(BB(S,HX)-CC(S,HX)
1 + AA(S,HX-1))
       DO 400 J=2, 1 PHI
       X=#X-J+1
       DO 400 S=1,61
400 TE(S,X) =AA(S,X) *TE(S,X+1)+BB(S,X)
C-4-----APPLY LEFT B.C.
       IP(IHIH.GT.2) GO TO 65
       DO 410 S=1,8
       A(S, 1) = \lambda(S, 3)
       H(S,1) = H(S,3)
       ¥ (5,1) = -¥ (5,3)
       U (S.1) = U (S.3)
        W(S,1) = W(S,3)
        R(S,1) = R(S,3)
       TZ(S,1) = TE(5,3)
   410 TI(S,1) = TI(S,3)
        GO TO 65
    52 CONTINUE
 C----- CASE WITH EACH END THEATED INDIVIDUALLY.
 C----- APPLY LEFT B.C.
        IF (XHIN.EQ.1) GO TO 12
        DO 11 S=1,8
        AA(S,IHIN) = AA(S, XHIN)/BB(S, XHIN)
     11 BB(S,XHIN) = (DD(S,XHIN)-CC(S,XHIN)+TE2(S,XHIN))/BB(S,XHIN)
        GO TO 13
     12 DO 341 S=1,8
        AA(S,1) = AA(S,1)/BB(S,1)
   341 BB(S,1) = DD(S,1)/BB(S,1)
     13 DO 351 X=X8P1, MX81
        DO 351 S=1,8
   \begin{array}{l} AA(S,X) = AA(S,I) / (BB(S,X) - CC(S,I) + AA(S,I-1)) \\ 351 & BB(S,X) = (DD(S,X) + CC(S,X) + BB(S,I-1)) / (BB(S,I) - CC(S,X) + AA(S,I-1)) \end{array}
 C-2----APPLY RIGHT B.C.
        DO 371 S=1,81
    371 TE(S,NX) = (DD(S,NX)+CC(S,NX)+BB(S,NX-1))/(BB(S,NX)-CC(S,NX)
1 + AA(S,NX-1))
        DO 401 J=2, #PHX
        X = HX - J + 1
        DO 401 5=1,81
    401 TE(S,X) = 11 (S,X) +TE (S,X+1)+BB (S,X)
  C----CHANGE STEP
     65 \text{ STEP} = 3-\text{STEP}
        IF (STEP.EQ.2) GOTO 85
 C----- ADJUST TIMESTEP SIZE IF HECESSARY.
```

•

```
CALL DTINE (DT, MI, N, TE, TI, TE2, N, PHI, A, E, EPS1, MIN, BDN, DX)
     -----COPT NEW SOLUTION
  425 CONTINUE
      DU 450 I=1,81
      DO 450 5-1,8
      A_2(S,X) = \hat{A}(S,X)
H_2(S,X) = H(S,X)
      ¥2(5,1) = ¥(5,1)
      U2(S,X) = U(S,X)
      $2(S,X) = ¥(S,X)
  TI2(S,X) = TB(S,X)
450 T12(S,X) = TI(S,X)
      ----- OUTPUT BOUTINE
C-
      IF ( ( (NPRT* (NTINE/NPRT) ). EQ. NTINE) . AND. (NPRT. NR. 999) )
      ICALL RITE (MTINE, TINE, N, NI, A, TE, TI, N, PHI, V, P, THO, DT, O, R, EP, 1418
      2, 898)
      IF (((BTAPE+ (BTINE/STAPE)). EQ. STINE). AND. (STAPE. NE. 999))
      ICALL TPOUT (WTAPE, BTI ME, TIME, M, MX, LO, BO, BO, TO, DT, A
      2, TE, TI, N, PHI, V, P, NBOUND, XLAS, U, 4, XMIN, RDN, RM)
      IF ((NTIME.GE. WTOTAL) .OR. (TIME.GE.THAI)) STOP
C-----END OF OUTPUT
       GO TO 75
      BID
       SUBROUTINE INIT (DT, DI, DEL, PCOLL, EP S20, PHI, F, N, A, TE, TI
      1, DELE1, DELE2, DELI1, DELI2, ABO, CS, TINE, THO, XLAS, RD, U, WO, XHIN)
                              ************
C******************************
       SUBROUTINE INPUTS INITIAL CONDITIONS.
С
                                                                                 .
                 BICHARD D. HILROT
                                          77-04-23
С
IMPLICIT MEAL (A-M, 0-2)
       REAL PHI (31) , V (30,60) , H (30,60) , A (30,60) , TE (30,60) , TI (30,60)
      1, NO, LO, KB, HI, LHDA, LLHDA, B(31), PB (30, 60), HINT (31), RD (30, 60)
      2, U (30, 60), NO (30, 60), L1, L2, L3, L4, L5, WCRIT
       INTEGER S, X, ININ
       LOGICAL TRBC, RDB
       CONNUM LO, NO, NO, TO, PWR1, PWR2, PWR3, T1, T2, T3, T4, THAX, VLAS, TDP1, TDP2
      *, BH
1, TS, AVR, AVX, PP, NX, H, NTOTAL, NBOUND, WTINE, NPRT, NTAPE, NXHIN, RDN, TRBC
      2/TSS/A 11, A1, A2, A3, DT HAX/LASP/L3, BLO, BS, BCBIT, TL, DTL, B RHAX
      3/PRT/NXI, MXP, 81, 8P, HS
       NARGLIST /INPOT/ NTINE, NTOTAL, NPET, NTAPE, PR, PUR1, PNE2, PNE3
      1, T1, T2, T3, T4, THAX, NBOUND, NX, H, LO, RU, NO, TO, RU1, W, VLAS
      2, BINT, TINT, NINT, PP, AVR, A11, A1, A2, A3, L1, L2, L3, L4, L5, BL0, BS, OTHAX
      3, NXHIN, NXHIN1, BON, TOP1, TOP2, NXI, NXP, 81, 87, 85, AVX, BENAX
      4, TRAC, BIT, R2T, R1H, R2H, BH
       TINE=0.0D0
       ILAS=0.0
       READ (5, IEPOT)
       IF (NTIME. ME. 0) GO TO 2
       READ (5, INPUT)
        281 - 28
        READ (5, INPUT)
        DO 6 I=1,#
     6 SINT(I) = 1.0
        L1=L1/L0
        L2=L2/L0
        L3=L3/L0
        L4=L4/L0
        L5=L5/L0
        11T = 117/20
```

121 - 121/1 111 - 114/10 ETU = 828/90 HCHT = 9.969+18/80 HEIU = HXETU1 IP (INIH.LT. HOORAD) XATA MOORED #10-110/10 85-85+10/80 KB=1.6D-12 C=3.00+10 #I=1.678-24+1E P0-10-18-70 80-8087 (8.8+3/10159+P0) V0-8087 (88+70/88) TE0=L0/T0 10-3. 19159-840+2" 210-20+A0+70+1.02-07 مرد 2751 - (20/L0) CS-C/VO LRDA=1.3562+28+((EB+T0)++3/(3.14159+10))++0.5 LLEDA=ALOG (LEDA) BLE1 = (1.90+21/LLBDA) = (T0++2.5+LQ) / (H0+T0+B0++2) DELE2 = (2.530+25/LLBDA++2) + (T0++3+B0++2) / (H0++2) BELE1 = (7.50+19/LLBDA) + (T0++2.5+LO) / (H0+Y0+B0++2) + SQIT (BH) BELI2 = (7.80+22/LLADA++2) + (20++ 3+ 80++2) / (80++2) SIGHA=0.7013+T0++1.5/LLADA EPS20=C++2+L0/(6.28319+SIGHA+80++2+70) PCOLL=3.270-09+80+LLBDA/(20++1.5)+L0/(70+84) AB0=9.748-36+80+24L0/10+ 1.5 DI=1.0/(II-# 00880) PUR1 = PER1/PU0 P112 = P112/PP0 PURS + PURS/PUG T1 - T1/TH0 T2 = T2/THO T3 = T3/THO IN = IN/INO 15 = 14 + 3.0/CS THAT - THAT/THO VLAS=VLAS/TO 201=201/20 1-1/10 BIST-BIST/BO TINT-TINT/TO C----- SPECIFY BEELL RADII. 11-1.0-801 AA= (#2/4) ++ (1.0/#-2)} DO 99 I=1,100 99 AA= ( ( ( BB+ ( AA-1.0 ) ) + W ) / W ) + 4 ( 1.0/ ( H- ]) ) R(1)=0.0 E (2) =E01 2(3)=201+¥ ## 1=#+1 BO 100 I=4,#P1 100 B (I) -B (I-1) + A&\* (B (I-1)-B (I-2)) --- SET INITIAL VALUE FOR DT.  ${\cal D}$ C---DINAL - DINAL/TRO C----- SET INITIAL VALUE OF DE AT A11 . VALUE ALLONED BY STABILIEY. c = (2.0+1.6667+TIST) + (2.0+BIST++2/BIST(1))

```
C - SQ42 (C)
        11 = 11
        IF (BDI) 95-201+8751
        37 - A11+ (85/C)
        LF (DT.GT.DTHAI) DT-DTHAI
        30 1 I=1,34
         30 1 5-1,8
         t8 (s , 1) = ži st
        TI(S.I) =TIST
         3 (S, X) -BINT (S)
         A (S, I) = (# (8+1) ++2-2 (S) ++2)
      1 T (S, I)=0.0
         DO 4 1-1,81
         20 4 S=1,#
         20(S,I) - 2(S+1)
         U (S, X) -0.0
      4 ¥0($,X)=8($+1)++2
         -- MAKE RADIAL PROFILE FOR TEMPERATURE AND DENSITY.
C-
         DO 7 5=1,8
      7 1 (5) = 0.5* (1(5)+1(5+1))
        AT1 = 1./(TI T-TO)
         AT2 = 1.0-AT1
         A#1 = 0.001
         A#2 = 1.0-A#1
         DO 8 3=1,8
         227 = 2(9) - 217
         228 - R(S) - R18
         INT = AT2/(1.+EIP(251/821)) + AT1
         X## = A#2/(1.+#IP(EIH/#2H)) + A#1
         DO 8 X=1,NX
         TE(S,I) = IST*TE(S,I)
TI(S,I) = IST*TI(S,I)
TI(S,I) = IST*TI(S,I)
      8 # (S, X) = X##+# (S,X)
         K=11/2
         CONST-NINT (8) * (TE (N, K) +TI (8, K) ) + BI #T**2
         po 3 S=1,8
      3 PHI (S) = SQRT (CONST-H (S,K) + (TE (S,K) +TI (S,K))) + A (S,K)
C----- POT IN INITIAL DENSITY PROFILE FOR GAS TARGET.
      / 8291 = 8241
         DO 5 J=1,#X
         X = SIP1-J
         \mathbf{X} = \mathbf{D}\mathbf{X}^{*}(\mathbf{X}^{-1})
         335=L1-S
         \mathbf{Y} = \mathbf{D}\mathbf{X} + (\mathbf{J} - \mathbf{1})
         TTT - (L4-T)
         X = (.999/(1.+EXP (235/L2))+.001)+(.999/(1.+EXP (YYY/L5))+.001)
         DO 5 S=1,8
      5 B45,J) - IN+B (S,J)
INITIALISE LOBAT.
С
         T1-0.0
          221-02/03
    SAITE (6,601)
601 PERMAT ('1', ' INITIAL CONDITIONS')
   WWWAT[''', ' LELTIAL COMDITIONS']
WRITE(6,602) L0,80,80,70,80,90,70
602 FORMAT('0L0+',175.2,T17,'80-',89.2,T32,'80-',29.2,T47,'T0-',
189.2,T62,'80-',89.2,T77,'90-',89.2,T92,'70-',89.2)
WRITE(6,603) WX,8,M0,D1,D7,980,T80
603 FORMAT(' HZ=',I3,T17,'8-',I3,T32,*A0-',1989.2
1,T47,'DI-',89.2,T62,'BT=',89.2,T77,'PN0-',89.2,T92,'T80-',89.2)
WRITE(6,604) DEL,SIGNA, EPS20, FC001, AB0
                      đ.
```

•}

1

A.

229

-

С

.

```
604 FORMAT(' DEL=', 1PE9. 2, T19, 'SIGMA-', E9.2, 236, 'HPS20-', E9.2
1, T53, 'PCOLL-', E9.2, T70, 'ABO-', E9.2)
WRITE(6,605) DELE1, DELE2, DELI1, DELI2, LLMBA
 605 FORMAT (' DELE1=', 1PE9. 2, 19, 'DELE2=', 19.2, 136, 'DELE1=', 19.2
1, 153, 'DELE2=', 29.2, 170, 'ELEDA=', 10.2)
       IP (RDS) BRITE (6, 606)
  606 POREAT ('O FULL RIDIAL DYBANICS ARE CALCULATED. ')
       IF (. NOT. RDN) WRITE (6,607)
  607 PUREAT ('O RADIAL PRESSURE BALANCE IS ASSURED.")
    2 BETURN
       END
       SUBROUTINE TPOUT (MTAPE, MTIME, TIME, N, MX, LO, NO, NO, DT, A
      1, TE, TI, N, PHI, V, P, SHOSHD, XLAS, U, N, XHIN, RDN, RA)
                                                   OUTPUT DATA FROM SHELL OFTO DISK OF TAPE IN UNFORMATTED FORM.
                                                                                    ۲
                                                                                     ٠
                                     77-04-25
            RICHARD D. MILBOY
С
                                          ------
REAL TIME, LO, RO, NO, TO, DT, A (30, 60), TE (30, 60), TI (30, 60), H (30, 60)
      1, PHI (31), V (30,60), P (120), XLAS, U (30,60), U (30,60)
       INTEGER XNIN
       LOGICAL RDS
       WRITE(3) WINPE, WINE, TIME, M., WI, LO, BO, NO, TO, DT, MBOUND, ILAS
             , IHIN, RDN, RM
      1
       WRITE(3) A
WRITE(3) TE
       WRITE(3) TI
       HRITE(3) H
       WRITE(3) PHI
       WRITE(3) V
       WRITE(3) P
       HRITE(3) U
        WRITE(3) W
       RETURN
       END
       SUBROUTINE RITE (WITHE, TIME, M. M. A. TE, TI, N. MI, V. P. THO, DT, U. H. BP
             , XNIN, RPB)
      1
       REAL PHI (31) , A (30, 60) , TE (30, 60) , TI (30, 60) , N (30, 60) , V (30, 60)
      1, B (11) , TINE, PRESS (31) , P (120) , THO , DT, U (30, 60) , EP (30, 60)
      2, # (30, 60) , VP (31)
        INTEGER S.X. ININ
        LOGICAL RPB
        COMMON /PRT/ NIL, NIF, H1, MF, H5
        XXXI = XXX
        BBZ = BXP
        IF (BII. EQ. 999) NHII = ININ
        IF (HIP. 1Q. 999) HHX = HX
        HHP=HP
        IF (NF. 20.999) 887=8
        TH=TIRE+THO
        DTS = DT+THO
        WRITE(6,601) WTINE, TIME, TH, DT, DTS
   601 FORBAT ('-WTI HE=', I4, ' TIHE=', 1PD 9.2, '
1, ' DT=', D9.2, ' OR', D9.2, ' SECO MDS')
                                                          OR', D9.2.' SECONDS'
        HRITE(6,602)
   602 FORMAT ('0
                      A*)
        DO 1 X-BEXI, BEX
      1 HRITE(6,603) (A(S,X),S=H1,HHF,HS)
   603 FORMAT (* *, 1 P200 10.2)
        WRITE (6,604)
   604 FORBAT ('O
                      18")
                                                          , *
                                                                       ٦
```

♠...

c

```
DO 2 I-BBLI, NHX
   2 WRITE(6,603) (TE(5,1),9-81,887,85)
    WRITE(6,605)
WRITE(6,605)
 605 PORMAT ('0
     DO 3 X-BBXI, BBX
   3 WHITE(6,603) (TI (8, 1), 5-41,447,45)
    WHITE (+, 606)
BRITE (6, 606)
 606 FORMAT (* 9
     DO 4 X-BBXI, BBX
   4 URITE(6,603) (B(S,X),S=81,887,85)
     VEITE(6,607)
                   ......
 607 PORMAT (10
     DO 5 X=NHXI, NHX
   5 HEITE(6,603) (V(S,X),S=E1,HEP,HS)
     17(RPB) 60 TO 9
 URITE(6,612)
612 PORMAT('0 VP')
     DO 13 X-BUXI, BUX
      DO 14 5-1,8
  14 VP(S) = U(S, I) / E(S, I)
  13 HEITE(6,603) (VP(S), S-E1, MEF, HS)
   3 HRITE(6,608)
9 HRITE(6,608)
 608 PORAAT (*0
     DO 6 1-8811,881
DO 7 5-1,8
   7 B (S) = PHI (S) / A (S, X)
    6 HRITE(6,603) (B(S),S=H1,HEF,HS)
      #HITE(6,609)
                   R BOUNDARIES")
 609 PORMAT('O
      DO & X-NNXI, NNX
    8 HRITE(6,603) (R(S,X),S=H1,HHP,H5)
 NETE(6,610)
610 PORMAT(*0
                   PRESSURE')
      DO 10 I-BHIL, NHY
      DO 11 5=1,8
   11 PRESS (S) = H (S, X) + (TE (S, X) +TI (S, X) ) + (PHI (S) /A (S, X)
   10 HEITE(6,803) (PEESS(S),S=#1,##7,#8)
  BRITE (6,614)
614 PORMAT (*0
                  - IP+)
      DO 15 X-MHXI,MHX
   15 HEITE(6,603) (EP(S,X), S-41, HEP, HS)
  HALTE (6,611)
611 PORMAT (*0
                   LASER POWER')
      BINP1=BI-BBII+1
      DO 12 1-1, #18P1
   12 HRITE(6,603) P(I),P(I+HX)
      RELATIN
      RND
      SUBROWTINE REST (DT, DI, DEL, PCOLL, EP S20, PHI, V, H, A, TE, TI
      1, DELE1, DELE2, DELI1, DELI2, ABO, CS, P, P2, TIME, THO, XLAS, R, U, WO
     2,1818)
                   C***********
     . READS OFF FORTERS WIT $2 TO RESTART PROGRAM AT RECORD BT.
С
                                          JONE 1977.
            RICHARD D. HILBOY
С
IMPLICIT BEAL (A-E,0-3)
      BEAL A(30,60), TE(30,60), TI(30,60), B(30,60), PHI(31), V(30,60)
+, P(120), LHDA, LLEDA, LO, NO, HI, KB, P2(31), PR(30,60), R(30,60), B(30,60)
      +, 40 (30, 60) , L1, L2, L3, BC BIT
       INTEGER ININ
```

:
```
LOGICAL TROC, ROS
  CORRON LO, BO, NO, TO, PVB1, PVB2, PVB3, T1, T2, T3, T4, THAE, VLAS, TOP1, TOP2
 •,11
 1, 15, AVR, AVI, PP, HI, H, BTOTAL, HBOUND, HEIHE, HPRT, HTAPE, HINLH, RDN, TROC
 2/TSS/A11,A1,A2,A3, DTHAX/LASP/L3, ELO, BS, MORIT, TL, DEL, HAHAX
 3/PRT/ WAI, WIF, M1, MF, MS
  NAMBLIST /IN PUT/ NTINZ, NTOTAL, NPBT, NTAPS, P.R. PUR1, PUR2, PUR3
 1, T1, T2, T3, T4, THAX, MOUND, NX, N, L0, 20, 80, 20, 80 1, N, VLAS, 28
2, BINT, TINT, PP, AVR, A11, A1, A2, A3, L1, L2, L3, RLO, BS, DTHAX
3, MININ, MININ1, RDN, TDP1, TDP2, MXI, MX P, H1, AP, MS, AVX, MRM AX, TR BC
1 READ (2) STAPE, MT, TINE, E, ST, LO, RO, BO, TU, DT, BOUND, ILAS
 1
        ,XHIN,RDN,RM
  READ(2) A
  2 EAD (2) TE
  READ(2) TI
  R JAD (2) I
  READ (2) PHI
  R XAD (2)
           ¥
  R EAD (2)
          2
  R EAD (2)
           ٠
  R RAD (2) 10
  IT (NT. ME. MTIME) GO TO 1
  E EAD (5, INPUT)
  DO 2 J=1,8X
  DO 2 I=1,8
  R(I,J) = SQRT(WO(I,J))
2 CONTINUE
  #X2=2+#X
                                                       ø
  DO 4 I=1,#X2
4 P2(I)=P(I)
  L3=L3/L0
  SCRIT = 9.960+18/80
  ELO-ELO/EO
  IS=IS+LO/RO
  DI=1.0/(#X-#BOUND)
  KB=1.60-12
  C=3.00+10
  HI#1.678-24# BH
  PU=#.0+ KB+T0
  B0=SQRT (8.0+3.1415+P0)
  VO=SQRT (KB+TO/RI)
 TEO=LO/VO
  10=3.14159+R0++2
 PE0=20+A0+T0+1.0E-07
 CS=C/V0
 LNDA =1.3562+28+((KB+T0)++3/(3.14159+80))++0.5
  LLHDA=ALOG (LHDA)
  DEL= 1. 9021+T0++2.5/(LLHDA+H0+V0+L0)
  DELE1 = (1.90+21/LLHDA) + (T0++2.5+LO) / (H0+Y0+H0++2)
  D&LE2 = (2.450+25/LLHDA++2) + (T0++3+80++2)/(H0++2)
 DELI1 = (7.80+19/LLHDA) + (10++2.5+L0) / (80+Y0+R0++2) +SQRT (RH)
 DELI2 = (7.60+22/LLHDA++2) + (T0++3+80++2) / (N0++2)
 SIGHA=8.7013+T0++1.5/LLHDA
 EP520=C++2+L0/ (6.283 19+5IGRA+80++2+T0)
  PCOLL=3. 250-09+80+LLEDA/ (20++1. 5)+L0/ (Y0+88)
 AB0=9.742-36+80++2+L0/10++1.5
                                        .
  PHE1 - PHE1/980
 PHR2 - PHR2/PHO
 PHE3 - PHE3/PHO
```

U

20

T1 = T1/TEO

232

74

•

V

```
12 - 12/200
       13 - 13/180
       14 - 14/100
       15 = 14+3.0/CS
       TEAT - THAT/THE
       VLAS=VLAB/70
       DTHAX - DTHAX/THO
      --- INITIALILE LUEAT.
C----
       TL - TISE
       DTL = DI/CS
  5 WRITE (6, 601)
601 PORMAT (*1', ' INITIAL CONDITIONS')
       HRITE(6,602) LO, 20, 20, TO, 30, PO, TO
  602 FORMAT ('ULO-', 1929.2, 117, 'RO-', 89. 2, 132, 'NO-', 89.2, 147, 'TO-',
      129.2,162,'B0=',89.2,177,'P0=',89.2,192,'Y0=',89.2)
  VRITE(6,603) UI, H, A0, DI, DT, PE0, TH0
603 PORMAT(' UI=',I3,T17,'B=',I3,T32,'A0=',1PE9.2
1,T47,'BI=',E9.2,T62,'DT=',E9.2,T77,'PU0=',E9.2,T92,'IH0=',E9.2)
       HRITE (6, 604) DEL, SIGEA, EPS20, PCOLL, ABO
  604 POBLAT (' DEL=', 1PED. 8, T19, 'SIGRAM', 89.2, T36, 'EPS20-', 89.2
1, T53, 'PCOLL=', 89.2, T70, 'AB0-', 89.2)
       VEITE(4, 605) DELE1, DELE2, DELI1, DELI2, LLEDA
   605 FORMAT (* BELE1=*, 1989.2, T19, 'DELE2=', E9.2, T36, 'DELE1=', E9.2
1, T53, 'DELE2=', E9.2, T70, 'LLHDA=', E9.2)
       IF (RD#)- URITE (6,606)
   605 FORMAT ('O POLL RADIAL DYNAMICS ARE CALCOLATED. ')
       IT (. 101. 201) URITE (6,607)
   607 PORMAT ("O RADIAL PRESSURE BALLUCE IS ASSUMED.")
        A RTURN
        ZED
       SQUAQUTINE LARAT (M.#X,STEP,TIME,T1,T2,T3,T4,MBOUND,TE15,XAIM
       1, PWR1, PWR2, PWR3, XLAS, YLAS, R0, AB0, P, P2, P2, R, H, A, T2, KP, DX, HXP1, T0)
٠
                            FIND LASER POWER.
C
      ۰
                    P IS POWER IN UNITS OF (NO+KB+TO+A0+YO)
                                                                                       •
С
      .
                                                                                       ٠
С
            E= (UNITS OF POWER) / (UNIT OF LENGTH) DEPOSITED IN PLASHA.
                     RICHARD D. HILROY HAY 15, 1977.
C
       ٠
             C *******
       INPLICIT REAL (A-H, 0-2)

EEAL P(120), PE(30,60), E(30,60), E(30,60), A(30,60), TE(30,60)

1, EP(30,60), PSI(120), P2(120), E(120), KA(120), LBDA, LLBDA(30,60), L3
       2, BCHIT, BCOB, TE15 (30, 60), KA1 (120)
        BEAL TOIP
        INTZGER S, X, STEP, XHIN
        COBBOB/LASP/L3, 2L0, BS, MCBIT, TL, DTL, BBMAI
        TU15 = T0+SQRT (T0)
C----- CALCULATE HR = NO. OF LASER TIMESTEPS TO TAKE.
        \mathbf{x}\mathbf{x}\mathbf{p}\mathbf{2} = \mathbf{x}\mathbf{x} + \mathbf{2}
        #X2 = 2*#X
        #X2#2 = #X2 - 2
        TDIF - TINE - TL
        HE = (TDIF/DTL + 0.5)
        IF (MA.LT.O) STOP
        # 2 - # I NO (#2, H MAX)
        IF (STEP. BQ. 1) #8-#8/2
        17 (SR. EQ. 0) RETURN
        DT = 0.000
        IF (STEP.EQ.2) DT=HE+DTL
        9L = 1L+31
 C------ LASER PROP. DIST. LIMITED BY BLEACHING NAVE VELOCITY - VLAS.
```

1

, 233

```
XLAS-XLAS+VLAS+OT
       #LAS= (XLAS/DI) + 2
       HIP2HI - HI+2-IMIN
       HLAS - HINO (HIP1, HLAS, HIP2HX)
C----- DEFINE PR (S, X). RELATIVE BADIAL INTENSITY.
C----- ASSUME GAUSSIAN PROFILE.
       DO 6 I-ININ, NI
       J = #XP1 - X
       \mathbf{Z} = \mathbf{D}\mathbf{X}^{\mathbf{+}} \left( \mathbf{J}^{-1} \right)
       SIGNA = RLO + BS+ABS (L3-2)
       SIG2 = 2.0+SIGHA++2
       RS = 0.0
       DO 6 5=1,8
       RS = Q.5+ (RS+R (S,X))
       EI11 = 25++2/SIG2
       IF (EXEL. GT. 100.0) EXEL = 100.000
       PR(S,X) = EXP(-EXZZ)
       RS = R(S,X)
     6 CONTINUE
C----- DEFINE LASER POWER AT FIRST X - POINT.
       IF (TINE.GT. T1) GO TO 15
       P1 = PUR1+TIME/T1
       GO TO 85
    15 CONTINUE
       IF (TINE.GT. T2) GO TO 16
        P1 = PHR1+ (PHR2-PHR1) + (TIME-T1) / (T2-T1)
       GO TO 85
    16 CONTINUE
       IF (TIBB.GT.T3) GO TO 17
        P1 = PWR2+ (PWR3-PWR2) + (TIHE-T2) / (T3-T2)
        GO TO 85
    17 CONTINUE
        IF (TIME.GT. T4) GO TO 18
        P1 = PHR3-PHR3+ (TIHE-T3) / (T4-T3)
        GO TO 85
    18 P1 = 0.0000
    85 CONTINUE
 C----- DEFINE PSI(X)
        DO 13 X-XMIN,NX
        PSI(X) = 0.0
        DO 13 5=1,8
        LNDA=18.7+18(S,1)+TO
        IF (TE (S, I) +TO. LT. 27.0) LHDA= 2.3+ (TO15+TE15 (S, I))
        IF (# (S, I).GT. (0.9999 *#CRIT)) GO TO 20
        HCOR = 1.0/SQRT(1.0-H(S,X)/HCRIT)
        GO TO 21
     20 BCOR=100.0
     21 CONTINUE
        LLNDA (S, I) = ALOG (LNDA) * HCOR
     13 PSI(X) = PR(S, X) + A (S, X) + PSI(X)
 C----- DEFINE KA (I) - LASER ABSORPTION RATZ AT I-POINTS.
        DO 1 X=XEIN, NX
        KA (I) =0.0
        DO 14 5=1,8
     14 KA (X) = KA (X) + H (S, X) + 2 PR (S, X) + A (S, X) + LLHDA (S, X) / TR 15 (S, X)
         KA (X) = KA (X) + ABO/PSI (X)
         KA1(X) = KA(X)
         IF (KA (X) . GT. (2.0/DX) ) KA (X) = 2.0/DX
       1 CONTINUE
  C----- INITIALIEE E(I) TO 0.0.
```

•

ļ

234

```
DO 7 X=1,942
     7 B(1) =0.0
---- FIND LASER POISE.
C---
       DO 2 IT=1,88
       23-21
       DO 3 X=2, MLAS
       PO-P (X-1) + (1.0-0.5+KA (#X+2-X) +DX) / (1.0+0.5+KA (#X+2-X) + DX)
       E(X-1) = E(X-1) + (P(X-1)-PO)/DX
       P (I-1)=P3
     3 23-20
        P (NLAS) = P3
        IF (BBOUND. EQ. 1) GO TO 2
        \begin{array}{c} 1 & F(BLAS, LT, BX) & GO & TO & 2 \\ DO & S & X = B(P1, BX2R2) \\ PO = P(X-1) & (1.0-0.5+K)(X-BX+2) + DX) / (1.0+0.5+KA(X-BX+2) + DX) \\ B(X-1) = B(X-1) + (P(X-1)-P0) / DX \end{array} 
        P (X-1)=P3
      4 23-20
        P (#X242) = #3
      2 CONTINUE
        I=HLAS
        IF (HBOUND. EQ. 2) I-MX282
        IF(STEP.EQ.1) GO TO 10
        DO 23 X=1,I
     23 P2 (X) = P (X)
        GO TO 26
     10 CONTINUE
        DO 22 X=1,I
     22 P(I)=P2(I)
     26 CONTINUE
         DO 5 I-ININ, NI
         DO 5 5=1,8
         EP1=AB0/(KA1(X)+PSI(X))+H(S,X)+PR(S,X)+LLHDA(S,X)/TE15(S,X)
EP(S,X) = EP1+E(HX+1-X)/HR
         IF (NBOUND. BQ. 2) EP (S, X) = EP (S, X) + EP 1+ E (HX-3+X) /HE
       5 CONTINUE
         RETURN
         TID
         SUBROUTINE DTIME (DT, NX, N, TE, TI, TE2, N, PHI, A, R, EPS1, MAIN, RDN, DX)
                                                                 ****************
  C***********************
               CALCULATE BAXIBUR ALLOWABLE TIRESTEP 2-D ROUTIBE CAN BAXE.
                                                                                         ٠
        ۰
  С
                                                                                         .
               STABILITY CONDITION IS BASED ON RADIAL DYNAMICS.
  С
        ٠
                                                    DECEABER 15, 1977.
                                                                                         .
                      EICHARD D. HILBOY
  С
        ٠
     C *
        INPLICIT BEAL (A-H,0-Z)

REAL TE(30,60),TI(30,60),TE2(30,60),B(30,60),E(30,60)

1,FINT(31),PHI(31),A(30,60)
 •
         INTEGER ININ,S,I
         LOGICAL RDE
         CORBON/TSS/A 11, A1, A2, A3, DTHAX
         DTO = DT
         HH1 = 2
          882 = 1
          IXS = 1
          DTST = 1.00+50
          DTTSH = 1.0D+50
          10 = 10
  C----- BAKE DT SATISFY RADIAL STABILITY CONDITION.
          DO 1 X=XHIN, NX, NXS
          5=1
```

٠

、 ·

BS = PNI(S)/A(S,X) C = (1.6667\*(TE(S,X)+TI(S,X))) + (2.0+BS++2/E(S,X)) C = SQRT(C) 17 (RDS) D1=R (5,1)+EP51 DTH = A1+DZ/C IF (DTH.LT.DTST) DTST = DTH IF (AN1.LT.2) GO TO 1 DO 2 5=2,881 BS = PHI(S)/A(S, X)C = (1.6667\*(TE(S,X)+TI(S,X))) + (2.0\*BS\*\*2/H(S,X)) C = SQET (C) IP (RDN) DZ= (R(S,X)-R(S-1,X)) + EPS 1 DIN = A1+DZ/C IF (DTR.LT.DTST) DTST = DTM 2 CONTINUE 1 CONTINUE , C----- LINIT DT SO ELECTRON TEMP. DOES NOT CHANGE TOO PAST. DO 3 X=XHIN, NX, NXS DO 3 5= 1 ,MM2 DTE = ABS(TE(S,X) - TE2(S,X))IF(DTE.LT. 1.00-20) DTE = 1.00-20  $DTE = \lambda 2 + TE2 (S, I) + DTO/DTE$ IF (DTH.LT.DTTEN) DTTEN - DTH 3 CONTINUE C----- PREVENT DT FROM INCREASING BY HORE THAN A FACTOR OF A3. DTB = A3+DTO C----- SET DT TO HOST RESTRICTIVE CASE. DT = AHIH1 (DTST, DTTEH, DTW) IF(DT.GT.DTHAI) DT=DTHAI RETURN END SUBROUTINE NDPTS (ININ, P, NI, NBOUND, NININ) ٠ CALCULATE ININ. С PREVENTS PROGRAM FROM DOING CALCULATIONS IN SECTIONS С ٠ WHERE THERE IS NO PLASHA BOTION. С ۰ . TO REDUCE CPU COSTS. С ٠ . 78-01-06 С . RICHARD D. HILROY. REAL P (120) INTEGER XHIN NIP1 = NX+2-NBOUND-NININ NIMININ-E+WINX-IN = IE K=#1-1 DO 1 J=#1,#XP1 IF(P(J).LT.1.0D-10) GO TO 2 1 K=J  $2 \parallel \mathbf{X} \mathbf{Z} = \parallel \mathbf{X} - \mathbf{X} + \mathbf{Z}$ HI2 = HI2 - HIBIHIF (XHIN.GT.NX2) XHIN-NX2 RETORN END

.

**4**}

```
.......
          TEST FOR CONSERVATION OF ENERGY IN 2-D E.H.D. ROUTINE.
                                                                         ۲
     .
С
                                                                         .
          MARCH, 27, 1977.
С
          BODIFIED APBIL, 30, 1977.
HODIFIED AGAIN (TO INCLUDE CRECK ON CONSERVATION OF MASS)
                                                                         ۲
С
     .
                                                                         .
С
          ON JANUARY, 18, 1978.
С
     ٠
           WRITTEN AND HODIFIED BY BICHARD J. HILBOY!
С
     ٠
                                                          .................
REAL & (30,60), TE (30,60), TI (30,60), B (30,60), PHI (31)
     1, Y (30, 60), LO, NO, LL NDA, P (120), N (30, 60), VP (30, 60)
      LOGICAL RDN
      COMMON A, TE, TI, N, PHI, V, LO, NO 1 VO, T, NS, NZ, DZ, DZL, GE11, GG, P, N, VP
      COBBON/CONST/ NBOUND, NBE, NBL, N1, N2, N3, N4, N5, N6, N7, N8, EPS1
       ---- READ I. WILL CALCULATE EMERGY OF EVERY I'TH RECORD.
C----
  WRITE(6,601)
601 PORHAT ('-', 'ENTER I - I3 PORHAT. - AND TP'/' ', ' I
                                                                   T7')
      READ (5, 501) II, CP
  501 PORMAT (13,810.3)
      GAM=5.0/3.0
      GH1I=1.0/(GAM-1.0)
      GG=GAN+GH1I
C-----CALCULATE DEL POR THERNAL CONDUCTIVITY OUT END.
      READ(3) HTAPE, HTIBE, T, HS, HZ, LO, RO, NO, TO, DT, NHOUND, XLAS
     1, ININ, RDN, RH
      THO = 1.02E-06+LO/SQRT (TO) +SQRT (EM)
      #8L=2
      # B #= 2
      IF (BBOUND. EQ.2) BBL= 1
       ¥1=¥BL+1
      #2=#BL+2
      ¥3=#Z-XBR-1
       14=12-182
       #5=#BR+2
       16=12-18L
      #7=2+#2-3-#8#
                                Ŧ
       #8=#4-#1
       BACKSPACE 3
       EPS1 = B0/L0
       DU1 = 0.0
       DG1 = 0.0
       LLHDA=ALOG(1.356E+28*((1.6E-12*T0) **3/(3.14159*#0))**0.5)
       Y0=SQRT(1.6E-12+T0/(1.67E-24+BA))
       DEL= 1. 9221+T0++2.5/(LLHDA+H0+V0+L0)
       IT=1
      IP(IX. NE. 1) IT=2
C-----CALCULATE BNERGY BTC. IN 1'ST RECORD.
       CALL RD (DU, DG, 64)
       f 2=T
       CALL ENG (U1,G1)
       DUT-DU
       DGT=DG
       TIBE = T+THO
       WRITE(6,602) TIME,U1,G1
   602 PORSAT( ', 'AT ', 1PE 10.3, ' SEC. U= ', E10.3, ' HASS = ', E10.3)
       GU TO(3,2),IT
     2 CONTINUE
 C-----FIND DU AT INTERNEDIATE TIMESTEPS.
       IH1=IX-1
       DO 1 I=1,IM1
```

```
CALL RD (DU, DG, 54)
      T 1=T
      DILTT=11-T2
      12-11
      DU1=DU1+0.5+DELTT* (DUT+DU)
      DG1=DG1+0.5+DELTT+ (DGT+DG)
      DUT=DU
    1 DGT=DG
    3 CONTINUE
    ------FIND ENERGY ETC. IN I'TH RECORD.
C---
      CALL RD (DU, DG, 54)
      T1=T
      DELTT=T1-T2
      12-11
      DU1=DU1+0.5+DELTT* (DUT+DU)
      DG1=DG1+0.5+DELTT+ (DGT+DG)
      DUT=DU
      DGT=DG
      CALL ENG (U2,G2)
      DIFF=02-01
      DIFG=G2-G1
      FRACT= (DIFF+DU1) +2.0/(01+02)
      PRACG= (DIPG+DG1) +2.0/(G1+G2)
      WRITE(6,603) DIFF, DU1, FRACT
  603 PORMAT (* *, *DIPP= *, 1PE10.3,*
                                         ENERGY LOST OUT END = ', E10.3,
          FRACTIONAL ERROR=', E10.3)
     11
  WRITE(6,604) DIPG, DG1, PRACG
604 PORMAT(' ', 'DIPP= ', 1PE10.3, '
                                         HASS LOST OUT END = ", E10.3,
     1. PRACTIONAL EBBOR=", E10.3)
      TIME - T+T80
      WRITE(6,602) TINE,02,G2
      01=02
      G1=G2
      DU1=0.0
      DG1=0.0
      IF(T.GT.TF) IT=3
      GO TO(3,2,4),IT
    4 STOP
       END
      SUBROUTINE RD(DU,DG,*)
REAL A (30,60), TE (30,60), TI (30,60), H (30,60), PHI (31)
      1, V (30, 60), LO, NO, P (120), N (30, 60), VP (30, 60)
      COMMON A, TE, TI, N, PHI, V, LO, NO, VO, T, NS, NJ, DZ, DZL, GE1I, GG, P, V, VP
       COABON/CONST/ NBOUND, NBR, NBL, N1, N2, N3, N4, N5, N6, N7, N8, EPS1
       READ (3, END=1) WTAPE, HTINE, T, NS, NZ, LO, RO, NU, TO, DT, MBOUND
       DZ=1.0/(NZ-HBOUHD)
       READ(3) A
       READ(3) TE
READ(3) TI
       READ(3) H
       READ(3) PHI
       ERAD(3) V
       READ (3) P
       EEAD(3) YP
       # #AD (3) W
       DO 7 J=1,#2
       DO 7 1=1,85
     7 V_{R}(I,J) = VP(I,J)/SQRT(H(I,J))
       DO 5 J=1,#Z
```

```
VPS1 = 0.5 + VP(1, J)
      DO 6 1=2,35
       VPS2 = 0.5 + (VP(I-1,J) + VP(I,J))
       VP(I-1,J) = VPS1
    6 VPS1 = VPS2
    5 V2 (85, J) = VPS1 -
      DU-0.0
       DO 2 I=1,#5
       DU=DU+ (A (I, NA) + (GG+H (I, H4) + (TE (I, H4) +TI (I, H4)) +0.5+H (I, H4) +
     1 (V (I, #4) ++2+ (BPS 1+VP (I, #4) ) ++2) ) +V (I, #4) +DT)
      2+DEL+A (I, #4) +TE (I, #4) ++2.5+0.5+ (TE (I, #4-1) -TE (I, #4+1) ) + (DT/DE)
     2 CONTINUE
       IF (BBOUND. BQ. 2) GO TO 4
       DO 3 I=1,#5
      DU=DU- (A (I, M 1) + (GG+M (I, M 1) + (TE (I, M 1) + TI (I, M 1)) + 0.5*M (I, M 1) + 1 (V (I, M 1) * 2+ (EPS 1*VP (I, M 1)) * 2) MW (I, M 1) * DT)
      2-DEL+A (I, #1) +TE (I, #1) ++2.5+0.5+ (TE (I, #1-1) -TE (I, #1+1)) + (DT/DZ)
    3 CONTINUE
C----- ADD LASER ENERGY INPUT.
    4 CONTINUE
       IF (NBOUND. KQ. 1) DU=DU-0.5* (P (NBR+1)+P(NBR+2)-P(N6)-P(N6+1))*DT
       IP (NBOUND. EQ. 2) DU=DU-0.5* (P (NBR+1) + P(NBR+2) - P (N7) - P (N7+1)) + DT
C----- CALCULATE HASS LOST FROM SOLENOID ENDS.
       DG = 0.0
       DO 8 I=1,85
     8 DG = DG + A(I, N4) + V(I, N4) + H(I, N4) + DT
       IF (MBOUND.EQ.2) GO TO 9
       DO 10 I=1,#5
    10 DG = DG - A(I,#1)+V(I,#1)+H(I,#1)+DT
     9 CONTINUE
       DU=DU/DT
       DG=DG/DT
       RETURN
  1 WRITE(6,601)
601 FORMAT (' ','ENDFILE ON UNIT #3')
       RETURN 1
       E ND
       SUBROUTINE ENG(U,G)
C-----CALCULATE TOTAL ENERGY & HASS IN SOLENOID.
       REAL A (30,60), TE (30,60), TI (30,60), H (30,60), PHI (31)
      1, V (30, 60), L0, N0, P (120), W (30, 60), VP (30, 60)
       REAL+8 UD, DBLR, GD
       COMMON A, TE, TI, N, PHI, Y, LO, KO, VO, T, NS, NZ, DZ, DEL, GN1I, GG, P, W, VP
       COMMON/CONST/ NBOUND, NBR, NBL, N1, N2, N3, N4, N5, N6, N7, N8, KPS1
       UD=0.0D0
       DO 1 J=#2,#3
       DO 1 1=1,85
        B=PEI(I)/A(I,J)
       U=A (I, J) + (GH1I+H (I, J) + (TE(I, J) +TI(I, J) ) +B++2+0.5+H (I, J)
      1* (V (I, J) **2+ (EPS1*VP (I, J)) **2))
        UD=UD+DBLE(U)
     TCONTINUE
        DO 2 J=#1,#4,#8
        DO 2 I=1, #5
        B=PHI(I)/A(I,J)
         U=A (I, J) + (GH1I+H(I, J) + (TE(I, J) + TI(I, J)) + B++2+0.5+H(I, J) 
       1+ (Y (I, J) ++2+ (EPS1+YP (I, J)) ++2))
        UD=00+0.500+DBLE(0)
     2 CONTINUE
C----- ADD ENERGY OF LASER BEAM.
```

```
٩.
         PERG-0.000
D0 3 J=85,86
     3 PENG-PENG+P(J)

IF (NOGHD.EQ.1) GO 20 5

DO 4 J=N2,87

4 PENG=PENG+P(J)

C CC-100,000
      5 CS=3.00+10/VO
         PING=PING/CS
         UD=UD+PENG
C---- CALCULATE FINAL VALUE OF ENERGY.
         U-SHGL (UD)
         U=U+D1
C----- CALCULATE TOTAL HASS IN SOLENOID.
         GD = 0.0
DO 6 J=02,03
DO 6 I=1,05
         G = \lambda (I, J) \bullet H (I, J)
      6 GD = GD+DBLE (G)
         DO 7 J=#1,#4,#8
DO 7 I=1,#5
      G = A(I, J) + H(I, J)
7 GD = GD+0.5+DBLE(G)
          G= SHGL (GD)
          G = G + D \bot
          RETURN
          END
```

٠

.

.

.

•

.

1. . .

.

240

`

•

É.

```
DRAH 3-D PLOTS FOR 2-D MAD ROUTINE.
     ۰
С
         BICHARD D. HILROT
                             1977
     .
С
REAL A (30,60), TB (30,60), TI (30,60), B (30,60), B (30,60)
     1, T (30, 60) , E (30, 60) , PHI (30) , E (62) , EX (62)
     2, D (100, 100) , ET (30, 60) , EI 1 (30, 60)
     3, P (120) , PT (30) , LO, HO , H (30, 60) , VPS ( 30, 60) , HBAT
      LOGICAL EDE
      CORNON/AREA1/DIST, PITCH, SIZE, EODE, BGH, HIR, HIQ, HIL
      KN2 - 60
      #2=30
      #18=4
      HIL=4
      #TQ=5
      HAX = 1.9992+19
      DIST = 5.0
      TAN = 45.0
      PITCH = 30.0
      SIZE = 5.0
      KODE = 0
      HGH = 2
      WEITE(6,601)
  601 FORMAT (* INPUT NO OF PLOTSETS - IS FORMAT')
      READ (5, 501) HPLTS
  501 PORMAT (13)
      WRITE(6,602)
  602 FORHAT ( INPUT DESIRED PLOT TIMES")
      READ (5, 502) (PT (I), I=1, #PLTS)
  502 FURNAT (10110.3)
      READ (3) BTAPE, MTINE, T. HS. HE, LO. RO. NO. TO. DT. HBOUND, ILAS
     1, XALN, RDN, RMS
                                               ં પુરુ
      THO-LO/SQRT (9, 582+ 91000)
DO 1 1=1, 1915
                                +SORTHER)
                                           ÷
                                                          -
                                  2
    1 PT (I) = PT (2) / 180
                                                • ]'
                            . X
Si
                                                               e ki
staji
sti
      152=15+2 L
      #22=#242
                                                :
                                                             .
                . :
      ##2=##+7
                                               ***
                                 า
      CALL PLOTS
                    4
      CALL ORGI (SIIE, SIIE)
                                       e
     CALL HPLT2 (84, 82, 832, 84, 852, 822, 88, 8843, 8843
1, 8, 72, 71, 8, 8, 9, 8, 981, 8, 82, 82, 82, 82
2, 882, 4, VPS, TAV, 88, 8083
      CALL PLOT (0.0,0.0,999)
                                            ÷
                                          Ë
      STOP
      2 20
     1
        A.F.
                             ...
           6
     ÷,
        $ 1. 9
```

,

. 994.98, 89218, 2, 86, 86, 56, 56, 86, 8 (4) 7=1, 63=30 IP (7.16.1.00 ), M 27 (3.47. 98 (87)) 60 20 20 8 840 (3) A 8 240 (3) TT , 8 EAB (3) 11 8 EAB (3) 8 8 8AD (3) 782 1 343 (3) T 1 (1) (1) 2 240 (3) 775 2 240 (3) 7 2 240 (3) 7 00 70 1 20 CONTINUE IF (ADS (PT (BP)-T) . GT. ABS (PT (BP)-TOL B) 60 10 21 BP=8P+1 2 240 (3) A 1 3A3 (3) TE 1 8A8 (3) TI R EAD (3) J R EAD (3) PHI 1 2AD (3) T R BAD (3) P EEAD (3) TPS 8 340 (J) V 60 TO 22 21 BACKSPACE 3 T-TOLD #2-#2+1 60 TO 22 23 #P-#PLTS+1 22 CONTINUE DEPINE D (5,3). C-4--BO 11 J=1,86 BO 11 I=1,86 11 B(I,J) =PHE(I)/A(I,J) PIRD B(0,5) C----DO 4 J=1,85 JJ = #2+1-J AT=0.0 AT0=0.0 DO 5 I=1,85 AT=AT+A (I,J) E (I,JJ) =0.5+ (SQET (AT0) + SQET (AT)) 111(1,JJ)=SQ 17 (AT) ATO-AT 5 CONTINUE 4 CONTINUE C----- PLOT CURRENT TIME IN SECONDS. TIME=T+L0/SQRT (9.58E+11+T0) +SQRT (BE) HEIP-ALOGIO (TIRE) ##£\$P=##XP-1 81P- 8817 1 BASE-TINE/(10.0++FEEP) HT= (SISE-2.0)/15.0 CALL STABOL(1.0, 1.0, HT, 'TINE-٠ X 10.0,14) YX=1.0+5.0+ET CALL NUBBER (VI, 1.0, MT, BASE, 0.0,2) TI-1.0+14.0+UT ¥1=1.0+0.5481

14

ŝ

- <u>-</u> -

λ.

State and

4

-

#2-0.5+#T CALL BOBBLE (VI, VI, NT, BIP,0.0,-1) CALL 086 (SISE, STEE) --- CONVERT QUANTITIES TO MORE MATCRAL SHITS. C-- 186 11 18 8V. V 18 10++7 CH/SBC. 8 18 66. 8 18 10++17 CH++-3. Y0 = 9.7882+05+502T(T0)/SQLT(LA) B0 = 6,3+18-09+5081 (20+T0) DO 12 J-1,8% 30 12 E=1,85 TE (1, J) = T0+TE (1, J) II (I,J) = TO+II (I,J) Y(I,J) = Y0 + Y(I,J)YPS(I,J) = Y0-80/L0+YPS(I,J)/RI1(I,J) 3 (I, J) = 30+8 (I, J) H(I,J) = H0+H(I,J) IP (8 (1, J) . GT, BHAX) 8 (1, J) - BHAX 12 CONTINUE DO 13 J=1,88 13 YPS (1, J) =0.0 ---- PLOT VARIOUS GEARTITIES C-/ CALL BBUD (SIIE, RI', RI, Z, M, BI, BI, BI, BI, BO, LO, PLOD (SIIE, RI', RI, SI, MOUND, TE, D, R, YAN, BO, LO, 'TE', 2) CALL PLODS (MS, BI, MR, BHI, MOUND, TI, D, R, YAN, BO, LO, 'TI', 2) CALL PLODS (SS, BI, SR, SHI, MOUND, TI, D, R, YAN, BO, LO, 'TI', 2) CALL PL305 (#5, #2, #2, ##2, MOUND, V, 8, 2, XAN, 20, 14, 'VS', 2) IF (808) •CALL PL3DS (US, #1, BR, HH2, MOUND, VPS, D, R, YAN, RO, LO, "VR", 2) YAU = 360.0-1AU CALL PL3DS (#5, #2, #2, #12, 4800 HD, 8, D, 2, YAN, 20, 10, '8', 1) CALL PL3DS (#5, 12, #8, #11, 8000 HD, 8, D, 2, YAN, 20, 10, '8', 1) YAN = 360.0-1AN CALL LASPLT (P, NI, NO, TO, RO, LO, NBOUSD IF (NP. LE. NPL15) 60 10 1 999 RETURN END SUBROUTINE RORG (Q, QN, R, NS, NI, NA, NAI, NAOURD) C------ SEBBOUTINE RE-ORGANIZES DATA TO RECTANGULAR GRID QB. C----- LINEAR INTERPOLATION IS USED. REAL Q(30,60), 2(30,60), QU(SE, SHI) C----- REVERSE ORDER OF 2"DD INDEX IN ARRAY Q. A = HA/2 DO 1 J=1,8 #1 = #\$+1-J DO 1 I=1,85 TEEP = Q(I,J)Q(I,J) = Q(I,S1) 1 Q(I,#1) = TERP C----- PIND QE. DEE = 1.0/(BBE-1) DE0 = 1.0/(12-880013) e DO 2 J=1,881 Z = (J-1)+BZH. L = 1/010 + 1 IF (L. 89. 85) 60 TO 7 101 = L+1 21 = (L-1) +930 12 - L+950 A2 = (2-21)/(22-21)A1 = 1.0 - A2 60 TO 8 7 L . #5-1

4

Ø

J

Í.

```
LP1 = #3
                ç
      Ă1 - 0.0
      7) = 1.0
11 = 1.0/(11-1)
                                     ß
        - 1
       0 2 I=1,##
      CONTINUE
      RH = (I-1)+DRH

BO = A1+R(K,L) + A2+R(K,LP1)

I7(BO.GT.RH) GO TO 4

17(BO.GT.RH) GO TO 4
      IT (K.GE. 85) 60 TO 5
      K = X+1
      GO TO 3
    4 KH1 - K-1
      IP(K. EQ. 1) GO TO 5
201 - A1+E(KE1,L) + A2+E(KE1,LP1)
       B = (1#-201)/(20-201)
      Q_{H} = A_{1} + Q_{0}(K_{H}1, L) + A_{2} + Q_{0}(K_{H}1, LP_{1})
Q_{P} = A_{1} + Q_{0}(K, L) + A_{2} + Q_{0}(K, LP_{1})
Q_{H}(I, J) = Q_{H} + B + (Q_{P} - Q_{H})
       GO TO 2
     5 QH(I,J) = 11+Q(E,L) + 12+Q(E,LP1)
     2 CONTINUE
       estury
       110
       SUBROUTINE ONGI (ILEN, YLEN)
C----- POR TENTROWII, BE-SCALE PLOT AND MALT UNTILL RETURN ENTERED.
       CALL TOLEAR(61)
       CALL TERASE (61)
       7-ARIS1 (30.0/XLES, 20.0/YLES)
       CALL PACTOR(P)
        CALL PLOT (2.0, 2.0, -3)
        GO TO 10
      1 CALL PLOT (3.0, 3.0, -3)
        10-0.0
        10-0.0
        IL=ILSE
        TL=TLES
        GO TO 10
        ENTRY ORG (XLEN, YLEN)
 CALL TCLEAR($2)
        READ (6,601) BOTHS
   601 FORMAT (IS)
        CALL TERASE (62)
        ETAD(6,601) BOTHG
        GO TO 10
      2 YI=6.0+XL
        Y1=6.0+1L
        T-TO+VT+TLES
 ....
        TL=TLES
        IF (T.GT.33.0) GO TO 3
        CALL PLOT (0.0, VI,-3)
         10=10+ VI
         IF (ILES.GT. IL) IL- ILES
        60 70 10
      3 YT-TO
        CALL PLOT (VI,VY,-3)
         IL-ILSS
. •
  ,
```

\*

10. 20-20+71 10-0.0 2 ς¢. 10 15TURS 220 SUBBOUTINE PLODS (US, NE, NR, NNS, NBOUND, Q, QR, R 1, TAN, 20, 20, / MESS/, HCHAR) REAL ( (30,60), OR (SR, SNE), R (30,60) COSHOS/AREA1/DIST, PLOCH, SIER, SOBL, SSH, HIR, HIQ, HIL. CALL FILT30 (0, #5,#2) CALL BORG (Q, QR, 2, 85, 83, 83, 88, 885, 8800 10) Lo J CALL CPLT3D (QR, NR, HUS, K, SEST, TAN, PITCH, SIXE, KODE, HGB, SC SC=1.0 1, UEE, HIQ, HIL, RO, NO, HESS, MCHAR) CALL 086 (SI38, SI38) STURN. 253 SUBROUTINE LASPLT (P, NI, NO, TO, NO, LO, NBOUND) PLOT LASES POURE VERSUE 2. - IN GH. ٠ . С RAY, 23, 1977. • С . 21 . REAL P(120), 1(120), 10, LO #=#2+1 IT (#800#88. EQ. 2) #= 2\*#8-2 #P1=#+1 ##2=#+2 IL29-6.0 TLES-4.0 € DO 2 I=1,8 2 P(I) =4.928-22+80+T0++1.5+80+80+P(I) AIDELT-STORED/ILES+LO DTIC=ILBE/8.9 CALL AXIS2 (0.0,0.0,'S',-1, ILEN,0.0,0.0, AIDELT, DTIC) CALL SCALE(P, TLEN, N, 1) THIN - P(MP1)YDELT = P(8P2) CALL AXIS2 (0., 0., "LASER PORER (64) ", 16, TLEN, 90., THIN, TDELT, -1.) DI=1.0/(HI-HBOUHD) DO 1 I=1,1 ŧ 1 I(I) = (I-1.5) +DI 3 (871) =0.0 1 (822) = NBOUND/XLE# CALL LINE (1, P. 8, 1,0, 1) CALL ORG (XLES, TLES) RETURN. 233 SUBROUTINE ROUD (SITE, RI1, RI, S, MS, MS, MS2, LO, RO) REAL RI (430,60), RI (822), 10, 1 (822) ---- PLOT E(S, I) : SHELL CENTERS. C-ILES - SISE TLAN - SISE/2.0 IDELT=1.0/ILES+LO INTIC-KLEN/4.0 CALL AXIS2 (0.0,0.0,'1',-1, 1LB8,0.0,0.0, XDELT, XDTIC) CALL AXIS2 (0.0, YLB8,'', 1, XLB8,0.0,0.0, XDELT, XDTIC) IBELT-1.0/ILEB+80 CALL AXIS3 (0.0,0.0, ', 1, TLES, 90.0,0.0, TDELT, TOTIC)

Ŷ

4

,

12-0.15 TI-0.8 ۵ YI= (ILSU/2.0)-0.5+87 CALL SYMBOL (VI, VY, NT, \* 8\*,0.0,1) CALL AXIS2 (1488,0.0,\* ',-1, YLEN, 90.0,0.0, I DELT, YMTIC) 11 (JIS+1) =0.0 11 (HE+2)=1.0/TLEB - DEFISE 1 - VECTOR. C. IS-1.0/82 2 (1)-0:5+IS . DO 6 1=2,88 6 2 (I) =2 (I-1) + X2 2 (#2+1) =0.0 1 (82+2)=1.0/XLEB DO 7 I=1,88 DO 8 J=1,85 8 RX (J) = RI1 (I, J) 7 CALL LIBE (1, 11, 15, 1,0,1) ACHAR=16 YT-0.8 ¥X- (XL88/2.0)-0.5+8CHAR+NT CALL STREOL (VI,VI,RT, 'SEELL BOUNDARIES', 0. 0, NCRAR) CALL ORG (SISE, SISE) RETURN LID SUBROGTINE CPLT3D (A, N, N, K, DIST , YAN, PITCH, SIZE, KODE, HGN, SCAL) C DEAR A PERSPECTAVE VIET OF A CONTOURED SURFACE. С С THELE SET OF SUBBOUTINES (6 OF THEN), HAVING THE HANES: CPLT3D, AUXO60, AUXO61, AUXO62, AUXO63, AND AUXO64, NEBE GIVEN TO STANFORD DI NOVARD JESPENSON OF IONA STATE UNIVERSITY. С С С C THEY MAYS BEES SYTEMSIVELY MODIFIED BECAUSE: C THE NAMES OF OUR PLOTTING SUBBOUTINES ARE DIFFERENT. С THE RETROP OF SCALING HAS BEEN CHANGED. С SORE NEW FEATURES NAVE BEEN ADD ND. С SOME OLD PEATURES HAVE BEEN HADE OPTIONAL. С С IN SPITE OF THE ABOUST OF BODIFICATION, THE ALGORITHM USED TO С DO THE PROJECTION, AND TO DETERMINE THE VISIBILITY OF THE POINTS С ARFLECTS THE HORK DOUR AT IONA. С С MODIFIED FOR USE AT STANFORD BY: BOBBET J. BEEBE, CAMPOS 'FACILITY, С STABPORD VHIVERSITT. С C DATE OF LAST REVISION: MAY 1, 1969 С С A IS THE 2-DIMENSIONED ABBAI CONTAINING THE С FUNCTION VALUES. С С BARBING ... THE CONTENTS OF THIS ARRAY ARE TRANSPORMED INSIDE THESE С BOUTINES (BORE PROPLE NOULD USE THE TERM "DESCROTED") -С С H IS THE BURBER OF BOUS IN THE ARBAY A. С С A IS THE BURDER OF COLUMNS IN THE ARRAY A. С С E IS A CODE THAT TELLS BEETHER TO DRAW THE GRID LINES: С K-1: ALONG THE N-BIRENSION ONLY. С H-2: ALONG THE M-DISENSION ONLY. С

4

7.

• 8.

40

## H-3: ALONG BOTH DINSESIONS.

C C SDISTS TELLS NOW THE ABAY THE SURFACE IS FROM YOUR SYR. С THE USIT OF REASUMENT DOS THIS BINEDSION IS THE C DIAGONAL OF A CUDE THAT ENCLOSES THE SUMPACE. THIS DORS NOT CHANNE THE SIZE OF THE FICTURE, DOT ONLY C С THE DISTORTION. С SDISTS >6 USUALLY BOB'T SHOW ANY DISTORTION DUE TO С PARALAI. С SDISTS <1 HAT CAUSE SUPPRESICTABLE SCALING ERBORS. С С TAN (IN DEGREES) SHOWS NOW THE OBJECT IS THENED ANAL PROM THE VIRER. С POSITIVE TAN WILL TERD TO BRING THE RIGHT EDGE INTO VIEW. С С THEO TAV SHORS THE SURFACE WITH A (1, 1) AT THE LEFT FRONT CORNER, С A (1, H) AT THE LEFT REAR CORNER, AND A (H, 1) AT THE RIGHT PROBT. C C PITCH (IN DEGREES) SHOWS NOW THE SUBFACE IS LOWERED OF PAISED AT THE FRONT LIGE. С C POSITIVE PIICE TENDS TO EXPOSE THE UPPER SURFACE. IF THE BAGHITUDE OF FITCH EXCEEDS 90 DEGREES, FORMI THINGS C С HAPPEN TO THE ORIENTATION OF THE PICTURE. С SIZE (IN INCHES) TELLS NOT LARGE TO MAKE THE PROJECTION OF THE "CUBE" С С THAT ENCLOSES THE SURFACE. С THIS "CODE" THAT BE HAVE BEED TALKING ABOUT HAS THE FOLLOWING ¢ DIMENSIONS: N, H, HAX(N, N). THE FUNCTION VALUES ARE SCALD TO FIT INSIDE THIS CUBE BEFORE ANY ROTATION IS BOBE. С APTER BOTATION, THE COBE IS SCALD TO FIT INSIDE С THE SQUARE PROJECTION PLANE (SIZE BY SIZE). BOTH С ROATSONTAL, AND VERTICAL SCAL PACTORS ARE COMPUTED, С AND BOTH ARE SET EQUAL TO THE SHALLER OF THE THO. С THIS REARS THAT, ALTROUGH THE PROJECTION HAT NOT ALWAYS С APPBAR TO BE THE SANE SIZE, IT WILL ALWAYS HAVE THE ¢ С SAUE SEAPE. C KODE TELLS WEETHER TO DEAR THE "HIDDEN" LINES: С C KODE-O: ASSURE THE SHAPACE IS OPAQUE, SO DOB'T DEAM THE "HIDDEN" LIPSS. С С KODE-1: ASSURE THE GURFACE IS TRANSPARENT, SO ALL THE LINES C ARE PLOTTED. C NGE TELLS BEETHER TO DRAN THE OUTLINE OF THE COBE TO HELP С OBIEST THE VIEWER. C AGN-O: DO NOT DRAW THE OUTLINE OF THE CUBE AT ALL. 5 C NGN=1: DRAW THE OUTLINE OF THE CUBE, BUT PUT IT IN ITS OWN THO-INCH PRANE JUST TO THE LEFT OF THE SURFACE PLOT. С DO NOT DRAW THIS CUBE PULL SIZE, BUT MAKE IT ABOUT C С THO INCHES ACROSS. AGH-2: DRAW THE OUTLINE OF THE CUBE SUPERINPOSED OF THE С С SURFACE PLOT. CPLT3D WILL NOT MIDE ANY OF THE ADGES С OF THE CUBE, REGARDLESS OF VHETHER OR NOT ANY OF THE LIBES IN THE SURFACE PLOT ARE HIDDEN. С С NGN=3: DRAW ONLY THE THREE EDGES OF THE CUBE THAT HEET AT THE С ORIGIN, SUPERINPOSED ON THE SUBFACE PLOT. c 🗮 С SCAL TELLS THE BOOTINES HOW TALL THAT SHOULD HAKE THE SURFACE, С RELATIVE TO THE MEIGHT OF THE CUBE. С

V

-

С

```
SCAL-O.: DO NOT SCAL THE DATA AT ALL, BUT TRUST THE USER THAT
THE DATA IS NOT SO HIGH THAT IT BURS OFF THE PAPER.
С
С
С
      SCAL= 1.: SCAL THE DATA SO THE TOP OF THE DATA JEST DEACHES
С
                 THE TOP OF THE CUBE.
                                                            ्य
С
С
     SCAL=.3: SCAL THE DATA SO THE TOP OF THE SURFACE IS
C
                 THREE TENTHS AS HIGH AS THE CUBE.
С
С
      SCAL=4.: SCAL THE DATA SO THE TOP OF THE SUBFACE IS
С
                 POUR TIMES AS HIGH AS THE CUME. (DANGEBOUS)
¢
С
   WARNING ... IT IS VERY EXPENSIVE TO DRAW OPAQUE SURFACES, BECAUSE
Ċ
C
                THE PROGRAM MAS TO DETERMINE THE VISIBILITY OF STREY
                POINT. THE COMPUTER TIME DOUBLES ON TRIPLES, DEPENDING
ON NON MANY LINE SEGRENTS ARE PARTIALLY VISIBLE.
            ... THIS IS NOT A STAND-ALONE PACKAGE, BUT IT IS INTENDED
                THAT THIS SUBROUTINE (S) WILL BE USED ALONG WITH OTHER
                 SUBBOUTINES FROM THE CALCORP PLOTTING PACKAGE.
                 THE OUTPUT BOUTINES HUST BE INITIALIZED BEFORE USING
                THIS PACKAGE. SEE THE WRITEUP FOR PROGRAM NUMBER CO23.
            ... DO NOT USE SUBBOUTINES, OR MAMED COMMON, WITH ANY
0000000
                 OF THE FOLLOWING NAMES:
                   CPLT3D, AUX060, AUX061, AUX062, AUX064
(THESE ARE SUBBOUTINES IN THE CPLT3D PACKAGE)
                         --02--
                   CO8024, CO8025
                   (THESE ARE FOR NAMED COMMON USED BY THE CPLT3D
                   PACKAGE) .
С
            ... DO NOT PORGET, THE CONTENTS OF THE ARRAY & GET CLOBBASTD.
С
       SUBROUTINE CPLT3D(A, N, H, K, DIST, YAW, PITCE, SIZZ, KODE, HGM, SCAL
      1, NIR, NIQ, NIL, RO, XO, /HESS/, NCHAR)
       COMMON /COM024/ ANGA , ANGB , HV , D, SH, SV
COMMON /COM025/SL , SH , SH , CX , CX , CZ , QX , QY , QZ , SD
DIMENSION H ( 10 ) , V( 10 ) , X(2) , Y(2) , Z(2) , XP( 8 ) ,
      : A (N,8), DZ (4)
C++++ ++++++++
       $DISTS=DIST
       ANGA = (YAN+270.) + .0174532
ANGB = PITCH + .0174532
       HY = SIZE
C DIRECTION COMPONENTS TO THE EVE.
                                                                  1
         JL = -COS (ANGA) + COS (ANGB)
SH = -SIN (ANGA) + COS (ANGB)
         SH = -SIN ( ANGB )
                                . # 8.
             IF ( ABS( $# )
                                                  GO TO 10
                                       1.0 )
    URITE( 6 ,
20 POBRAT( '1', 201 ,
                                   20
                                        )
                                 20(1+1) , / 101, YOU ARE ATTEMPTING TO LOD
      :K STRAIGHT DOWN ( OR UP ) AT THE SURFACE * )
                 GO TO 2150
    10 CONTINUE
        X(1) = 1
         X(2) = #
         T(1) = 1
         ¥(2) = #
```

218

1

```
T=#I10 (#, #)
   FIND THE DIAGONAL OF THE "CUBE".
D = H ++ 2 + H ++ 2 + T ++ 2
С
         D = SQRT (D)
SCL = SDISTS * D
   COORDINATES OF YOUR STR.
С
         CI = -SL + SCL
         CT = -SH + SCL
    CI = -SH + SCL
COORDINATES OF THE PROJECTION PLANE.
С
          QX = CX + D + SL
          QT = CT + D * SH
          Q1 = C2 + D + 11
 2060 CALL AUX060 ( A, H, H, Z , KODE)
С
        DI(1) = I(1)
DI(2) = I(2)
                                                                2
         VUQ = NIQ
         CALL SCALE (D1, VUQ, 2, 1)
         Z(1) = DS(3)
         I(2) = VUQ = DI(4) + I(1)
С
                                                                ۰.
         BAHGB= (3(2)-3(1))
         $=1.
         IF( SCAL . ME. 0 ) S-T/RANGE+SCAL
   SCAL THE SURPACT TO MAKE A "CUBE".
С
          DO 30 I = 1 , I
          DO 30 J = 1 , 8
          A ( I , J ) = ( A ( I , J ) - I ( 1 ) ) * $
     30 CONTINUE
         QHI#=2(1)
         Q8AX=2 (2)
          2(1) = 0.0
          I(2) = T
          ALL AUXO61 (X,Y,Z,XP,H,V,KODE)
DO 2130 I = 1,8
H(I) = ((XP(I) - QX) + SH - (H(I) - QY) + SL) + SD
V(I) = (V(I) - QE) + SD
ONF(HTP)
  2080 CALL AU1061
  2130 CONTINUE
  2100 CALL AUXOGO ( H, 8, 1, H (9) , KODE)
2120 CALL AUXOGO ( V, 8, 1, V (9) , KODE)
IF ( HGH .EQ. 0) GO TO 2140
         S=HV
     \mathbf{a}
         IF (BGH .EQ. 1) S=1.5
SH = S/ (H (10) -H (9) )
SV = S/ (V (10) -V (9) )
         SH = SIGH ( AMIN1 (SH, SV), SH )
         SV = SIGN(SH, SV)
         IF (HGH. EQ. 1) CALL PLOT (0.0, 2.0, -3)
CALL AIDR (H, V, M, SV, HESS, NCHAR, TAN, MIR, MIQ, MIL, RO, 10, QHIN, QHAX)
          IF(MGH .#2. 1) GO TO 2140
 2139
         CALL PLOT (AINT((H(10)-H(9))+SH+2.),-2.05,-3)
   2140 CALL AUXO63 ( X, Y, A, H , H , H , Y , K , KODE)
   2150 CONTINUE
             RETURN
         EED
 С
         SUBROUTINE AUXOGO ( A, N, H, Z , KODE)
 DINENSION Z(1), A(N, N
C FIND THE MAX, AND MIN Q
                                    Z(
```

ħ

.

```
1050 Z(1) = A(1, 1)
 1060 Z(2) = Z(1)
С
 1080 DO 1190 J = 1 , R
С
 1100
         DO 1180 I = 1, I
С
        I(1) = AMIN1(I(1), A(I,J))
        Z(2) = ABAX1(Z(2), A(1,J))
С
 1180
          CONTINUE
 1190
         CONTINUE
 1230
                     RETURN
                              ZH D
   SUBROUTINE AUXO61 ( X, I, Z, IP , H , V , KODE)
FIND THE CORNERS OF THE BOTATED CUBE.
С
С
        DIMENSION X(1),Y(1),Z(1),H(1),Y(1),XP(1)
С
   050
        L = 0
   070
        DO 180 I = 1, 2
С
           DO 170 J = 1, 2
   090
С
   110
            DO 160 K = 1, 2
С
              L = L + 1
   130
   140
              CALL AUXO62 ( X(I), Y(J), Z(K), XP(L),
      1
                      H(L), V(L), KODE)
   160
            CONTINUE
           CONTINUE
   170
   180 CONTINUE
   190 RETURN
          LND
   SUBROUTINE ADX062 ( X, Y, Z, XP, JP, ZP, KODE)
FIND THE LOCATION OF A POINT IN THE ROTATED CUBE.
С
         COMMON /COMO24/ ANGA, ANGB, HV, D, SH, SV

COMMON /COMO24/ ANGA, ANGB, HV, D, SH, SV

COMMON /COMO25/SL, SH, SN, CX, CY, CZ, QX, QY, QZ, SD

SK = D / ( (X - CX) + SL + (Y - CY) + SH + (Z - CZ) + SW)

IP = CX + SK + (X - CZ)

TP = CT + SK + (Y - CT)
          2P = C2 + 3K + (2 - C2)
                           RETURN
                                            ZID
        SUBROUTINE AUXO63 (X,Y,A,M,H,H,Y,K,KODE)
С
     DRAN THE FIGURE.
          COMBON /CON024/ ANGA , ANGB , HV , D, SH,SV
Combon /Con025/SL , SH , SN , CX , CY , CZ , QX , QY , QZ , SD
С
         DIMENSION I(1), Y(1), H(1), V(1), A(N,H)
         INTEGER UP , DOWN , PEN , P , Q
           INTEGER P1 , PO
С
С.
С
      END = 1.0 / 16.0
CAN DSE 1 / 32 OR 1 / 64 FOR FINER INTERPOLATION
С
С
С
```

ð

```
٠
      82
            - 3
      DOWN = 2
      SH = HV / (H (10) - H (9))
SV = HV / (V (10) - V (9))
      SH = SIGN (AMIN1 (SH, SV), SH)
      ST = SIGH(SH, ST)
      ) #8 = N
       #¥ = ¥
  080 IF ( K - 1 ) 100 , 120, 100
С
  100 IF ( K - 3 ) 1110, 120, 1110
С
С
  DRAW LINES ALONG THE T-AXIS
  120 CONTINUE
      L = 0
      LD = 1
        DD = 0.5 + LD
С
  140 DO 1060 J = 1, 8
      Q = 0
      \bar{\mathbf{Y}}\mathbf{J} = \mathbf{J}
                                                           1
  160
      DO 1030 I = 1, XH
С
            L = L + LD
        XI = L
       CALL AUXO64 ( A ,XI ,YJ , X , H , P ,KODE)
        PEN = UP
                                                        ⊲tt
            IF (P) 510, 520, 530
  510 CONTINUE
          IP (Q) 540, 550, 540
  520 CONTINUE
          IF (Q) 610, 1020, 610
  530 CONTINUE
          IF (Q) 540 , 550 , 540
  540 CONTINUE
         PEN = DOEN
        GO TO 170
  550 CONTINUE
                   .EQ. 1 ) GO TO 170
           IF ( I
           DI = DD
       TO = L - LD
         T = TO + DI
         P1 + Q
       IF ( ABS( DI ) .LT. END ) GO TO 570
  500
        CALL AUXO64 ( A , T , YJ , W , H , PO , &ODE)
DI = DI = 0.5
        TO = T
         P1 = P0
         T = T - DI
         GO TO 560
  565 T = T + DI
GO TO 560
570 CONTRUE
T = TO
IF ( FT = P )
580 CONTRUE
                        170 , 170 , 580
   590 CONTINUE
       2P = A(L-LD, J) + (T-L+LD) + (A(L, J) - A(L-LD, J)) / LD
       CALL AUX062 (T, TJ, ZP, XP, NH, TV, KODA)
```

•

251

3

. 🖝

3

,

```
HH = ( ( XP-QX) * SN- (HH - QY ) * SL ) * SD

VV = ( VV - QZ ) * SD

HH = ( HH - H (*) ) * SH

VV = ( VV - V(*) ) * SH

C) L D COM ( HH - H * ST
          CALL PLOT ( NH , VV , PEH )
 600 PEN = 5 - PEN
        GO TO 170
                                   .
  610 CONTINUE
           PIN - DONN
            DI = DD
          TQ = L - LD
        T = TO + DI
             P1 = Q
  620 IF (ABS(DI) .LT. END ) GO TO 630
CALL AUXO64 (A, T, YJ, H, H, PO, KODE)
          DI = DI + 0.5
                                      GO TO 625
          IF ( PO . BQ. 0 )
           70 = 1
         P1 = PO
         T = T + DI
          GO TO 620
          T = T - DI
  625
             GO TO 620
          CONTINUE
  630
         IF ( P1 + Q ) 600, 600, 590
CALL AUX062 ( XI, YJ, A( L, J ), IP, RH, VV, KODE)
VV = ( VV - QZ ) * SD
     .
  170
         HH = ( ( XP-QX) + SH- (HH - QY ) + SL ) + SD
         HH = ( HH - H(9) ) * SH
VV = ( VV - V(9) ) * SV
CALL PLOT ( HH , VV , PEN )
  190
  200
         Q = P
 1020
 1030 CONTINUE
С
С
         L = L + LD
LD = -LD
             DD =-DD
С
 1060 CONTINUE
С
С
 1090 IF (K - 3) 2060, 1110, 2060
С
C DRAW LINES ALONG THE X-AXIS.
 1110 CONTINUE
                                   .
С
        L = 0
        LD = 1
        DD = 0.5 * LD
DO 2040 I = 1 , M
  1140
           I = I
          Q = 0
           DO 2020 J = 1 , NH
  1160
                   \mathbf{L} = \mathbf{L} + \mathbf{L}\mathbf{D}
                 YJ = L
          CALL AUXO64 ( A ,XI ,TJ , M , H , P ,KODE)
          PSH = UP
                  IF (P) 1510, 1520, 1530
```

```
IP (Q) 1540 , 1550 , 1540 6
1510 CONTINUE
      CONTINUE
1520
                      1610 , 2010 , 1610
         IF ( Q )
1530 CONTINUE
             IF (Q) 1540 , 1550 , 1540
         CONTINUE
1540
           PEN - DOWN
          GO TO 1170
               CONTINUE
1550
           IF ( J . EQ. 1 ) GO TO 1170
            DI = DD
       TO - L - LD
          T = TO + DI
          P1 = Q
1560 IF ( ABS( DI ) .LT. END ) GO TO 1570
         CALL AUX064 (A, XI, T, M, M, PO, KODE)
DI = DI + 0.5
         LT ( PO .EQ. 0 ) GO TO 1565
           10 = T
          \begin{array}{rcl} \mathbf{P1} &= & \mathbf{P0} \\ \mathbf{T} &= & \mathbf{T} &= & \mathbf{DI} \end{array}
            GO TO 1560
                                                 .
 1565 T = T + DI
               GO TO 1560
           CONTINUE
 1570
           T = TO
         IF ( P1 + P ) 1170 , 1170 , 1580
        CONTINUE
 1500
        CONTINUE

2P=A (I, L-LD) + (T-L+LD) + (A (I, L) - A (I, L-LD))/LD

CALL AUX062 (XI, T, ZP, XP, HH, VV, KODE)

HH = ( (XP-QX) + SH - (HH - QY) + SL) + SD

VV = ( VV - QZ) + SD

HH = ( HH - H (9) ) + SH

VV = ( VV - V (9) ) + SV

CALL DIOT ( HH - VV - PRN )
 1590 CONTINUE
           CALL PLOT ( HH , VV , PEN )
  1600 PEN = 5 - PEN
            GO TO 1170
  1610 CONTINUE
             PEN = DONN
              DI = DD
           TO = L - LD
          T = TO + DI
               P1 = Q
          IF (ABS(DI) .LT. END ) GO TO 1630
 1620
           CALL AUXO64 ( A , XI , T , W , H , PO , KODE)
DI = DI + 0.5
                       .EQ. 0 ) GO TO 1625
            IF ( PO
             T0 - T
           P1 = PO
           T = T + DI
            GO TO 1620
            T = T - DI
   1625
               GO TO 1620
             CONTINUE
   1630
            T = TO
           IP ( P1 + Q ) 1600 , 1600 , 1590
CALL AUIO62 ( XI, TJ, A ( I, L ) , XP , HH , YY , KODE)
HH = ( ( XP-QI) +SH- (HH - QI ) +SL ) + SD
   170
```

```
VV = ( VV - QI ) • SB

HR = ( HH - H(9) ) • SR

VV = ( VV - V(9) ) • SV
 1180
 1190
           CALL PLOT ( BM , YY , PRM )
 2010
         Q = P
           CONTINUE
 2020
С
         L = L + LDLD = - LDD\Theta = -DD
 2040 CONTINUE
С
 2060 CONTINUE
С
 2130 BETURN
           END
        SUBROUTINE AUXOGA (I, XI, YJ, H, H, P, KODE)
C SEE IP A POINT IS VISIBLE.
          CORNON /CORO24/ ANGA , ANGB , NY , D, SN, SV
CORNON /CORO25/SL , SN , SN , CX , CX , C2 , QI , QI , QI , SD
          INTEGRE CUR , CNT , F
REAL [ , J , II , JJ
DINENSION & (W.W)
         IF ( KODE . EQ. 1) GO TO 78
         II = II
         JC = TJ
                                                                                  • .
         ZB = Z (IR, JC)

IF ( XI .BQ. IR ) GO TO 2

ZB = Z(IR, JC) + (XI - IR) + (Z(IR + 1, JC) - Z(IR, JC))
               GO TO 4
      2 IP (YJ .EQ. JC) GO TO 4
ZB = 2(IR, JC) + (YJ-JC) + (Z(IR, JC+1) - 2(IR, JC))
       4 CONTINUE
           XEND = 0.0
           DX = 0.0
          THULT = 0.0
           ZBULT = 0.0
          IP (XI .EQ. CX ) GO TO 10
YHULT = (YJ - CY ) / (XI - CX )
ZHULT = (IB - CZ ) / (XI - CX )
          DX = 1.0
             XEND - 8 + 1
             IF ( 11 . LT. CI ) GO TO 10
             DX = -1.0
           XEND = 0.0
   10 CONTINUE
          TEND = U.O
            DI = 0.0
             IBULT = 0.0
            IF (YJ .EQ. CY ) GO TO 20

HULT = (XI - CX ) / (YJ - CY )

IF ( ZHULT .EQ. 0.0 ) ZHULT=(XB - CZ ) / (YJ - CY )
            DY = 1.0
            YEND = ¥ + 1
            IF ( YJ .LT. CY ) 60 TO 20
            DY = -1.0
            TEND = 0.0
       20 CONTINUE
           CUB = 0
           CHT = 0
```

κ.

- 1

-

```
-
                                 ۰.
                               ``
        2 - 4
                                 ۹.
         X8 - XI
         18 - 1J
                                                            ۱
     30 CONTINUE
        II = AINT( IN )
        JJ = AINT( YN )
        ISTEP - DI
        ISTER - DT
          IF (XB . EQ. II) GO TO 40
IF (DX .LT. 0.0) X3TEP = 0.0
        GO TO 45
     ...
          IF ( YB
                  . BQ. JJ ) 60 TO 45
          IP ( DY .LT.
                        0.0) ISTEP = 0.0
     45 CONTINUE
          I = II + ISTEP
          J = JJ + YSTEP
          IF ( I
                 -EQ. IEND ) GO TO 80
-EQ. TEND ) GO TO 80
          IP ( J . IQ.
       XB = CX + XHULT + ( J - CY )
        13 - CI + YHULT + (I - CI )
         IF ( DI .LT. 0.0 ) 20 TO 55
          IF ( IB . LT. I ) GO TO 60
    50 XB = I
        60 10 65
    55
       IF ( IB .LT. I ) GO TO 50
    60 TB - J
    65 CONTINUE
       IN - CI + INULT + ( IN - CX )
       IR = I
       JC = J
        IP ( IB . #R. J ) GO TO 70
IDI = I - DI
       25 = 2(IR, JC) - DX + (XB - I) + (2(ID1, JC) - 2(IR, JC))
       GO TO 75
    70 \text{ JDY} = J - DY
       25 = 2(_IR,JC ) - DI + ( IB-J ) + (2( IR,JDT ) - 2( IR,JC ) )
    75 CONTINUE
       SG# = 1
       IP ( 28 .LT. 25 ) SGH = -1
       CUN - CUN + SGN
       CHT = CHT + 1
       IP ( IABS ( CUS )
                            - EQ. CHT ) GO TO 30
        GO TO 90
78
       P=1
       GO TO 95
    80
       CONTINUE
       P = 1
          IP ( CUM ) 84 , 86 , 90
   84 P = -1
             GO TO 90
    96
       CONTINUE
              IP (IB .LE. CI )
                                      GO TO 90
       P = -1
   90 CONTINUE
95
        *****
      RHD
      SUBROUTINE AXDR (E, V, SE, SV, /SESS/, ECHAP, YAB, SIR, HIQ, HIL, BO, 10
     1,QHID,QHAX)
      DIBENSION R(10), V(10), X(8), Y(4)
C----- PUT THE COORDINATES OF EACH OF THE & CUBE CORNERS IN 161.
```

đ

•

DO 1 I-1,8 I(I) - (B(I)-B(D))+88 I(I) - (V(I)+V(D))+89 1 CON нŤ CALL PLOT (2(2) , 1 (2) , 3) CALL PLOT (1 (4) AT (4) , 2) CALL PLOT (1 (34 . 8 (3) . 2) CALL PLOT (2 (7) . 1 (7) . 2) CALL PLOT (1 (0) , T (0) , 2) CALL PLOT (I (4) ,I (4) , 2) CALL PLOT (# (4) , T (6) , 3) CALL PLOT (2 (6) , T (6) , 2) CALL PLOT (X (2) . T (2) , 2) -- DRAW THE MARIUS AZIS. CALL AXPL(1, 5, Z, Y, 0. 0, B0, 'RADIUS', -6, HIR, 1.0) IP (YAN. GT. 180.0) GO TO 2 -- DRAU THE APPROPRIATE ANIS FOR G.LT. TAN. LE. 45. CALL AIPL (5, 7, 2, 7, 0, 0, 20, 2', -1, NIL, 1. 0) CALL AXPL (1, 2, X, Y, QHIN, QMX, MESS, MCHAR, MIQ, -1.0) C----- DRAN LINES TO FINISH BOX. CALL PLOT (X (5) , Y (5) , 3) CALL PLOT (1 (6) , T (6) , 2) CALL PLOT (1 (1) , T (1) , 3) ۰ می CALL PLOT (I (3), I (3), 2) RETURN 2 CONTINUE -- DRAW THE APPROPRIATE ALLS FOR 325.LT. TAN.LT. 360. CALL AXPL (3, 1, 1, 1, 10,0.0, 12', -1, #1 4, 1.0) NCH - - NCHAR CALL AIPL (5, 6, 1, 7, QMIN, QMAI, HISS, HCH, HIQ, -1. 0) --- DRAN LINES TO PINISH BOX. C----CALL PLOT (1(5), 1(5), 3) CALL PLOT (1 (7) . 1 (7) . 2) CALL PLOT (1 (1), 1 (1), 3) CALL PLOT (X(2), T(2), 2) RETURN Z ND SUBROUTINE AXPL(I, J, I, T, CHIN, QHAI, /HESS/, HCHAB, HIQ, DTS) DINENSION I(8), T(8)  $\mathbf{D}\mathbf{X} = \mathbf{X}\left(\mathbf{J}\right) - \mathbf{X}\left(\mathbf{I}\right)$  $\mathbf{DI} = \mathbf{Y}(\mathbf{J}) - \mathbf{Y}(\mathbf{I})$ . AXALU = QUIU AXDELT - (QUAX-QUIN) /AXLIN ١ DTIC = AXLEN/NIQ-098 C CALL AXIS2 (X (I), Y (I), HESS, BCHAR, AXLEN, ANGLE, AXHIN, AXDELT, DTIC) 209 SUSACUTINE FILTID (9, 85, 82) C+ C . • FILTER MET JETTER FROM OUTPOR OF 2-5 AND ADOTINE. C ٠ ARCHARD D. MILBOT 78-02-14 \_ BEAL Q (30,60) INTRAS S.I <u>A</u> C-- FILTER IN I-DIR BOTION #Z#T = #Z - 1 DO 1 5-1,45



÷

ş

.

.

**~**.

-در ,

••

ı

.

.

.

ŧ

÷.

.

÷

ſ

r

٩

۰

.

c

.

1

۰t,

15 300 2-0 SZM121 70-02-09 1000000000000 -----REAL A (30,60), T2 (30,60), T2 (30,60), S(30,64), PHZ (31), T (30,64) , TFS (30,60), P (130), B(200 2, H2 (500), T27 (500), Z27 (500), T2 (500) 3, H1 (500), T27 (500), T17 (500), T (500) 4, H1 (500), T2 (500), T17 (500) 4, H1 (500), T2 (500), T17 (500) 5, H1 (500), T2 (500), T17 (500) 7, H1 (500), T2 (500), T1 (500) 1 3 ٠ INTRESS XINT, BINT, TINT LOGICAL BOR ILEE . 4.4 TLES = 5,0 TIN - 5.8-7 TINC - 5.8-7 TFIN # 30.8-6 TING = 1.52-0 . . HAAT + 5.8+14 ٠. XINT . 5 d, 1101 = 3TINT - 5 10 - 50.0 **#**\$3 READ(3) STAPS, WTINE, T, NE, NE, LO, RO, NO, TO, DT, NBOWND, XLAS 1, IAI, 80 8, 88 MCESPACE J Y0 = SQRT (9, 562+ 11+T0) / SQRT (RA) TH0 = L0/SQRT (9, 562+ 11+T0) + SQRT (RA) DE - LO/(NE-MOONND) XI = 10/0X + 0.5 KI = HI+1-KI 17 - . XX = 1 2 = -1.0E+50 CALL PLOTS CALL ORGI (TLES, TLES) 2 KK =KK+1 IF (T.GT. TFIN) GOT TO 96 " TD = TIN + (KK-1)+TINC 150 • 1 TOLD . T E EAD (3, EBD-99) STAPE, STI ME, T, SS, NE, LO, RO, NO, TO, DT, NBOUND 7 - 7.780 IF(T.LT. 1.8-30) 2=1.08-30 IF (T.GT. TD) 00 TO 3 READ(J) A READ (3) TE RKAD(3) TI ٩ 2 ZAD (3) READ (3) PHI R BAÐ (3) . . 2 X A D (3) P 2 SAD (3) VPS -2 SAD (3) 1 ų, IT = IT+1 HT (IT) = # (1, KI) +HO TET (IT) = TE (1, KI) +TO TIT (IT) = TI (1, KI) +TO ١. IT (IT) = T 60 TO 1 Ż

?

٩,

\* 🖕

258

ï

>

Ba-

5

.02. ADD (19-906.00) .00 TO 4 22 28 1240 (3) TI 1240 (3) T ----2 240 (3) 2 240 (2) ۳. . 12A.0 (3) 405 1840 (3) 4 17 - 17+1 IT = IT = I (1, KI) = 00 TET(IT) = 1 (1, KI) = 00 TET(IT) = TI (1, KI) = TO TII(IT) = TI (1, KI) = TO TT (IT) = T = T = 5 = T0 5 A BACKAPACE 3 2 - 2018 00 20 S 19 2728 - 4.0 C---TINDOT HEID-ALGOIO(TINE) HEID-SEID-1 EXPORT? 9 AAB-TISE/(10.0++##17) HT= (YLEH-2.0)/15.0 CALL STHEOL(1.0, 1.0, HT, 'TIHE-YZ=1.0+5.00HT CALL HOMBER (VZ, 1.0, HT, BASE, 0.0, 2) YZ=1.0+0.00HT YY=1.0+0.00HT CALL HOMESR (VZ, VT, HT, SIP, 0.0, -1) CALL ONG (ILEH, FLEN) ---- BEFINZ VECTORS POE PLOTTING BARIAL PLOTS. AT=0.0 x 10',0\_0,14) ٩ C-12-0.0 111-0.0 80 10 T=1,85 AT = AT+A(I,EE) R(I) = 9087(8.5\*(AT+AT1)) 10 AT1 - AT 10 6 I-1,85 I(I) = 1(I)+10 b1(1) = B(1, K1) = B TE1(1) = TE(1, K1) = TE TE1(1) = TE(1, K1) = TE TE1(1) = TE(1, K1) = TE CALL XPLOT (88, X, 81, 781, 781, 7188, 7188, 8187, 88AX, 8187, '8', -1, 80) --- BEFISE VECTORS DOB PLOETING AXIAL PLOTS. DO 7 I-SDOURD, 98 C-J=##+1-I 

• •

.

œ'

259

· 🍆 👝

CO BARS PLOTS TO A PURCESSU OF TIME. CALL EPLOR (IT, 97, 97, 982, 952, XLON, XLON, MENT, MAX, 2102, "9",-1,2200) -- FLOT ALLAL VELOCITY VS L. CALL VELOT (01, 1, VELT, 1110, 1101, 1197, 44, 10) CALL BLOT (0.0,0.0,990) STOP 233 SUBROUTINE ANDR (ILES, YLES, /IAI/, MIT, IAIN, IDELE, IDT , THIN, TOOLE, TOT, NAIN, NDELE, NDELE, NDE REAL HAIN, NO SLE, NOT CALL AIIS2 (0.0,0.0,IAI, WTIT, ILEN, 0.0,IMIS, IDALT, IDE) CALL AIIS2 (0.0,0.0,'T', 1,YLEN, 90.0, THIN, TDALT, TDT) CALL AXIS2 (MLEN, 0.0,'Y', -1,YLEN, 90.0, NMIN, NDALT, NDT) CALL PLOT (0.0, TLB. 3) CALL PLOT (IL SU, TLBU, 2) 111111 21D SUBROUTINE X PLOT (SP, I, N, TE, TI, ILES, FIST, SINT, SUBROUTINE X PLOT (SP, I, N, TE, TI, ILES, XLST, SINT, XLST, XLST, /XAZ/, NTIT ۰ , 10) BEAL X (500) , N (500) , TE (500) , TI (500) , NAAX, NAIN, NPALT, NDT, SC (4) INTERE XENT CALL FILETS (28, 8P) CALL FILTID (TI, BP) CALL FILTID (N. NP) TELS = 1.250 THAX = -1.850 DO 1 I-1,82 IF (TS(I).LT.TSIN) THIN-TS(I) IF (T1(I).LT.TNIN) THIN-TS(I) IF (T1(I).LT.TNIN) THIN-TI(I) IF (TS(I).GT.TNAX) THAX-TS(I) IF (TI (I) . GT. THAX) THAX-TI (I) IF (N (I) . GT. 3 MAX) N (I) = NHAX 1 CONTINUE SC (1) = THIN SC (2) = THAX \* CALL SCALE (SC, TLBS, 2, 1) XHIN = 0.0 IDELT- XO/ILES INT - ILES/XINT T#IN = SC (3) TB&LS = SC (4) TPT = -1.0 ----PORLY - BRAX/TLES BOT - -TLEN/HENT TS(8P+1) = SC(5) TS(8P+2) = SC(4) TI(8P+1) = TE(8P+1) TI (##+2) =TE (##+2) H (HP+1) = HAIH H (HP+2) = SDELT X (NP+1) = XAIN X (HP+2) = XDELT CALL AIDR (ILBP, YLBP, IAX, MIT, XHIS, SDELT, IDT , THIN, TOSLE, TOT, JALS, HORLE, UPT) CALL LINE (1, 28, 89, 1, 0, 0) CALL LIBE (1, 11, 19, 1, 0, 0) CALL LINE (1, 8, 89, 1, 0, 0) 1

ŧ

2.5

ر

260

13 e

ONST (MAR, TLEN) STOLE IN A WAT TO AINIMIZE PLOPTER PAPER WASTE. STOLET, SB-SCALE PLOT AND BALT BETILL BETER BUTER SD. 784 C-C. **4**(61), CALL SHE HA(61) CALL CHIMAE (677 P-AREN (D0.0/ILES, 20.0/ILES) CALL PACTOR (P) CALL PACTOR (P) CALL PACTOR (2.0, -3) GO TO TO GALL CALL PLOC(3.0, 3, 0, -3) X0=0.0 10-0.0 IL-ILES 11-1180 60 10 10 BITRY 006 (ILEU, TLEP) CALL TO THE ALL DOT 1'ST CALL TO THES BOUTINE. E EAD (6, 601) 80296 601 POBEAT (15) CALL TERASE(62) E BAS (6, 601) BOTHE GO TO 18 2 YE=6.0+1L VI-6.0+1L T-TO+TY+TLEE TL=TLEB **£**3 IF(Y.GT.33.0) 00 TO 3 CALL PLOT (0.0, VY,-3) YO=YO+YT IF (XLIN, GT. XL) XL-XLEN GO TO 10 3 71-10 🖷 : E : S CALL PLOT (VI, VI,-3) IL-ILBE XO=XO+VI 10 252035 \$ END SUBBOUTINE LASPLT (P. HI. HO, TO, BO, LO , BOUND, XLEN, YLEN, XINT, XO) -----.......... PLOT LASER POWER VENUS 2. - IN GN. Č C ٠ C • BAY, 23, 1977. C • RICEADD 6, BILLON C • RICEADD 6, BILLON . . 794- 1 REAL P (120), I (120), NO, LO, ISTES (120) INTERNA XINT NS = 0.25 1L0 = 0.0035 #=#2+1 IF ( NOOUND. EQ. 2) #= 2485-2 481=4+1 872=8+2 80 2 I+1,8 2 P (1)=0.528-22+80-20++1.5-20+80+P (I) ALBRIE-SBOSSB/ILES-LO PTIC=ILSI/II IT

. . ʻ- ik 262 CALL ASTR: 00.0.0.0.1. CALL SCALE (P. T.L. 8, 8, 1) TLBU, 0.0, 0.0, ANDELS, DELC) . 1819 - P(891) 19817 - P(892) CALL ATTS2 (0., 0., "LASER DOURN (GI) ", 104 LEU, 90., THIN, TOBLE,-1.) 54-1.0/(HE-BBOGUD) 80 1 2-1,8 . 1 8(I)=(I-1.5) +08 8 (#21) =0.0 L (MP 2) = 19099 3/2L29 CALL LINE (5, P, 8, 1, 0, 1) CALL PLOT (0.0, TLED, 3) C £ 35+136 (I0-3(I) +L0) 101/ (6. 20345168A++2) +1.29 10 (171) 10 (199 2) 122 (XLEB, 0.0, 'INTERSITY', -9, TLEE, 90., THIE, TOBLE, -1.) ALUE (1, 1988, 8, 1, 0, 1) A ... CALL ODS (ILBS, TLES) ASTUDA -... 212 SUBBOUTINE FILTID (Q, #) С . FILTER JITTER FROM RESULTS BEFORE PLOTTING. ٠ 78-02-14 C • RICHARD B. HILBOY BEAL Q(1) ##1 = #-1 22 = 2(1) DO 1 1=2,881 Q1 - Q(I) Q(I) = 0.5+Q(I) + 0.25+(Q2+Q(I+1))02 - 01 1 CONTINUE BETVEN. 212 SUBROWITHE VPLOT (N.X. VPLT, XLEN, TLEN, XLNT, KK, YO ٠ PLOT VELOCITY FOR SEVERAL TIMES. C • RICHARD D. HILBOY C 78-02-15 REAL VPLT (100, 10), V (100), SC (4), I (500) INTERSE XINT 1 - 11-1 -- FIND MAI. AND MIN. OF V AND SCALE. C-VEAL = -1.250 6 VALU = 1.250 DO 1 J=1,E BO 1 I=1,B IF (VPLT (I, J) .LT. VHIN) VHIN - VPLT (I, J) IF (VPLT (I,J) .GT. VEAX) VEAX = VPLT (I,J) 1 CONTINUE SC(1) = VAIN SC(2) = VEAX CALL SCALE (SC, TLES, 2, 1) VALU = 9C (3) • . . .....

7 • . 1 •; Ľ • 263 Ð •**;**,,, £Ť , VALLE - State) IRLE - 0.5 IDELT - 10/XLEN XAT - XLES/XLEN YBT = -1.0 CALL AXIB2(0.0,0.0, 'I', -1, XLEB,0.0, XBX M, XBRLT, XPT) CALL AXIB2(0.0,0.0, 'V', 1, XLEB,0.0, YELE, YBRLT, YBRLT, YPR, S) CALL PLOT(0.0, YLEB, S) CALL PLOT(XLEB, TLEB, S) CALL PLOT(XLEB, TLEB, S) CALL PLOT(XLEB, TLEB, S) CALL PLOT(XLEB, 0.0, 2) BO 2 J=1, E BO 3 I=1, B 3 Y(I) = YPLT(I, J) CALL FILTID(Y, B) Y(B+1) = YELE Y(B+2) = YDELT CALL LINE(I, Y, B, 1, 0, 0) VOT = -1.0 . . 11. CALL LINE (X, V, H, 1, 0, 0) <u>.</u>99 ų, 2 CONTINUE . . 🕁 RETURN RED . • Υ. ۰. \* 33 · J . ٠ 7 • 9 ۰,7 . ~ 7 .

- 2 -READ DATA FROM DISK ON TAPE AND PRINT RESULTS. Ē BEAL & (30,60), TE (30,60), TI (30,60), H(30,60), PHI (31) · 1, V (30, 60) , P (120) , B (30) , A (30) , PRESS (30) , NO, LO, B (30, 60) , B (30, 60) INTEGER I, S, IAIN BATTE (6,621) 629 POBRAT (' IMPOT RECO, RECE, AND SKIP BATE'/' 8.7 1 8 848 (5,501) #8CO, #RCH, #54 501 FORMAT (315) 101 BEAD (3, END=102) FTAPE, STINE, T, H, NY, LO, RO, NO, TO, DT, NR, YLAS, YHIN - BEAD(3) A 8 EAD (3) 78 8 EAD (3) 75 2 XAD (3) 8 BEAD (3) PHI E ADD (3) T TEAD (3) P E EAD (3) A ,∗ **¥∰** 's · • • ÷ . ب BEAD(3) W IP (BELAS. #2. BBCO) GO TO 101 "BCO-BBCO-BSO IP(BTIBL.GT. BER) GO TO 102 WRITE(7,622) WTINE, T, DT, MR DT=', 89.2, ' #R=', OP[3] 622 FORMAT('-HTINE=',I4,' TINE-',1989.2,' WRITE(7,623) H, NI, WTAPE,LO,RO, BO,TO 623 POBMAT(\* H=\*,I3, \* HI=\*,I3, \* HTAPE,\*,I3, \* LO=\*, 1PE9.2, \* RO=\* 1,89.2, \* HO=\*,89.2, \* TO=\*,89.2) ¥¥X=60 BRII=INIB 83=12 ##\$=1 HHI=1 DO 13 X-BHIL, NHX DO 13 S-MAI, NO, NAS ¥ (S, X) =SQRT (¥ (S, X) ) . 13 U(S,X) = U(S,X) / H(S,X) 609 PORMAT (\* 1, 1 P20E 10.2) WRITE(7,604) 604 FORBAT (\* 0 787) DO 2 X=NHXI, NHX 2 WRITE(7,603) (TE(S,I), S-MAI, MA, AMS) WRITE(7,605) 605 PORAT('O 1I') DO 3 X-BENI, NEX 3 #RITE(7,603) (TI(S,X),S=BHI,HE,HES) WHITE (7, 606) 606 PORBAT (\* 0 811 DO 4 X-BEEL, MEX 4 HELTE(7,603) (#(S,X),S=BHI,RE,EES) BRITR(7,607) **T**') 607 PORPAT (\*\* BO S X-BUXI, BUX 5 28372(7,603) (Y (S, I), S- BHI, HA, HAS) WRITE(7,608) 77 \*> 608 PORAT (40 DO 6 I-BHXI, SHI 6 HRITE(7,603) (U(S,X),S=HHI,HH,HHS)

1

5 4

, ن≜

And a subscription of the second

•

•

1.144 

۲

-

4

•

Ý

.

265

1

.

• •

103 RETURN -239 1

>

PLOT BEAS INTENSITIES VS/ X AT VARIOUS TIMES. BEARS ARE BEAT PREQUENCT SIIMD & COLLISIONALLY DAMPED. BICHARD D. AILROY SEPT. 13, 1977. С . С ۰ ۲ С ۲ C+++++++ ESAL TE(1900), TI(1000), E(1000), L1(1000), L2(1000), E1(1000) 1, I2(1000), K1(1000), K2(1000), K3(1000), K4(1000), LLEDA, L0, LP, LT, H0 2, K10, L40, L20, I 10, I20, I01(1000), I02(1000) #0=4,03+16 L10=10.6 120-10.615 TE0=100.0 TI0=100.0 I 10= 1.0E+11 I 20= 1.0E+10 1P=900 #PLTS=3 L0=100.0 LP=20.0 LT=LO+LP DX=LT/(81-1) ##1=##+ (LO/LT) ## 1# 1=##1+1 3P21=32-1 C----- FIND K1. LLEDA=ALOG (ARIH1 (2.3\*780\*\*1.5,12.0\*780)) K10=8.67E-38+80++2+L10++2+LLEDA/TE0++1.5 C----- INITIALISE SOME ARRAYS. ۰, DO 1 I=1,#P 8 TE(I) = TE0 TI (I) =TIO H (I) = HO-L1(I)=L10 L2(I)=L20 1 CONTINUE DO 2 I=1, #P1 X= (#P1P1-I) + DX K1 (I) = K10 K3 (I) = K10+ (L20/L10) ++2 I1(I)=I10+BIP(-K10+X) 12(1)=1.0 2 CONTINUE DO 3 I-MP1P1,MP K1(I)=0.0 X2(I)=0.0 K3(I)=0.0 **F**. K4 (I) =0.0 I1(I)=I10 12(1)=1.0 3 COPTINGE 12(1)=120 DO 9 I=1,88 IO1(I) =I1(I) 9 102(I)=12(I) -- PIND R2 AND K4. C---CALL BPER (L1, L278, TE, TI, 2, K2) DO 4 1=1,821  $K_2(E) = K_2(1)$   $K_4(E) = K_2(1) + (L10/L20)$ 

.

.

1

1

4 CONTINUE C----- INITIALISE PLOTTING AND BRAN ANIS. CALL PLINT (LT. DI. MP) STOTAL-SP-2 HPLT=(HTOTAL/HPLTS)-1 C----- CALCULATE INTENSITIES. DO 5 J=1, HTOTAL DO 6 I=1, HP#1 Z=0.5+(K1(1)+K2(I)+I02(I))+DI 6 I1 (I) = I01 (I+ 1) + (1-I) / (1+I) DO 7 1=2,87 3=0. 5+ (K3 (I) -K4 (I) +I01 (1) +DX 7 12(1)=102(1-1)+(1-2)/(1+2) DO 8 I=1, #7 IO1(I)=I1(I) 8 102(I)=I2(I) C----- PLOT IP DESIRED. IF(((J-1)/HPLT)+BPLT.EQ.(J-1)) CALL PLT(I1,I2) 5 CONTINUE CALL PLOT (0.0,0.0,999) STOP END SUBBOUTINE PLINT (LT, DI, NP) RALLET, I1 (1000) , I2(1000) , X (1000) , LININ, LINAX, LI1 (1000) , LI2(1000) BHESEC (2) /0, 1000000/ 7 ALIBINS.O LINAX-14.0 CALL PLOTS CALL BRASE (61) CALL THAIT (0, OMESEC) CALL PLOT (20.0, 10.0,-3) P=AHIH1 (30.0/XLEH, 20.0/YLEH) CALL FACTOR (7) GO TO 2 1 CALL PLOT (3.0,3.0,-3) 2 CONTINUE C----- DRAN THE AXIS. DELTLO=LIBIH-LIHAX CALL LOBAX (0.0,0.0,'I', 1,YLEN, 90.0, LININ, DELTLG, 0.0) AXDELT=LT/XLE# CALL AXIS2 (0.0,0.0, 'X',-1, XLEN,0.0,0.0, AXDELT, 1.0) CALL PLOT (0.0, YL BB, 3) CALL PLOT (XLEN, TLEN, 2) CALL PLOT (XLEN, 0.0,2) C----- SET UP I - VECTOR. DO 3 I=1,#P 3 I (I) = (I-1) + DI60 20 999 BATRY PLT(I1,I2) --- TAKE LOG OF INTENSITIES. C---DO 4 I=1,8P LI1(I) = ALOG 10 (I1(I)) LI2 (I) =ALOG10 (I2 (I)) IP (LI1 (I) . LT.LININ) LI1 (I) -LININ IF (LI2 (I) . LT. LININ) LI2 (I) -LININ IF (LII(L).GT.LINAX) LII(I) = LINAX IF (LIZ (I) .GT.LIHAX) LIZ (I) =LIMAX 4 CONTINUE 2

(،

٠.

1

J

267

2
LI1(NP+1) - LININ LI2(NP+1) - LININ LI1(89+2) - (LIULI-LISIM/ILSU\* L12(10+2) - L11(10+2) I (BP+1) = 0.0 I (BP+2) = 12/1188 CALL LINE (I,LI2, BP,1,0,0) CALL LINE (I,LI1, BP,1,0,0) 777 1.87Vbu 212 SUBSOUTIES BYAR (L1, L2, 8, TE, TI, SP, ALPEA) CALCULATE BEAT PERQUENCY AIXING BATES. C . С POTH REACTION & ION NOTION IS ACCOUNTED FOR. . Ĉ C SLECTRON - ION COLLISIONS ARE FULLY ACCOUNTED FOR. POUDEBONDTIVE PORCE ON BOTH BLECTBORS AND IONS. ۰ AUGUST, 14, 1977. C BICHARD D. HILBOT **C** • ..... COMPLEX J/ (0.0, 1.0)/, KE (1000), KI (1000), EP (1000), EDE (1000) 1, ADI (1000), ECE (1000), ECI (1000), JE (1000), JI (1000), X, B (1000) EELL L1 (1000), L2 (1000), H (1000), TE (1000), TE (1000), EJ (1000), KJ (1000) 1, FI (1000), EZ (1000), ALDEM (1000), KI, EZ C=3. 000 10 10 13 AND K3. DO 1 1=1, 3P 5X-1 ¥1(I)=1.8658+15/(L1(I)) #2(1) = 1. 8858+15/(L2(1)) ¥3 (I) =#1 (I) -#2 (I) 172-5, 638+4+ 5027 (I (I )) 19 (UPE. 62. U2 (1)) 60 10 8 K1=5005 (U1 (L) \*\*2-UPE\*\*2) /C # 3-SQRT (#2 (E) ++2-17E++2) /C 1 K3(I)=K1+K2 60 TO 9 & JP-KK-1 -- FIND LINEAR SUSCEPTABILITIES. 9 CALL LSUS (N3, K3, N, TE, TT, MP, KB, KI) ---- CALCULATE COLLISIONAL CORRECTION PACTOR. DO 7 I=1,88 ¥EI=2.53E-05+# (I) / (TE (I) ↔ 1.5) NPB-5, 632+4+ SQRT (N (I)) B(I)=1,0-J+V3(I)+VEL/(WPE++2) KE (I) = B (I) \* KE (I) KI (I) =B (I) +KI (I) 7 CONTINUE C----- SPECIFY DIBLECTRIC CONSTANT EP. DO 2 I=1,87. 2 EP (I)=1.0+KE (I)+KI (I) ----- SPECIPY- POUDE BONOTIVE 'PIELDS'. 708 I1=12=1.0 90 3 I=1,8P X=KJ (I) +L1 (I) +L2 (I) +J BBE(1)=3,118-16+1 0 3 BAL(1) -1.692-19+1 - SPECIFI CORLORD FLELDS # 6 1-1, ## #CE(1) --KB(1) \* (ED2(1)\* (3.04 D6 4 1-1, 1P 4 BCI(I) =- HI(I) + (BDI(I) + (1.04 C-- SPECIFY BLECTBON AND TON CURRENT DO 5 1=1,88

268

`• **1** 

\$

• 3

```
203063(1)/12.566
(1)20001=(1)26
(1)20001=(1)26
(1)20001=(1)26
                                                                                 ٠
            DERCEPT DEAT THE
                                                   e lates.
        10 6 2+1, SP
        1-2.00-7-81(1)/43(1)
                                            (JI(I)))
        A318 - X-88AL (802 (1)
        AJSE - IMEAL (BCI (I)
                                            (JE(I)))
                                         AJ1I = 1+REAL (EDI (E
AJ2I = 1+REAL (ECE (E
                                        Si (JI (II))
        A38 - A318+A228
        AJI = AJ1I+A321
        ALPHA (I) = A 30+A 31
HRITE(4, 601) L2 (I), A 318, A 328, A 38, A 311, A 321, A 31, ALPHA (I)
С
   601 PORLAT (* *, 198812. 4)
     6 CONTINUE
        152989
                         ¥
        233
        C4
       . CALCULATE LINEAR RESCEPTABILITIES FOR BOTH ELECTRONS & PARTONS.
C
      CONFLEX 42(1000), KI (1000), J/ (0.0,1.0)/, F0
BEAL U3 (1000), KI (1000), J/ (0.0,1.0)/, F0
BEAL U3 (1000), KI (1000), J (1000), TE (1000), TI (1000)
С
                                                                                          .....
C4
        -- 7189 KL.
C
        1 1 1-1, 1P
        #P-5.638+4+SQET (# (I))
        10-5.938+7+8981(TR (I))
        A0-#3(I)/(#3(I)+70)
        IF (10.67.50.0) 60 20 A
EE (I) = 2.'0* (WP* 10/W3 (I) + 2* (1.0+ 3* 10* F# (10))
        40 TO 1
      3 KE(I) -- (WP/W3(I)) ++2
      1 CONTINUE
        -- 7100 XI.
C
        DO 2 1=1,87
        ¥P=1.315+3+8987 (# (1))
        TO=1.388+6+SQRT (TI (I))
        10-#3 (I) / (E3 (I) + TO)
         IF (10.GT.50.0) 80 70 4
        XI (I) =2. 0+ (# P+ A0/#3(I) ) ++24(1.0+J+ A0+P0 (A0))
        60 TO 2
      4 KI (I) -- (##/#3(I)) ++2
      2 CONTINUE
         STTU BE
         IID.
        COMPLEY PURCTION FO(A0)
         -- CALCULATE PO AS DEFISED ON PAGE 178 OF STIL.
C---
        COMPLEX J/(0:0,1.0)/
REAL POSPL(12,5)
       ATA POSPL/0.2,0.5,0.8,1.0,1.2,1.4,1.7,2.0,2.5,3.0,3.5,4.0,
1.3895,0.7176,1.064,1.076,1.015,0.913,0.7451,0.6027,0.4462,0.3565
2,0.2992,0.2567,1.003,1.299,0.5478,-0.2165,-0.4162,-0.5635,-0.442,
       3-0.0110,-0.2316,-0.1393,-0.0931,0.0,0.0.0.0.967,-3.4911,-0.3204;
       4-0.6701,-0.06606,0.1709,0.231,0.1287,0.55938-1,0.03642,0.0,1.097
5,-4.976,5.272,-0.5696,1.007,0.2633,0.06673,-0.06821,-0.0485,
6-0.01301,-0.02426,0.0/
         103-10
         IF (AGE. 61. 10.) AOD- 10.0
```

...

ni.

Danie - Pacfill, 1/ Princes FL (1, 3) + Pacfil (1, 3) + EL+ Pacfil (1, 4) + EL+ EL+ FU Statistics

12 7945664 (1.0 -16.5/ (20-60) -0.75/ (20-40) - 1. 670/ (20-40) - 4. 570/ (20-40) ) 00 90 11

6

÷,

90 96-2.04J0 (18-8.066678/2010 20 443+.266667810+45-.076798899977 00 90 11

11 81.00-1.778-830 (- 1010) 74-76-8400

د

121420

.

SINULATE INTERACTION OF TWO ANTIPARALLEL BEARS IN A PLASMA . C · С . HITE A DIFFERENCE PRODENCY OF THE ORDER OF THE IOS-ACOUSTIC . С . . С ۰ С . DEVELOPED BY RICHARD &. HILROY. OCTOBER 24 1977. . C+++ COMPLEX +16 D(100), A1(100), A2(100), H(100), K1, K2, A1BHD, CPT, CDT2 3 1, CW3, CTEAX, J/ (0. 0D0, 1. 0D0) /, T, CKIB1, CKIB2 COMPLEX CHPLY REAL BO, LD1, LD2, K3, KA1, KA2, KIB1, KIB2, LLADA C----- SPECIPY INDEPENDENT VARIABLES. 10=2.58+17 0 TE0=160.0 TI0=120.0 LD1=10.2470 ٧ LD2=10.2605 ZAAI=80.0 -----¥54X=6.08+11 PLEN=3.0 3 XP=99. • 1 BIST=41 #P¥=900 BDT=10 #F#T-30 HINP=5 BREST=0 P1=1.8E+10 P21 = 1.0P22 = 1.210 P23 = 1.0P24 = 1.0 P25 = 1.0 221 - 0.0 222 = 1.5223 = 3.0324 = 4.0 225 = 5.0 C=3.0E+10 C----- CALCULATE SOME DEPENDENT VARIABLES. #3=1.885E+15+(1./LD1-1./LD2) DI=PLES/(SP-1) #1=1.885E+15/LD1 #2=1.885#+15/LD2 HP=5.632+04= SQRT (80) C1 = C+SQRT(1.0-(WP/W1)++2) DT = DZ/(2.0+C1)DT2 = 2.0+DT THAX = ZHAX/C1  $K\lambda 1 = SQRT(H1++2-HP++2)/C$ KA2 = SURT (#1++2-#P++2)/C \*3 × KA1 + KA2 T= (0.0D0,0.0D0) STINE=0 C----- CALCULATE INVERSE BREMSSTRAHLUNG COEFICIENTS. LLHDA = ALOG (12.0+TEO) IP (TEO. LT. 27.0) LLBDA = ALOG (2.3+T #0++1.5) GAN = 2.9E-06+N0+LLHDA/ (TE0++1.5) KIB1 = GAH+#P++2/(2.0+#1++2)

J

EIB2 - GAL+HP++2/(2.0+824+2) // -- FREPARE TO CALL PCOR - THE MAIN SHEROWILLE C CALL 987 (9, 9197, 9445, 63, 682, 520, 210, 10, 828) 9 11 = SQAT (6. 28327/(# 1+KA 1) 12 = SQRT (6.20387/(#2\*KA2)) DO 1 I=1,88 # (I) = (0.000, 0.000) A1(1) = 11+5 (AT(P1) + E12(-KIB1+ (THAT)) 1 CONTINUE -- SPECIFY INITIAL PROFILE FOR PULSE. C---DO 2 I=1,#P -Z = (I-1)+DI IF(2.GT.122) GO TO 3  $\lambda_2(I) = P21 + ((I-I21)/(I22-I21)) + (P22-P21)$ GO TO 6 3 CONTINUE IP(1.GT.123) GO TO 4 A2(I) = P22 + ((1-122)/(1-1-12)) + (P23-P22) GO TO 6 • A CONTINUE IF (2.GT. 224) 60 10 5 A2(I) = P23 + (41-123)/(124-123))\*(P24-P23) GO TO 6 5 CONTINUE IF(2.GT.125) STOP  $\lambda 2(1) = P24 + ((1-124)/(125-124)) + (P25-P24)$ 60 TO 6 A2(I) = I2+CDSQRT(A2(I)) 2 CONTINUE IF WEEST. .... ICALL REST (MREST, WTIME, T, A1, A2, N) A188D = X1+SQRT(P1) K1=-J+#P++2/(2.0+#1+#0) K2=#1+K1/#2 CW3=CHPLX(W3,0\_0) 2 CTEAX=CEPLX (THAI,Q.Q) CDT=CHPLI(DT,0.0) CUT2=CEPLX (DT2,0.0) CKIR1 = CRPLI(KIB1,0.0) CAIJ2 = CHPLI(KIB2,0.0) CALL PCOB (A1, A2, H, D, CDT, CDT2, HDT, HINT, HP, K1, K2, A1BHD, CW3, CTHAX 1, W1, W2, MPRT, MTAP, T, MTIBE, KA1, KA2, CKIB1, CKIB2) RETURN ZED SUBROUTINE PCON (A1, A2, E, D, DT, DT2, FDT, MINT, NP, K1, K2, ABSD 1, W3, THAX, W1, H2, HPBT, HTAP, T, WTIBE, KA1, KA2, KIB1, KIB2) C \* \* \* ..... С . CALCULATE BEAH COMPRESSION DUE TO TWO ANTIPARALLEL BLECTROHAGHETIC BEAMS WITH A DIFFERENCE FREQUENCY С . С CLOSE TO THE MATURAL ION-ACOUSTIC FREQUENCY. С TRANSIENT GENAVIOUR IS INCLUDED. С RICHARD D. BILROY OCTOBER 23, 1977. ۰ ٠ C \*\* ISPLICIT COMPLEX#16 (A-H,O-Z) COMPLEX+ 16 A1 (100) , A2 (100) , B (100) , D (100) , K1, K2 1, A10 (100), F (100), IJ/ (0.0D0, 1.0D0)/, KIB1, KIB2 REAL W1,W2,KA1,KA2 BDT1=BDT EDT=1

DO 9 I=1,#P

٤

)

12.4

۹

272

1

١.

ø

9 + 10(I) = 1(I)1 CONTINUE AIBID = CDEXP(«KIBA+ (THAX-T))+ABBD stiss=stiss+1 IF (NTINE.GT. NP+2) NDT-NDT1 DTN-DT+NDT T=T+DT# C----- ADVANCE A WBY DT - NDT TIERS. A1(1) = A1889 DO 3 J=1, NDT ATTRI=ATEP DO 2 1=2,#P IN1=I-1 AII=AI(I) A1(I) = A TIR 1+ DT+K1+ (# (IR 1) + A2 (IE1) + #(I) + A2(I)) - DT+KIB1+A1IE1 ATTH1=ATT 2 CONTINUE 3 CONTINUE -- ADVANCE A2 BY DTN. " ## = HIBO((HIINE+1)/2,#P) DO 4 I=1,88 A2(I)=A2(I)+DTH+K2+BCOBJG(H(I))+A10(I)-DTH+KIB2+A2(I) 4 CONTINUE C----- BAKE OLD AT EQUAL TO NEW AT-1 DO 5 I=1,#P 1 A10(I)=A1(I) 5 CONTINUE C----- BECALCULATE # (I). H H=SINO (NTIBE/2, NP) IF (BH. LT. 2) GO TO 8 DO 7 I=2,## HEN-MINO (I, HINT) DO 6 J=1,888 F (J) = A1 (I-J+1) + DCOBJG (A2 (I-J+1)) +D (J) +CDEXP(IJ+83+(J-1)+DT2)+.3989 6 CONTINUE CALL SIBP (ANS, P, BHB, DT2) B(I)=ABS 7 CONTINUE 8 CONTINUE IF ( (HTIHE. EQ. 1) . OB. ( ( (HTIBE/HTAP) + HTAP) . EQ. HTIHE) ) 1WBITE(3) T, BTIME, SP, DT2, A1, A2, S, W1, W2, KA1, KA2 IF((NTIME.EQ.1).OR. (((WTIME/WPAT) \* WPAT) . EQ. WTIME)) ICALL BITE (A1, A2, N, HTINE, NP, T, E1, N2, KA1, KA2) IP (CDABS (T) . LT. CDABS (TBAX) ) GO TO 1 CALL BITB (A1, A2, N, WTIBE, NP, T, W1, W2, KA1, KA2) RETURN END SUBROUTINE DET (D, HINT, WHAX, K3, DT, TE, TI, H, H PU) ............ C\*\*\*\*\*\*\*\*\*\* D (K, T) -CALCULATE D; · • RICHARD D. HILROY OCTOBER 24, 1977. . COMPLEX+16 D (100), ANS COMPLEX DW (1000) , P (1000) , J/(0.0, 1.0) / REAL 23(1000), T(100), K3,8 C---- CALCULATE DW; D(K,W).

DO 1 I=1,8P8 #3 (I) =- #HAX+ DEL#+ (I-1) IF (ABS (#3 (I)). LT. 1.0 E+08) W3 (I) = 1.0 E+08

DELH=2.0+WRA X/ (#PH-1)

<u>\_</u>

1

C-

С

С

CONTINUE CALL DEV (NPN, 13, 13, TE, TI, MAN) - CALCULATE D; D(K,T). C DO 2 JJ=1, SINT TT=DT+ (JJ-1) Ť (JJ) =ŤŤ DO 3 I=1,8P8 WT=WJ(I) +TT P(I)=0.3989+DU(I)+CEXP(-J+UT) 3 CONTINUE CALL SIMPI (ANS, P, NPN, DELN) D(JJ)=AMS 2 CONTINUE D(1) = (0.000, 0.000)C----- BAKE PLOTS OF DU AND D IF DESIED. CALL PLT (#3, T, DW, D, #PW, MINT) RETURN END SUBROUTINE DEN (NP. N3.K3.TE.TI.N.DE) C\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\* D (K, N) CALCULATE DE; • С C • BICHARD D. MILBOY OC OCTOBER 24, 1977. COSPLEX DE (1000), EN 1000), EI (1000), B, J/(0.0, 1.0)/,C REAL N3 (1000), N, K3 --- FIND LINEAR SUSCEPTADULITISS. C-CALL LSUS (W3,K3,W,TE,TI, W, KE,KI) ---- CALCULATE DW. C-YEL=2. 538-05+8/(TE+-1. 5) ¥P #= 5. 63 E+04 +5Q #T (#) C=-K3++2+9.716E+04 DO 1 I=1, #P B=1.0-J+W3(1)+VEI/(VPE++2) DW(I)=C\*KE(I)\*(1.0+B\*KI(I))/(1.0+B\*KE(I)+B\*KI(I)) 1 CONTINUE RETURN IND SUBBOUTINE LSUS (W3,K3,W,TE,TI,NP,KE,KI) .......... C. . CALCULATE LINEAR SUSCEPTABILITIES FOR BOTH BLECTRONS & PROTONS. С AUGUST, 14, 1977. ٠ RICHARD D. HILBOT С C\*\*\*\*\*\*\* .... COMPLEX KE (1000), KI (1000), J/ (0.0,1.0)/, PO REAL #3(1000), K3, # C----- FIND KR. ¥P=5.63E+4+SQRT(#) 10=5.932+7+50RT (TE) DO 1 I=1,#P A0=#3(I)/(K3+VO) IF (A0.GT.50.0) GO TO 3 KE(I)=2.0+(WP+A0/W3(I))++2+(1.0+J+A0+F0(A0)) GO TO 1 3 KE(I) =- (MP/W3(I)) \*\*2 1 CONTINUE C---- FIND KI. WP=1.31E+3+SQRT(#) TO=1.382+6+50RT(TI) DO 2 I=1,#P A0=#3(I)/(K3+¥0) IF (A0.GT.50.0) GO TO 4

KI(I) = 2.0 + (WP+A0/W3(I)) + +2 + (1.0 + J + A0 + P0(A0))GO TO 2 4 KI (I) =- (WP/W3(I)) ++2 2 CONTINUE RETURN ٩ . 2 I D COMPLEX PUNCTION PO(AAO) C----- CALCUNTE FO AS DEFINED ON PAGE 178 OF STIX. COMPLEX J/(0.0,1.0)/ REAL PUSPL (12,5) DATA POSPL/0.2,0.5,0.8,1.0,1.2,1.4,1.7,2.0,2.5,3.0,3.5,4.0,  $1 \quad .3 \\ \{95, 0.7199, 1.064, 1.076, 1.015, 0.913, 0.7451, 0.6027, 0.4462, 0.3565, 0.2992, 0.2587, 1.003, 1.299, 0.5474, -0.2165, -0.4162, -0.5635, -0.532, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.5635, -0.532, -0.54162, -0.54162, -0.5635, -0.532, -0.532, -0.54162, -0.54162, -0.54162, -0.54162, -0.54162, -0.54162, -0.54162, -0.54162, -0.54162, -0.54162, -0.54162, -0.54162, -0.54162, -0.54162, -0.54162, -0.55162, -0.54162, -0.55162, -0.54162, -0.55162, -0.54162, -0.54162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.55162, -0.5516$ 3-0.4114,-0.2316,-0.1393,-0.0931,0.0,0.0,0.907,-3.4911,-0.3244, 4-0.6701,-0.06606,0,1709,0.231,0.1287,0.5593E-1,0.03642,0.0,1.097 5, -4. 976 , 5. 272, -0. 56 96, 1. 007, 0. 263 3, 0. 04673, -0. 06821, -0. 0485, 6-0.01301,-0.02428,0.0/  $\lambda 0 = \lambda BS(\lambda A 0)$ AOE=AO IP (AOE. GT. 10.) AOZ- 10.0 AOES=AOE+AOE SPLINE APPROXIMATION FOR FO С IP (AO. LT. 0.21) GO TO 90 IT (A0: GT. 3.25) GO TO 12 DO7 I=1,12 IF (AU. LT. POSPL (I, 1)) GO TO 8 CONTINUE 7 I=I-1 8 DA=A0-FOSPL(I,1) F0=F0SPL(I,2)+F0SPL(I,3)+DA+F0SPL(I,4)+DA+F0SPL(I,6)+DA++3 20=1+20 GO TO 11 12 f0=J/A0\* (1.0+0.5/(A0\*A0)+0.75/(A0\*\*4)+1.875/(A0\*\*6)+6.563/(A0\*\*8)) GO TO 11 90 F0=2.0+J+(10-0.666667+10++3+.266667+10++5-.0761905+10++7) GO TO 11 11 DAMF=1.772+E XP (- AO ES) P0 = (AA0/A0)+P0 PO=PO+DAMP RETURN END SUBROUTINE SIMP (ANS, F. H. DT) C++++++++ INTEGRATE P USING SIMPSON'S RULE. С ۰ c、 . IF # IS BOT ODD INTEGRATE TO #-1 AND ADD RENAINDER. . RICHARD D. HILBOY OCTOBER 23, 1977. С . IMPLICIT COMPLEX+16 (A-H, 0-Z) COMPLEX+16 P (100) IF (N. EQ. 2) GO TO 1 GO TO 2 1 A#S= (0.5D0,0.0D0) + D1+ (P(1)+P(2)) GO TO 6 2 JTEST=1 IF(((#/2)+2).EQ.#) #TEST=2 #1=# IP (NTEST. EQ. 2) N 1=N-1 ۹. ##1=#1-1 ##2=#1-2 71=7(1)+7(81)

.

275

ંબ

٩.

• \_ ^ , ,

```
F2= (0. 000, 0. 000)
      73= (0.000,0.000)
      DO 3 1=2,881,2
    3 72=72+7 (I)
      DO 4 I=3,842,2
    4 73=73+7(1)
      A#S= (DT/(3.000,0.000)) + (P1+ (4.000,0.000) +P2+ (2.000,0.000) +P3)
      GO TO (6,5), NTEST
    5 A #S=A#S+ (F (# 1) +F (#)) +DT+ (0.500,0.000)
    6 RETURN
      E I D
      SJBROUTINE SINPI (ANS, P.N., DT)
С
     ٠
           INTEGRATE P USING SIMPSON'S RULE.
           IF # IS NOT ODD INTEGRATE TO #-1 AND ADD REMAINDER.
С
     ٠
C
     .
              RICHARD D. BILBOY OCTOBER 23, 1977.
C **
   *****************
      COMPLEX#16 ANS
CERLEX F(1000), F1, F2, F3
IF(N.EQ. 2) GO TO 1
      GO TO 2
    1 \text{ A} = (0.5, 0.0) + DT + (P(1) + P(2))
      GO TO 6
    2 #TEST=1
      IF(((#/2)+2).EQ.W) #TEST=2
      11=1
      IP (#TEST.EQ. 2) # 1=#-1
      38 1-1
      ¥#2=#1-2
      21=2(1)+2(#1)
      P2=(0.0,0.0)
      F = (0.0, 0.0)
      DO 3 I=2, NN1,2
    3 22=22+2(1)
      DO 4 I=3, #82,2
    4 P3=P3+P(I)

A MS= (DT/(3.0,0.0)) * (P1+ (4.0,0.0) * P2+ (2.0,0.0) * P2)

G0 T0 (6,5), NTEST

G0 T0 (6,5), NTEST
    6 RETURN
      END
      SUBROUTINE REST (MREST, MTINE, T, A1, A2, H)
С
     .
            READ DATA FROM DISK TO ALLOW BOUTISE TO RESTART.
     •
                 RICHARD D. HILBOT
                                        OCTOBER 26, 1977.
Ċ
C**************
      COMPLEI+16 A1(100), A2(100), B(100), I, DT2
    IP (NTIME.NE. NREST) GO TO 1
      BACKSPACE 2
      READ(2) T, HTIME, NP, DT2, A1, A2, E, W1, W2
                                                                                  {
      ARTURS
  999 HRITE(6,601) HREST
  601 POBLAT ("1", "RECORD MREST=", IS, " IS NOT ON UNIT 2.")
      STOP
      ZND
      SUBROUTINE PLT (#3, T, DW, DT, MPW, MPT)
      COMPLEX+16 DT (100)
      COMPLEX DW (1000) ,122.
      REAL W3 (1000), T(100), RW (1000), IW (1000), IT (1000), F (2000)
```

V.

¢

```
ES #C (2) /0, 1000000/
         IIE
               .
                  5.
         YINT = 6.0
         CALL PLOTS
         CALL ORGI (ILEN, YLEN)
         CALL TERASE (61)
         CALL TWAIT (0, OHESEC)
       1 CONTINUE
         DO 2 I=1, HPW
         RW(I) = REAL(DW(I))
       2 IN(I) = AIMAG (DW (I))
                                     1
         DO 3 I= 1,8PT
         222=DT(I)
       3 IT(I) = BEAL (325)
45 -
  C----- SCALE QUANTITIES TO BE PLOTTED.
         DO 4 I=1, MPN
         2(I) = RH(I)
       4 P(I+MPM) = IM(I)
         1PH2 = 2+1PH
          CALL SCALE (N3, XINT, NPW, 1)
        CALL SCALE (T, XINT1, MPT, 1)
          CALL SCALE (P, YINT, NPN2, 1)
          CALL SCALE(IT, TINT, NPT, 1)
          #3 (#P#+2) = #3 (#P#+2) +XI#T/XLE#
          T (NPT+2) = T (NPT+2) *XINT 1/XLEN
          F (MPW2+2) = F (MPW2+2) * TINT/YLEM
          IT (NPT+2) = IT (NPT+2) *TINT/TLEN
          RH (NPN+1) = P (NPN2+1)
          IN (NPN+1) = P (NPN24.1)
          RW(MPW+2) = P(MPW2+2)
IW(MPW+2) = P(MPW2+2)
   C----- DRAN THE AXIS FOR PREQUENCY SPECTRUM PLOT.
          AXAIH = W3(HPW+1)
                                                 t
          AXDELT = #3 (#P#+2)
          DTIC = ILES/IINT
       · CALL AXIS2 (0.0,0.0, W', -1, XLEW, 0.0, AXHIW, AXDELT, DTIC)
          CALL PLOT (0.0, YLEN, 3)
          CALL PLOT (XLEN, YLEN, 2)
          CALL PLOT (XLEN,0.0,2)
           AXHIB = BH(BPW+1)
           AXDELT = RW(MPH+2)
           DTIC = -TLES/XINT
          CALL AXIS2 (0.0,0.0, 'D', 1, TLEN, 90.0, AXHIN, AXDELT, DTIC)
    C----- PLOTL THE PLOT.
           CALL LINE (N3, RH, MPN, 1,0,0)
           CALL LINE (WJ, IW, MPH, 1,0,0)
           CALL ORG (ILEN, TLEN)
         ---- DRAN AXIS FOR T PLOT.
    C----
           AXHIN = T(NPT+1)
           AXDELT = T(HPT+2)
           DTIC = ILEN/IINTI
           CALL AXIS2 (0.0,0.0, 'T', -1, XLEB, 0.0, AXHIN, AXDELT, DTIC)
           CALL PLOT (0.0, YLEN, 3)
           CALL PLOT (XLES, YLES, 2)
           CALL PLOT(ILEN,0.0,2)
                                                          Ň
           AIMIN = IT (NPT+1)
           AXDELT = IT (#PT+2)
```

4

...

DTIC = -YLEN/XINT CALL AXIS2(0.0,0.0,"D",1,YLEN,90.0,AXNIN,AXDELT,DTIC) C----- PLOT THE PLOT. CALL LINE(T, IT, MPT, 1,0,0) CALL PLOT (0.0,0.0,999) RETURN END SUBROTINE ORGI(XLEN, YLEN) C----- NOVE ORIGIN IN A WAY TO HIMIHIZE PLOTTER PAPER WASTE. C-----POR TEXT BOWIY, RE-SCALE PLOT AND HALT UNTILL RETURN BUTERED. CALL TCLEAR(61) CALL TERASE(61) P=AHIN1 (30.0/XLEW, 20 CALL PACTOB(P) CALL PACTOR(P) CALL PLOT (2.0, 2.0, -3) GO TO 10 1 CALL PLOT (3.0, 3.0, -3) X0=0.0 Y0=0.0 XL=XLEN YL=YLEB GO TO 10 ENTRY ORG (XLEN, TLEN) C----- ENTER HERE FOR ALL BUT 1'ST CALL TO THIS BOUTINE. CALL TOLEAR(62) E EAD (6,601) HOTEG 601 PORMAT (15) CALL TERASE (62) Ĵ GO TO 10 F VI=6.0+IL ¥Y=6.0+TL Y=YO+YY+YLEN YL=YLEB IP(Y.GT.25.0) GO TO 3 CALL PLOT (0.0, VT,-3) ¥0=¥0+¥¥ IP(XLEN.GT.XL) XL=XLEN GO TO 10 3 ¥1=-10 CALL PLOT (VX, VY,-3) XL=XLEN X0=X0+¥X Y0=0.0 10 RETURN END SUBROUTINE RITE (A1, A2, N, NTINE, NP, TINE, N1, N2, KA1, KA2) CUMPLEX+16 A1(100) , A2(100) , # (100) , TIME COMPLEX 11 REAL P1, P2, T, KA1, KA2 C=3.0E+10 T1=TIME T=BEAL (T1) WRITE(6,601) "MTIME,T 601 PORMAT ( 0', ' HTINE=', IS, ' T=', 1PE 10.2,' SEC.') MP1=MP #SKIP=10 DO 1 I=1, MP1, MSKIP P1 = #1+KA1+ (CDABS (A1(I)) ++2/6.2838+07 P2 = W2\*KA2\* (CDABS (A2(I))) \*\*2/6. 28 38+07 XH=CDABS(H(I))

.

•

# -

•

•

4

`

WRITE(6,602) I,P1,P2,IH 1 CONTINUE 602 FORMAT(\* \*,I5,1P3E10.2) RETURN RND

7

٠

.

279

•

.

```
PLOT THE OUTPUT FROM PCOA - PULSE COMPRESSION.
                                                                      .
С
     ٠
                                                                      ٠
                   RICHARD D. MILROY OCTOBER 26, 1977.
     ٠
С
......
     CONFLEX# 16 T, DT2, A1 (100) , A2 (100) ,# (100)
      CONPLEX 1
      BEAL P1(102) , P2(102) , B1(102) , I (102) , KA1, KA2
      XLEN = 3.5
      YLE# = 3.5
      IINT = 6.0
      LPHIN = 7
      LPHAX = 13
      LENIE = 12
      LEBAX = 17
      C = 3.05+10
      # EAD (3) T, #TINE, #P, DT2, A1, A2, #, # 1, #2, KA1, KA2
      BACKSPACE 3
      C1 = KA1/V1+C++2
      CALL PLOTS
      CALL ORGI(ILEF, TLEE)
      ALX=0.0
      YT4800
      TE-10.0
    1 CONTINUE
      CALL SETAL (ALX, ALT, 1, $ 100, $ 100)
  100 BRITE(6,601)
  601 FORMAT ('OIMPUT I')
      R EAD (5, 501) DIST
      IF (DIST.LT.0.0) GO TO 999
  501 FORMAT (#10.3)
      TPLT = DIST/C1
      IP (TPLT.LT.T.R) REWIND 3
C----- BEAD IN VALUE PRON DISK CLOSEST TO DESIRED PLOT TIME.
    2 TOLD=TR
      READ (3, END=8) T
      TR=CDABS(T)
      IF (TE.LT.TPLT) GO TO 2
      IF ((TPLT-TOLD) .GT. (TR-TPLT) ) GO TO 7
    8 BACKSPACE 3
    7 BACKSPACE 3
      BEAD (3) % T, HTIHE, NP, DT2, A 1, A 2, N, W 1, W2, KA 1, KA2
C----- DEFINE I - VECTOR.
      TE=CDABS(T)
      DT=CDABS (DT2)
      DZ=C1+DT
      DO 3 I=1,#P
      X(I) = (I-1) = DZ
    3 CONTINUE
C----- DEFINE T - VECTORS.
      DO 4 I=1,#P
      P1(I) = 81*KA1*(CDABS(A1(I))) **2/6.2838+07
      P2(I) = H2*KA2*(CDABS(A2(I))) **2/6.283#+07
      #1(I)=2.0+CDAB5(#(I)) +
                                1.0
    4 CONTINUE
C----- TAKE LOG OF QUANTITIES TO BE PLOTTED.
      DO 5 I=1,#₽
      P1(I) = ALOG10 (P1(I))
      P2(I)=ALOG10(P2(I))

P1(I)=ALOG10(H1(I))
                                                  -
```

280

`

÷

S CONTINUE --- HALE SUBS ALL QUARTITIES STAT UTTALE GRAPE ADDUBARIES. C--DO 6 1-1,8P IF(P1(1).LT.LPHIN) P1(1)-LPHIN IF(P1(I). CT. LPHAN) P1(I)=LPHAN IT(P2(I).LT.LPHIN) P2(I)-LPHIN IT(P2(I).GT.LPHAN) P2(I)-LPHAN IT(#1(I).LT.LBAIN) B1(I)-LBBIN IF (N1(I) .67. LBBAX) B1(I)-LBBAX 6 CONTINUE c------- SCALE THE VUCTORS TO BE PLOTTED. CALL SCALE (I, XINT, NP, 1) X (##+2) = X (##+2) \* XX #7/XLR# P1(87+1)-LPEIS P2(#P+1)=LPHIB #1(#P+1)=L##I# P1 (SP+2) = FLOAT (LPHAX-LPHIN) / YLEN P2 (#P+2) = P1 (#P+2) #1 (#P+2) = FLOAT (LHEAX-LURIE) /ILEE •-C----- PLOT THE AILS. AXHIH -X (NP+ 1) AIDELT=1 (MP+2) DTIC - XLEB/XINT CALL AXIS2 (0.0,0.0," I (CH) ", -6, XLEB, 0.0, AXAIB, AXDELT, DTIC) CALL PLOT (0.0, TLEN, 3) CALL PLOT (ILEN, TLEN, 2) DELTLE=FLOAT (LPHIN-LPHAI) AXBINI=FLOAT (LPHIN) CALL LOGAT (0.0,0.0, 'P (N/CH++2)', 11, TLEN, 90.0, AXRINI, DELTLG, 3.0) DILTLG=PLOAT (LURIN-LUSAI) AININI-PLOAT (LUNIN) CALL LOGAI (ILZE, 0.0, 'S', -1, TLEE, 9J.0, AIMINI, DELTLG, 3.0) C----- BRITE POSITION OF BEAMPRONT IN LOVER LEFT CORRER. DIST = C1+TE VI=0.0 VI-0.5 #T=0.1 CALL NUMBER (VI, VY, NT, DIST, 0.0, 1) C----- DRAN THE LINES. CALL LINE (X, P1, HP, 1,0, 1) CALL LIBE (X, P2, SP, 1,0, 1) CALL LINE (X, N1, NP, 1, 10, 3) C----- HOVE'ORIGIN UNLESS ON TEXTRONII. CALL ORG (ILES, TLES) GO TO 1 999 CALL PLOT (0.0.0.0.999) STOP END SUBROUTINE ONGI(ILEN, YLEN) INTEGER ONESEC (2) /0, 1000000/ CALL TELBAR (61) CALL TEBASE (61) CALL TWAIT (0, OURSEC) P=AMIN1 (30.0/ILRH, 20.0/ILRH) CALL PACTOR (P) CALL PLOT (3.0, 2.0, -3) GO TO 10 1 CALL PLOT(5.0,5.0,-3)

0.1 3

and the second states

٠

```
10-0.0
                   10-0.0
                   IL-ILBS
                   TL-TLES
       YL-YLES

GO TO 10

ENTRY ORG(ILEN,ILEN)

C-----ENTRE WERE FOR ALL DUT 1'ST CALL TO THIS DOUTINE.

CALL TCLEAR(62)

BEAD(6,601) NOTHE

601 PORMAT(IS)

CALL TRANSE(62)

CALL TRANSE(62)

CALL TRANSE(62)

CALL TRANSE(62)

CALL TRANSE(62)
               GO TO 10
2 VI-6.0+1L
VI-6.0+1L
.
                   T-TO-VT-TLES
                   TL=TLEB
IF(T.GT.20.0) GO TO 3
                   CALL PLOT(0.0, VT,-3)
       . ~
                   10-10+11
       4
                   IF (ILBR. GT. IL) IL-ILBS
                60 TO 10
3 VI-- IO
    .
                   CALL PLOT (V1, V1, -3)
XL=XL88
                   10=10+VI
                   Y0=0.0
              10 BETURN
                   213
                 • . .
```

,

٤

.

202

1

ð

•

\*\*

.

C....... С • WILL PLOT ALPEA VS. ANY VARIABLE. (DEPENDING OF 'COOS'). • C COOS . . . . . C . -+1 08 -1 PLOT ALPEA VS. TE. . C -+2 01 -2 PLOT ALPHA VS. TI. ۲ . -- 3 08 -3 PLOT ALPHA VS. 8. . С C ------. PLOT ALPEA VS. L2. (SEE TI-TE/L.) -- 5 08 -5 PLOT ALPUA VS. TE. С . . IF (CODE.LT.O) DO BED ALLS IS BADE. PLOTS OVELAT. С RICHARD &. BILMOT A060 ST, 15, 1977. C . **C++** BEAL L1(1000), L2(1000), B(1000), TE(1000), TI(1000), T(1000), T(1000) 1, ALPUA (1000) , L2MIU, L2MAI, NO, L20, L1 9, ND, L20, L2MIAP, L2MAIP, N (1000) 2, 31(1000),41(1000) INTEGER CODE, C1, ON ESSC (2) / 0, 100000 4/ STIRS-0 17-798 871-9P ILE8-3.5 1680-3.5 LTRAILS LTIBID-0 LTIBAI-4 LBAID-16 LIRA I-19 LUBIS-10 LUBA 1- 14 L2819-10.6 L2841=10.606 C----- LININP AND LINARP USED FOR I-ALIS LINITS HUEN CODE=4. L2HISP-10.6 128AIP-10.606 LYEIN-15 LTBAI=-10 LT28IN--16 LY28AX=-11 L10=9.6 . ALI-0.0 ALT-800.0 TED-FLOAT (LTERAI-LTERIN) /NP TID-FLOAT (LTIBAI-LTIRIN) /NP SD-PLOAT (LURAI-LURIS) / SP L20- (L28AX-L28I8)/## CALL PLOTS CALL ORGI MLES, TLEN CALL TERASE(64) ---- PAUSE ONE SECORD UNILE TERTPONIS SCREEN IS ERASED. C----CALL THAIT(0, ONESEC) 1 CONTINUE 17-571 NTIRE-NTIRE+1 ALT=ALT-60.0 . CALL SETAL (ALX, ALY, 1, 5 100, 5 100) 100 CONTINUE WRITE(6,601) 601 POMAT ('0', 'INPUT CODE 1-TE; 2-TI; 3-N; 4-L2; 5-T; 6-N;') READ (5,501) COBE 501 PORMAT (13)

.

Þ

IF (TABS (COOR) . 07.0) 00 TO 2 CALL PLOT (0.0,0.0,799) STOP ٤ 2 CONTINUE 502 POREAT (3810.3) 60 TO 16 1 12 WRITE(6,603) 603 PORMAT (' INPUT TRO, NO,, L20') 2 2AD (5,502) 120, 80, L20 GO TO 16 13 WRITE (8,604) 604 FORMAT (' 'IMPUT TEO, TIO, L20') READ (5,502) TEO, TIO, L20 60 TO 16 14 BEITE (6,605) 605 PORMAT (' INPUT TRO, TIO, NO') READ (5, 502) TEO, TIO, HO GO TO 16 15 BRITE(6,610) 610 POBRAT(\* INPUT NO, L20, TE/TI\*) READ (5,502) 80,120,8 16 CONTINUE C----- IF (CODE.GT. 0) BRASE SCREEN & RE-WRITE INITIAL CONDITIONS. IF(CODE.LT.V) GO TO 70 CALL TERASE (670) C----- PAUSE ONE SECOND WHILE TERTBONIX SCREEN IS BRASED. CALL TWAIT (0, ONESEC) ALY=740 CALL SETAL (ALI, ALY, 1, 670, 670) HRITE(6,601) WRITE(6,606) CODE 606 PORMAT(1 1,13) GO TO(61,62,63,64,65,64),C1 61 WRITE(6,602) WAITE (6,607) TIO, NO, L20 607 FORMAT (' ', F7.0, 1919.1,0PF8. 4) GO TO 70 62 BRITE(6,603) URITE(6,607) TEO, NO, L20 GO TO 70 63 HEITE(6,604), WRITE(6,608) TEO,TIO,L20 608 FORMAT(' ',2F7.0,F8.4) GO TO 70 64 JRITE(6,605) WRITE(6,609) TEO,TIO,HO 609 PORMAT(' ',2P7.0,1PE9.1) GO TO 70 65 WRITE(6,610) #-WRITE(6,611) #0,L20,R 611 PORMAT(' ', 17E9.1, 0Pr8.4,P6.2) 70 CONTINUE C----- BOVE ORIGIN AND PLOT NEW AXIS IF DESIRED. IF (CODE.LT.0) GO TO 3

٤.....

1.38

.

```
IF (HTIBE.SE. 1) CALL ORG (ILEE, TLEW)
      CALL AXPL(C1, XLEN, YLEN, LYNIN, LYNAX, LYRNIN, LYENAX, LYININ
      1, LTIMAX, LUNI N, LUNAX, LONINP, LONALP, EN MIN, LUNAX
     2, LY24IS, LY28AX)
    3 CONTINUE
C---
    ---- SPECIFT PLASEA PARAMETERS.
      IF((C1.EQ.1).OR. (C1.EQ.5)) GO TO 4
       DO 5 I=1,#P
    5 TE(1)=TE0
    4 CONTINUE
       IF ( (C1. EQ. 2) .OR. (C1. EQ. 5) ) GO TO 6
       DO 7 I=1, #P
    7 TI (1)=TIO
    6 CONTINUE
       IF (C1. EQ. 3) GO TO 8
       DO 9 I=1, NP
    9 ¥(I)=#0
    8 CONTINUE
       IF((C1.EQ.4).OR.(C1.EQ.6)) GO TO 10
       DO 20 I=1, #P
   20 L2(I)=L20
    10 CONTINUE
      DO 21 I=1, #P
   21 L1(I)=L10
C----- SPECIPY VARYING PARAMETER.
      GO TO (31,32,33,34,35,43),C1
   31 DO 36 I=1, NP
   36 TE(I)=10_0++ (LTEHIE+TED+I)
       GO TO 40
   32 DO 37 I=1,#P
   37 TI(I)=10.0++ (LTIHIH+TID+I)
       GO TO NO
   33 DO 38 I=1,#P
                          •
   38 # (I) =10.0** (LWHIW+WD*I)
       GO TO 40
   34 DO 39 I=1,#P
   39 L2(I)=L2HI#+L2D+I
       GO TO 40
   35 DO 42 I=1,8P
       TE(I) = 10_0++ (LTEHI#+TED+I)
   42 TI (I) =TE (I) / E
       GO TO 40
   43 DLW = PLOAT(LWHAI-LWHIN) / WP
       QO 44 I=1, #P
       W(I)=10.0++(LWHIH+DLH+I)
   44 L2(I) = 1.885E+15+L10/(1.885E+15-#(I)+L10)
   40 CONTINUE
C----- CALCULATE HIXING RATE (ALPHA).
      CALL BFBB (L1, L2, N, TE, TI, NP, ALPHA, NE, NI)
C---- SPECIPY Y-VECTOR FOR PLOTTING
       DO 41 I=1,#P
       Y(I)=ALOG10(ALPHA(I))
       IP(Y(I).GT.LYMAX) Y(I)=LYMAX
IP(Y(I).LT.LYMIN) Y(I)=LYMIN
       BE(I) = ALOG10 (BE(I))
       HI (I) = A LOG 10 (HI (I))
       IF (NE(I).GT.LT2HAX) NE(I)=LT2HAX
       IP (#B(I) . LT. LT2HIN) #B(I)=LT2HIN
       IP (NI (I) .GT.LY2MAX) NI (I) = LY2MAX
IP (NI (I) .LT.LY2MIN) NI (I) = LY2NIN
```

4

285

ſ

```
41 CONTINUE
      T(#P+1)=LTHIF
      I (NP+2) = FLOAT (LIBAI-LIBIN) /ILES
      BE(SP+1)=LY2BEE
      HI (HP+1)'=LY2HIH
      NE(NP+2) = PLOAT (LY2HAX-LY2NIN) /YLEN
      HI (NP+2) = PLOAT (LY2BAX-LY2EIN) /YLBB
C----- SPECIPT I-VECTOR FOR PLOTTING.
       GO TO (51,52,53,54,51,66),C1
   51 I (HP+1) =LTERIN
       X (HP+2) = PLOAT (LTER AX-LTERIN) /XLBH
       DO 56 I=1, NP
                                 -5
   56 X (I) = 1 LOG 10 (TE (I))
       GO TO 60
    52 X (#P+1) = LTINI#
       X (#P+2) = PLOAT (LTIM AX-LTIMIN) /XLEN
       DO 57 I=1,#P
    57 I (I) = A LOG 10 (TI (I))
       GO TO 60
    53 I (#P+1) = L#HI#
       X (HP+2) = PLOAT (SWHAX-LHHIH) /ILES
       DO 58 I=1, #P
    58 X (I) = ALOG 10 (# (I) )
       GO TO 60
    54 X (#P+1) = L28I #P
        X (MP+2) = (L2HAXP-L2HIMP) /XLBM
        DO 59 I=1,#P
    59 I(I)=L2(I)
        60 TO 60
     66 X(NP+1)=LNHIN
        X (NP+2) = PLOAT (LUMAX-LUMIN) /XLEN
        DO 67 I=1, 8P
    67 I(I) = ALOG10(V(I))
     60 CONTINUE
 C---- DRAN THE LINE.
        CALL LINE(X, Y, MP, 1,0,0)
        ICHREP = MP/20
        CALL LINE (I, NE, NP, 1, ICHREP, 1)
        CALL LINE (X, HI, NP, 1, ICHNEP, 3)
        CALL TOLEAR(61)
        GO TO 1
         END
        SUBROUTINE AXPL (C1, XLEN, YLEN, LYHIN, LYHAX, LTENIN, LTENAX
       1, LTININ, LTIBAX, LUBIN, LUBAX, L28INP, L28AXP, LVBIN, LUBAX
        2, LT2HIN, LT2HAX)
        REAL LININP, LINAXP
        INTEGER C1
 _C---- PLOT THE Y-AXIS.
         AXBIN=LYMIN
         DELTG=LYMIS-LYMAX
         CALL LOGAX (0.0,0.0, * ALPHA* , 5, YLEN, 90.0, AIHIN, DELTG, 3.0)
         AXHIN=LY2HIM
         DELTG=LY2MIH-LY2MAX
         CALL LOGAX («LEN, 0.0, "N", -1, YLEN, 90.0, AXHIN, DELTG, 3.0)
  C----- PLOT THE 1-AXIS.
         GO TO (1,2,3,4,1,6),C1
       1 AININ-LTERIN
         DELTG=LTEBIE-LTEMAX
         CALL LOGAX (0.0,0.0, 'TE', -2, XLEN, 0. 0, AXMIN, D2LTG, 0.0)
CALL LOGAX (0.0, YLEN, '', 1, XLEN, 0.0, AXMIN, D2LTG, 0.0)
```

```
. GO TO 10
    2 AIRIB-LTIMIN
      DELTG-LTININ-LTINAX
      CALL LOGAX (0.0,0.0, 'TI', -2, XLEH, 0. 0, AXHIH, DELTE, 0. 0)
      CALL LOGAI (0.0, TLEN, ' ', 1, TLEN, 0.0, AININ, DELTG, 0.0)
      GO TO 10
    3 AXREM=LUNIN
      DELTG=LEEIN-LEEAX
      CALL LOGAX (0.0,0.0, * * , - 2, XLEX, 0.0, AXHIN, DELTG, 0.0)
CALL LOGAX (0.0, YLEN, * * , 1, ILEN, 0.0, AXHIN, DELTG, 0.0)
      GO TO 10
                                                                    f
    4 AIBIN-LOBINP
      DELTG= (L2MAXP-L2MINP) /XLEM
      CALL AXIS2 (0.0,0.0, 'L2', -2, XLEB, 0. 0, AXHIN, DELTG, 1.0)
CALL AXIS2 (0.0, YLEB, '', 1, XLEB, 0.0, AXHIN, DELTG, 1.0)
      GU TO 10
    6 AXHIN=LUMIN
      DELTG=LWHIN-LWHAX
      CALL LOGAX (0.0,0.0, 'N',-1, XLEN, 0.0, AXHIN, DELTG, 0.0)
CALL LOGAX (0.0, YLEN, '', 1, ILEN, 0.0, AXHIN, DELTG, 0.0)
   10 RETURN
      EXD
      SUBROUTINE ORGI(ILEN,ILEN)
C------ BOVE ORIGIN IN A MAY TO BINIBLES PLOTTER PAPER WASTE.
C----- FOR TEXTRONIX, RE-SCALE PLOT.
      CALL TOLEAR(61)
      CALL PLOT (20.0, 10.0, -3)
      P=ABIN1 (30.0/ILEN, 20.0/ILEN)
      CALL PACTOR(P)
     - GO TO 10
    1 CALL PLOT (3.0, 3.0, -3)
       X0=0.0
       Y0=0.0
      XL=XLEN
       TL=YLEN
       GO TO 10
      ENTRY ORG (XLEN, YLEN)
CALL TOLEAR(62)
       GO TO 10
     2 VX=5.0+1L
       ¥¥=5.0+¥L
       Y=Y0+VY+YLEB
       YL=YLEN
       IF(Y.GT.25.0) GO TO 3
       CALL PLOT (0. 0, VY,- 3)
       Y0=Y0+VY
       IF (ILEN.GT.XL) XL= XLEN
       GO TO 10
     3 Y.Y =- YO
       CALL PLOT(VX,VY,-3)
       XL=XLEE
       IO=IO+VI
    10-0.0
       END
       SUBROUTINE BPHR (L1, L2, N, TE, TI, NP, A LPHA, NE, NI)
**********
          CALCULATE BEAT PREQUENCY HIXING RATES.
С
      ۰
          BOTH ELECTRON & ION BOTION IS ACCOUNTED FOR.
                                                                                  ٠
      ٠
С
```

ELECTRON - ION COLLISIONS ARE PULLE ACCOUNTED FOR. PONDERONOTIVE PORCE ON BOTH ELECTRONS AND LONG. C ۲ . С • RICHARD D. NILROY AUGUST, 14, 1977. • С C+4 COMPLEX J/ (0.0, 1.0) /, KE (1000) , KI (1000) , EP (1000) , EDE (1000) 1, EDI (1000), ECE (1000), ECI (1000), JE( 1000), JI (1000), I, B (1000) EELL L1 (1000), L2 (1000), E (1000), TE (1000), TI (1000), N3 (1000), X3 (1000) 1, 11 (1000) , 12 (1000) , ALPHA (1000) , K 1, K2, NE (1000) , NI (1000) C=3.02+10 --- FIND NJ AND KJ. \ C DO 1 I=1,#P KK=I ~#1(I)=1.8852+15/(L1(I)) #2(1)=1.8852+15/(L2(1)) W3 (I) = W1 (I) - #2 (I) #P#= 5. 638+4+ SQRT (# (I)) IP (WPE.GE.W2 (I)) GO TO 8 K1=SQRT (W1 (I) +2-WPE+2) /C K2=SQET (#2 (I) ++2-#PE++2) /C 1 K3(I) = K1 + K2GO TO 9 8 XP=KX-1 --- PIND LINEAR SUSCEPTABILITIES. C---9 CALL LSUS(#3, K3, H, TE, TI, HP, KE, KI) --- CALCULATE COLLISIONAL CORRECTION FACTOR. C---DO 7 I=1,#P VEI=2.53E-05+# (I) / (TE (I) ++ 1.5) HPE=5.638+4+ SQRT (# (I)) B(I) =1.0-J+W3(I) +TEL/(WPE+2) KE(I)=B(I)+KE(I) KI (I)=B (I) \*KI (I) 7 CONTINUE C----- SPECIFT DIELECTRIC CONSTANT BP. DO 2 I=1, #P 2 EP(I)=1.0+KE(I)+KI(I) ---- SPECIFY PONDEBONOTIVE 'FIELDS'. C---FOR I1=12=1.0 DO 3 I=1,#P NPE2 = 3.172+09+8(I) C1 = C\*SQRT(1.0-WPE2/(W1(I)++2))C2 = C+SQRT(1.0-WPR2/(W2(L)++2)) X=K3(I)+L1(I)+L2(I)+J+C/SQRT(C1+C2) EDE(I)=3.11E-16+X 3 EDI(I) -- 1.698-19\*X C----- SPECIPY COULONS FIELDS. DO 4 I=1,#P ECB(I) -- KB(I) + (EDB(I) + (1.0+KI(I)) - KI(I) + EDI(I)) / (BP(I) + B(I)) 4 ECI(I) =-KI(I) + (EDI(I) + (1.0+KE(I)) - KE(I) +EDK(I))/(EP(I)+B(I)) C----- SPECIFY ALECTRON AND ION CURRENTS. DO 5 I=1,#P x=J+#3(I)/12.566 JE(I)=I+ECE(I) 5 JI (I) = I + BCI (I) -- SPECIFY BEAT PREQUENCY HIXING BATES. C----DO 6 I=1,#P I=2.02-7+W1(I)/W3(I) A31E = X\*EEAL (EDE (I) \*CONJG (JE (I) )) A32E = X+REAL (BCI (I) +CONJG (JE (I))) A311 = I\*REAL (EDI (I) \*CONJG (JI (I))) A32I = X+REAL (ECE (I) +CORJG (JI (I))) A38 = A318+A328

288

÷\*

```
A31 = A811+A321
       ALPUA(I) = AJE+AJI
       JE (I) = IJ (I) +2. 0+CA DE (ECE (I) ) / (6. 838-09+8 (I) )
II (I) = IJ (I) +2. 0+CA DE (ECI (I) ) / (6. 038-09+8 (I) )
  ERITR(6,601) L2(1),A31E,A32E,A3E,A311,A321,A31,ALPHA(1)
601 FORBAT(' ', 1POE12.4)
С
     6 CONTINUE
       STTORN.
       209
       SUBROUTINE LSUS (83, K3, N, TE, TI, NP, KE, KI)
• CALCULATE LINEAR SUSCEPTABILITIES FOR BOTH ELECTRONS & PROTONS. •
• RICHARD D. HILROY AUGUST, 14, 1977. •
С
С
COMPLEX EE (1000), KI (1000), J/ (0.0,1.0)/, FO
       REAL W3 (1000) , K3 (1000) , N (1000) , TE ( 1000) , TE (1000)
    ---- FIND KE.
C-
       DO 1 I=1,#P
       #P+5.63E+4+5QET (# (I) )
       V0=5.93E+7+SQRT (TE (I))
       A0-W3(I)/(K3(I) 40)
IP(A0.GT.50.0) GO TO 3
       EE(I) =2.0+ (WP+A0/W3(I)) ++2+(1.0+J+A0+P0(A0))
       GO TO 1
     3 KE (I) =- (NP/N3(I)) ++2
     1 CONTINUE
       -- FIND KI.
      DO 2 I=1, NP
NP=1_318+3*SQRT(N(I))
       TO=1. 382+6+ SQRT (TI (I))
       A0=#3(1)/(K3(1)+TO)
       IF (A0.6T.50.0) GO TO 4
       KI(I) = 2.0 + (WP+A0/W3(I)) + 2 + (1.0 + J + A0 + P0(A0))
       GO TO 2
     4 KI(I)=-(WP/W3(I))++2
     2 CONTINUE
       RETURN
       2 ND
       COMPLEX PUNCTION PO(A0)
C----
      --- CALCULATE FO AS DEFINED ON PAGE 178 OF STIL.
       COMPLEX J/(0.0,1.0)/
       REAL FOSPL(12,5)
       DATA POSPL/0.2,0.5,0.8,1.0,1.2,1.4,1.7,2.0,2.5,3.0,3.5,4.0,
      1 .3895, 0.7199, 1.064, 1.076, 1.015, 0.913, 0.7451, 0.6027, 0.4462, 0.3565
     2, 0. 2992, 0. 25 87, 1. 003, 1. 299, 0. 5474, -0. 2165, -0. 4 162, -0. 5635, -0. 532,
      3-0.4114,-0.2316,-0.1393,-0.0931,0.0,0.0,0.987,-3.4911,-0.3284,
      4-0.6701,-0.06606,0.1709,0.231,0.1287,0.55938-1,0,03642,0.0,1.097
      5, -4.976 , 5.272, -0.5696, 1.007, 0.2633, 0.06673, -0.06821, -0.0485,
      6-0.01301,-0.02428,0.0/
       ADE=AO
       IF (AOE.GT. 10.) AOE= 10.0
       AUES=AUE+AUE
       SPLINE APPROXIMATION FOR PO
С
       IF (A0. LT.0.21) GO TO 90
IF (A0. GT. 3.25) GO TO 12
       DO7 I=1,12
       IF (A0. LT. POSPL (I, 1)) GO TO &
   7
       CONTINUE
       I=I-1
       DA-AO-POSPL(T. 1)
```

 $\langle \cdot \rangle$ 

· · · - 、

 $\begin{array}{c} PO=POSPL (I,2) + POSPL (I,3) + BA + POSPL (I,4) + DA + BA + POSPL (I,5) + BA + + 3 \\ PO=J=PO \\ GO TO 11 \\ 12 PO=J/A O + (1.0+0.5/(AO + AO) + 0.75/(AO + + 1.875/(AO + + 6) + 6.563/(AO + + 8)) \\ GO TO 11 \\ 90 PO=2.0+3+(AO - 0.666667 + AO + 3), 266667 + AO + 5 - .0761905 + AO + + 7) \\ GO TO 11 \end{array}$ GO TO 11 11 DARP=1.772\*EXP (-A0ES) ¢ • PU=PO+DAEP RETORN TND . -. -1 1

.

3

.

.

.

.

-

.

1.1

.

٠

-

.

.

٠.

.

• 7

• •

290

.

#### VITAE

Ĺ

MAME: Richard Douglas Milroy

1

## DATE AND PLACE OF BIRTH: March 20, 1951, Hardisty, Alberta, Canada

•

DEGREES:

B.Sc. Physics (Honors) Univ. of Alberta April, 1973 M.Sc. Elect. Eng. Univ. of Alberta March, 1976 Ph.D. Elect. Eng. Univ. of Alberta Sept., 1978 WORK EXPERIENCE: Geophysicist; Geophysical Services Inc. 640-12 Avenue, S.W. Calgary, Alberta May 1973 to Sept., 1974 Part-time Univ. of Alberta 1974 - 1978 Teaching Assistant AWARDS: University of Alberta Honor Prize 1969

National Research Council of Canada 1974, 1975, 1976, 1977 Postgraduate Scholarship

National Research Council of Canada 1978 Postgraduate Fellowship

#### PUBLICATIONS:

R.D. Milroy, C.E. Capjack, C.R. James and J.N. McMullin, "MHD Simulation of a Beat Frequency Heated Plasma." Can. J. Physics, <u>54</u>, No. 21, 2140-2146, 1976.

R.D. Milroy, C.E. Capjack, and C.R. James, "A Plasma-Laser Amplifier in the 11-16 µm Wavelength Range." Plasma Physics, <u>19</u>, No. 10, 989-995, 1977.

C.R. James, R.D. Milroy, and C.E. Capjack, "An Infrared Plasma Parametric Amplifier." Physics & Chemisrry of Material Processing, N A&T-BO "HaYKa" (In Press. Accepted September 1977).

J.N. McMullin, R.D. Milroy, and C.E. Capjack, "A Two-Dimensional Magnetic Flux Shell Model for Magnetohydrodynamic Simulations." (Submitted to Phys. Fluids) R.D. Milroy, J.W. McMullin, and C.E. Capjack, "Two-Dimensional Hydrodynamic Simulations of a Laser Heated Gas Target Plasma." (In Preparation - To be submitted to Can. J. Phys.)

1

R.D. Milroy, C.E. Capjack, and C.R. James, "Plasma Parametric Amplification of Ultra-Short Laser Pulses." (In Preparation - To be submitted to Can. J. Phys.)

## PATENT

A United States patent is being applied for through Canadian Patents and Development Limited for "Parametric Amplifier for Amplifying Laser Radiation", R.D. Milroy, C.E. Capjack, and C.R. James. **29**2

.